DIGITALISATION, SOCIAL CAPITAL AND POLITICAL PARTICIPATION

An Investigation of Technological Affordance and Intensity of Social Media Use.

Alessandro Grillo

A thesis submitted in partial fulfilment of the requirements of Kingston University London for the degree of Doctor of Philosophy

March 2021

ABSTRACT

Social Network Sites (SNSs) have attracted the attention of academics and practitioners in the field of Political Marketing, Media Studies, Sociology and Political Science for their capacity to boost social, political and civic engagement. Yet, extant research has yielded contrasting empirical evidence on the impact of different SNSs on Social Capital and Political Participation, possibly due to lack of consideration of the differences among SNSs affordances and their intensity of use.

The aim of this study is to enhance the understanding of the relationship between Social Capital and Political Participation, by considering the moderating effects of the technological affordances of different SNSs (Facebook and Twitter), their intensity of use and their combined and not combined use. The broad setting of the research is the growth of "mediatisation" and "digitalisation" of social interactions in the UK. The underlying frameworks of this thesis are the Resource Mobilisation Theory from Political Science, the Social Capital Theory and Social Network Theory from Sociology, in an optic of technological determinism through the lens of technology affordance.

To fulfil the aim of this study, a quantitative research strategy with a cross-sectional design were employed. Data were collected through an online, self-completion survey employing measures well established in previous academic research. The sample was randomly selected from a consumer panel owned by a reputed UK-based market research agency, resulting in 1212 valid responses. Partial Least Squares Structural Equation Modeling (PLS-SEM) through the SmartPLS software was used to test the moderation effects of the type of SNSs used, their intensity of use and their combined or not combined use on the relationship between Social Capital and Political Participation. A series of parametric two-way ANOVAs in SPSS and non-parametric two-way ANOVA in R were employed to test for the interaction effects of the type of SNSs used and their intensity of use on Social Capital and Political Participation.

Results show that the type of SNS employed and their intensity of use moderates the relationship between online Social Capital and Political Participation. Moreover, the research findings indicate that an interaction effect between the type of SNSs used and their intensity of use must be considered when analysing the effects of SNSs on Social Capital and Political Participation.

Given the relevance and use of Social Capital in different disciplines, this thesis contributes to theoretical knowledge in different academic fields. First, this thesis advances prior work in the domain of Engagement and Mobilisation Theories by expanding the Resource Mobilisation Theory framework through moderation mechanisms working in accordance with the principles of Technology Affordance and Technological determinism. Second, this research advances knowledge in the field of Social Capital and Political Participation by delving into the connections between different SNSs, Online (Bridging/Bonding) Social Capital, Political Participation (online/offline) and their relationships, introducing interaction effects that help to explain the contrasting empirical evidence in extant literature. Third, this thesis introduces a novel perspective to Political Marketing studies, by employing Social Capital as the main element of investigation to advance knowledge in engagement and targeting theories through different SNSs meanwhile maximising social, ethical and democratic engagement. Fourth, this thesis expands on previous SNSs studies in the domain of Media and Communication that have argued that social media affordances facilitate specific political behaviour. Integrating the Affordance thesis with the intensity of use of SNSs clarifies several competing theories related to media usage.

ACKNOWLEDGEMENTS

When I started my PhD I was not fully prepared for the long and complex journey waiting ahead of me. Only now, I am aware of the intense desire, spasmodic dedication, and endless commitment that it requires. My PhD has been like a rollercoaster. It had its exciting ups and terrifying downs where I learned more about myself, my ambitions and anxieties, my strengths and my weaknesses. I understood who and what is important to me and how my "Social Capital" helped me to survive this complicated journey. Hence, here I would like to thank all those people and organisations who jointly, willingly or unwillingly, contributed to produce my Social Capital.

To begin with I would like to thank from the bottom of my heart my principal and only thesis supervisor Prof. Francesca Dall'Olmo Riley for her steadfast encouragement, precious advice and incredible patience demonstrated during my Ph.D. This thesis would certainly not exist without her support. She is among my best role model as a researcher, mentor, and person. Also, a special thanks must go to Dr. Mark Passera. He was part of the supervisory team at the very beginning of this life-changing experience. He was the one who helped me framing my initial research proposal and found a partially-funded studentship.

Besides my supervisory team, I would like to thank Professor Stavros P. Kalafatis and Professor Arthur Money for their insightful comments on the statistical chapters of my thesis.

Also, I gratefully acknowledge the funding received towards my PhD from Kingston University. It allowed me to work part-time and focus my efforts on the PhD.

Next, I need to thank my partner of a lifetime Eleonora and my three years old son Liam. Eleonora, currently a researcher and till a few years ago a PhD student herself, has been by my side throughout this journey, living every single second of it. Without her I would not have had the strength to endure this journey. She has been a true and great supporter during my ups and downs. Never judgmental, she always trusted me and made me realise that is not the lack of knowledge, but rather the lack of trust and the lack of will that make the difference between success and failure. I feel that we both learned a lot about life and each other during our PhDs. I will never forget our discussions about research methods and statistical techniques of analysis that helped me to clear up most of my doubts. I could

have never completed my PhD without her and her precious insights on how to use R. A big thanks also to my sweet little Liam for being such a good son these past years and making it possible for me to finish what I started. His sudden and unexpected hugs and cuddles meanwhile I was sitting at my desk working on my PhD gave me the strength to go ahead. He simply made the darkest days of this journey the brightest ones.

I am forever indebted to my father Angelo and my mother Lisa for everything they have done for me during these past years. Their moral and physical support was the wind that filled the white sails (pages) of this ship (my thesis) and helped me reaching this harbour. The professional and personal success they achieved in their respective lives inspired me to do always more and always better.

I cannot avoid mentioning my brothers who helped me to settle down in London and start this journey. The many funny moments that I shared with them helped me to relax when facing stressful moments in my PhD.

A huge thank also to my dearest friends Giancarlo, Giovanna, Rosario and Domenico who listened to my outbursts and never complained. This journey would have been much more difficult without them and our long conversations that encouraged me in numerous ways.

Finally, I would like to thank all my fellow doctoral students at Kingston University and my colleagues at Kaplan College London, Kingston University, King's College London and Inseec University for the stimulating "nerd" discussions but especially for sharing enjoyable coffee and lunch breaks. Not to mention all the afterwork ludic activities we enjoy together. My PhD journey would have been much harder without them.

Thank you my Social Capital!

DECLARATION

I, Alessandro Grillo, declare that the work submitted in this thesis is my own and that all references are cited accordingly.

Signed: <u>*Alessandro Grillo* (student)</u>

Date: <u>31/03/2021</u>

TABLE OF CONTENTS

ABSTRACT	2
ACKNOWLEDGEMENTS	4
DECLARATION	6
TABLE OF CONTENTS	7
LIST OF FIGURE	13
LIST OF TABLES	16
LIST OF ABBREVIATIONS	24
1. INTRODUCTION	27
1.1. THE RESEARCH PROBLEM	27
1.2. THE RATIONALE FOR RESEARCH	33
1.3. AIM, CONCEPTS, THEORETICAL FRAMEWORK AND RESEARCH QUESTIONS	35
1.4. PROPOSED CONTRIBUTIONS TO THEORY AND PRACTICAL IMPLICATIONS	37
1.5. RESEARCH PROCESS AND METHODOLOGY	39
1.6. RESEARCH STRUCTURE	40
1.7. LITERATURE REVIEW AIMS AND METHODOLOGY	43
2. LITERATURE REVIEW ON POLITICAL MARKETING	45
2.1. EVOLUTION OF POLITICAL MARKETING	49
2.1.1. POLITICAL MARKETING AS A "MERE PROCESS": NARROW APPROACH ERA (1976-1987)	51
2.1.2. POLITICAL MARKETING: THE BETWEEN ACTS ERA (1988-200)3) 54
2.1.3. POLITICAL MARKETING: THE WIDE APPROACH ERA (2004 -	
2020)	57
2.2. TOWARDS A SOCIETAL WELFARE AIMED DEFINITION	60
2.3. POLITICAL MARKETING AND ITS CRITICISMS: THEORETICAL AI	ND
EMPIRICAL JUSTIFICATIONS FOR A NEW APPROACH	63

	2.4.	SUMMARISING: IDENTIFIED GAPS	.71
3.	LI	TERATURE REVIEW ON PARTICIPATION	.74
	3.1.	PARTICIPATION IN ETHICS	.76
	3	.1.1. VIRTUES AND THEIR REVIVAL AS EMBEDDED IN A	
		COMMUNITARIAN PERSPECTIVE FOR THE COMMON GOOD	.76
	3	.1.2. CIVIC VIRTUE AND THE COMMON GOOD	. 79
	3.2.	PARTICIPATION IN THEORIES OF DEMOCRACY	. 81
	3	.2.1. DEMOCRACY AS COMPETITION: PARTICIPATION IN	
		REPRESENTITIVE DEMOCRACY AND COMPETITIVE ELITISM	. 82
	3	.2.2. DEMOCRACY AS PARTICIPATION: DELIBERATIVE DEMOCRAC	ĽΥ
		AND SOCIAL CAPITAL	. 83
	3.3.	POLITICAL PARTICIPATION CONCEPTUALISATION	. 88
	3	.3.1. DEFINING CRITERIA OF POLITICAL PARTICIPATION	. 88
	3.4.	PARTICIPATION IN MEDIA STUDIES: THE INCONSISTENT INTERNE	Т
		EFFECT	.94
	3	.4.1. DEFINING SNSs AND THE INCONSISTENT EFFECT ON POLITICA	٨L
		PARTICIPATION	.97
	3	.4.2. REASONS BEHIND INCONSISTENT EMPIRICAL FINDINGS	100
		3.4.2.1. SNSs' TECHNOLOGICAL AFFORDANCE	100
		3.4.2.2. SNSs' INTENSITY OF USE	105
	3.5.	SUMMARISING: IDENTIFIED GAPS	107
4.	Lľ	TERATURE REVIEW ON SOCIAL CAPITAL	109
	4.1.	SOCIAL CAPITAL LEVEL OF ANALYSIS: COLLECTIVE VERSUS	
		INDIVIDUAL PERSPECTIVE	113
	4.2.	SOCIAL CAPITAL QUALITY OF ANALYSIS: CULTURAL VS	
		STRUCTURAL VIEW	116
	4.3.	CONCEPTUALISATION OF SOCIAL CAPITAL	124
	4.4.	SOCIAL CAPITAL AND THE INTERNET	129
	4	.4.2. THE INCONSISTENT SNSs EFFECT	132

4.4.3. REASONS FOR INCONSISTENT EMPIRICAL FINDINGS134
4.5. SOCIAL CAPITAL, POLITICAL PARTICIPATION AND THE INTERNET
4.6. SUMMARISING: IDENTIFIED GAPS
5. RESEARCH AIM, QUESTIONS AND THEORETICAL FRAMEWORK146
6. RESEARCH STRATEGY, DESIGN, METHODS, MEASURES, SAMPLING
AND TECHNIQUES OF ANALYSIS158
6.1. RESEARCH STRATEGY
6.2. RESEARCH DESIGN
6.3. RESEARCH METHODS: RESEARCH INSTRUMENT DEVELOPMENT. 162
6.3.1. APPROPRIATENESS OF THE RESEARCH INSTRUMENT FOR THIS
RESEARCH: RATIONALE, ADVANTAGES AND ISSUES162
6.3.2. DESIGN THE SURVEY INSTRUMENT: ONLINE MODE OF
ADMINISTRATION: RATIONALE, ADVANTAGES AND ISSUES 163
6.3.3. DESIGN THE SURVEY INSTRUMENT: STRUCTURE OF THE
QUESTIONNAIRE164
QUESTIONNAIRE
QUESTIONNAIRE1646.3.3.1. MAIN QUESTIONNAIRE BODY1656.3.4. LENGTH OF THE QUESTIONNAIRE AND RESPONSE RATES1696.3.5. COMMON METHOD BIAS PRECAUTIONS1706.4 OPERATIONALIZATION OF RESEARCH CONSTRUCTS1726.4.1.OPERATIONALISATION OF SOCIAL CAPITAL1736.4.2. OPERATIONALISATION OF OFFLINE AND ONLINE POLITICAL178
QUESTIONNAIRE1646.3.3.1. MAIN QUESTIONNAIRE BODY1656.3.4. LENGTH OF THE QUESTIONNAIRE AND RESPONSE RATES1696.3.5. COMMON METHOD BIAS PRECAUTIONS1706.4 OPERATIONALIZATION OF RESEARCH CONSTRUCTS1726.4.1.OPERATIONALISATION OF SOCIAL CAPITAL1736.4.2. OPERATIONALISATION OF OFFLINE AND ONLINE POLITICAL1786.4.3. OPERATIONALISATION OF MODERATOR VARIABLES180
QUESTIONNAIRE1646.3.3.1. MAIN QUESTIONNAIRE BODY1656.3.4. LENGTH OF THE QUESTIONNAIRE AND RESPONSE RATES1696.3.5. COMMON METHOD BIAS PRECAUTIONS1706.4 OPERATIONALIZATION OF RESEARCH CONSTRUCTS1726.4.1. OPERATIONALISATION OF SOCIAL CAPITAL1736.4.2. OPERATIONALISATION OF OFFLINE AND ONLINE POLITICAL PARTICIPATION1786.4.3. OPERATIONALISATION OF MODERATOR VARIABLES1806.4.4. OPERATIONALISATION OF CONTROL VARIABLES182
QUESTIONNAIRE
QUESTIONNAIRE
QUESTIONNAIRE1646.3.3.1. MAIN QUESTIONNAIRE BODY1656.3.4. LENGTH OF THE QUESTIONNAIRE AND RESPONSE RATES1696.3.5. COMMON METHOD BIAS PRECAUTIONS1706.4 OPERATIONALIZATION OF RESEARCH CONSTRUCTS1726.4.1. OPERATIONALISATION OF SOCIAL CAPITAL1736.4.2. OPERATIONALISATION OF OFFLINE AND ONLINE POLITICAL PARTICIPATION1786.4.3. OPERATIONALISATION OF MODERATOR VARIABLES1806.4.4. OPERATIONALISATION OF CONTROL VARIABLES1826.4.4.1. OPERATIONALISATION OF POLITICAL INTEREST1826.4.4.2. OPERATIONALISATION OF INTERNAL POLITICAL183

6.4.4.4. OPERATIONALISATION OF SOCIO-DEMOGRAPHIC
VARIABLES: AGE, EDUCATION AND GENDER 185
6.5. STATISTICAL METHODS OF ANALYSIS
6.5.1. RESEARCH QUESTIONS Q1, Q2 AND Q3: THE USE OF PLS-SEM. 188
6.5.1.1. DEFINING AND ASSESSING THE MEASUREMENT MODEL
6.5.1.2. DEFINING AND ASSESSING THE STRUCTURAL MODEL 199
6.5.1.3. ANSWERING RESEARCH QUESTIONS Q1, Q2, AND Q3 AND
TESTING FOR MODERATION
6.5.2. RESEARCH QUESTION Q4: TWO-WAY ANOVA
6.5.3. RESEARCH QUESTION Q5: NON-PARAMETRIC TWO-WAY
ANOVA
6.6. SAMPLING
6.6.1. TARGET POPULATION
6.6.2. WEB PANEL SAMPLE ADVANTAGES, LIMITATIONS AND
SOURCES OF ERROR215
6.6.3. SAMPLE SIZE
6.6.3.1. SAMPLE SIZE IN PLS-SEM
6.6.3.2. SAMPLE SIZE FOR THE TWO-WAY ANOVA ANALYSIS 220
6.6.3.3. SAMPLE SIZE FOR THE ART TEST
6.7. PILOT STUDY: TESTING FOR QUESTIONNAIRE CLARITY.
COMPREHENSIVENESS, ACCEPTABILITY AND FOR PSYCHOMETRIC
PROPERTIES OF MEASURES
6.8. SUMMARISING
7. DATA ANALYSIS
7.1. INVESTIGATING THE MODERATING EFFECT OF TYPE OF SNS USED
7.1.1. DATACOLLECTION AND SCREENING
7.1.1.1. MISSING DATA

7.1.1.2. SUSPICIOUS RESPONSE PATTERNS	
7.1.1.3. DESCRIPTIVE STATISTICS, DATA DISTRIBU	JTION AND
OUTLIERS	
7.1.2. ASSESSING PLS-SEM RESULTS OF THE REFLECT	TIVE
MEASUREMENT MODEL FOR Q1	
7.1.3. ASSESSING PLS-SEM RESULTS OF THE STRUCTU	JRAL MODEL
FOR Q1	
7.3.1.1. ASSESSMENT OF MODERATION EFFECTS	THROUGH
PERMUTATION TEST	
7.2. INVESTIGATING THE MODERATING EFFECT OF INT	ENSITY OF USE
OF SNSs	
7.3. INVESTIGATING THE MODERATING EFFECT OF THE	E COMBINED USE
OF SNSs	
7.4. ASSESSING RESULTS OF TWO-WAY FACTORIAL AN	OVA TO ANSWER
RESEARCH QUESTION Q4.	
7.4.1. MEETING THE TWO-WAY FACTORIAL ANOVA A	SSUMPTIONS.
7.4.2. TWO-WAY ANOVA FOR ONLINE BONDING SOCI	AL CAPITAL 298
7.4.3. TWO-WAY ANOVA FOR ONLINE BRIDGING SOC	IAL CAPITAL.304
7.5. ASSESSING RESULTS OF TWO-WAY FACTORIAL AN	OVA TO ANSWER
RESEARCH QUESTION Q5.	
7.5.1. MEETING THE TWO-WAY FACTORIAL ANOVA A	SSUMPTIONS.
7.5.2. ART TEST FOR OFFLINE POLITICAL PARTICIPAT	TION
7.5.3. ART TEST FOR ONLINE POLITICAL PARTICIPATI	ION
8. RESULTS INTERPRETATION, DISCUSSION AND CON	CLUSIONS.
IMPLICATIONS FOR THEORY AND PRACTICE AND F	URTHER
RESEARCH.	
8.1. RESULTS INTERPRETATION, DISCUSSION AND CON	CLUSIONS 323

REFERENCES	353
8.4. FURTHER RESEARCH	351
8.3. LIMITATIONS	347
8.2. CONTRIBUTION TO THEORY AND PRACTICE	344
8.1.5. RESULTS INTERPRETATION AND DISCUSSION FOR Q5	338
8.1.4. RESULTS INTERPRETATION AND DISCUSSION FOR Q4	334
8.1.3. RESULTS INTERPRETATION AND DISCUSSION FOR Q3	330
8.1.2. RESULTS INTERPRETATION AND DISCUSSION FOR Q2	327
8.1.1. RESULTS INTERPRETATION AND DISCUSSION FOR Q1	323

LIST OF FIGURES

Figure 1.1. Common key concepts recurring in different fields and their link to Social
Capital
Figure 1.2. Conceptual Framework of the Research
Figure 1.3. Outline of the Thesis
Figure 1.4. Flow chart overview of the literature underpinning the research
Figure 2.1. Conceptual Map Chapter 248
Figure 3.1. Chapter 3 Conceptual Map75
Figure 3.2. Political Participation Conceptualisation Criteria
Figure 4.1. Social capital application fields109
Figure 4.2. Research trends in Social Capital
Figure 4.3. Conceptual Map Chapter 4
Figure 4.4. Lin's Model of Social Capital119
Figure 4.5. Structural Theory of Social Action Components and Interactions120
Figure 4.6. Nahapiet and Goshal (1998) Model of Social Capital123
Figure 4.7. Most Frequently Used Social Capital Concepts126
Figure 5.1. Model with Moderation Effects152
Figure 5.2. Model with Moderation Effects and Control Variables
Figure 6.1. Research Process
Figure 6.2. Visual Analogue Response of Survey Questions
Figure 6.3. ISCS (Internet Social Capital Scale)175
Figure 6.4. Fixed Choice Option for Time Spent on SNSs for this Study (Copy)181
Figure 6.5. Reflective Measurement Models Vs Formative Measurement Models
Operationalisation
Figure 6.6. General Measurement Model
Figure 6.7. General Structural Model
Figure 6.8. Interaction Term Moderation Approach
Figure 6.9. MGA Approaches in PLS-SEM

Figure 6.10. Structural Model of the Two-Stage Moderation Analysis of Int_SNS_Use.
Figure 6.11. Minimum Sample Size Computed in G*Power for PLS-SEM Moderation Analysis
Figure 6.12. Sample Size Computation for the Two-Way ANOVA analysis
Figure 6.13. Fixed Choice Response Options for Time Respondents Spent on SNS in Ellison et al. (2007)
Figure 6.14. Fixed Choice Option for Time Spent on SNSs for This Study
Figure 6.15. Online Bonding Scale
Figure 7.1. A Systematic Procedure for Applying PLS-SEM
Figure 7.2. Structural Model Research Question Q1231
Figure 7.3. Proposed Measurement Model for Research Question Q1231
Figure 7.4. Measurement Model with no Control Variables for Research Question Q1.
Figure 7.5. Proposed Measurement Model with Control Variables
Figure 7.6. Structural Model Research Question Q2273
Figure 7.7. General Measurement Model for Q2274
Figure 7.8. Simple Slope Plot Brid_SC \rightarrow Off_PP when Moderated by Int_SNS_Use (Twitter Sample)
Figure 7.9. Simple Slope Plot Brid_SC \rightarrow On_PP when Moderated by Int_SNS_Use (Twitter Sample)
Figure 7.10. Structural Model Research Question Q3
Figure 7.11. Pairwise Comparisons of Combined and Non-Combined Use of SNSs Across Different Levels of Intensity of SNSs use
Figure 7.12. Pairwise Comparisons of Combined and Non-Combined Use of SNSs Across Different Types of SNSs used
Figure 7.13. Proposed Measurement Model for Research Question Q3284
Figure 7.14. Outliers Investigation for Bonding Social Capital for each combination of Type of SNS Used and Intensity of Use of the SNSs considered

Figure 7.15. Outliers Investigation for Bridging Social Capital for each combination of
Type of SNS Used and Intensity of Use of the SNSs considered
Figure 7.16. Plot of the Estimated Marginal Means for the Interaction Effect of Type of
SNSs used and Intensity of use of SNSs on Online Bonding Social Capital
Figure 7.17. Plot of the Estimated Marginal Means for the Interaction Effect of Type of
SNSs used and Intensity of use of SNSs on Online Bridging Social Capital
Figure 7.18. Outliers Investigation for Offline Political Participation for each combination
of Type of SNS Used and Intensity of Use of the SNSs considered
Figure 7.19. Outliers Investigation for Online Political Participation for each combination
of Type of SNS Used and Intensity of Use of the SNSs considered
Figure 7.20. Plot of the Estimated Marginal Means for the Interaction Effect of Type of
SNSs used and Intensity of use of SNSs on Offline Political Participation
Figure 7.21. Plot of the Estimated Marginal Means for the Interaction Effect of Type of
SNSs used and Intensity of use of SNSs on Online Political Participation

LIST OF TABLES

Table 2.1. Political Marketing Eras. 50
Table 2.2. Historical account of definitions of Political Marketing
Table 2.3. Ethical Concerns of Marketing Applications to Politics
Table 2.4. Criticisms to Political Marketing Practice and Theory. 68
Table 3.1. Models of democracy
Table 3.2. Participation in democratic theories and Social Capital. 87
Table 4.1. Conceptualisation of the Major Exponents of Social Capital125
Table 5.1. Gaps in extant literature. 146
Table 5.2. Main and Subsidiary Research Questions
Table 6.1. Survey Measures' Format, Types and Measurements. 166
Table 6.2. Online Bonding Social Capital Indicators and Respective Labels
Table 6.3. Online Bridging Social Capital Indicators and Respective Labels178
Table 6.4. Offline Political Participation Indicators and Respective Labels
Table 6.5. Online Political Participation Indicators and Respective Labels
Table 6.6. Political Interest Indicators and Respective Labels. 183
Table 6.7. Political Efficacy Scale Indicators and Respective Labels. 184
Table 6.8. Political Knowledge Questions Battery. 185
Table 6.9. Research Questions and Statistical Techniques of Analysis
Table 6.10. General Guidelines to Choose the Measurement Model Mode
Table 6.11. Bias-Corrected and Bonferroni-Adjusted Confidence Intervals of the CTA- PLS analysis for Bond_SC of the Facebook Users Only Sample ($n = 101$)194
Table 6.12. Measurement Model Assessment Problems and Solutions

Table 6.13. Interaction Term Approaches Characteristics
Table 6.14. Target Population of the Study 215
Table 6.15. Number of Groups for MGA analysis
Table 6.16. Sample Size Pilot Study
Table 7.1. Sample Count
Table 7.2. Demographic Characteristics of E_S, F_U, T_U and F+T_U Samples235
Table 7.3. Sample Count of Heavy, Mild and Light users of F_U, T_U and F+T_U236
Table 7.4. Sample Percentage of Heavy, Mild and Light users of F_U, T_U and F+T_U.
Table 7.5. Summary Convergent Validity and Internal Consistency ReliabilityAssessment for Facebook Users Sample ($n = 432$)
Table 7.6. Discriminant Validity Summary Measurement Model F_U with no ControlVariables (n = 432)
Table 7.7. Summary Convergent Validity and Internal Consistency ReliabilityAssessment for T_U (n= 383).
Table 7.8. Discriminant Validity Summary Measurement Model T_U with no Control Variables (n = 383).
Table 7.9. Bias-Corrected and Bonferroni-Adjusted Confidence Intervals of the CTA-PLS analysis for Off_PP of F_U (n = 432)
Table 7.10. Bias-Corrected and Bonferroni-Adjusted Confidence Intervals of the CTA-PLS analysis for Off_PP of T_U (n = 383)
Table 7.11. Summary Measurement Model Assessment Metrics for F_U Sample with no
Control Variables and Off_PP_3 Indicator Removed (n= 432)247
Table 7.12. Discriminant Validity Summary Measurement Model F_U with no ControlVariables and Off_PP_3 Indicator Removed (n= 432).248

Table 7.13. Summary Measurement Model Assessment Metrics for T_U Sample with no
Control Variables and Off_PP_3 Indicator Removed (n = 383)249
Table 7.14. Discriminant Validity Summary Measurement Model T_U with no Control
Variables and Off_PP_3 Indicator Removed (n= 383)250
Table 7.15. Internal Consistency Reliability Summary F_U and T_U with control
variables and Off_PP_3 removed
Table 7.16. Convergent Validity Summary F_U and T_U with control variables and
Off_PP_3 removed
Table 7.17. Discriminant Validity Summary F_U and T_U with control variables and
Off_PP_3 removed
Table 7.18. VIF Values of the Structural Model of F_U with no Control Variables (n =
432)
Table 7.19. Coefficients of Determination of the Structural Model of F_U with no Control
Variables (n = 432)
Table 7.20. BCa Confidence Intervals of the R^2 of F_U with no Control Variables (n =
432)
Table 7.21. Effect Size f^2 of the Structural Model of the Sample of F_U with no Control
Variables (n = 432)
Table 7.22. Path Coefficients of the Structural Model of the Sample of F_U with no
Control Variables (n = 432)
Table 7.23. Confidence Intervals BCa of the Path Coefficients of the Structural Model of
F_U with no Control Variables ($n = 432$)
Table 7.24. Predictive Relevance Q2 of the Structural Model of the Sample of Facebook
Users with no Control Variables (n = 432)255
Table 7.25. q2 Effect Size of the Sample of Facebook Users with no Control Variables (n
= 432)

Table 7.26. In-Sample and Out-of-Sample Model Selection Criteria and Model Fit
measures of the Sample of Facebook Users with no Control Variables ($n = 432$) 256
Table 7.27. CMB Assessment Through Multicollinearity for the Sample of Facebook
Users with no Control Variables (n = 432)
Table 7.28. Model Comparison with and without Controls Variables for F_U (n = 432).
Table 7.29. F test for Significance of the Change in R2 Between the Models Without and
With Control Variables of F_U (n = 432)
Table 7.30. Model Comparison with and without Controls Variables for T_U (n = 383).
Table 7.21. E test for Significance of the Change in D2 Detween the Models Without and
Table 7.51. F lest for Significance of the Change III K2 between the Models without and With Control Variables of T. II $(n - 383)$ 260
with Control variables of 1_0 (if $= 383$)200
Table 7.32. Compositional Invariance Constructs of Facebook Users ($n = 432$) and
Twitter Users ($n = 383$)
Table 7.33. Equality of Composite Mean Values for Facebook $(n = 432)$ and Twitter $(n = 432)$
383) Samples
Table 7.34. Equality of Composite Variances Values for Facebook $(n = 432)$ and Twitter
(n = 383) Samples
Table 7.35. Permutation Test Results for Significant Differences between Facebook (n =
432) and Twitter (n = 383) users
Table 7.36 Comparisons of Facebook and Twitter Users Across Different Levels of
Intensity of SNSs Use
Table 7.37. Internal Consistency Reliability Summary of F_U and T_U with Control
variables and Off_PP_3 Removed
Table 7.38. Problematic Constructs with $AVE < 0.5$ in the Heavy, Mild and Light Sub-
groups of Facebook and Twitter Users
Table 7.39. Outcome of Indicators Elimination Process. 267

Table 7.40. Q2 and R2 Values and their Significance for Facebook and Twitter UsersAcross Different Levels of Intensity of SNSs Use.268
Table 7.41. Path Coefficients and their Significance Across Levels of Intensity of SNSs Use per Type of SNSs used
Table 7.42. Problematic VIF values found through CMB Assessment of Facebook Light Users Sample with Control Variables and Item Off_PP_3 Removed ($n = 99$)270
Table 7.43. Problematic VIF values found through CMB Assessment of Twitter Mild Users Sample with Control Variables and Item Off_PP_3 Removed ($n = 95$)270
Table 7.44. Mann-Whitney Ranks Comparing Total Time Spent on Facebook (n = 432)and Twitter (n = 383)
Table 7.45. Mann-Whitney Test Statistics Comparing Total Time Spent on Facebook (n= 432) and Twitter (n = 383)
Table 7.46. Indicators with Communality Values Below 0.5 in Facebook and Twitter Users Samples. 275
Table 7.47. Summary Statistics of Predictive Relevance and Accuracy for F_U and T_U Samples.
Table 7.48. Model Comparison with and without the Moderator Int_SNS_Use for F_U (n= 432)
Table 7.49. Model Comparison with and without the Moderator Int_SNS_Use for T_U(n = 383)
Table 7.50. Sample sizes for Facebook, Twitter and, Facebook and Twitter combined users when accounting for Intensity of use of SNSs
Table 7.51. Summary Internal Consistency reliability for all F+T_U Samples. 285
Table 7.52. Problematics Constructs with AVE < 0.5 in the Mild Facebook and TwitterCombined Users Sample (n = 78)
Table 7.53. Summary predictive Accuracy and Relevance for all F+T_U Samples286

Table 7.54. Path Coefficients and their Significance Across Levels of Intensity of SNSs
Use per Type of SNSs used
Table 7.55. Permutation Test Results for Significant Differences between $F+T_U$ (n =
397) and F_U (n = 432)
Table 7.56. Permutation Test Results for Significant Differences between $F+T_U$ (n =
397) and T_U (n = 383)
Table 7.57. Summary Pairwise Permutation Tests Across Different Levels of Intensity of
Use
Table 7.58. Summary Pairwise PLS-MGA Comparisons for the Variable of Interest
Across Different Levels of Intensity of Use
Table 7.59. Non-Parametric Confidence Set Approach for the Comparison between the
Samples of Facebook and Twitter Combined Heavy Users $(n = 41)$ and Facebook Heavy
Users (n = 108)
Table 7.60. Sensitivity Analysis of the Multigroup Pairwise Comparisons Results Across
Different Statistical Methods of Analysis 203
Different Statistical Methods of Analysis.
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296 Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging Social Capital for Type of SNSs Used Across different Levels of Intensity of Use. 297
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296 Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging Social Capital for Type of SNSs Used Across different Levels of Intensity of Use. 297 Table 7.63. Means and Related Variances of Online Bridging Social Capital for each
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296 Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging Social Capital for Type of SNSs Used Across different Levels of Intensity of Use. 297 Table 7.63. Means and Related Variances of Online Bridging Social Capital for each 298
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296 Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging Social Capital for Type of SNSs Used Across different Levels of Intensity of Use. 297 Table 7.63. Means and Related Variances of Online Bridging Social Capital for each 298 Table 7.64. Descriptive Statistics for Online Bonding Social Capital Scores by Type of
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296 Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging Social Capital for Type of SNSs Used Across different Levels of Intensity of Use. 297 Table 7.63. Means and Related Variances of Online Bridging Social Capital for each 298 Table 7.64. Descriptive Statistics for Online Bonding Social Capital Scores by Type of 298 SNSs Used and Intensity of Use of SNSs. 299
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use. 296 Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding and Bridging Social Capital for Type of SNSs Used Across different Levels of Intensity of Use. 297 Table 7.63. Means and Related Variances of Online Bridging Social Capital for each 298 Table 7.64. Descriptive Statistics for Online Bonding Social Capital Scores by Type of 298 Table 7.65. Summary of Two-Way ANOVA for Online Bonding Social Capital. 299
Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSs Used and Intensity of Use.

Table 7.67. Pairwise Comparisons of Facebook and Twitter Users Across Each Level of
Intensity of Use of SNSs
Table 7.69 University Test for Differences in Online Donding Social Capital Detwoon
Table 7.68. Univariate Test for Differences in Online Bonding Social Capital Between
Light, Mild and Heavy Users Across Each SNS
Table 7.69. Pairwise Comparisons Across Each Level of Intensity of Use of SNSs by type
of SNSs used 303
51 51 55 used.
Table 7.70. Descriptive Statistics for Online Bridging Social Capital Scores by Type of
SNSs Used and Intensity of Use of SNSs
Table 7.71. Summary of Two-Way ANOVA for Online Bridging Social Capital305
Table 7.72 Univariate Test for Differences in Online Bridging Social Capital Between
Eacabook and Twitter Users Across Each Level of Intensity of SNS Use
Tacebook and Twhter Users Across Each Level of Intensity of Sivs Use
Table 7.73. Pairwise Comparisons of Facebook and Twitter Users Across Each Level of
Intensity of Use of SNSs
Table 7.74. Univariate Test for Differences in Online Bridging Social Capital Between
Light, Mild and Heavy Users Across Each SNS
Table 7.75. Pairwise Comparisons Across Each Level of Intensity of Use of SNSs by type
of SNSs used
Table 7.76. Games-Howell Pairwise Comparisons of Different Levels of Intensity of Use
across Eacebook and Twitter Users for Online Bridging Social Capital
across Facebook and Twitter Users for Onnine Bridging Social Capital
Table 7.77. Confidence Intervals and p Values for Online Bridging Social Capital robust
to heteroskedasticity
Table 7.78. Skewness and Kurtosis of Offline and Online Political Participation of Groups
of SNSs users Split According to Type of SNSs Used and Intensity of Use
Table 7.70 Leven 2's Tast for Home can site of Varianess of Office Dalitical Datisiantian
Table 7.79. Levene's Test for Homogeneity of Variances of Offline Political Participation
Ior Type of SNSs Used Across different Levels of Intensity of Use
Table 7.80. Descriptive Statistics of Offline Political Participation Scores by Type of
SNSs Used and Intensity of Use of SNSs

Table 7.81. Summary of the ART test for Offline Political Participation. 315
Table 7.82. Robust Results of the ART two- way ANOVA of Offline PoliticalParticipation between Facebook and Twitter Users at each Level of Intensity of Use. 315
Table 7.83. Pairwise Comparisons of Levels of Offline Political Participation within the Robust ART framework
Table 7.84. Games-Howell Pairwise Comparisons of Different Levels of Intensity of Useacross Facebook and Twitter Users
Table 7.85. Confidence Intervals and p Values for Offline Political Participation robust to heteroskedasticity
Table 7.86. Descriptive Statistics of Online Political Participation Scores by Type of SNSs Used and Intensity of Use of SNSs.
Table 7.87. Summary of the ART test for Online Political Participation
Table 7.88. Results of the ART two- way ANOVA of Online Political Participationbetween Facebook and Twitter Users at each Level of Intensity of Use.320
Table 7.89. Pairwise Comparisons of Levels of Online Political Participation within the ART test framework
Table 7.90. Games-Howell Pairwise Comparisons of Different Levels of Intensity of Use across Facebook and Twitter Users
Table 7.91. Confidence Intervals and p Values for Online Political Participation robust to heteroskedasticity
Table 8.1. Findings Summary for Research Question Q1. 324
Table 8.2. Findings Summary for Research Question Q2. 329
Table 8.3. Findings Summary for Research Question Q3. 331
Table 8.4. Findings Summary for Research Question Q4. 335
Table 8.5. Findings Summary for Research Question Q5. 339

LIST OF ABBREVIATIONS

AIC Akaike's Information Criterion AICc Akaike's Information Criterion Corrected AICu Akaike's Information Criterion Unbiased **ART** Aligned Rank Transform **BIC** Bayesian Information Criterion AMA American Marketing Association **AVE** Average Variance Extracted **BCa** Bias Corrected Accelerated Confidence Intervals Bond_SC Bonding Social Capital Brid_SC Bridging Social Capital C-L Cross Loadings CA Cronbach's Alpha **CB** Covariance Based Approach **CET** Cognitive Engagement Theory **CI** Confidence Intervals **CMB** Common Method Bias Comb_Use_SNSs Combined use of SNSs **CPM** Comprehensive Political Marketing **CR** Composite Reliability **CTA-PLS** Confirmatory Tetrad Analysis in Partial Least Square **DV** Dependent Variable *df* Degree of Freedom **E_S** Entire Sample f^2 f Squared (Effect Size) F-L Fornell-Larcker Criterion **F_U** Facebook Users **F**+**T**_**U** Facebook and Twitter Users FCA Full Collinearity Assessment G-D Good Dominant Logic **HF_U** Heavy Facebook Users HF+T_U Heavy Facebook and Twitter Users

HT_U Heavy Twitter Users **HTMT** Heterotrait-Monotrait Ratios Int_SNS_Use Intensity of Use of SNSs **ISCS** Internet Social Capital Scale **ITM** Interaction Term Moderation **Kurt** Kurtosis L_O_L Lowest Outer Loading LF_U Light Facebook Users LF+T_U Light Facebook and Twitter Users LT_U Light Twitter Users LV Latent Variable Max Maximum Value Min Minimum Value MAD Mean Absolute Deviation MF_U Mild Facebook Users MF+T_U Mild Facebook and Twitter Users MGA Multi Group Analysis **MICOM** Measurement Invariance of Composite Models MMM Marketing Mix Management paradigm MT_U Mild Twitter Users **MV** Measured Variable **n** Sample Size Off_PP Offline Political Participation **On_PP** Online Political Participation **OLs** Outer Loadings **PEff** Political Efficacy **PInt** Political Interest **PKnow** Political Knowledge **PLS** Partial Least Squares PLS-MGA Multi-Group Analysis in Partial Least Square Q² Stone-Geisser's coefficient q^2 Effect Size Predictive Relevance \mathbf{R}^2 Coefficient of Determination

 \mathbf{R}^{2}_{adj} Adjusted Coefficient of Determination **RBV** Resource Based View **RM** Relationship Marketing RMS_{theta} Root Mean Square Residual Covariance **RMSE** Root Mean Squared Error S-D Service Dominant Logic SC Social Capital **SD** Standard Deviation SE Standard Error **SEM** Structural Equation Modeling **Skew** Skewness **SNSs** Social Networking Sites SPSS Statistical Package for Social Sciences SRMR Standardised Root Mean Square Residual T_U Twitter Users V_L_O_L Value Lowest Outer Loading VAR Visual Analogue Response **VIF** Variance Inflation Factor

1. INTRODUCTION

Consistent with best practice (Bryman, 2012) this chapter starts with a definition of the research area and of the research problem (Section 1.1). Based on the research problem, the rationale of the research is discussed (Section 1.2). Then, aims and objectives of the research are set, and research questions defined (Section 1.3). A discussion of the implications of the research for theory and practice follows (Section 1.4). Next, the research process and methodology are highlighted (Section 1.5). Finally, an outline of the thesis structure is provided (Section 1.6).

1.1. THE RESEARCH PROBLEM

In recent years, citizens' disengagement with traditional forms of political participation has become a major concern for almost all western democracies and scholars alike, given the startling constant decline of voters' turnout, party membership and affection of citizens with politics (Peters, 2017; Van der Meer, 2017), except for some rare events like the U.S.A. presidential election of 2020 (Pew Research Center, 2021). Declining levels of traditional forms of Political Participation challenge the notion that democracy is in "good shape" since people's engagement in politics is understood as something positive and necessary for a proper functioning of the democratic process (Livingstone, 2013).

At the same time, there is evidence of political news exchanged through the Internet and particularly through Social Network Sites (SNSs) like Facebook and Twitter (Gil de Zuniga et al., 2017; Park, 2019; Tang and Lee, 2013; Xenos et al., 2014), of online petitions with millions of signatures, of mass protests and even revolutionary movements coordinated through SNSs like the *Arab Spring*, the anti-austerity protest in Greece, the *Indignados* in Spain and the *Gilets Jaunes* in France (Boulianne, 2015; Carson and Elstub, 2019; Theocharis and Lowe, 2015; Valenzuela et al., 2018). Arguably, this evidence suggests that new forms of political engagement are arising in the online world, taking over the more traditional types of participation, allowing the Internet to be recognised as an instrument of political mobilisation (Ekman and Amna, 2012; Fox, 2014; Van Deth, 2014).

Therefore, the claim that democracy is in danger because of declining levels of traditional forms of Political Participation may be exaggerated (Theocaris and Van Deth, 2018). Rather, different and new forms of participation (online) must be taken into account as indicators of a healthy democracy in studies regarding political engagement (Ekman and Amna, 2012) which may be facilitated by the use of SNSs. Indeed, since the success of the U.S.A. Facebook-based presidential campaign of Barack Obama in 2008, heavily SNSs-based political campaigns started to appear all around the world, as a potential remedy to the declining engagement of the electorate with politics (Earl and Kimport, 2011). Arguably, this is due to the ability of SNSs to lower transaction costs, remove physical obstacles to participation, and generate new forms of engagement that seem to have a positive incremental effect also on traditional political activities offline (Bimber et al., 2005; Van Laer and Van Aelst, 2010; Theocharis and Lowe, 2014).

Considering the relevance of SNSs use for increasing levels of citizens Political Participation, several academic disciplines have engaged with the study of online platforms. For instance, communication studies have advocated the intensive and extensive use of the SNSs for establishing one-to-one bidirectional dialogue, transparency, trust, intimacy and higher levels of responsiveness from voters (Press and Williams, 2010). Political Scientists have argued in favour of more direct kinds of democracy (Deliberative and Participatory) through online forums where citizens can make their opinion really "count", restoring that feeling of individual influence over politics which appeared to be lost (Livingston, 2013). Political Philosophers have widely criticised *deontological* and *consequentialist* ethical approaches to politics, regarding them as part of the fundamental causes of the current "democratic malaise". Rather, they advocate the revival of Virtue Ethics and Communitarianism (Hartley and Watson, 2014), which are embedded in a logic of common good as arising from society, through means of debating and deliberation in online "agora". Even Political Marketing academics, following the Service-Dominant (S-D) logic of mainstream Marketing (Vargo and Lusch, 2004, 2008), have placed the concepts of interactions, co-creation of values, exchanges, bidirectional dialogue, long-term relationships, trust and intimacy at the centre of the relationships between political institutions and society at large (Egan and Lynch, 2012; O'Cass, 2009; Peterson and Godby, 2020).

Notably, concepts like trust, intimacy, reciprocity, long-term relationships and bidirectional dialogue routinely and consistently recur in the extant literature of different disciplines when dealing with SNSs (Figure 1.1). Sociologists in particular have demonstrated how reciprocity, trust, civic virtue among citizens and a strong embeddedness in social networks' ties among citizens are indispensable elements of a healthy society characterised by high levels of Political Participation (Putnam, 2000; Norris, 2002; Lin et al., 2008). These concepts are relevant and central to the construct of Social Capital, which in this study is regarded as the instrument to investigate the relation among SNSs, the "*democratic malaise*" and Political Participation.

Figure 1.1. Common key concepts recurring in different fields and their link to Social Capital.



Source: Created by author.

Indeed, many competing perspectives exist about the mechanisms through which SNSs use might affect Political Participation (for a comprehensive discussion see Boulianne, 2015). One of the most accredited perspective (employed in this research) focuses on the role of social networks ties as generator of resources like Social Capital that can be purposely mobilized to perform political actions (Gil de Zuniga et al., 2012; Molyneux et al., 2015). Social Capital exists within social networks in the forms of trust, obligations, expectations of reciprocity, sharing informational behaviour and knowledge provision (Coleman, 1988; Putnam, 2000). Yet, different types of social ties can produce different types of Social Capital (Putnam, 2000). Strong ties (relationship with parents, relatives and close friends) generate Bonding Social Capital defined by high level of trust, reciprocity and emotional support created among like-minded individuals and, as such, may provide homogenous types of resources and information. In contrast, weak ties (relationship with acquaintances) generate Bridging Social Capital, that is characterised by weak emotional and affective support, while it provides heterogeneity of information and resources as it is created among people with different cultural background and beliefs (Putnam, 2000).

Although, to some extent, Online and Offline Social Capital seems to have similar effects on increasing citizens' levels of political engagement, extant studies have originated mixed and opposing findings over the effect of different forms of Social Capital (Bonding/Bridging) on Offline and Online Political Participation.

Indeed, while some studies have shown a positive effect of Online Social Capital on Offline Political Participation (Gibson and McAllister, 2013; Skoric et al., 2009), others found no significant association between them (Gil de Zuniga et al., 2012). With respect to Online Political Participation, several studies have identified a positive relationship with Online Social Capital (e.g., Skoric et al., 2009; Valenzuela et al., 2012). However, the literature provides mixed and inconsistent findings with respect to what type of Social Capital (Bonding/Bridging) may facilitate online political behaviour.

Likewise, studies on the relationship between the Internet and Bonding and Bridging Social Capital have produced contradictory results. Indeed, some authors found no significant effects (e.g., Papacharissi and Mendelson, 2011; Petersen and Johnston, 2015; Uslaner 2004). Meanwhile, many others found either significant negative effects (Helliwell and Putnam, 2004; Putnam, 2000; Nie et al., 2002; Norris, 2000; Zhang et al., 2010) or positive effects (Burke, Marlow, and Lento, 2010; Ellison et al., 2007; Gil de Zuniga et al., 2012; Ho et al., 2003; Kaigo, 2012; Resnick, 2002; Steinfield et al., 2008; Valenzuela et al., 2009; Yoder and Stutzman, 2011). The same pattern of inconsistent and contradictory findings holds true for Offline and Online Political Participation, and its relationship with SNSs use. While Gil de Zuniga et al. (2012) and Valenzuela et al. (2009) found no relationship between SNSs use and Offline Political Participation, Vitak et al. (2011) found a weak negative relationship. In contrast, Bode (2012), Park et al. (2009), Skoric and Poor (2013), DiGrazia et al. (2013) and Towner (2013) found that SNSs use is a statistically significant indicator of offline political behaviour. Furthermore, Bode (2012) and Vitak (2012) found SNSs use to be a significant predictor of political activities online. However, Gil de Zuniga et al. (2012) and Skoric and Poor (2013) found no significant relationship.

The contrasting empirical evidence regarding the impact of SNSs on Social Capital, Political Participation and their respective relationship has been ascribed by scholars to several reasons. Two of the most prominent are the specific technological affordance of each SNS, and the time spent on them.

With respect to the technological affordance, different SNSs have different functions, settings and algorithms which affect users' perceptions of the needs that each SNS can satisfy (Brandtzaeg, 2012; Petrocchi et al, 2015; Phua et al., 2017; Valenzuela et al., 2017). Furthermore, research has shown that different types of needs are better satisfied by different types of Social Capital (Putnam, 2000). While Bonding Social Capital satisfies the need for emotional and psychological support, reinforced by the high levels of trust, reciprocity and solidarity provided by closed homogenous groups, Bridging Social Capital bridges across different heterogeneous social context and as such it is better at satisfying the need for novel and diverse information and at providing large social support (Somma, 2009). Hence, Bonding Social Capital seems to facilitate collective actions which need greater moral support like protesting, marching etc. (Centola, 2011; Putnam, 2000). Rather, Bridging Social Capital facilitates actions that do not require high moral support, like voting or contacting politicians (Gladwell, 2010; Morozov, 2011; Somma, 2009). Therefore, the unique affordance of each SNS can reinforce or weaken the relationship between Online Bonding and Bridging Social Capital and Online or Offline Political Participation to different extents and through different processes (e.g., strengthening social ties, decreasing communication costs, facilitating acquisition of political information, etc.).

Moreover, the time spent on the SNSs may affect different types of Social Capital (Bonding/Bridging) and Online/Offline Political Participation. Indeed, each SNS has specific affordances which, in time, would facilitate the dominance of a particular type of Social Capital. Yet, as discussed above, each type of Social Capital is better at satisfying different needs and would help individuals to perform some social and political actions over others (Ellison and Vitak, 2015). Hence, a logical link between SNSs' affordances and the time spent on them and their effect on Social Capital and Political Participation may exists.

Furthermore, it is understood that nowadays a large majority of people use two or more SNSs simultaneously (Duggan, 2015; Hampton et al., 2011) to satisfy different needs (Brandtzæg, 2012; Huberman et al., 2009; Petrocchi et al., 2015; Phua et al., 2017). Yet, since different SNSs have different affordances, it is conceivable that the simultaneous use of different SNSs can affect the relationship between Online Social Capital and Political Participation too.

However, a review of the literature (see Chapter 4, Section 4.5) reveals that extant studies investigate the relationship between Online Bridging and Bonding Social Capital and Online and Offline Political Participation on one SNS and then generalise its effects to other platforms without taking into consideration how their respective technological affordances, their simultaneous use or their intensity of use could affect such relationship. This limitation in extant research has increased the level of uncertainty over the effects of SNSs on Social Capital and Political Participation, resulting in the development of several competing perspectives based on the contrasting empirical evidence reported above (see Chapter 3, Section 3.4.1 and Chapter 4, Section 4.4.2). Such lack of clarity is surprising, given that politicians and their political parties rely heavily on SNSs to engage with the electorate and implement their political strategies. Given the above background, shedding light on which type of SNSs could help politicians to better address and engage with the electorate according to their specific affordance, their combined or not combined use and their intensity of use is fundamental. Furthermore, investigating the heterogeneity of the SNSs with respect to their specific affordances and intensity of use may allow for more nuanced connections between Online (Bridging/Bonding) Social Capital and Political Participation (Online/Offline). This would ultimately help to explain the contrasting empirical evidence in extant literature, by appraising competing media

theories related to media usage (e.g., Time Displacement Theory and Channel Complementarity Theory) (See Chapter 3, Section 3.4.2 and Chapter 4, Section 4.4).

The above discussion provides the contextual background to this thesis and is the base for defining the rationale for conducting the current research.

1.2. THE RATIONALE FOR RESEARCH

As discussed in Section 1.1. with the development of the Internet and SNSs, social interaction has started to take place also online (Quan-Haase and Young, 2010). Over time, people have increasingly embraced SNSs to fulfil their informational (Gil de Zuniga et al., 2017; Xenos et al., 2014), emotional, and social needs (Gan and Wang, 2015; Hołyst, 2017; Zhan et al., 2016). Hence, studying SNSs and their effects on different social settings and human behaviours has become a pressing issue for scholars and for practitioners alike in a variety of disciplines and contexts. In particular, the effects of SNSs on Social Capital and Political Participation have been the focus of interest of sociologists and political scientists, who have investigated whether SNSs could increase levels of citizens' political engagement (Boulianne, 2015; Boulianne and Theocharis, 2020; Chae et al., 2019; Skoric et al., 2016). Meanwhile, communication experts and political marketers alike have focused on the targeting efficacy and efficiency of SNSs with respect to inactive and disengaged people and on how different digital platforms could be better used for targeting specific segments of the population like millennials that are hard to reach offline (Hughes, 2018).

Nonetheless, as highlighted in Section 1.1. extant research has provided contrasting empirical evidence with respect to the effects of SNSs on Social Capital, Political Participation and their relationship. Skoric et al. (2016) remark that early research suggested the necessity to advance knowledge on how different types of Social Capital (Bonding/Bridging) created online could affect Political Participation online and offline (e.g., Hampton, 2011; Skoric et al., 2009; Zhong, 2014). However, by focusing on the heterogeneity of Social Capital and Political Participation the effect of the heterogeneity of the SNSs on their relationship was neglected. This is problematic because, as discussed in Section 1.1, each SNS has unique technological affordances (Ellison and Vitak, 2015; Enli and Skogerbø, 2013) which can facilitate or hinder the relationship between Social Capital and Political Participation through several

mechanisms (see Section 3.4.2 and Section 4.4.3). Furthermore, the time spent on SNSs could affect the relationship between SNSs, Online Social Capital and Political Participation. For instance, given that Facebook's affordances facilitate the maintenance and reinforcement of strong ties (Petrocchi et al., 2015; Nadkarni and Hofmann, 2012), the time spent on it may affect the relationship between Online Bonding Social Capital and participatory behaviour that need high personal psychological and moral support like protesting and marching (Valenzuela et al., 2017). Rather, Twitter which is centred around information and opinions sharing and is better at fulfilling informational needs through weak ties can create large social support and develop a sense of reciprocity (Huberman et al., 2009; Petrocchi et al., 2015; Valenzuela et al., 2017). In time, such sense of reciprocity may affect the relationship between Online Bridging Social Capital and participatory behaviour that needs social support, like creating petitions or joining civic movements.

The above discussion suggests that the technology affordance, the combined or not combined use and the intensity of use of SNSs could be differentiating elements of analysis and could explain, to some extent, the contrasting empirical findings identified through the literature review. In fact, contrasting and inconsistent empirical findings regarding the effect of an independent variable (Online Social Capital) on the same dependent variable (Political Participation) raise the question of whether interaction or moderating effects can explain such differences (Field, 2013; Memon et al., 2019).

Given the relevance of Social Capital to several disciplines, research that fills the above gaps can lead to advancements in Sociology, Politics, Communication studies and Political Marketing. For the latter, this research will help to inform the decision of political parties and their marketers in relation to the implementation of effective engagement and targeting strategy with the electorate, through the appropriate SNSs for the type of behaviour sought after by political institutions. Indeed, many scholars advocate that research should distinguish between SNSs based on their unique affordances (Boulianne 2019; Halpern et al., 2017; Valenzuela et al., 2017; Woo-Yoo and Gil de Zuinga, 2014), their combined or not combined use (Petrocchi et al., 2015) and their intensity of use (Boulianne, 2015; Ellison et al., 2007; Skoric and Kwan, 2011; Skoric and Poor, 2013; Skoric et al., 2016; Steinfield et al., 2008) and should examine

possible paths of influence of SNSs on Social Capital, Political behaviour and their relationship. Yet such research is lacking.

The above rationale for advancing knowledge informs the aims and objectives of this research as discussed in the next Section.

1.3. AIM, CONCEPTS, THEORETICAL FRAMEWORK AND RESEARCH QUESTIONS

This research aims to investigate the impact of the affordances of different types of Social Network Sites, namely Facebook and Twitter, their intensity of use and their combined or not combined use on sharing, cooperative and mobilising behaviours of people in the context of Social Capital and Political Participation. The broad setting is the growth of "*mediatisation*" and "*digitalisation*" of social interactions through SNSs, under the assumption of "*technological determinism*" and "*technology affordances*". The former is centered on the belief that technology "per se" can affect individuals' actions, interactions and even psyche regardless of any socio-demographic characteristics of respondents and any reasons of usage of the technology (Hauer, 2017). The latter is defined as "*the perceived and actual properties of the thing, primarily those fundamental properties that determine just how the thing could possibly be used*" (Norman, 1988: p. 9).

The main constructs employed in this research to investigate the above are: (1) **Online Social Capital**, differentiated in its Bonding and Bridging dimensions (Putnam, 2000; Williams 2006), mirroring cooperative and sharing behaviour in social interactions; and (2) **Political Participation** differentiated in its online and offline form and representing mobilised behaviour (Ekman and Amna, 2012; Fox, 2014; Van Deth, 2014).

The underlying frameworks of this research which brings together the above concepts are: (1) the *Resource Mobilisation Theory* (Zald and McCarthy, 2002) according to which resources such as financial capital, education, skills and even Social Capital (Bourdieu, 1986; Bourdieu and Wacquant, 1992) may affect levels of Political Participation; (2) the *Social Capital Theory* (Putnam, 2000) which corroborates the claim that association and interaction in social networks facilitate the creation of Social Capital;

(3) the *Social Network Theory* (Granovetter, 1973; Wellman 1983) according to which participation in social networks may promote Political Participation.

To account for inconsistent findings in the extant research about the effect of SNSs on Online Social Capital, Political Participation and their relationship, the "Technology Affordance" principle is recalled (Norman, 1988) and the type of SNSs employed and their intensity of use are tested for moderation within the Resource Mobilisation Theory framework. Also, given that different SNSs have different affordances and that many people use more than one SNS (Petrocchi et al., 2015), the combined use of Facebook and Twitter is tested as moderator of the relationship between Online Social Capital and Political Participation. This will help to answer the first three research questions:

- **Q1.** Does the type of SNSs used moderate the relationships between Online Social Capital and Political Participation? To what extent?
- **Q2.** Does the intensity of use of SNSs moderate the relationships between Online Social Capital and Political Participation? To what extent?
- **Q3.** Does the combined use of SNSs moderate the relationships between Online Social Capital and Political Participation? To what extent?

The conceptual framework employed to investigate the above is presented in Figure 1.2.

Finally, the type of SNS used is tested for interaction with their level of usage, to investigate differences in the amount of Online Bonding/Bridging Social Capital and Online/Offline Political Participation reported by users of different SNSs applying the Social Capital Theory and the Social Network Theory, respectively. This will help to answer the following research questions:

- **Q4.** To what extent do levels of Online Social Capital differ across different types of SNSs employed at different levels of their intensity of use?
- **Q5.** To what extent do levels of Political Participation differ across different types of SNSs employed at different levels of their intensity of use?
However, given that Online Social Capital is conceptualised through the two dimensions of Bonding and Bridging (Putnam, 2000; Williams, 2006) and that Political Participation is examined in its Online and Offline forms (Ekman and Amna, 2012; Fox, 2014; Van Deth, 2014) each main research question has several subsidiary questions as reported in Chapter 5, Table 5.2.





1.4. PROPOSED CONTRIBUTIONS TO THEORY AND PRACTICAL IMPLICATIONS

Given the relevance and use of Social Capital in different disciplines (see Section 1.1) this thesis aims to contribute to theoretical knowledge in different academic fields.

First, this thesis advances prior work in the domain of Engagement and Mobilisation Theories by expanding the Resource Mobilisation Theory framework (McCarthy and Zald, 1977) through moderation mechanisms working in accordance with the principles of Technology Affordance and Technological determinism (see Section 1.3).

Second, this research advances knowledge in the field of Social Capital by delving into more nuanced connections between different SNSs, Online (Bridging/Bonding) Social Capital, Political Participation (Online/Offline) and their relationships, introducing interaction effects that may explain the contrasting empirical evidence in the extant literature presented in Section 1.1. In particular, this thesis provides a framework for understanding how different types of social connectedness relate to different political actions, in accordance with the heterogeneity of the SNSs specific affordances, their intensity of use and combined or not combined use.

Third, this thesis introduces a novel perspective to Political Marketing studies which so far have been heavily criticised for their lack of engagement with theories of democracy, ethics and sociology (Henneberg et al., 2009; O'Shaughnessy, 2001; Ormrod et al., 2013). This shift is advocated by many academics (Henneberg et al., 2004; Ormrod et al., 2013; Savigny and Wring, 2009) and is in line with the Comprehensive Political Marketing (CPM) approach (Lees-Marshment, 2001, 2003) which encourages the creation and integration of Political Marketing Theories with concepts originating from other disciplines. Employing as the main element of investigation the concept of Social Capital, originating from the sociological field, provides a new theoretical framework for researchers in Political Marketing that could advance knowledge in engagement and targeting theories meanwhile maximising social, ethical and democratic engagement (see Chapter 3 and Chapter 4).

Fourth, this thesis expands on previous SNSs studies in the domain of Media and Communication that have argued that social media affordances facilitate specific political behaviour. Integrating the Affordance thesis with the intensity of use of SNSs would allow to appraise competing media theories related to media usage (e.g., Time Displacement Theory and Channel Complementarity Theory) in different SNSs and for different types of Political Participation (online/offline).

With respect to the practical implications, this research will help to inform the decision of political marketers in relation to the implementation of effective engagement

and targeting strategy of the electorate through the right SNSs for the right type of behaviour. Indeed, understanding whether a specific SNS can enable or inhibit offline or online political behaviour can facilitate marketers' strategic choices with respect to the objectives of their political campaign. Also, understanding how use of different social media affects different types of Social Capital would further help Political Marketers for their targeting strategies, given that different types of Social Capital (Bonding/Bridging) have been found to affect Online and Offline Political Participation differently (Boulianne 2015; Skoric and Zhu, 2016).

However, although the focus of this thesis is on Political Marketing, its significance expands also to other sectors. This is because the positive implications of Social Capital for practice are well documented in the economics sphere (Aldridge et al., 2002; Halpern, 2001; Putnam et al., 1993), as well as in marketing (Antoniadis and Charmantzi, 2016; Jones and Taylor, 2012; Ritchie and Sridharan, 2007), for educational attainment (Aldridge et al., 2002; Israel et al., 2001), in public health and societal wellbeing (Coulthard et al., 2001) and even for problem solving (Boyte, 1995; Sirianni and Friedland, 1997). For instance, the UK government has a specific section of its website dedicated to Social Capital (https://www.gov.uk/government/statistics/social-capital-in-the-uk) which is regarded as a measure of national well-being and is used to inform decision-making in terms of resource allocation, public health, societal wellbeing, and cohesion (ONS, 2017). Understanding which SNSs could facilitate the development of Social Capital may introduce changes in government use of SNSs and push policies for the development and social affordability of one SNS over another (or maybe both) to improve social well-being and cohesion.

1.5. RESEARCH PROCESS AND METHODOLOGY

Following Bryman (2012) the research process of this study was divided into the following eleven phases: (1) carrying out the literature review; (2) establishing a theoretical framework; (3) defining the research strategy; (4) highlighting the research design; (5) contextualising the research method; (6) operationalising the constructs in the theoretical framework; (7) determining the appropriate statistical analysis; (8) determining sample characteristics and size; (9) data collection; (10) data analysis and (11) summarising the results and drawing conclusions.

Briefly, to fulfil the aim of this study and answer the research questions presented in Section 1.3, this research used a quantitative research strategy employing a crosssectional design which is commonly adopted in the field of SNSs, Social Capital and Political Participation studies (Boulianne, 2015; Skoric et al., 2016). Data was collected through an online self-completion survey hosted on Qualtrics. The questionnaire was piloted and designed to minimise common method bias, social desirability and acquiescence biases following the recommendations of MacKenzie and Podskoff (2012). The measures employed to operationalise the concepts included in the proposed theoretical framework originate from previous peer-reviewed academic articles. The use of well-established scales increases the reliability of the research and facilitates comparability and replicability of the study, while also establishing face and measurement validity (Bryman, 2012). The sample was randomly selected from an online consumer panel owned by the data collection agency Pureprofile, resulting in 1212 responses. The use of a web panel was considered appropriate because it helped to reach online people that were the target population of this research (Rea and Parker, 2014). PLS-SEM Moderation analysis with the SmartPLS software (Hair et al., 2017) was employed to answer research questions Q1, Q2, Q3 and relating sub-questions. Parametric two-way ANOVAs in SPSS (Field, 2013) and robust two-way ANOVA in R (Luepsen, 2017) were employed to answer, respectively, research questions Q4 and Q5 and relating subquestions (Table 5.2 in Chapter 5).

1.6. RESEARCH STRUCTURE

As mentioned above, this study follows a 11 stages approach to research (Bryman, 2012) which are distributed across 8 Chapters (Figure 1.3).

Chapter 1 introduces the research problem, relevance, aims, theoretical framework, methodology and structure used to meet the research objectives.

Chapter 2 provides a systematic literature review of Political Marketing following its historical evolution. The review highlights reasons behind the need for a shift in both practice and theory, towards a more "*comprehensive*" and "*holistic*" approach to Political Marketing. This approach is oriented at a societal welfare perspective which fits the aims of this research and advocates the inclusion of concepts like Social Capital in Political Marketing Studies.

Chapter 3 reviews the current debate in academia around the issue of *'participation'* according to the three main fields at the centre of this work, namely *Ethics*, *Democratic Theories* and *Media Studies*, and their link with Social Capital. A theoretical and critical excursus of those fields helps to define Political Participation and to identify gaps related to media studies and the use of SNSs in politics.

Chapter 4 provides an account of the extant Social Capital literature and its relationship with the Internet, SNSs and Political Participation studies. Criticisms and different theoretical points of view over the content of the construct of Social Capital are provided. A clear stand in the light of the aim of this study is chosen and supported by theoretical arguments, with the final aim to provide a definition which will inform the operationalisation of Social Capital in this research. Gaps in the extant literature of Online Social Capital and its relationship with SNSs and Political Participation studies are provided.

Chapter 5 harnesses the specific literature relevant to the development of the conceptual model and presents the theoretical framework and relative constructs, together with the research questions.

Chapter 6 outlines and justifies the research approach, design and methods of sampling, data collection and data analysis used. This chapter presents the operationalisation of the constructs employed in the proposed model and provides details of the development and structure of the questionnaire, pinpointing the steps taken to enhance the reliability and the validity of the research.

Chapter 7 details the process of preliminary data analysis including data screening, descriptive statistics, and tests for common method bias. Then, it provides results and discussion of the analysis carried out to answer the proposed research questions.

Chapter 8 presents a summary of the findings, followed by a discussion of the research questions. It highlights the theoretical and practical implications of this study and its limitations, with recommendations for future research.

Figure 1.3. Outline of the Thesis.

Chapter 1	Introduction			
 Introducing the research rationale, aims, questions, methodology and structure. 				
Chapter 2	Literature Review on Political Marketing			
•Stage 1 - Carrying out the	e literature review> Political Marketing and the Comprehensive			
	Approach towards a Societal Welfare Definition .			
Chapter 3	Literature Review on Political Participation			
•Stage 1 - Carrying out the	e literature review> Political Participation in Ethics, Theories of			
	democracy and Media studies. Relationship with Social Capital.			
Chapter 4	Literature Review on Social Capital			
•Stage 1 - Carrying out the	e literature review> Social Capital Online in Media and Communication			
	studies. Relationship with Political Participation.			
Chapter 5	Research Aims, Questions and Model			
•Stage 2 - Establishing res	earch aims, questions and theoretical framework.			
Chapter 6	Research Startegy, Design, Methods and Techniques of Analysis			
•Stage 3 - Defining the res	search strategy> Quantitative Approach.			
•Stage 4 - Highlighting the	e research design> Cross-Sectional.			
 Stage 5 - Contextualise t 	he research method> Online Survey.			
 Stage 6 - Operationalisin 	g the constructs> Measures employed in previous studies.			
•Stage 7 - Determining the appropriate statistical analysis> PLS-SEM - Two-Way ANOVA.				
•Stage 8 - Determining sample characteristics and size> Web Panel Sample (n = 1212).				
•Stage 9 - Data collection> Questionnaire hosted on Qualtrics.				
Chapter 7	Data Analysis, Results and Discussion			
•Stage 10 - Data analysis and Discussion> Questions Q1 - Q2 - Q3> PLS-SEM Moderation				
	Question Q4> Two-Way parametric Anova			
Question Q5> Two-Way non parametric Anova				
Chapter 8	Summary and Conclusions			
•Stage 11 - Summarising the results and drawing conclusions.				

Source: created by author following Bryman (2012) research stages.

1.7. LITERATURE REVIEW AIMS AND METHODOLOGY

This thesis follows a deductive approach to research. Therefore, the scope of the literature review is to identify gaps in the extant literature and to pinpoint a theoretical framework under which define concepts to answer the proposed research questions (Bryman, 2012). This study relies on the three interrelated concepts of Political Participation, online Social Capital and online Social Networks, to build a theoretical framework in line with the research topic. The need to link those different concepts coming from different disciplines like Political Science, Sociology and Media Studies is dictated by the desire to overcome the criticisms to Political Marketing of not adequately engaging with ethical questions, theories of democracies and sociological perspectives (Henneberg, 2004; Henneberg et al., 2009; O'Shaughnessy, 2002; Ormrod et al., 2013).

Given the multidisciplinary and broad nature of this research, an interdisciplinary review of the literature was conducted. Indeed, the research is underpinned by four streams of literature, namely Political Marketing, Political Participation, Social Capital and SNSs use as illustrated in Figure 1.4. Notably, the SNSs use stream underlies the other three streams and is incorporated in their respective Chapters. Hence, the literature review is organised in three main Chapters (see Section 1.6.) and follows a *funnel approach* (Webster and Watson, 2002). For each Chapter, and corresponding literature stream, gaps are identified.

The literature has been conducted using a multi-keyword search initially employing Google Scholar, JSTOR and ProQuest databases and later focusing on the relevant academic journals identified through the databases. Specifically, ProQuest was employed because it is considered the leading literature review database (Levy and Ellis, 2006), while Jstor is regarded as one of the most trusted source for academic publications (Jstor, 2016). Key words and academic journals used for each stream of literature are reported in Appendix A.



Figure 1.4. Flow chart overview of the literature underpinning the research.

Source: created by author.

2. LITERATURE REVIEW ON POLITICAL MARKETING

The literature review on Political Marketing follows a historical evolution of its theoretical conceptualisation linking it to theories of Political Science, Sociology and Media Studies. The aim is to provide a theoretical framework which engages with those disciplines and through which Political Participation can be approached. This is because:

- many scholars have fostered a great degree of historical reflection (Dholakia 2012a, 2012b; Fullerton 2011; Hunt, 2010), arguing that history not only may add context and richness to our understanding of a field, but most importantly, it helps scholars avoid unnecessary repetition in research (Tadajewski and Saren, 2008) and frame the contribution to Political Marketing knowledge that we make.
- 2. So far, Political Marketing research has been informed and led mainly by mainstream marketing concepts originating from economic theories and has somewhat neglected the importance of values, ethics and ideology in the political sphere and in society at large (Henneberg, 2008; Lees-Marshment, 2019a, b; Ormrod et al., 2013; Savigny and Wring, 2009; Taylor, 2006; Yoon et al., 2005).
- 3. The relevance of Political Science and Social Studies to Political Marketing is confirmed by O' Cass (2009) who claims that "Two activities politics and marketing often dominate the social psyche of many societies, having a major influence over individuals and the broader citizenry" (p. 190) and by Butler and Harris (2009) for whom "the significance of politics and marketing [...] for societies demands the integrated engagement of research in both" (Butler and Harris, 2009: p. 149).
- 4. As far as Social and Media Studies are concerned, scholars acknowledge that those discipline have exercised a fundamental influence on Politics and especially on Political Communication studies leading these two fields to investigate attitudes, beliefs and opinions of voters (Bennett and Iyengar, 2008; McNair, 2017; Ryfe, 2001). Indeed, most of the earliest and important researchers in communication (among the others Lasswell, Lazarsfeld, Lewin and Hovland) were trained in the social psychological tradition and largely influenced Political Communication studies to the point that "*the two most common instrument of data*"

collection in political communication, surveys and experimental studies, were developed by social psychologists" (Ryfe, 2001: p. 409).

Importantly, the inclusion of other disciplines is dictated by a general concern over the "democratic malaise" resulting from a decline in civic engagement and participation of citizens in the political process. Arguably, Political Marketing as it is understood nowadays and the marketisation of politics with no regards for social or ethical implications are some of the several causes which have fostered such "democratic malaise" (Savigny and Wring, 2009). Furthermore, the inclusion of the aforementioned disciplines in studies of Political Marketing is in line with the concept of Comprehensive Political Marketing (CPM) suggested by Lees-Marshment (2001; 2003), who suggests that the study of Political Marketing should encompass the political and management science literatures, allowing for greater cross-disciplinary research. Arguably, an interdisciplinary research approach to Political Marketing is required (Perannagari and Chakrabarti, 2020) and "certainly makes for a more complex mission, but a comprehensive engagement offers real theoretical reward" (Butler and Harris, 2009: p. 154). Indeed, interdisciplinary may represent almost a forced choice for Political Marketing, since "the significance of politics and elections for societies demands the integrated engagement of research in politics and marketing" (Butler and Harris, 2009: p. 149) and other relevant disciplines like Sociology and Communication Studies (Ormrod et al., 2013). Yet, "existing literature has neglected the interdisciplinary nature of the discipline" (Perannagari and Chakrabarti, 2020: p. 10).

For all mentioned so far, the following Sections, provide a portrait of the development of the understanding of the nature of Political Marketing, with a specific focus on the emergence of Political Marketing as an academic discipline and political practice strategy, thus spanning from the original broadening of marketing debate introduced by Kotler and Levy at the end of 60s, to contemporary definitions which highlight differences between mainstream marketing and Political Marketing. Through this process the need for a shift in the conceptualisation and research practices of Political Marketing is justified arguing that "*Political Marketing in its current form is informed by a set of economic assumptions that are antithetical to democracy and serve to disconnect the public from the process of politics*" (Savigny, 2008: p. 29). This historical approach

will engender a definition of Political Marketing which is better suited to the focus of this study based on relationships and interactions among people, with an emphasis on Political Marketing at a service to society and to the "common good". The final aim is to identify theoretical and practical gaps in Political Marketing research. Figure 2.1 provides a conceptual map of this Chapter.





2.1. EVOLUTION OF POLITICAL MARKETING

The concept of Political Marketing is strictly correlated with the "marketing of persons, organizations and ideas" which stemmed from the broadened concept of Marketing introduced by Kotler and Levy (1969) claiming that "*marketing is a pervasive societal activity that goes considerably beyond the selling of toothpaste, soap and steel*" (p. 10). Up to that moment, and from a theoretical standpoint, Politics was a term associated with ideas and ideologies rather than Marketing. However, Political Marketing as a practice, or as the art of persuading and spinning (Jones, 1997), or as the enactment of presentation and "*doing what will play well with the electorate*" (Palmer, 2002: p. 354) or as the packaging of politics (Franklin, 1994) could go back to Machiavelli (Harris, 2001) or even Aristotle (Butler and Harris, 2009; Savigny, 2009). Therefore, in practice, there has always been a Marketing of political people (Kotler, 2005) and marketing techniques in politics were already observable before the broadened marketing debate, but they remained practices without an explicit framework in the general political strategy of candidates and parties.

However, even nowadays it appears difficult to provide a clear framework for Political Marketing since there is no agreement on its definition (Lock and Harris, 1996; O'Cass, 2001; Ormrod et al., 2013; Scammell, 1999). Therefore, the following Sections present a historical account of the evolution of Political Marketing with the aim of clarifying any confusion and to provide a better understanding of what it was, what it is and what it could be.

Particularly, the following sections show how Political Marketing has experienced several transformations in its conceptualisation, switching from a *narrow* view, where it was regarded as a mere process in its *instrumental/tactical function*, to a more general *wide holistic approach*, according to which Political Marketing is regarded as an organisational philosophy (Henneberg 2002) rather than as a mere transposition of Mainstream Marketing (Maarek, 2007). The change from the narrow approach 'era' and the wide holistic approach 'era' occurred through a transition period that in this thesis will be identified as the "Between Act" era. Table 2.1. provides a summary of the main features of each era, followed by a detailed discussion in the next three Sections.

NARROW APPROACH ERA	THE BETWEEN ACTS ERA	WIDE HOLISTIC APPROACH ERA
1976 - 1995	1996 - 2003	2004 -2020
NORTH AMERICAN SCHOOL	SCANDINAVIAN SCHOOL	EUROPEAN SCHOOL
INSTRUMENTAL/TACTICAL FUNCTION	MANAGERIAL/STRATEGIC FUNCTION	ORGANISATIONAL PHILOSOPHY
G-D LOGIC	RELATIONSHIP LOGIC	S-D LOGIC
TRANSACTIONAL APPROACH	INTERACTION APPROACH	NETWORK APPROACH
VOTERS ORIENTATION	VOTERS AND RELEVANT STAKEHOLDERS ORIENTATION	SOCIETY AT LARGE
ATTRACT NEW VOTERS	RETAIN EXISTING VOTERS	ESTABLISH MANTAIN AND ENHANCE RELATIONSHIP WITH SOCIETY
SHORT-TERM ORIENTATION	LONG-TERM ORIENTATION	LONG-TERM ORIENTATION
FOCUS ON ELECTORAL CAM- PAIGN	PERMANENT CAMPAIGN	PERMANENT CAMPAIGN
"ADOPTION" OF MARKETING TOOLS "IMITATION"	"ADOPTION" OF MARKETING TOOLS AND CONCEPT	"ADAPTATION" "CREATION"
VALUE DISTRIBUTION	VALUE CREATION	VALUE CO-CREATION
MASS COMMUNICATION	TARGET COMMUNICATION	SELECTIVE/PURPOSIVE COMMUNICATION PUSH/PULL
REINFORCEMENT FUNCTION	MOBILISING FUNCTION	MOBILISING FUNCTION
MEDIATED /UNILATERAL COMMUNICATION	MEDIATED/BILATERAL COMMUNICATION	DIRECT BIDIRECTIONAL COMMUNICATION
TV/RADIO/NEWSPAPER	TV/RADIO/NEWSPAPER	INTERNET/TV/RADIO/ NEWPAPER
MOBILISING MASS	MOBILISING POLITICALLY ENGAGED	MOBILISING POLITICALLY ENGAGED/DISENGAGED
INTELLECTUAL CAPITAL	POLITICAL CAPITAL	SOCIAL CAPITAL
DEONTOLOGICAL/ CONSEQUENTIALIST ETHICS	DEONTOLOGICAL/ CONSEQUENTIALIST ETHIC	VIRTUE ETHICS CIVIC VIRTUE COMMUNITARIANISM
REPRESENTATIVE DEMOCRACY	REPRESENTATIVE DEMOCRACY	DIRECT DEMOCRACY

Table 2.1. Political Marketing Eras.

Source: created by author following Ormrod et al. 2013.

2.1.1. POLITICAL MARKETING AS A "MERE PROCESS": NARROW APPROACH ERA (1976-1987)

A first theoretical formalisation of Political Marketing is provided by Shama (1976) who regarded it as a "*process*" (p. 766) based on transactions happening at election time between political candidates and voters as unique stakeholders of the political market.

This transaction-based approach to Political Marketing, or what is called "Narrow approach" (Henneberg, 2008), focused on the similarities with commercial marketing, mirroring theories and underlining the use of marketing tools in election campaign based on the Marketing Mix Management paradigm (MMM) (O'Leary and Iredale, 1976). The MMM Theory (McCarthy, 1960) is commonly referred to as Transaction Marketing because, in exchange terms, it focuses on frequent, low-risk, singular purchases and new customers acquisition rather than "customer loyalty" or longterm relationships. Most importantly, the MMM concept was rooted in a "Good logic", where the production and consumption processes were independent from each other. As it will be argued later, this separation between production and consumption does not adequately fit politics, which is more similar in nature to the service industry, where production and consumption are not necessarily independent and separate functions (Ormrod et al., 2013). Moreover, Shama's definition was rooted in the marketing concept of that time for which "the consumer, not the company is in the middle" (keith, 1960: p. 35). Such concept was translated into Political Marketing terms, suggesting that voters were at the centre of the political process and as such they were the only fundamental stakeholder which candidates should have listened to and targeted, providing them with the product they wanted, in order to win elections (Shama 1976; Mauser, 1983). This is in line with the concept of "Market Orientation" proposed by Shama (1976), who argued that in the 1970s politicians can be seen to acquire such form of advanced campaigning focused on voters rather than on candidates' and parties' as it was in the period 1940-60 with the "Candidate Orientation", or in the 60s with the "Sales Orientation". For Shama (1976), such shift was technologically rooted in the large diffusion of television in the houses of people and mirrored theories in Political Communication (Chaffee, 1975). Indeed, the political agenda at that time was mainly concerned with election campaigns, the messages conveyed to citizens and the television as the principal mean for meeting voters. Put it in Laswell (1948) terms, Political Communication at that time was still involved in analysing "who/says what/in which channel/to whom/with what effect" (p. 37).

The focus on "Electoral Marketing" was essentially due to a shift from a perspective in which campaigns could not persuade voters, to the principle that campaigns matter, as they could "enlighten the public" (Gelman and King, 1993) and shape voters' preferences (Campbell, 2000).

This was in line with the "*Realist*" view of democracy provided at that time by the political analyst Andrew Gamble (1974) who considered voters as the fundamental stakeholders in the political market and as passive receivers of unidirectional messages created by political parties and their candidates to market the value created by their offerings. This was in line with Political Communication and Sociology theories, which regarded the television, the most used media of communication in electoral campaigns at that time, as a centralized, mainly unilateral means of communication, antagonistic to interpersonal networks (Klapper, 1960).

Therefore, in this "Era" Political Marketing adopted Marketing Techniques and mirrored developments in Commercial Marketing Theory, like McCarthy's (1960) Marketing Mix, adapted by O'Leary and Iredale (1976) to the political marketplace. Moreover, Political Marketing was mainly influenced by the political communication theory rooted in the antagonist effect of Media and was centred on the concept of "persuasive campaign" dominated by the technological developments of that time (large diffusion of TV use).

However, Bagozzi (1975, 1978), argued that as marketing lays on the general idea of *exchange*, social relationships should do the same. In fact, although there is a lack of monetary exchange, social exchange involves the symbolic transfer of both tangible and intangible entities. Following Bagozzi's suggestion, Farrel and Wortmann (1987: p. 298) explicitly embedded the notion of exchange into Political Marketing, defining it as "*a system of exchange...of something of value*" (Table 2.2). However, Farrel and Wartmann's (1987) definition, still focused on election time, where voters are the main exchange partner and promotional activities prevail in a perspective of tactical/instrumental function of Political Marketing that seems to restrict the scope of analysis to communication instruments (Butler and Collins, 1996).

The idea of marketing in Politics started to be recognised and be accepted in this era in a process that lasted 9 years, starting from the formalization of Shama (1976) and culminating in the attempt of the American Marketing Association (AMA) in 1985 to integrate social and political issues in the mainstream marketing thought, by adding the word "Ideas" to the definition of marketing:

"Marketing is the process of planning and executing the conception, pricing, promotion and distribution of ideas, goods and services to create exchanges that satisfy individual and organizational objectives" (cited in Fine 1992: p. l).

Although through such definition the AMA rebutted the simplistic belief that marketing was just about firms providing their customers with what they wanted, it was still regarded as a process focused on tactical activities like pricing, promotion and distribution, which not only reduces the variety of instruments available in the Political Marketing Mix, but also ignores its strategic element (Harrop, 1990; Wartmann 1989). Furthermore, "placing the formal campaign as the central issue is akin to focusing only on the sale negotiation as the core of marketing" (Butler and Harris, 2009: p. 151). On the basis of the MMM paradigm, for which there were always new segments of the market to target, this short-term orientation of parties within the political marketplace was unsurprising. Yet, because of its narrow consideration of Political Marketing and thanks to the progressive affirmation of the new "service" logic, the Instrumental/Managerial school experienced strong criticisms, which progressively led to the affirmation of a different conceptualisation of Political Marketing. Also, the spread of new forms of more direct types of democracy, made possible by creation and diffusion of more interactive means of communications like the Internet, contrasted the representative democratic system of this period and the unilateral communication process with a passive receiver. Consequently, a change in Political Marketing theories and practice was on the way.

2.1.2. POLITICAL MARKETING: THE BETWEEN ACTS ERA (1988-2003)

As mentioned in the previous Section, the MMM paradigm, being closer to the G-D rather than to the S-D logic, not only does not fit well the political marketplace but also creates some paradoxes.

The first paradox stems directly from the MMM paradigm itself, which makes the narrow approach too technical and incomplete (Gummesson, 1987). According to Dixon and Blais (1983), the MMM paradigm provided a manipulative approach to consumers who were seen as "fish waiting for a bait" (Gummesson, 1999). In Political Marketing terms, this implied for parties to be distant from voters since they did not listen to their voice, but rather tried to convince them of the validity and goodness of their offers (bait). Therefore, most political parties were not just interested in adjusting to social changes, but rather they would try to influence the direction and intensity of such changes (Johansen, 2005). This meant that Political Marketing, as conceived through the MMM approach, was prone to ignore the importance of voters' participation in the political process and it was likely to contribute to the deprivation of voters of a sense of joint responsibility for the democratic process and outcome (Scammell, 2003).

The second paradox is that, from a political perspective, the MMM paradigm created distance and centralization of power structures (Dalton and Wattenberg, 2000). The MMM paradigm does not embed any (or little) notion of participation or co-creation of value with respect to customer and this, translated into political terms, implies that parties move away from voters, which in turn lose their influence over policy and candidate selection processes and regard politics as a politicians' responsibility consequently decreasing their participation as volunteers and or campaign activists (Dalton and Wattenberg, 2000; Scammell, 2003). This translate into civic and political disengagement (Putnam, 2000) which, from a marketing perspective, is equivalent to loss of market share.

Therefore, in marketing terms, if parties want to gain electoral market shares, they must increase voters' participation and turnout. It is in this context that the concept of Relationship Marketing (RM) started to emerge in services marketing, as proposed from a group of Scandinavian scholars (Gummesson, 1987). According to this paradigm the focus is on "value creation" rather than on "value distribution" (as it was in the MMM

paradigm), which means that value for consumers is created by the "interaction" between them and the firm in the production process (Grönroos, 2000). Hence, the customer is no longer considered as a passive "value destroyer". This process implies the establishment of long-term relationships with customers, a concept not new in Political Marketing. In fact, Blumenthal (1980) developed the concept of "Permanent Campaign", later revisited by Klein (2005), for which governing with public approval requires a constant political campaign aimed at building long-term relationships with stakeholders at large. With the affirmation of this approach, interaction becomes a central marketing concept taking over exchanges, which remains still important but could not exist without good interactions (Grönroos, 2006). It is in this context that Lock and Harris (1996, pp. 21-22) provided a definition of Political Marketing which took into consideration not only exchanges of value, but also interactions at both aggregate and individual level between political organisations and the environment in which they operate. This constituted a first step towards a "wide interpretation" of Political Marketing, by looking closely at strategies for positioning, communication and interactions not only with voters but with the environment in general (Table 2.2). However, it is only with Henneberg (2002) that the long-term nature of such relationships and the notion of mutual interaction make their appearance, mirroring RM's definitions (Table 2.2). Moreover, Henneberg's (2002) definition takes into consideration more than one stakeholder, contrary to the transactionbased approach. Specifically, Henneberg (2002) mentions "society" at large which is in line with the reasoning that "a priori linking of specific stakeholder groups to the definition of Political Marketing can be too restrictive" (Ormrod et al., 2013: p. 17). This is mainly because political actors must rely on more than just the favour of voters at election time (Kotler, 1975; Newman, 1994). Rather, they need to interact and exchange value with competing candidates/parties, mass media, members of their own parties, private and public corporation and so forth (Newman, 1994; Henneberg, 2002; Ormrod, 2005, 2007; Dean and Croft, 2001) and not just at election time.

From a relationship marketing perspective, even the focus on mediated communication started to be questioned (Swanson and Mancini, 1996; Blumler and Kavanagh, 1999), because the media were seen as a filter between politicians and voters, which increased the gap between the two, fostering the idea of politics as politicians' responsibility which, ultimately, encouraged disengagement (Webb, 2000). On the contrary, RM in politics should try to fill the gap between services provider (parties) and

users (voters) by establishing an image of trustworthiness through the fulfilment of promises and creation of intimacy (Ormrod et al., 2013). This mirrors the assumption of Grönroos (1994) in marketing literature for which "a *firm pursuing a relationship marketing strategy.... develops over time more and tighter ties with its customers. Such ties may, for example, be technological, knowledge-related or information-related, or social in nature. If they are well handled they provide customers with added value, something that is not provided by the core product itself. Of course, price is not unimportant but is often much less an issue here. Thus, relationship marketing makes customers less price sensitive" (p.11).*

Ultimately, developing long lasting relationships, RM represented the opportunity for parties to increase Political Participation and social involvement of citizens (Ormrod et al., 2013). In fact, the involvement of the public in political activities is relevant to the concept of *value-consumption* developed by Dermody and Scullion (2001: p. 1088). Accordingly, it is no more the transcendent value of a product that determines its value, rather the customer experience in consuming it. This attribution of value applied even to non-tangible experiences like the voting experience itself (Dermody and Scullion, 2001). Therefore, Political Marketing must be seen as a "*process of signification and representation*" (Dermody and Scullion, 2001: p. 1090) which "*can make politics meaningful to the electorate by understanding the different meanings ascribed to political consumption*" (p.1095).

The above discussion brings to light implications for political communication, "*particularly in the dyadic creation of message meanings between sender and receiver*" (Dermody and Scullion, 2001: p. 1095). Indeed, exchanges were not anymore characterised by one-off transactions between an active and creative seller and a passive customer (Gamble, 1974), but rather by multiple exchanges with active people (Ford et al. 2003). Such multiple exchanges were facilitated also by the affirmation and diffusion of the Internet as a mean of communication. Indeed, the many actors involved in the political communication process could personally and more easily interact through blogs, websites, e-mail and chats. Communication was faster and no longer unilateral, but rather personal, symmetrical and two-sided (Wellman, 2004).

In the Political Science field, this raised hope also for more direct and open forms of democracy in Habermas' (1989) terms (see Chapter 3). Indeed, through the Internet,

citizens had potentially equal opportunities to express themselves, to hear and to be heard through direct discussion and deliberation on virtual "*agora*". Yet, such "possibility" remained a hope because the penetration and diffusion of the Internet among the population was still very limited and those that had access to it were the richest and most educated who already participated intensively in politics (Wellman, 2004). Therefore, the Internet at this stage seemed to aggravate the political participatory gap between engaged and non-engaged citizens to the point that Margolis and Resnick (2000) stated that: "*There is an extensive political life on the Net, but it is mostly an extension of political life off the Net*" (p. 14). Hence, in terms of communication studies, the media were still seen as reinforcement agent (rather than as a mobilising one) able to replicate and strengthen participatory patterns already established offline.

Moreover, at this stage, Political Marketing research was still embedded in the ontological assumption that parties could be considered as businesses and voters as consumers. Such economic assumption has led to several criticisms and is regarded as one of the major factors contributing to social and political disengagement (Section 2.3). Hence, we can speculate that in this Era, Political Marketing was not seen under a positive light.

2.1.3. POLITICAL MARKETING: THE WIDE APPROACH ERA (2004 - 2020).

Although Political Marketing was initially embedded in the transaction-oriented framework (O'Shaughnessy, 1990), gradually academics started to accept the relationship-based approach proposed by Grönroos (1990), with the belief that this shift would have helped research in Political Marketing (Bannon, 2005). However, most recently, a new dominant logic stepped into the Marketing field; as "*markets and the study of marketing continually change and adapt to new empirical realities*" (Butler and Harris, 2009: p. 158) so must do Political Marketing. Such new logic was proposed by Vargo and Lusch (2004) who questioned the theory of marketing based on exchanges of value with a new perspective which emphasises co-creation of value, intangible resources and sustainable relationships. Shifting from a G-D to a S-D logic, Lusch and Vargo (2006) suggested rebuilding marketing on the human knowledge, skill, information, connectivity and relationship within human society. RM is considered by them as a steppingstone

between the G-D Logic and their S-D Logic. This is the reason why we considered RM at the basis of the "Between Acts Era".

Some new aspects of the S-D logic framework have grabbed the attention of Political Marketing scholars.

Specifically, the S-D Logic discards any distinction between goods and services and regards the latter from a perspective of value rather than as a category of market offering (Edvardsson et al., 2005), that is, service is considered as the foundation of exchange with all stakeholders (Vargo and Lusch, 2006). In political terms, this implies the extension and nurture of relationships with society at large and not just with voters. Such emphasis on society at large is fostered also by the AMA which in 2007 defined marketing as: "*the activity, set of institutions and processes for creating, communicating, delivering and exchanging offerings that have value for customers, clients, partners and society at large*" (cited in Hughes and Dann, 2009: p. 244).

Furthermore, the notion of 'offerings of value' appeals particularly to Social and Political Marketing. In fact, the S-D logic assumes that the value of an exchange may only be determined by the user through the consumption process, thus recognising that businesses do not provide value in the exchange process per se, rather they make value propositions to customers (Vargo, 2008). This last concept fits particularly well the political context, where political parties' offerings represent a proposition of value to voters which can be implemented only if the political exchange is successful in the parliamentary and governmental marketplace where politicians can fulfil their promises through the implementation of laws and measures concretising the value of their offer propositions (see Henneberg and Ormrod, 2013). In other words, value creation occurs "when a potential resource is turned in a specific benefit" (Lusch et al., 2008: p. 8) and it is created from "economic actors who exchange a variety of resources that go beyond good and money" (Michel et al., 2008: p. 154). In this sense, the S-D logic embraces the Resource Based View (RBV) and points towards intangible dynamic resources (Lusch and Webster, 2011), a concept that fits well the political context. Lynch et al. (2006) and O' Cass (2009) considered this concept from a Political Marketing perspective, concluding that when a political party has the capacity to deploy tangible and intangible resources and possesses good organisation processes, it has a competitive advantage over others political actors.

A further striking point of the S-D logic is communication, in here it implies conversation and dialogue which are fundamental elements for building long-term relationships in the political marketplace (Ballantyne and Varay, 2006). Accordingly, "promotion will need to become a communication process characterized by dialogue, asking and answering questions" (Vargo and Lusch, 2004: p. 13). Although not explicitly mentioned by Vargo and Lusch (2004), it seems logical to extrapolate that the Internet and Social Media are required both as a technology and as a methodology, in order to effectively exploit the networking service dominant logic. Based on such networking logic, "the final force that has created an inflection point in the movement toward collaboration is network ubiquity. Increasingly, everyone and everything is connected to each other and each thing" (Lusch et al., 2007: p. 10) and: "It is especially important with conversation and dialogue that all parties are transparent and truthful with each other. This approach to communication should include not only customers, but also employees and other relevant stakeholders that may be affected by service exchange. All stakeholders need to be part of the market dialog. Thus, marketing conversation will (should) occur as an integral part of the marketplace – the market itself will be part of the conversation. Thus, the marketplace itself becomes part of the marketing which no single firm can control". This last argument is in line with Gueorguieva's (2008) argument about the potential of the Internet in allowing people to create their own contents, thus, reducing the level of control politicians have over their own image or that of competitors.

Arguably, the S-D logic should prove theoretically attractive to Political Marketing (Butler and Harris, 2009). A two-way dialogue, a direct interaction without filters, the possibility for people to influence standards of service production and participate in the process of co-creation of values should appeal to Political Marketing, especially if considered as philosophy at the service of society.

Furthermore, all these concepts, especially co-creation of values and direct interaction, appeal also to a participatory kind of democracy (see Chapter 3) which is often seen as one of the means through which to cure the "*democratic malaise*" (Savigny and Wring, 2009). Particularly, in political terms, for co-creation of values and significant interaction to happen, the actual low levels of Political Participation must be increased. Hence, the new S-D Logic may be seen as a means through which to achieve such outcome. Therefore, if Political Marketing embraces such logic it could also be

appreciated as a means of improving Political Participation and therefore as an instrument aimed at a societal benefit.

2.2. TOWARDS A SOCIETAL WELFARE AIMED DEFINITION

Although, over the years, Political Marketing Theory has been mirroring developments in mainstream Marketing (Henneberg, 2002; Baines et al., 2003), Ormrod et al. (2013) argue that several fundamental differences exist between political and commercial marketing. As Dean and Croft (2001) state: "*traditional marketing frameworks do not fit neatly into a Political Marketing configuration*" (p.1197) and this "*natural diversity of political contexts, structures and processes complicates the process of theory development in Political Marketing*" (Butler and Harris, 2009: p. 149). Arguably, a shift from "imitation" and "adoption" towards "creation", "integration" and "adaptation" is needed in Political Marketing theory and research practice.

However, until now, research in Political Marketing has been carried out on epistemological and ontological assumptions coming from the economic field (Savigny and Wring, 2009) and few studies have taken into consideration Political Marketing as a discipline whose aim should be that of benefiting society. Arguably, this is one of the reasons that compels Political Marketing to defend and justify itself from critics (see Section 2.3). However, this is not the only reason and any "guilt by association suspicion of Political Marketing research probably derives from a perceived connection with dubious marketing practices both of politicians and businesses" (Butler and Harris, 2009: p. 150). Therefore, not only a theoretical shift from adoption to adaptation (Ormrod et al., 2013) and from imitation to integration (Lees-Marshment, 2003) is needed, but also a change in Political Marketing practices from questionable and suspicious, to good and beneficial for society, is advocated by scholars.

These changes may path the way to a new direction for Political Marketing, regarded as a means for better democracy (Quelch and Jocz, 2007). Therefore, is essential to provide a definition which originates from Political Marketing Theory, but that at the same time reflects the influence of Political Science, Sociology and Communication studies over the years, so that to highlight advances in research and start to consider Political Marketing as beneficial for society. The proposed definition builds upon several

historical definitions provided over the years (Table 2.2) and tries to strike a balance between the needs of politicians, marketers, citizens and society at large:

"Political marketing is a perspective from which to understand phenomena in the political sphere (Ormrod et al., 2013: p. 18), which includes those "set of activities, processes or political institutions used by political organizations, candidates and individuals" (Hughes and Dann, 2009: p.244) and is "concerned with reciprocated exchanges of value with their environment" (Winther-Nielsen, 2011: p. 29) "through interactions in the electoral, parliamentary and governmental markets"(Ormrod et al., 2013: p. 18) in order to " establish, maintain and enhance long-term political relationships" (Henneberg, 2002: p.103) "in ways that benefit" (AMA, 2004) "society at large" (Henneberg, 2002; Hughes and Dann, 2009).

The last part of the definition is the most relevant to this study since it may help to overcome criticisms to Political Marketing (Section 2.3). The argument is that people could really benefit if the political and public realms were guided by the best of Political Marketing, which should be a discipline led not only by economics and marketing principles and frameworks but also by Political Science and Social Theories. Undoubtedly, since Political Marketing is mostly grounded on economics principles, might alter the basis of democracy and therefore attracts a wide array of criticisms (Savigny and Wring, 2009). However, this is particularly likely if Political Marketing is mainly (or only) thought of as based on mainstream marketing and directed at satisfying personal needs, wants and preferences, ultimately reinforcing materialism and individualism (Quelch and Jocz, 2007).

Therefore, in the following discussion, marketing will not be considered as an economic function or as a business practice but rather as a social force aimed at societal welfare (or the benefit of society at large). This concept is nothing new in the Marketing discipline given that researchers have often tried to understand marketing's role in improving societal welfare. In fact, as early as 1934 Breyer claimed that "*marketing is not primarly a mean for generating profits for individuals. It is, in the larger, more vital sense, an economic instrument used to accomplish indispensable social ends*" (cited in Quelch and Jocz, 2007: p. 7). From this societal perspective, marketing should contribute to improve standards of living and even foster national economic development (Kotler and Keller, 2012). This is a concept that must be particularly relevant to Political

Marketing, since the nature of the political process is such that the implementation of "promises of value" has consequences on society as a whole (Egan, 2005).

ERA	AUTHOR/S	DEFINITION
NARROW	Shama (1976)	"Political marketing is the process by which political candidates and ideas are directed at the voters in order to satisfy their po- litical needs and thus gain their support for the candidate and ideas in question" (p.766)
APPROACH	Farrel and Wortmann (1987)	"a system of exchange in which two or more actors each possess "something of value" that can be traded. The "sellers" offer representation to their "customers" in return for support" (p. 298)
	Harrop, (1990)	"political marketing is not simply an investigation of promo- tional activities such as press advertisements, party political broadcasts, and election addresses. It is a much broader area, covering whether, where, and how a party positions itself in the electoral markef" (p. 277)
THE BETWEEN ACTS	Lock & Harris (1996)	"the study of the processes of exchanges between political enti- ties and their environment and amongst themselves, with par- ticular reference to the positioning of both those entities and their communications dealing with strategies for positioning and communications, and the methods through which these strategies may be realized, including the search for information into attitudes, awareness and response of target audiences" (pp. 21-22).
	Henneberg (2002)	"political marketing seeks to establish, maintain and enhance long-term political relationships at a profit for society, so that the objectives of the individual political actors and organisa- tions involved are met. This is done by mutual exchange and fulfilment of promises" (p.103).
	Hughes & Dann (2009)	"political marketing is a set of activities, processes or political institutions, used by political organisations, candidates and in- dividuals, to create, communicate, deliver and exchange prom- ises of values with voters-consumers, political party stake- holders and society at large" (p. 244)
HOLISTIC APPROACH	Winther-Nielsen (2011) Ormrod et al. (2013)	"political marketing is concerned with reciprocated exchanges of value between political entities and their environment" (p. 29) "Political marketing is a perspective from which to understand phenomena in the political sphere, and an approach that seeks to facilitate political exchanges of value through interactions in the electoral, parliamentary and governmental markets to man-
		age relationships with stakeholders" (p. 18)

Table 2.2. Historical account of definitions of Political Marketing.

Source: Adapted from Ormrod et al., 2013.

In political terms, one of the ways through which this objective may be achieved consists in fostering high levels of participation of citizens in the political sphere. Therefore, with the help of new communication technologies, Political Marketing should focus on promoting and facilitating political and civic engagement among citizens. However, such perspective and function of Political Marketing have not been addressed adequately (if at all) in extant research. Arguably, this may represent a serious flaw which has brought several criticisms to the field, accused of not engaging consistently and effectively with Ethics, Political Science and Sociology (Henneberg et al., 2009; Ormrod et al., 2013; Perannagari and Chakrabarti, 2020; Quelch and Jocz, 2007; Savigny, 2008). For this reason, the next section presents an excursus over the criticisms brought by academics and practitioners in the fields of politics and marketing, with the aim of showing that a shift in both theory and practice of Political Marketing is necessary to overcome such criticisms.

2.3. POLITICAL MARKETING AND ITS CRITICISMS: THEORETICAL AND EMPIRICAL JUSTIFICATIONS FOR A NEW APPROACH

The basic premise to this study is that "*Political Marketing in its current form is informed by a set of economic assumptions that are antithetical to democracy and serve to disconnect the public from the process of politics*" (Savigny, 2008: p. 29). Therefore, marketing in politics represents an "*ideology of disconnection*" (Taylor, 2006).

Moreover, developments in Political Marketing theory have highlighted critical differences between political and commercial marketing (Baines et al., 2003; Butler and Harris, 2009; Henneberg and Ormrod, 2013; Lock and Harris, 1996) leading to the conclusion that "*traditional marketing frameworks do not fit neatly into a Political Marketing configuration*" (Dean and Croft, 2001: p. 1197). Therefore, theoretical concepts need to be adapted or integrated into the political context rather than adopted as they are (Lees-Marshment, 2001; Ormrod et al., 2013; Perannagari and Chakrabarti, 2020). Researchers must bring their studies to multidisciplinary level (Henneberg and O'shaughnessy, 2007) and adopt a more comprehensive approach to Political Marketing research, integrating concepts from different fields of study (Lees-Marshment, 2003).

This brings us to the first and most relevant criticism to this thesis that is, politics is not a marketing domain (Henneberg, 2004). Arguably, the development and use of a new approach based on "adaption" rather than an "adoption" of marketing concepts in the political sphere seems advisable. In fact, the straight application of marketing techniques to politics has been considered as unethical especially for "*those who believe that politics has a higher purpose than commercial profitability or who acquaint marketing with style rather than substance*" (Baines and Egan, 2001: p. 27), with "*packaging [...] spin [...] rather than political content*" (Ormrot et al., 2013: p. 61).

The second criticism to Political Marketing Theory and research regards the assumption that voters may be understood as consumers (Savigny and Wring, 2009), embedded in a rational choice context. Taylor (2006) argues that such assumption is "*an ideology - perhaps the ideology - of disconnection, of disintegration*" (p. 87) because it idealises a society where individuals are disconnected from the communities in which they live, from their cultural background and their values. Arguably, this criticism might be overcome by the adoption in Political Marketing of a *Communitarian* perspective to research through the concept of Social Capital (Chapter 3, Section 3.1.2 and Chapter 4). In fact, being a multidimensional concept, Social Capital includes in his analysis cognitive elements (shared codes, language and culture) relational aspects (trust, reciprocity and norms) and structural factors (networks) of a society which embed individuals in their social context.

Thirdly, the "*marketisation*" of politics, when closely tied to economic notions, promotes a concept of Political Participation strictly (but not solely) related to the mechanism of elections. Hence, participation comes to be regarded just as a mean through which political actors (competitors) may achieve their goals, namely winning the election and gaining power (Savigny and Wring, 2009). However, in political terms, participation means much more and represents the path towards policy legitimacy, empowerment of people and better democracy (Carpentier and De Cleen, 2008). Moreover, from a social perspective participation is not considered just as an instrumental means through which civic virtues may be fostered among citizens and the public good may be pursued over the self-interest. However, regardless of the perspective (the individualistic self-interested or the one aimed at the societal welfare) declining participation among citizens and consequently low levels of participation should be key areas of concern for politicians,

sociologists and marketers. Moreover, since the marketing of a "product" is affected not only by its brand identity in comparison with competitors, but also by the overall reputation and standing of the whole industry (Gummesson, 1999; Mortimore, 2002), low levels of participation and increasing disaffection towards politics are problems that involve the political system and its actors in general.

Many authors (Seyd and Whiteley, 1992; Shaw, 1994; Dalton and Wettenberg, 2000) ascribe such disaffection towards politics to the strict implementation of a "manufacturer good marketing" perspective into the political context which, in their view, has led to a strong centralization of powers structure. Centralised politics, inevitably, "requires the increased use of external professional consultants and market research" (Johansen, 2005: p. 99) which means that the focus of politics will be on mediated communication (Swanson and Mancini, 1996; Blumbler and Kavanagh, 1999), which in turn increases the gap between politicians and citizens since, especially in the past (Section 2.1.1), media represented a filter which exacerbated the distance between the two to the point that their traditional role of co-producers of policies and new representatives has been denied (Peterson and Godby, 2020). In other words, "people have effectively been deprived of the notion that they have some responsibility for the political outcome and have been led to believe that politics is the responsibility of politicians and that, rather than citizens, they are invited to act as self-interested shoppers on the look-out for service delivery" (Johansen, 2005: p. 98). The establishment of such marketing led ethos in the political marketplace has been facilitated by the affirmation of a liberal elitarian form of democracy (Savigny and Wring, 2009), for which the only role of voters is to elect rulers who will unilaterally establish what is good for society (Held, 2006). This is because citizens, from the elitarian perspective, are considered as individuals with limited knowledge and interest in political affairs (see Chapter 3).

The debate presented above underpins a narrow approach to Political Marketing (Section 2.1.1) and, arguably, is at the basis of low levels of Political Participation. However, scholars have considered many other factors as contributing to the democratic malaise in western countries. Many authors embrace a cultural and technological perspective (Lloyd, 2004; Norris, 2000) while others focus on apathy (Habermas, 1989; Henn et al., 2002; Marshall, 2001), declining levels of Social Capital (Putnam, 2000) or on the "Marketing-savvy" voters that perceive the marketing strategy "*behind the*

political campaign [...]as demeaning to the high principles of politics and democracy" (Butler and Harris, 2009: p. 151).

Considering what discussed so far, this study proposes a new perspective from which conducting Political Marketing research, which looks closely to the servicedominant logic with an emphasis on the improvement of citizens' life and the societal welfare in general, through the investigation of Political Participation and the use of the Internet as a means for a better democracy but engaging also with Political Science and Sociology and particularly with the concept of Social Capital. Therefore, the focus of investigation will not be on "what determines a citizen's vote" (Johansen, 2005: p. 102) but rather on what mechanisms may improve Political Participation in general. Indeed, examining the factors that determine a citizen's vote means focusing on "winning the single battle" (Johansen, 2005: p. 102) which implies addressing election and short-term relationships at the advantage of parties and political candidates. All these aspects are at the basis of the heavily criticised narrow approach to Political Marketing and are also considered as one of the causes of the democratic malaise experienced by western democracies. Rather, Political Marketing research should aim at a "holistic" long-term approach and should preserve "the benefit of society at large". Indeed, as the role of modern marketing is to improve people's lives (a concept particularly relevant to Social Marketing) so it should be for Political Marketing (Kotler et al., 2002). Indeed, focusing on mechanisms which may improve Political Participation, not just at election time, may be regarded as the attempt to improve societal welfare since high levels of Political Participation are deemed essential for a healthy democracy (Held, 2006). In turn, a healthy democracy is considered fundamental for a healthy society and its general welfare. This reasoning is in line with the definition of Political Marketing provided in this study. The main argument is that recent developments in marketing theory, particularly RM and the S-D logic, must serve as means for a new wave of conceptualisation which should take into consideration also social and ethical arguments in the political context and provide a route to a more participatory type of democracy (Chapter 3). This approach "could help bridge the conceptual gap between political science and marketing, as they both struggle to analyse contemporary democratic process" (Johansen, 2005: p. 102). However, as Ormrod and Savigny (2012) and Henneberg et al. (2009) argued, very few attempts have been made to integrate and adapt marketing concepts to politics in a perspective which considers implications for

democracy and, thus, society at large. Perannagari and Chakrabarti (2020) in their bibliometric meta-analysis of Political Marketing studies for the period between 1996 and 2018 further confirmed this trend. This, inevitably, has raised ethical concerns about the applicability of marketing activities in the political sphere (Lees-Marshment, 2019b), as summarised in Table 2.3.

Field	Ethical Concern
Policy	Marketing encourages a focus on short term actions the market wants now, rather than more effective long-term solutions.
	Emphasising the need to ensure policies appeal to the market reduces policy innovation.
Leadership	Market research encourages political leaders to listen to public opinion, which is problematic due to the lack of information, expertise, and objectivity.
	Consumerists voters do not judge governments fairly.
	Prevent politicians showing leadership on unpopular but socially beneficial policies such as climate change.
Representation	Knowing how to appeal to the most influential markets give
	some people more power than others.
	Marketing can encourage false and inauthentic stereo-typical
	appeals such as photo opportunities between politicians at cultural events or eating ethnic foods.
Citizenship	Treating voters like consumers discourages them to consider
	ethical, moral and community aspects.
	Marketing politics like a product encourages focus on policies
	that produce tangible outcomes (in the short-term), rather than what actually works (in the long-term).
Participation	Voters are more important than members, reducing the value
	of party participation.
Authenticity	Research leads candidates to change their policies, presentation
	and even personalities, rendering politics to a false shopping exercise.

Table 2.3. Ethical Concerns of Marketing Applications to Politics

Source: adapted from Lees-Marshment (2019b: pp. 251-252)

Moreover, the shift from a G-D logic to a S-D logic, the pursuit of co-creation of value and bidirectional dialogue embedded in a network relationship perspective could be facilitated by the diffusion of new technologies and especially the Internet. This is because the Internet and all its derived products (i.e., SNSs) can be considered the most interactive medium to date. They may enhance users' participation and generation of content and thus may fit well with the S-D logic where bilateral and direct dialogue, co-

creation of value and interactions are dominant aspects. The Internet may appeal not only to Marketing but also to Political Science and Sociology. Particularly, from a political science perspective, an informed public is essential to the proper functioning of democracy (Delli Carpini, 2000). Since the Internet is a mass of information it might be regarded as a powerful democratising agent. Arguably, this may lead to increasing citizens' awareness and involvement in self-governance which from a sociological perspective means that the Internet could enhance Political Participation and voter learning (Tolbert and Mc Neal, 2003; Drew and Weawer, 2006). Hence, the study of the Internet and its tools like Social Network Sites, blogs, micro-blogging and so forth, may contribute to overcome criticisms on Political Marketing at two levels: practical and theoretical. Indeed, apart from the ones already discussed, Henneberg (2004) and Ormrod et al. (2013) identified other criticisms relevant to Political Marketing which the use of new communication technologies by political parties and candidates may help to overcome (Table 2.4).

CRITICISMS TO POLITICAL MARKETING				
BOLITICAL MARKETING DRACTICES	POLITICAL MARKETING			
FOLITICAL MARKETING FRACTICES	THEORY/RESEARCH			
1. Elections can be bought (only rich	1. Research is not focused on politics but			
candidates can afford to run).	on communication tactics and campaign			
2. Politics has no more content since it is	with a short-term orientation.			
obsessed with "spin" and "packaging".	2. Research is unsophisticated and does			
3. Political marketing has caused more	not engage with ethics and political			
populism.	frameworks.			
4. Parties and politicians are focused on	3. Politics is not a marketing domain.			
selling. Politics is something totally	4. No theoretical and ethical frameworks			
different.	exist that can allow valuable discussion			
5. Political marketing uses manipulative	on Political Marketing.			
methods and therefore Political				
Marketing practices are not compatible				
with how voters should make an				
informed voting decision.				
6. Too much personal negative campaigns.				

 Table 2.4. Criticisms to Political Marketing Practice and Theory.

Source: adapted from Henneberg (2004) and Ormrod et al. (2013).

For instance, "one of the most damming criticisms [...] is the accusation that democratic election can now be bought" (Henneberg, 2004: p. 229). The reasoning behind this claim is that "modern elections are won by the candidate or party with the most

professional campaign" (Ormrod et al., 2013: p. 60). Yet, in order to implement a professional campaign, parties must hire and pay for marketers, ads expert, media slots and cover all the necessary expense to conduct a professional campaign (Franklin, 1994; Newman, 2001). Thus, it has been argued that only the party with more financial resources wins the election, regardless of the validity of the political argument (Wray, 1999).

However, several objections can be brought to such reasoning, based on empirical evidence related to the use of social media. For instance, the Internet may help to drastically reduce campaign costs with no detriment to the effectiveness of the campaign. An example is Beppe Grillo's Five Star Movement in Italy. This party exploited the web and social media for carrying out its political activities, employing the Internet not only as a means of spreading its own political message, but also as the principal instrument for its fundraising activity, avoiding the high cost of advertising on TV, billboards, newspapers and organization of events. The outcome of their strategy was that at the 2013 general election the new Five Star Movement gained 26% of votes in the Lower House (coming second to the established Democratic Party - PD) and 24% of votes in the Senate, beating Silvio Berlusconi's PDL, the Centre Union, Lega Nord and other minor parties. The Five Star Movement electoral campaign costs was equal to 348.506,49 Euro (www.beppegrillo.it, 2013) against the 6,5 million Euro of PD (politica.nanopress.it, 2013). This evidence suggests that online social platforms can be effective in increasing a party's potential through the effective use of scarce resources, the generation of valuable information and the promotion of greater responsiveness in the political process (Kotler and Kotler, 1999), ultimately facilitating the process of establishing long term relationships with core stakeholders. Ultimately, new communication media are changing the way Political Marketing is considered by politicians, professional political marketer and even academics.

Furthermore, online political engagement provides an opportunity for all political stakeholders and even citizens to persuade others to participate in political activities. Given the high cost involved in traditional types of Political Participation, the use of social media enables users to engage in a timely and direct manner, at a relatively low cost and with higher levels of efficiency than can be achieved through traditional communication tools (Kaplan and Haenlein, 2012). This makes it desirable to use social media as an approach to the promotion of political behaviour, assisting the need for greater public

involvement in politics. This ultimately would benefit society at large, as it will enhance democratic practices through large scale discussion and deliberation on virtual agora (see Chapter 3). Moreover, the 'big data' originating from social media allow for more effective market segmentation, helping politicians to better understand how different voters segments want to participate in the political process, ultimately creating more desirable individualised options for involvement (Lees-Marshment, 2019b).

The study of SNS within the Political Marketing frameworks can help overcome several criticisms. "*Political marketing is a pervasive force in our lives* [...] *and along with problems there is also the potential for it to strengthen the functioning of political system and society. What makes the differences is how Political Marketing is practiced*" (Lees-Marshment, 2009: p.247). However, not only good practices, but also more conceptual rigour is needed. Particular attention must be given to ethical consequences of Political Marketing practices on democratic systems (Ormrod et al., 2013). Also, Political Marketing research must address ethical concerns, by fostering and designing better democratic frameworks drawing from Political Science, sociological concepts like Social Capital and the study of the new technologies like the Internet and Social Media.

2.4. SUMMARISING: IDENTIFIED GAPS

From the literature review undertaken in this section we can identify several gaps in Political Marketing theory and research.

1. Political Marketing theory. Extant theories of Political Marketing have mostly considered citizens as perfectly informed, rational, involved and knowledgeable individuals acting according to a cost-benefit analysis (Savigny and Wring, 2009; Taylor, 2006), rather than as misinformed and disengaged people as suggested by empirical evidence (see Chapter 1). Arguably, this assumption of voters as rational decision makers which stems from the economics approach to politics in the tradition of Anthony Downs's (1957) work "Economic Theory of Democracy", has not helped to solve the empirical problem of low levels of political engagement among citizens. Empirical evidence suggests that grounding Political Marketing only on theories and concepts originating from the economics disciplines might not be effective in enhancing citizens' participation in the political process. This instrumental approach to Political Marketing as a mere transposition of marketing principles and tools to politics, not only has raised several criticisms (Henneberg, 2004; Lees-Marchment 2019b; Ormrod et al., 2013) but also it has been considered as one of the main causes of citizens' disaffection towards politics (Savigny and Wring, 2009). Very rarely Political Marketing as a discipline has engaged with theories of democracy (Henneberg et al., 2009; Perannagari and Chakrabarti, 2020), ethical questions (Henneberg and O'Shaughnessy, 2007; Lees-Marshment and Raynauld, 2020; Ormrod et al., 2013) and theories of participation (Scammell 2003). Indeed, as Brown et al. (2018) argue, Political Marketing theory has, so far, been essentially static, "backward even" (p. 1359), to the extent that theory in the field is still bound to early marketing concepts. Recent developments in marketing theory, particularly the relationship approach and the S-D logic, must serve as means for a new wave of conceptualisation which takes into consideration social and ethical arguments in the political context and provides a route to a more participatory democracy. This "could help bridge the conceptual gap between political science and marketing, as they both struggle to analyse contemporary democratic process" (Johansen, 2005: p. 102). Therefore, this study investigates the problem of voters'

disengagement in politics focusing on theories originating from politics (*Mobilisation Theories*), ethics (*Virtue Ethics*) and sociology (*Social Capital*), and "*adapting*", rather than "*adopting*", concepts from the economics and marketing fields like trust, reciprocity, long term relationships and intimacy.

2. Political Marketing Empirical Research. To date, empirical research in Political Marketing has been focusing on the welfare of political parties and relevant stakeholders and for this reason it has been the object of criticism. The most prominent criticism is that politics has a higher purpose than commercial profitability and should benefit society at large rather than just parties and relevant lobbies (Baines and Egan, 2001).

Although it can be argued that focusing on the problem of enhancing Political Participation eventually will benefit parties, it may be retorted that it will also benefit society at large, thus overcoming some of the criticisms brought to the field. Indeed, although some discrepancies do exist among theories of democracy with regards to which levels and intensity of participation are desirable and good for democracy, engagement in politics is understood as something positive and necessary in democratic societies (Livingstone, 2013). Thus, investigating this problem and trying to solve it can be regarded as an attempt to benefit society rather than just political parties.

Moreover, the lack of a clear ethical position with respect to democratic fundamentals and political practices holds back the research field of Political Marketing (Henneberg and O'Shaughnessy, 2007). For this reason, the underpinning of the current study will be the Ethical Theory of Virtue, abandoning traditions of the last decades based on the *deontological* and the *consequentialist* approach. Furthermore, such theory is in line with the construct of Social Capital, which has at his basis the concept of civic virtue.

Another gap in Political Marketing research is that so far it has been grounded on epistemological and ontological assumptions originating from the positivism perspective grounded in the economic field (Savigny and Wring, 2009), thus regarding political parties as corporations and voters as rational decision makers. This creates a sort of paradox in terms of scientific methodology. In fact, those assumptions assume that parties "are" rather than "behave" as if they were businesses (ontological claim) (Savigny, 2008). However, from a positivism
perspective, models are used to describe and potentially predict what happens in reality, because reality is already there, and the world exists independently of our knowledge of it. So, we cannot "create reality". As such, the basic assumption that parties must be considered as businesses and voters as consumers creates a dissonance with the principle of positivism. Firstly, it must be investigated and discovered in the real world whether they (parties and voters) effectively "are" businesses and consumers. Only after such observation has been validated in the real world it can be assumed also in the theoretical field. Thus, a shift is needed also in the approach of empirical research. For this reason, the fundamental point of view of this study is that knowledge, although starting from our senses (Hume cited in Hay, 2002), "is mediated by the concepts we use to analyse it, so there is no way of classifying, or even describing, experience without interpreting it" (Marsh and Furlong, 2002: p. 23). Furthermore, social phenomena are fundamentally different from natural phenomena since they do not exist independently from the social context in which they are embedded. This last assumption will be the leading principle of this research which will consider attitudinal and cultural norms (reciprocity, trust and shared values) of the social context in which people are embedded at the centre of the theoretical framework of this study. Specifically, this research transcends the rational choice perspective of self-interested citizens aiming at the maximisation of their own benefit and investigates cultural aspects and structural dimensions of relationships, as elements able to affect the decision-making process of citizens through the concept of Social Capital.

3. LITERATURE REVIEW ON PARTICIPATION

As highlighted in the previous Chapter, one of the gaps identified in the field of Political Marketing is that it needs to engage with theories of democracy ethical questions and theory of participation, to provide itself with legitimacy and bring relevant improvements in the political field and in society.

In particular, the issue of Political Participation has been of critical interest for sociologists and political scientists in the last decades, since it has become axiomatic that wide participation means better democracy (Fox, 2014; Held, 2006; Pateman, 1970; Van Deth, 2014; Verba et al. 1995). However, what is meant by Political Participation and how this can affect democracy has been the cornerstone of a long debate in both Social and Political Science (Fox, 2014; Held, 2006; Van Deth, 2014). Therefore, following the suggestions of Henneberg and O'Shaughnessy (2007), Henneberg et al. (2009), Lees-Marshment (2019b), Lees-Marshment and Raynauld (2020), Ormrod et al. (2013) and Perannagari and Chakrabarti (2020) when approaching Political Marketing research, it is important to delineate the concepts of democracy and Political Participation and show how these are interconnected and grounded on ethical theories.

The literature review of this Chapter has multiple outcomes. First, it discusses and delineates the ethical approach underlying this study, namely "Virtue Ethics" (Section 3.1). Second, it reviews the concept of citizens' political participation according to different theories of democracy and establishes a link among the ethical framework of this research, Political Participation, Political Marketing and Social Capital Theory (Section 3.2). Third, a comprehensive definition of the concept of Political Participation is provided drawing, upon criteria identified through an extensive overview of the literature (Section 3.3). Fourth, the relationship and the impact of new digital media like the Internet and the SNSs on Political Participation is addressed following the principle of "Technology Affordance" under the logic of "Technological Determinism" and the Intensity of use of SNSs (Section 3.4). Finally, gaps are identified (Section 3.5). Figure 3.1, provides a conceptual map of the chapter and highlights the main themes emerging from the review of the relevant academic literature.



Figure 3.1. Chapter 3 Conceptual Map.

Source: created by author.

3.1. PARTICIPATION IN ETHICS

Engaging with ethical theories is important for Political Marketing, since the lack of a clear ethical position with respect to democratic fundamentals and political practices holds back the research field (Brown et al., 2018; Henneberg and O'Shaughnessy, 2007; Lees-Marshment, 2019b). The aim of this section is to pinpoint the ethical framework under which this study will be conducted, namely Virtue Ethics and its political translation that is Communitarianism, which are in line with the advocated shift in Political Marketing towards a societal welfare definition (Chapter 2, Section 2.2). Indeed, as it will become evident from the review, societal welfare and Communitarianism provide the natural context for the concept of Social Capital.

However, the scope here is not to provide a comprehensive account of all the possible ethical theories existing in political philosophy, rather to provide a critical analysis of the reasons for choosing a particular framework, acknowledging that "*the application of ethical frameworks does not generate any final answer, as no ethical debate is ever final; ethical questions can only be taken further, not answered*" (Ormrod et al., 2013: p. 107).

3.1.1. VIRTUES AND THEIR REVIVAL AS EMBEDDED IN A COMMUNITARIAN PERSPECTIVE FOR THE COMMON GOOD.

Taking an ethical perspective is important for understanding what is considered good or not good, fair or unfair, socially desirable or avoidable for the common sake of the collective. Generally speaking, ethical questions can be observed not only in politics and sociology but even in the marketing field, especially in social marketing, through social issues related campaigns. For instance, anti-smoking campaigns apart from pinpointing how smoking negatively affects the smoker, stress the consequences that such decision (smoking) has on other members of the community, such as children and family. The same can be said for campaigns on safe driving, pollution and so forth. The negative consequences that a specific human behaviour may have on the rest of the community are something central to the concept of Social Capital especially in Coleman (1988) who defined those as bad externalities. From an ethical point of view, and from the Social Capital perspective, those negative externalities should be avoided or at least reduced to a minimum.

Hence, exploring participation from an ethical perspective means to understand how politicians ought/should act (normative approach) to encourage citizens participation in politics and avoid the negative externalities that non-participation could bring to society.

However, a critical evaluation of the *Deontological* (Kantian) and *Consequentialist* ethical frameworks in which modern democracies are embedded is needed, to understand how to engage citizens in the democratic process (Hartley and Watson, 2014).

Particularly, the deontological approach, grounded on the Kantian thought, stated that actions should flow from elemental principles which represent universal moral rules for the action of all individuals in a society (O'Shaughnessy, 2002). Universal, from the Kantian perspective, means that those rules are always valid irrespective of time and contexts. Such moral absolutism presents several shortcomings. First, very little is mentioned on how society should arrive at these rules (O'Shaughnessy, 2002). Second, in its moral absolutism, the deontological approach does not leave space for conceptualization and define rules and norms with no regard to circumstances. To put it into Political Marketing terms, considering as an example the criticism of negative advertising in politics (Chapter 2), a person or society may suggest prohibiting it since for them negative advertising might represent a morally unacceptable practice. But, as it can be argued, negative advertising represents information that can be crucial for the assessment of the quality of the service provider, in this case the political candidate. Indeed, it could represent a fundamental element on which voters may base their expectations and may achieve an informed opinion. Furthermore, a deontological approach to Political Marketing would hinder even theoretical research. In fact, for the convinced Kantian, once a model has been elevated to universal rule there will be no alternative or possible deviation from that model. We can just think about the ongoing debate on whether to treat voters as consumers, embedded in a context of political consumerism, or as citizens, thus as not perfectly rational decision makers (Lees-Marshment, 2009: p. 13; 2019b: p. 45). From a deontological perspective, if the model of decision making based on objective information and full rational deliberation was

elevated to universal rule, no further discussion could take place, hindering debates, research and thus the creation of knowledge.

As far as the consequentialist tradition is concerned, it states that the morally right action is the one with the best overall consequences (Ormrod et al., 2013). In political terms, following the recent concerns on the consequences of Political Marketing practices for society (Henneberg, 2004), the fundamental question could be related to the ethical values of Political Marketing activities. First, the answer to such question may vary on the basis of how Political Marketing is conceptualised and according to our view of democracy. Looking back at the first stage of Political Marketing history (Table 2.1, Chapter 2, Section 2.1), we could say that if the consequences of the application of marketing tools in politics, as a result of an instrumental approach, have been the centralisation of power, the marketisation of politics, the professionalization of campaigns and the disengagement of citizens from politics, then the application of marketing to politics is unethical according to the consequentialists. On the other hand, if we consider good government as responsiveness to public opinion, then there would be some vindications of the marketing application to politics. However, the consequentialist tradition does not fit Political Marketing as conceptualised in this study. For instance, if we look at political campaigns, consequentialism may justify a short-term approach, since in marketing terms it means maximising popularity in opinion polls without considering other consequences. But even when other consequences are taken into consideration, such process of considering all the possible eventualities of an action will become too time consuming, and from both a marketing and a political perspective, where promptness and good responsiveness to the public are central issues, this would be a considerable shortcoming (Ormrod et al., 2013).

Rather, "*ethics would benefit from an adequate moral psychology, such as that found in ancient Greek ethics, where one can look for norms in human virtues*" (Timpe and Boyd, 2014: p. 1). Indeed, differently from the deontological and consequentialist approach, Virtue Ethics grounds moral action in the real agent and thus takes into consideration emotions and personal motivation of acting (Timpe and Boyd, 2014). Therefore, virtue ethics focuses on "being", while deontological and consequentialist perspectives focus on "doing". As we have seen earlier in this review, this reasoning is in line with a service marketing logic where the features of the service provider are fundamental elements for consumers to develop an assessment of the possible quality of

the service and since the political context is closer to a service marketing perspective, the features (virtue) of the service provider (candidate/party) are essential elements to allow citizens to reach an informed opinion in political terms. Moreover, virtue ethics takes into account emotions, attitudes and personal motivation of people to explain their actions and this matches not only the considerations brought earlier in this work that voters cannot be considered as perfectly rational individuals, but also the multidimensional construct of Social Capital which is regarded in this study as the set of cultural, relational and structural aspects which include trust and reciprocity as fundamental variables of the concept (see the definition of Social Capital in Chapter 4).

Many authors (e.g., Henneberg et al., 2009; O'Shaughnessy, 2002; Timpe and Boyd, 2014) consider *Communitarianism* as the logical translation of Virtue Ethics in the political sphere. "*Communitarianism locates virtue within the context of a parochial social setting* [...] *ethical tradition and sensitivities are seen as arising out of community*" (O'Shaughnessy, 2002: p. 1086), that is the "good" of democracy is the one favoured by society whose actions should not be guided by universal and abstract principles. Thus, if virtue is what the community teaches, then it can be claimed that any community/culture have its own standard of virtue. In Political Marketing terms, this means that some practices may be more acceptable in some countries than in others. Also, it means that the research context matters and that attitude, beliefs and shared codes and values must be taken into consideration when conducting research within the political field. These concepts are embedded in the notion of Social Capital (Chapter 4).

However, for the aim of this thesis, the focus will not be on all virtues in general (Moral, Intellectual, Theological) but rather on "Civic Virtue" and on the idea that reciprocity and trust (elements of Social Capital) are internalised norms of individuals which make them virtuous and are central to social cooperation and political life (Hartley and Watson, 2014).

3.1.2. CIVIC VIRTUE AND THE COMMON GOOD

The key element of civic virtue is virtue itself, which comes from the Greek word "*Arete*", meaning "*excellence*" related to certain activities or actions. However, although used as synonyms, virtue is a distinctive form of "arete": not just excellence but moral excellence (Newstead et al., 2018) and specifically excellence of character or of intellect and as such it is like a "*blueprint*" for a person for how to think, act and behave.

Civic Virtue refers to the connection between citizens and the community at large and how individuals ought/should behave (normative sphere) in the civic and social context to be considered good citizens. Following the classical republican tradition in both its Aristotelian and Roman forms, civic virtue is a "*disposition to further public over private good in action and deliberation*" (Burtt, 1990: p. 24). Therefore, the emphasis is on the public interest which must be defended and preferred over people's self-interest.

Although there are different approaches to what can be considered good at both individual and common levels, the Communitarian perspective is one of the most accepted by political scientists (Ormrod et al., 2013). Communitarians define "*the individual good in whole or in part in terms of its contribution to the common good*" (Plauche, 2006: p. 7). As Cicero claimed, "*we are not born for ourselves alone, but our country claims for itself one part of our birth, and our friends another*" (quoted in Honohan, 2003: p. 30).

For Communitarians Civic Virtue means that everyone is committed to the common good and that he/she is willing to do his/her part in realising it. This, inevitably, ties the concept of Civic Virtue to Political Participation and to the Aristotelian view for which Political Participation is a responsibility of the good and virtuous citizen. Hence, Communitarians call also for a more participatory behaviour of citizens in the political process which is regarded as a core element of a good and robust democracy (Held, 2006; Conway, 2000; Green, 2010; Bakker and De Vreese, 2011) since it improves the awareness of citizenry, enhances Government responsiveness, provides polity with legitimacy and fosters people to support policy implementation.

From the relevance of Political Participation stems the necessity for Political Marketing to engage not only "*with theories of democracy in order to provide itself with legitimacy*" (Henneberg et al., 2009: p. 181) but also with studies of Political Participation, since as Scammell (2003) argued, Political Marketing may be valuable, even essential, for fostering voters' interest and participation. Hence, in the following section, participation as understood in various theories of democracy will be analysed.

3.2. PARTICIPATION IN THEORIES OF DEMOCRACY

As Carpentier and De Cleen (2008) have argued, rather than being an academic construct, the term Political Participation is understood as a part of a political-ideological struggle between minimalist and maximalist approaches. This contrast between those two approaches may be better understood through the lens of theories of democracy.

Held (2006) classified and distinguished among models of democracy on the basis of how these articulate citizens' participation. Specifically, Held identified several types of democracies (Table 3.1) and classified them within two broader categories:

- 1. *Liberal Representative Democracy*, which tries to control and limit people's selfgovernment and participation through the feature of representation and power delegation.
- Direct Democracy, which defends a direct notion of citizens' participation in the decision-making process.

DEMOCRACY				
REPRESENTATIVE	DIRECT			
 Competitive Elitism Legal Democracy Protective Democracy 	 Classical Democracy Developmental Republicanism Participatory Democracy Deliberative Democracy 			

Table 3.1. Models of democracy

Source: Adapted from Held (2006).

However, the following review focus only on the general concept of *Liberal Representative Democracy* and particularly on the *Competitive Elitism Theory* and on one forms of direct democracy, namely *Deliberative*. The discussion will be articulated in a perspective of Political Participation and on how the aforementioned models deal with it. Also, a link to Social Capital and new Media Technology will be provided. The focus on those specific forms of democracy is motivated by the following reasons:

- 1. they represent the broad spectrum of the contemporary debate in democratic theories (Henneberg et al., 2009), that categorise democracy under two headings:
 - a) Democracy as competition (minimalist approach to participation).
 - b) Democracy as participation (maximalist approach to participation).

- 2. Competitive elitism, by common agreement, has been the most influential model for western democracies for more than fifty years from the beginning of the twentieth century (Carpentier, 2011; Henneberg et al., 2009; Scammell, 2000).
- 3. Deliberative democracy stems directly from criticisms to competitive elitism.

3.2.1. DEMOCRACY AS COMPETITION: PARTICIPATION IN REPRESENTITIVE DEMOCRACY AND COMPETITIVE ELITISM

In the past, many scholars have advocated a liberal *Representative Democracy* as the most realistic (Held, 2006) and effectively viable "*method*" of governance in which the only role of people is to elect governments (Schumpeter, 1976: p. 269).

Representative Democracy Theory has influenced an important theoretical approach, namely, Competitive Elitism (Held, 2006), whose most famous advocate was the Austrian philosopher and economist Joseph Schumpeter (1976). His approach is based on Elite Theory, whose central aspect is the stratification of society between rulers and ruled, which Schumpeter based on "*realism*": a ruling class is present in all political contexts (Henneberg et al., 2009; Held, 2006). In Schumpeter's (1976) view, citizens act in condition of semi-ignorance (limited knowledge and interest) and their only role is to elect rulers which will establish what is the common good by themselves (Held, 2006). Therefore, the highest level of participation to which citizens may aspire to is about voting in free and fair elections, which are regarded as an essential and valuable part of the democratic process.

From a Political Marketing perspective, and especially from the *Instrumental* school perspective (Chapter 2), Competitive Elitism is very attractive, since Schumpeter (1976) was among the first political theorists to consider elections as something resembling the selling process in the commercial market (Street, 2003). Accordingly, elections were embedded in a logic of competition where the electorate was an uninterested mass which parties needed to mobilise through appropriate techniques that he called "psycho techniques" like spinning, rhetoric, communication practices, etc. Those techniques were regarded as indispensable tools for making democracy work and not as means for corrupting voters as they are considered nowadays (Ormrod et al., 2013). Arguably, this approach fits closely the instrumental, narrow, selling oriented school of

Political Marketing that it is at the basis of most of the criticisms discussed in Chapter 2, and it is regarded as the cause of the current "democratic malaise" (Henneberg et al., 2009; Klein, 2006; Savigny and Wring, 2009). Specifically, Competitive Elitism is focused on a top-down offering system, designed according to needs and beliefs of producers rather than of "society at large" and is centred on the extensive use of marketing's tools and techniques to "sell the president", in a short-term orientation (political campaigns) towards the electorate. Marketization and personalisation of political campaigns, short-term orientation and the identification of Political Marketing with an instrumental approach have created a sort of distance between the electorate and parties, making citizens believe that politics was not something of interest to them (Scammel, 2003).

Therefore, competitive elitism does not fit the concept of Political Marketing proposed in this study for two main reasons. First, because it is sceptical that meaningful long-term relationships with citizens, based on political interaction, are realistically sought after by the electorate which is instead regarded as a mass of uninterested people. Second, because Competitive Elitism is more concerned with leadership than with society (Henneberg et al., 2009).

The aforementioned factors and empirical evidence showing decreasing levels of voters' turnout, increasing disaffection and lack of trust in political parties (Mair, 2005; Franklin, 2004) have given rise to theories favouring more direct forms of participation (see next Section), challenging the legitimacy of representative democracy (Michels, 2011).

3.2.2. DEMOCRACY AS PARTICIPATION: DELIBERATIVE DEMOCRACY AND SOCIAL CAPITAL

Supporters of Direct Democracy in the 1970s advocated the maximum participation of citizens in their self-governance even in sectors beyond the political context (such as households, workplaces, neighbourhoods, schools). In Pateman's (1970) words (cited in Hilmer, 2010: p. 45): "the notion of a participatory society requires that the scope of the term political should be/must be extended to cover spheres outside the national government".

Direct democrats considered the representative liberal democracy and its consequent delegation of decision-making power as the leading factor of citizens' alienation from politics (Carpentier 2011). This theory derived directly from Rousseau's concept of "*social contract*" (Michels, 2011: p. 277) for which the participation of each citizen in political decision making is vital to democracy, since it spurs individuals to work together for influencing the decision-making process and developing "*a more equitable and humane society*" (Macpherson, 1977: p. 94).

By the 1990s several theories of Direct democracy developed and, among others, *Deliberative Democracy*. Deliberative democrats argue that "*rational public deliberation among free and equal citizens about matters of common concern*" (Hilmer, 2010: p.51) is the essence of democratic legitimacy (for a more comprehensive discussion see Dryzek and List, 2003 and Hendriks et al., 2007). Accordingly, a deliberative process assumes (Michels, 2011):

- *Equality* of citizens who have equal voice and equal opportunity to persuade others.
- Inclusion of different interests, even those of minorities.
- Mutual respect.

However, according to Mutz (2006), critics of deliberative democracy argued that it may actually discourage Political Participation for several reasons. First, some people may be unwilling to publicly take a stand that can be opposed by friends and associates. Second, it is utopistic to find a place where all citizens can gather to deliberate. Third, not all people have the skills needed for deliberating. However, if we think of modern technology and especially the Internet, the reluctance to stand publicly might be overcome by the possibility of anonymity or pseudo-identity that this new mean of communication offers (Collins and Butler, 2003). Furthermore, public debate and reciprocal reason giving may be practically implemented through the creation of web pages, blogs, or networks groups on Social Media virtually at no cost and for everyone. Lastly, inclusion in social networks per se (even the digital ones) encourages the development of the skills needed for deliberation and reciprocal reason-giving. This last point is particularly relevant to the theory of *Social Capital* (Putnam, 2000), that claims that citizens' participation encourages *civic skills* and *Virtues* through the inclusion in social networks and voluntary associations. In Putnam's (2000) view, those two forms of involvement are a sort of school for democracy, since through them people may learn how to speak publicly and debate on issues, allowing for their voice to be heard. Thus, citizens learn civic skills and become familiar with what Putnam (2000) considers the three most important civic virtues:

- 1. Participation in public life.
- 2. Trustworthiness.
- 3. *Reciprocity* (giving and taking).

Hence, Social Capital is closely related to the theory of Civic Virtue (Skoric et al., 2009: p. 416) and is in line with the ethical stand taken in this thesis (Section 3.1). However, as mentioned above, Social Capital Theory argues that civic virtue is most powerful when embedded in a dense network of reciprocal social relations. Those social relations help people develop civic skills, which in turn help them to actively participate in politics through norms of trust and reciprocity embedded in long-term relationships established in social networks. Therefore, Social Capital Theory emphasises some salient aspects of participation which contribute to a better democracy and that are common to the deliberative form of democracy (

Table 3.2) and it also appeals to Political Marketing and its *Comprehensive* and *Holistic* perspectives (see Chapter 2).

Indeed, trust and reciprocity are at the basis of Relationship Marketing but also of the S-D logic which, as argued in Chapter 2, should inform Political Marketing theory and practices. Specifically, Social Capital could be relevant to the S-D logic which emphasises the co-creation of value and encourages dialogue. Indeed, from a political perspective, the co-creation of value might occur if citizens actively participate in politics through debates and deliberative actions. However, from a Social Capital perspective, people may learn to debate and to deliberate through the inclusion in social networks and voluntary association (Putnam, 2000), that is through participation in public life. Participation will help citizens to co-create political offering (policies) which may benefit society at large. This last point, benefit to society at large, is in line with the definition of Political Marketing provided in this study. Furthermore, given the previous premises that Political Marketing should engage with Virtue Ethics and Communitarianism, which are grounded in the concept of Civic Virtue, Social Capital seems the perfect framework for merging these worlds. Indeed, Social Capital is closely related to Civic Virtue (Putnam, 2000) and it appears to be closely related to Communitarianism. In particular, reciprocity and trust (two elements of Social Capital) will be considered in this research as internalised norms of individuals which may define them as virtuous. Hence, because of its affinity with Virtue Ethics, Theories of Democracy and the S-D logic, the application of Social Capital in studies of Political Marketing could help to partially overcome ethical criticisms of political scientists who regard marketing as a means though which politics "*will be transformed from what should be a quest for a common vision of the just noble and good*" (Ormorod et al., 2013: p. 75) into the private self-interest pursuit of consumerism.

MODEL	YEARS	PARTICIPATION APPROACH	FOCUS OF PARTICIPATION	SALIENT ASPECTS		
Liberal Representative (Dahl, 1956; Schumpeter 1952)	1950s	Narrow (voting, membership)	Methods (election)	• Election as only mean to influence politics		
Participatory Democracy (Pateman, 1970; Barber, 1984; Macpherson, 1977)	1970s	Wide (need for representation)	Sectors (workplace, household, politics, etc.)	 Influence decision making Develop Civic skills Legitimacy 		
Deliberative Democracy (Boheman, 1996; Habermas, 1996; Mutz, 2006)	1990s	Wide	Modes (deliberative polling, deliberative forums, Citizens' juries, etc.)	 Inclusion Influence decision making Deliberation Legitimacy 		
Social Capital Theory (Bourdieu, 1980; Nan Lin, 2008; Putnam, 2000;	2000	Wide	Modes, Sectors, Methods	 Inclusion Develop Civic skills Develop Civic Virtue Influence decision making 		
<i>inclusion</i> : citizens' participations contributes to the inclusion of singular individuals in the policy process, allowing their voices to be heard.						
influence decision making: possibility to effectively influence decision making.						
<i>civic skills</i> : debating in public and running meetings.						
deliberation: rational decision based on public reasoning and assessment						
legitimacy: increasing legitimacy of decisions and subsequent policies.						

Table 3.2. Partici	pation in demo	cratic theories an	nd Social Capital.
--------------------	----------------	--------------------	--------------------

Source: see references in table.

3.3. POLITICAL PARTICIPATION CONCEPTUALISATION

Providing an exhaustive definition of Political Participation is not an easy task especially in the contemporary research context where academics have started to consider alternative participatory activities to justify the constantly decreasing levels of electoral turnout in western democracies (Dalton, 2006; Held, 2006; O'Toole et al., 2003; Norris, 2002; Putnam, 2000). Indeed, Political Participation has been associated with so many activities that Van Deth (2014) have coined the expression "*endless expansion*" (p. 349).

Therefore, providing an exhaustive definition of Political Participation is quite a difficult task and an inadequate conceptualisation could lead to misleading results in empirical research. As Hay (2007) remarked, "those with the most restrictive and conventional conceptions of Political Participation identify a strong and consistent pattern of declining Political Participation and engagement over time, whilst those with a more inclusive conception discern instead a change in the mode of Political Participation" (p. 23).

Although this study acknowledges such influencing effect of a broadened and too inclusive definition of Political Participation on the output of the research, in this thesis a wider conceptualisation of Political Participation is adopted, including both conventional and non-conventional forms, offline and online, of political actions. Reasons for this choice are provided in the next Sections.

3.3.1. DEFINING CRITERIA OF POLITICAL PARTICIPATION

Political participation has been defined in many ways over the years. Drawing on decades of research in the field and on several overviews and of the literature (see Brady, 1998; Conge, 1988; Ekman and Amna, 2012; Fox, 2014; Van Deth, 2001, 2014), it is possible to identify a series of criteria around which the concept of Political Participation has been built:

- 1. Action Vs No Action.
- 2. Active Vs Passive behaviour.
- 3. Voluntary Vs Forced activity.

- 4. Individual Vs Group activities.
- 5. Amateur Vs Professional behaviour.
- 6. Political Vs Civic sphere.
- 7. Conventional Vs Non-conventional participation.
- 8. Legal Vs Illegal activities.
- 9. Intent Vs Influence.
- 10. Online Vs Offline.

The above criteria contributed to the conceptualisation of Political Participation employed in this study, as illustrated in Figure 3.2 and explained in the following paragraphs.

Figure 3.2. Political Participation Conceptualisation Criteria.



Source: created by author.

In particular, following most of the extant definitions of Political Participation (e.g., Almond and Verba, 1963; Barrett and Brunton-Smith 2014; Norris 2002; Parry et al., 1992; Pattie et al., 2004; Van Deth 2014; Whiteley, 2011) this research acknowledges that participation must involve actions (Criterion 1). Nonetheless, although some activities like "boycotting" certain products or "not voting" cannot be considered as instances of actions, they are sometimes regarded by people as "highly political acts" (Hay, 2007: p. 26). Therefore, abstention from carrying out specific activities like voting is considered a form of Political Participation when it represents the intent, and has the effect of, influencing the civic and political sphere (Ekman and Amna, 2012). Rather, passive behaviour (Criterion 2) like holding certain values and attitudes or watching political TV programs is not considered participation because those activities do not affect directly political or civic outcomes (Van Deth 2014; Verba et al., 1995), but are forms of "political engagement" (Barret and Brunton-Smith, 2014). According to Fox (2014), Pettie et al. (2004), Van Deth (2014) and Verba et al. (1995) Political Participation must be purposeful and "voluntary" (Criterion 3). Thus, any action forced by law, rules or threats will not be taken into consideration. This choice is in line with the theory of Virtue Ethics which underpin this research framework (Section 3.1.1). In fact, since virtue is a matter of character or of the inclination and choice to habitually act in a specific manner, a virtuous behaviour implies the voluntary aspect of the action.

Notably, the first three criteria explain the first part of the definition of Political Participation as: *"voluntary and purposeful activities"*.

Following the most traditional definitions of Political Participation, voluntary and purposeful activities can be undertaken by the single individual (Burretand Brunton-Smith, 2014) or collectively (**Criterion 4**) as in the tradition of authors working in the field of Civil Society and Social Capital (Macedo et al., 2005; Putnam, 2000; Verba et al., 1995). Also, almost all the definitions of Political Participation refer to "citizens" (**Criterion 5**), when discussing either single or collective behaviour (Almond and Verba, 1963; Barber, 1984; Callahan 2007; Michels 2012; Stoker, 2006; Van Deth, 2014). This is to differentiate Political Participation from the professional and paid activities of politicians, civil servant and public officers which do not have the characteristics of voluntary activities. Furthermore, the conceptual area of Political Participation will not be restricted only to "government, politics and the state" (Van Deth, 2014: p. 352), but rather it will encompass also civic forms of participation (**Criterion 6**) according to the

definition of Norris (2002), who claimed that "*organised activities* [...] to impact civil society or which attempt to alter systematic patterns of social behaviours" can be considered as forms of Political Participation. Several reasons lead to this choice. First, boundaries between civic and political engagement are not clearly delineated in the literature (Boulianne, 2009) because "politics and civil society are interdependent" (Macedo et al., 2005: p.6) and it is not clear where the boundary between the two lies (Martin, 2012). For this reason, many studies involve an approach where "any organised action, or social behaviour or any activity aimed at change or influencing collective life" (Van Deth, 2014: p. 352) are embedded in the definition of Political Participation. This thesis follows this tradition.

Second, it has been argued that there is a difference between the subject to be targeted and the aim of political actions (Parry et al, 1992). In fact, even if the target of Political Participation could be considered "*always the government*" (Parry et al, 1992: p. 7) or "*generally the state*" (Pattie et al. 2004: p. 78), political activities do not need necessarily to be aimed at the government, rather citizens may interact with other institutions to influence the governmental output, such as lobbies, interest groups or even other political institutions like the European Union. Therefore, actions of Political Participation cannot be restricted to government, rather they should be expanded to society at large.

According to the above reasoning concerning criteria 4, 5 and 6, Political Participation includes purposeful and voluntary activities "*individually or collectively carried out by citizens which target and affect the political and civic spheres*".

Also, following Dalton (2008); Dalton et al. (2010), Ekman and Amna (2012), Marsh and Kaase (1979) Norris (2002), Parry et al. (1992) and Van Deth, (2014) this thesis includes in the definition of Political Participation both conventional and unconventional activities (<u>Criterion 7</u>). The former relates to the classic institutionalised modes of participation like voting, contacting politicians, working for a party and all the activities related to the electoral process (Rucht, 2007). The latter are defined as any "episodic, public" (Tarrow and Tilly, 2007: p 438) non-institutionalised political action (Dalton, 2008) like boycotting, marching, protesting, striking and occupying buildings even when those actions are directed towards "*non-governmental institutions [...] but in the attempt to influence political outcomes*" (Teorell et al., 2007: p. 336). Such non-institutionalised set of activities can be:

- 1. Legal (e.g., signing petitions or participating in lawful demonstrations).
- 2. Illegal, which can further be categorised as:
 - a) Violent (e.g., set building or things on fire, shooting);
 - b) Non-violent (e.g., unofficial strike, peaceful occupation of building).

Following a long tradition in research, this study includes in the conceptualisation of Political Participation both legal and illegal activities (<u>Criterion 8</u>) (Almond and Verba, 1963; Ekman and Amna 2012; Huntington and Nelson, 1976; Norris, 2002; Parry et al., 1992; Pattie et al., 2004; Van Deth 2014; Verba et al., 1995). To understand why, it is helpful to refer to the genesis of democracy and particularly to the teleological and praxialist arguments.

Focusing on the teleological argument means to understand what the ultimate goal of a specific behaviour is (from the Greek word "*telos*" which means ultimate end or goal). In political terms, this means to understand whether participants are trying to influence (<u>Criterion 9</u>) through their actions the political and civic sphere. Any behaviour aiming at that outcome (regardless of how it has been achieved), can be considered Political Participation. Such goal-oriented approach in defining Political Participation is really common in the literature (among the others, Almond and Verba, 1963: p. 117; Huntington and Nelson, 1976: p. 3; Norris, 2002; Verba et al., 1995: p. 38).

Alternatively, some researchers (Diemer, 2012; Munroe, 2002; Riley et al., 2010) capitalise on the lawful nature of the praxis (from the Greek word "*praxis*" which means accepted practice or custom) through which the political intent is manifested. However, when the ultimate goal of a political action is the common good of society, the lawfulness of the praxis is usually ignored as in the ancient Greek tradition. Indeed, following Gagarin and Fantham's (2010) discussion about ancient democracy in Greece, it seems that after the oligarchic revolution of 411 B.C. Athenians approved a decree which stated that the killer of any over-thrower of the democracy could not be held liable for any penalty. The underlying telos was that of protecting democracy (a public concern) from tyrannical power and to achieve that telos any praxis, even murder, was considered an acceptable form of action. Accordingly, it is possible to qualify as acceptable praxis any actions that serve a specific political telos on issues of public concern for the common

good of society. Therefore, the telos above the praxis. Accordingly, the definition of Political Participation should include both forms of praxis: legal and illegal. After all, this view has been widely embraced in academic practice because it has been recognised that societies are constantly changing and evolving in both technological and sociological terms and such changes have allowed reconsidering the legal, the philosophical and the orthodox boundaries within which studying political actions. Especially technological changes like the development and diffusion of the Internet have significantly affected the political behaviour of citizens (Dalton, 2009) and have allowed new forms of Political Participation. Indeed, Van Laer and Van Aelst (2010) have identified a wide range of participatory acts which would be not available without the Internet. In particular, SNSs have affected the way in which people gather and share political information, with almost no regard for geographical boundaries and time constraints (Dalton, 2009). Furthermore, it has been demonstrated how the Internet could serve as a mobilising mean, like for instance when protesters started a campaign against tuition fee changes in England in 2010 (Teocharis, 2012) or when Obama fostered people to volunteer for its electoral campaign (Bakker and De Vreese, 2011). Therefore, there is general agreement that the Internet affects Political Participation and how people are targeting state and government, including actions less reliant upon traditional political institutions (O'Neill, 2010) and aimed at different parts of society. Arguably, this upholds a broadened definition of Political Participation, which includes all those activities performed by citizens either offline or online, with the aim to affect civic and political realms (Criterion 10). Holding on to traditional definitions of Political Participation anchored exclusively to the offline political sphere limits the understanding of societies that have experienced the aforementioned changes: "If our definitions do not follow the structural trends occurring in reality, in the end we will simply be left out, using outdated categories and concepts" (Hooghe, 2014: p. 341).

Moreover, political actions must not be led by the intrinsic and/or the explicit motivation (intention) of individuals to affect political outcomes. Indeed, as Hooghe (2014) claims: "*Participants can have multiple motivations*" (p.339), but "*it does not really matter what the motivation is*" (p.340), as long as the final result of the action is that of affecting in some ways the political context. Indeed, accepting "political intentions" as fundamental element of the definition would imply that activities carried out purely for personal and private reasons or for reasons of emotional support would not

be considered as acts of Political Participation. Still, they will probably affect the political outcome. Moreover, it is difficult to practically determine what the intention of people are when conducting empirical research. Different intentions could be present, and a survey is possibly not the most suitable research instrument to uncover all those intentions or to understand whether the political one is dominant among the others.

According to the above discussion, criteria 7, 8, 9 and 10 finalise the definition of Political Participation employed in this study as:

"the set of all voluntary and purposeful activities individually or collectively carried out by citizens which target and affect the political and civic spheres through both legal or illegal praxis, online or offline".

Notably, conventional and unconventional activities are introduced in the definition of Political Participation by the word "*all*".

Based on the above definition of Political Participation, the possible items to include in the operationalisation of the construct have been identified and reported in Appendix B. This is meant to facilitate the operationalisation of Political Participation and the identification of suitable measures and scales for this study.

3.4. PARTICIPATION IN MEDIA STUDIES: THE INCONSISTENT INTERNET EFFECT

Direct democracy models of participation establish a strong connection between citizens' Political Participation and media participation considering the last as potential democratising agents, since they may enhance citizens' involvement in the media sphere contributing to the creation of the "informed citizens", that is a fundamental element for a healthy democracy (Curran, 2011; Held, 2006). Indeed, Jenkins (2006: p. 252) suggests that the Internet could be more democratic with respect to other media in four different dimensions:

- 1. *In its contest*, covering the kind of information that individuals need to implement their civic and political choices.
- 2. *In its effects*, mobilising sectors of the population which were previously inactive or less engaged with public issues.

- 3. *In its values*, enhancing public debate and therefore fostering a sense of belonging to a community in line with theories of Participatory Democracy and Social Capital.
- 4. In its process, enlarging access to content production and distribution.

However, some authors (e.g., Carpentier and Dahlgren, 2014; Ekstrom et al., 2011) argue that there is a long history of media and participation which shows that technology on its own, although able to foster participation and to enhance exposure to news and public issues, does not necessarily lead to a change in social or democratic processes (Couldry, 2010; Curran et al., 2012). However, Harrison and Barthel (2009) retort that even if history reminds us that there has been always a sort of direct media participation of society, what is crucial with the Internet is that it "*enables vastly more users to experiment with a wider and seemingly more varied range of collaborative, creative activities*" (p. 174). Moreover, thanks to specific innovations in other fields (smart phones, laptops, etc.) the Internet is potentially available at any time and it is placed in almost every context of our everyday life (Press and Williams, 2010), therefore incorporating an intrinsic interactive potential which goes beyond the limits of time and space of the 'old' media (Jenkins, 2006).

Nevertheless, there is no agreement among academics on whether the Internet may improve, or not, Political Participation. Indeed, three theoretical stands exists in the extant literature and each of them is supported by empirical evidence.

A first school of thought, regarded as the "Optimists", considers the Internet as a mobilising agent of citizens traditionally characterised by lower levels of political engagement and activity such as young people, individuals with lower socio-economic status or isolated. This is because the Internet offers additional and convenient pathways to participation facilitated by the low cost of interaction with political and civic institutions and the ease of access to relevant political information (Coleman and Blumler, 2009; Christensen and Bengtsson, 2011; Delli Carpini, 2000; Gibson, et al., 2005; Hamilton and Tolbert, 2012; Johnson and Kaye, 2003; Kavanaugh, et al., 2008; Kim and Kim, 2007; Krueger, 2002; Norris, 2003; Rojas and Puig-i-Abril, 2009; Tolbert and McNeal, 2003; Ward, et al., 2003). However, the majority of studies within this tradition found that the strongest mobilising effect of the Internet was associated mainly, and sometimes exclusively, with voting (Gibson and McAllister, 2013; Bimber, 1999; 2001;

Johnson and Kaye, 2003; Norris, 2003). Moreover, although some studies found positive effects, authors highlighted that such effects were very small (e.g., Bimber, 2003; Jennings and Zeitner, 2003) and concluded that the Internet "*has not changed levels of political engagement in any substantial way*" (Bimber, 2003, p. 224).

A second stream of the literature, the "Pessimists", considers the Internet as a medium which can contribute to declining levels of political and civic participation by inducing citizens to engage in online recreational activities, drifting them away from more meaningful forms of civic and Political Participation (e.g., Nisbet and Scheufele, 2004; Scheufele and Nisbet, 2002; Zhang and Chia, 2006). This perspective is rooted in Putnam's (2000) work on Social Capital (Chapter 4). Putnam (2000) provided empirical evidence of the decrease of Social Capital in the American society, arguing that such decline was due to the concomitant growth of the use of TV for entertainment purposes, which displaced time that could be otherwise invested in more meaningful civic or political activities. Several scholars have applied this "Time Displacement" logic to the Internet, suggesting that it generated passivity by absorbing energies that citizens would, otherwise, invest in more meaningful civic and political activities (Diani, 2001; Kraut, et al., 1998; Lusoli and Ward, 2004; Nie and Erbring, 2002; Rash, 1997; Turkle, 1996). Empirical research testing the time displacement logic found either no significant effects of the Internet on Political Participation (Zhang and Chia, 2006; Tolbert and McNeal, 2003) or a negative relationship (e.g., Gibson et al., 2000; Kwak et al., 2004; Moy et al., 2005; Shah et al., 2001a; Shah et al., 2001b).

Finally, the "*Normalisers*" claim that the Internet has a supplementary effect on Political Participation, rather than a mobilising one, as it reinforces participatory trends of those citizens who are already interested and active in politics (Calenda and Meijer, 2009; Calenda and Mosca, 2007; Dutta-Bergman and Chung, 2005; Kenski and Stroud, 2006; Kim, 2006; Krueger, 2006; Moy, et al., 2005; Norris, 2001, 2002; Polat, 2005; Wang, 2007). Online political activities are perceived in this sense as an extension of offline ones (Calenda and Mosca, 2007; Tedesco, 2004). Hence, the Internet has a "Reinforcement" effect which is perceived by some scholars in a negative light. Indeed, by strengthening existing patterns of political engagement and participation, the Internet fortifies established power structures and widens the knowledge gap between politically active and less active citizens (Bimber, 2001, 2003; Bonfadelli, 2002; Brundidge and Rice, 2009; Delli Carpini and Keeter, 1993; Lindner and Riehm, 2011; Schlozman, et al., 2010; Tedesco, 2004; Weber, et al., 2003).

The above discussion highlights that "*empirical research has not revealed any clear pattern*" (Vissers et al., 2012: p. 153) of the effects of the use of the Internet on Political Participation. This observation also applies to SNSs studies, which are examined in the next Section.

3.4.1. DEFINING SNSs AND THE INCONSISTENT EFFECT ON POLITICAL PARTICIPATION

Kaplan and Haenlein (2010) define SNSs as "Internet-based applications that build on the ideological and technological foundations of the Web 2.0, and that allow the creation and exchange of user-generated content" (p. 62). As such, it is possible to include in this definition SNSs like Facebook and MySpace, media-sharing sites like YouTube and Flickr, micro-blogging platforms such as Twitter and Weibo, and blogs. Indeed, all those platforms allow for the sharing of user-generated content.

The arrival of the SNSs "has changed the nature of everyday online interactions among citizens enabling new forms of personalized public engagement that no longer require significant organizational resources nor common identity and ideology" (Skoric et al., 2016: p. 1819). Indeed, empirical evidence shows that SNSs have facilitated the creation, organisation and implementation of online movements which migrated in the offline world, like "The Arab Spring" in 2011, the "Umbrella Revolution" in Hong Kong, the "Anti-Austerity" protests in Greece or the "Indignados" in Spain (Boulianne 2015; Skoricand Zhu, 2016; Valenzuela et al., 2017). This evidence has spurred researchers to suggest that SNSs have changed the ecology of the Internet and that a new comprehensive assessment of the role of the Internet in civic and political life is needed (Scoric and Zhu, 2016). However, although SNSs may have changed the ecology of the Internet, they did not change the uncertainty of its effects on Political Participation. Indeed, similarly to Internet research, contrasting evidence has been produced in relation to the impact of SNSs on Political Participation.

A first stream of the literature supports the positive effect of SNSs on Political Participation. For instance, the pioneering study of Williams and Gulati (2007) found that the number of Facebook friends of political candidates at the 2006 U.S. midterm elections was positively related to final vote shares and that SNSs could effectively affect offline political behaviour. Similarly, Kim and Geidner (2008) found that SNSs usage enhanced voting probability by increasing individual and collective rewards originating from it, such as Social Capital, civic duty and political efficacy. They also found that SNSs engaged younger voters who, usually, are rather disengaged from politics. These two studies identified a positive relationship between SNSs usage and voting but, by concentrating solely on this offline activity, they neglected the rest of the Political Participation spectrum (Section 3.3.2). However, other more comprehensive studies found similar results. For instance, Bode (2012) and Vitak (2012) found that Facebook use was a significant predictor of Online Political Participation. Also, the use of microblogs like Weido (Chan et al., 2012) and political blogs in general (Gil de Zuniga et al., 2009; Lewis, 2010) was found to be positively and significantly related to online political behaviour. Moreover, Offline Political Participation was also found to be significantly predicted by SNSs use (Bode, 2012; Park et al., 2009; Skoric and Poor, 2013). All these studies and many others (e.g., Bachman, et al., 2010; Holt et al., 2013; Towner, 2013 and Xenos, et al., 2014) provided evidence in support of the political mobilisation hypothesis of the SNSs.

In contrast with these findings, other authors found either limited (Baumgartner and Morris, 2010; Valenzuela et al., 2009), no (Effing et al., 2011) or negative (Ancu and Cozna, 2009; Fenton and Barassi, 2011; Theocharis and Lowe, 2016) effects of SNSs on Political Participation. In their study exploring the reasons why members of the public visited MySpace profiles of 2008 U.S. primary candidates, Ancu and Cozna (2009) determined that campaign involvement was negatively related to the use of MySpace for social interaction and was not related to the use of MySpace for information seeking and entertainment purposes. Fenton and Barassi (2011) contended that differences between individual and collective forms of participation must be considered when assessing the political potential of SNSs, as they tend to promote individualism and personal affairs, disconnecting individuals from the public terrain of common good and collective political endeavour. Furthermore, Theocharis and Lowe (2016) found that maintaining a Facebook account had negative consequences on offline and online forms of political and civic participation. Yet, other authors rejected the idea that SNSs negatively affect Political Participation, suggesting that these platforms have limited or no effects on Political Participation. Specifically, Baumgartner and Morris (2010) found that people who rely on SNSs for news are not more likely to participate in politics by traditional means. Similarly, studying Facebook, Valenzuela, et al. (2009) stressed the limited effect of SNSs over youths' Political Participation and argued that these platforms are not likely to exert a mobilising effect on young, disengaged citizens. Effing et al. (2011) found that Social Media did not significantly influence voting behaviour during the 2010/2011 local elections in the Netherlands, although politicians with higher presence in social media gained relatively more votes in the 2010 national elections.

Finally, some studies found empirical evidence of the normalisation effect of the SNSs which can strengthen the engagement patterns of those people who are already involved and interested in politics (Mascheroni, 2012; Rainie and Smith, 2012). For instance, Carlisle and Patton (2013) found that "individuals in general engaged in limited political activity via Facebook during the 2008 presidential campaign" and that "political interests play a significant role in determining whether an individual is more or less engaged in Facebook" (p. 891). Hence, people who are more interested are those who are more likely to be engaged and politically active. This reinforcement effect exacerbates existing inequalities in real life that are translated and carried over into the online world creating a digital divide. Also, Vesnic-Alujevic (2012) established that those citizens already engaged in politics offline were also more likely to participate in political activities on Facebook. Similarly, Gustafsson (2012) found that the use of SNSs per se does not drive previously inactive people to Political Participation. Rather, only members of interest organizations view SNSs as valuable tools for participation, whereas respondents who are not members of such organisations, refrain from sharing political views with their friends. Finally, Vitak, et al. (2011) pinpointed the strong link between political interest, offline Political Participation and Facebook participation, suggesting that those who are already engaged in politics seek multiple outlets and modalities of expression for their political behaviours and that SNSs use alone do not act as a mobilising agent.

This literature review highlights the contrasting empirical findings in extant research regarding the effect of the Internet and SNSs on Political Participation (see Appendix C for a full list of the studies consulted). Scholars have adduced several reasons as justification for these inconsistent empirical findings which are discussed in the next Section.

3.4.2. REASONS BEHIND INCONSISTENT EMPIRICAL FINDINGS

The literature review on the effect of Internet Use and its different tools (SNSs, blogs, micro-blogs, etc.) on Political Participation has highlighted opposing and inconsistent findings in extant empirical studies. Scholars ascribe such differences in research outcomes to several reasons, such as: (1) the contextual setting of the study (Anduiza et al., 2012; Norris, 2001), (2) the purpose of using the SNSs (e.g., Bakker and de Vreese 2011; Bode, 2012; Boulianne, 2009; 2015; Dimitrova et al., 2014; Gil de Zuniga et al., 2012; Kenski and Stroud 2006; Quintelier and Visser's, 2008; Rojas and Puig-i-Abril, 2009; Tang and Lee, 2013; Vitak et al., 2011; Xenos et al., 2014), or (3) the methodological operationalisation of Political Participation and in particular the lack of distinction between online and offline forms of political actions (Boulianne, 2009; 2015). However, relevant to this work are the justifications related to the unique "technological affordance" of each SNS and its intensity of use. These are addressed in the next two Sections.

3.4.2.1. SNSs' TECHNOLOGICAL AFFORDANCE

Affordance is defined as "the perceived and actual properties of the thing, primarily those fundamental properties that determine just how the thing could possibly be used" (Norman, 1988, p. 9). As McGrenere and Ho (2000) explain, affordance identifies possible characteristics and limitations of the technology which in turn defines how it should and should not be used.

An extended stream of the Media Studies literature focuses on the affordances of SNSs. Treem and Leonardi (2012) suggested that SNSs afford high visibility of information, persistent, instantaneous and continuous editability of the content, as well as aggregation among users. Likewise, Boyd (2010) described SNSs affordances as persistency and replicability of contents that can be easily retrieved and have the potential to target a large audience. Moreover, SNSs afford the creation and maintenance of relationships with large social networks which would be almost impossible to establish and manage offline (Ellison et al., 2014).

Scholars have often ignored the different intrinsic characteristics of the specific SNS considered (e.g., Facebook, Twitter, MySpace, etc.) and have generalised the effect of a specific SNS to the entire category (Dimitrova, et al., 2014; Woo-Yoo and Gil de

Zuniga, 2014). Arguably, this could be at the basis of the inconsistent empirical findings discussed earlier. Woo-Yoo and Gil de Zuniga (2014) and Skoric et al. (2016) highlighted the value of a differential approach and observed that different SNSs attract different types of individuals with respect to their social-economic status. Facebook and blogs facilitate Political Participation of more knowledgeable people, whilst Twitter use has a stronger relationship with Political Participation only among the most educated respondents. Furthermore, while blogs and microblogs can positively affect Offline Political Participation (Gil de Zuniga et al., 2013), Facebook was found to have no impact (Gil de Zuniga et al., 2012; Valenzuela et al., 2009) or even a negative effect (Vitak et al., 2011). Arguably, such opposing findings could be attributed to the different characteristics and affordances of different types of SNSs which therefore can affect different types of Political Participation and may indeed provide confounding results if analysed in the aggregate (Valenzuela et al., 2017).

For instance, the specific affordances of Facebook and Twitter (the most used SNSs in the UK, www. yougov.co.uk, 2018) could be the reason for the inconsistent findings highlighted by the literature review. Their different affordances include:

- 1. Network structure.
- 2. Types of information provided.
- 3. Types of social ties created.
- 4. Communication patterns.
- 5. Users' anonymity.

Considering *network structure*, Facebook is a symmetrical social media based on reciprocal approval that is, a user can connect to another only if there is the consensus and approval of both parties. As a result, Ellison et al. (2007) found that Facebook users have an online network that mostly mirrors the offline one as people tend to connect to those that they already know. Arguably, Facebook helps to maintain and nurture existing social ties rather than create new ones. On the other hand, Twitter allows more unidirectional connections, because reciprocal approval is not needed to follow and contact other users. Arguably, Twitter may facilitate the connection with people who are not know personally to the user, such as celebrities, politicians, news organisations, etc.

Ultimately, this may significantly enlarge the size of the social contacts within the Twitter network compared to Facebook.

With respect to the type of information provided, the bigger and more heterogenous social network of Twitter provide the potential to expose users to greater and more heterogeneous type of political information. In contrast, Facebook users are exposed to like-minded individuals and to news with which users have already shown greater affinity. Indeed, Bucher (2012) found that Facebook's "algorithm is based on the assumption that users are not equally connected to their friends. Some friends...count more than others...and...are those with whom a user interacts with on a frequent basis or on a more intimate level" (p. 1168). Thus, the algorithm shows news and feeds of friends (or pages) that users interact most frequently with. Some scholars have expressed concerns that people who are exposed predominantly to like-minded information online may become more polarised as they are enclosed in an "echo chamber", where there is a limited and restricted view of political issues which does not foster political debate and therefore does not facilitate political engagement (Hayat et al., 2016; Himelboim et al., 2013). Hence, the type of information and news obtained on Facebook is rather homogenous and somehow limited in amount compared with Twitter, whose algorithm rewards immediacy and recurrence, rather than interaction and affinity (Valenzuela et al., 2017). Indeed, the algorithm of Twitter shows posts to its users mostly on a real-time basis and displays more content from the sources that tweet more often, such as news companies, political groups, and social organizations. Consequently, Twitter users are potentially exposed to more novel and timely information, provided also by sources with whom they have no strong relationship. Research has shown that the exposure to diverse opposing and heterogeneous news sources, even if accidental, is a predictor of Political Participation as it may improve deliberative democracy in many ways, such as enhancing mutual understanding, promoting political tolerance, generating political debates, critical reasoning and elaboration of news (Gil de Zuniga et al., 2012; Mutz, 2006; Valenzuela et al., 2009; Vitak et al., 2011). In addition, Golbeck et al. (2010) found that politicians "are primarily using Twitter to disperse information, particularly links to news articles about themselves (...) and to report about their daily activities" (p. 1612). Arguably, the amount of political information is greater and more diverse on Twitter than on Facebook. As a result, Facebook could have weaker effects on people's political behaviours compared to Twitter.

Looking at the *types of social ties* there are several differences between Facebook and Twitter. Indeed, by strengthening and facilitating the maintenance of strong *relationships* (e.g., those with relatives and close friends), or what Putnam (2000) defines "strong ties", Facebook facilitates the creation and maintenance of Bonding Social Capital which is defined by high level of trust, reciprocity and emotional support created among like-minded individuals that can provide homogenous resources and information (see Chapter 4). In contrast, by allowing users to connect with people they do not know, Twitter facilitates the creation and maintenance of "weak ties", which generate Bridging Social Capital, characterised by weak emotional and affective support, while providing heterogeneity of information and resources (Putnam, 2000). Bonding Social Capital has been found to positively affect Offline Political Participation, meanwhile Bridging Social Capital seems to affect mostly Online Political Participation (Skoric et al., 2009; Zhong, 2014). Arguably, Facebook could be a better predictor than Twitter of Offline Political Participation, whilst Twitter could facilitate Online political activities better than Facebook.

The two SNSs show differences also in terms of *communication affordance*. Indeed, Twitter communication among users is centred on brief messages no longer than 280 characters (140 until 2017), as opposed to Facebook, which allows users to upload lengthier and more elaborated posts. These different communication structures may also explain why the two SNSs may have different effects on Political Participation across platforms. Indeed, weak and strong ties will have the same amount of space in Twitter to develop their message and because they must be short, users mostly focus on the informational value of the posts. Rather, on Facebook, users are exposed to longer and more elaborated messages and people are more likely to read the messages posted by closer contacts, with whom they feel more emotionally connected (Kaun and Stiernstedt, 2014). As a result, Facebook is more focused on reciprocal social interaction compared with Twitter, which is more centred around information and opinions sharing (Huberman et al., 2009; Petrocchi et al., 2015). Hence, the former is better at satisfying the needs for belongingness, social connectedness, interaction and self-presentation (Nadkarni and Hofmann, 2012), while the latter is better at fulfilling informational needs (Petrocchi et al., 2015). Indeed, previous research has shown that these differences may explain why people tend to learn more about political issues in one network over another (Woo-Yoo and Gil de Zuniga, 2014). Moreover, an informed citizen is at the basis of Participatory types of democracy (see Section 3.2) and it is a prerequisite for political engagement (Held, 2006). Therefore, Twitter may be a more suitable platform to increase levels of Political Participation.

Facebook and Twitter show differences also in terms of *anonymity*. Indeed, Facebook enables users to create an account only if they provide some information about themselves, meanwhile Twitter allows its users to remain anonymous as they do not need to disclose themselves to express their opinions (Huberman et al, 2009; Woo-Yoo and Gil de Zuniga, 2014). Arguably, anonymity would encourage greater participation online because it resolves the social desirability bias, allowing people to publicly take a stand on political matters that can be opposed by friends and associates (Collins and Butler, 2003; Towne and Herbsleb, 2012). Furthermore, anonymity has been argued to increase information flow. Indeed, anonymous interactions may allow people to transcend their stereotypes. In addition to the suppression of certain sensory information provided by the Internet (voice, smell, facial expression, etc.) anonymity may allow individuals to overcome emotional reactions that would interfere with their relationships. This would allow avoiding any judgmental fear for the information posted, even when this is against the society principles (Resnick, 2002).

Arguably, all those differences in the specific affordances of Facebook and Twitter may allow for different levels and types of Political Participation (Online and Offline). Yet, very few studies have investigated and compared the nuanced influence that these different SNSs have over participatory behaviours. Those that have done so have often failed to differentiate between online and offline Political Participation, analysing them as one unique construct or just focusing on a specific dimension of it. Therefore, such studies either did not account for structural differences of the Political Participation construct or failed to carry out any differential comparisons of the SNSs effects on different forms of Political Participation.

Indeed, after searching on ProQuest, Google Scholar and Jstor database for studies addressing this issue a total of 17 studies were identified (for the full list and more details about the search process see Appendix D). Of these, only two studies (Halpern et al., 2017; Woo-Yoo and Gil de Zuniga, 2014) have compared the effects of different types of SNSs on Online and Offline Political Participation.

3.4.2.2. SNSs' INTENSITY OF USE

Many studies investigated how the intensity of use of SNSs may be related to Political participation. Theoretically, given that Facebook's affordances facilitate the recreational and entertainment use of the network (Bode, 2012), more time spent on it may lead to the dominance of entertainment content on one's newsfeed (Lu et al., 2018). In this case, following the time displacement logic discussed earlier, a high amount of time spent on Facebook can hardly contribute to political mobilisation. Indeed, Vitak, et al. (2011) found the intensity of Facebook use to decrease general Political Participation, in terms of both online and off-line activities. In contrast, Bakker and de Vreese (2011) stressed that the Internet offers digital possibilities to engage in political activities, thereby intensive use of the Internet does not necessarily lead to decreased Political Participation. Rather, researchers should differentiate between different uses of SNSs (recreational, political, getting news, etc.) and between online and offline forms of Political Participation as the outcomes could vary significantly when accounting for such differentiation.

Extant research mirrors both points of views and provides inconsistent results. For instance, Baumgartner and Morris (2010), found that time spent on social networking sites, measured in days per week, was significantly associated with online political expression. Similarly, a Pew Center survey indicated that Facebook users who used the online platform many times a day were more likely to participate in online political activities compared to those who did not (Hampton et al. 2011). Likewise, Skoric and Kwan (2011), Skoric and Poor (2013) Skoric et al. (2016), Vitak et al. (2011) and Yung and Leung (2014) reported a positive link between Political Participation and the generic intensity of Facebook use. Arguably, such positive outcomes are understandable as political engagement on SNSs naturally requires users to spend time on them. However, the effect size of such relationships in those studies has always been small and mostly affected by other factors, such as the specific types of social media use (Chan et al., 2012; Mou et al., 2013), respondents' political interest and efficacy and the "contagion effect" of SNSs, which work by engaging people in politics by imitation simply because users see their friends performing political activities on the SNSs (Vitak, 2011).

On the other hand, Chunly (2019) showed that intensity of Facebook use was not related to online political behaviour. This finding is supported by several other studies that were unable to establish a link between time spent on social media and online Political Participation (Dimitrova and Bystrom 2013; Gil de Zuniga et al., 2012; Zhang et al. 2010). Other studies found a negative association between intensity of SNSs usage and online Political Participation (Yang and DeHart 2016). One possible reason is that some heavy SNSs users were so intensely absorbed by the social networking and entertaining aspects of this digital medium that they found online politics uninteresting or unworthy of pursuing (Yang and DeHart, 2016).

With respect to offline Political Participation, previous studies showed mixed and inconsistent findings too. For instance, while Bode (2012) found that time spent using Facebook is negatively related to voting, Skoric and Kwan (2011) found a positive relationship, but Chakraborty (2016) and Quintelier and Visser (2008) did not find any significant effects. Using the 2010 Pew Internet and American Life Project "Social Side of the Internet" survey, Zhang and Gearhart (2015) found that that general use of the Internet and SNSs increased online participation but not offline participation. Similar findings were found by Baumgartner and Morris (2010) and Towner (2013).

While the effects of the intensity of use of Facebook on Online and Offline Political Participation have been addressed, very few studies have investigated the effect of the intensity of use of Twitter. Indeed, a search on ProQuest, Google Scholar and Jstor yielded only 3 studies relevant to this thesis (see Appendix E). Those studies found a positive and significant relationship between time spent on Twitter and online (Hopke et al., 2016; Varnali and Gorgulo, 2015; Yang and DeHart, 2016) and offline (Hopke et al., 2016) Political Participation.

Given the logical link between SNSs' affordances and the time spent on them (Lu et al., 2018) and their effect on Political Participation, and that academic studies focusing solely on technology affordances or intensity of use of SNSs yielded contrasting empirical evidence, it seems logical to argue that an interaction effect between technology affordances and intensity of use could explain earlier opposing findings. Indeed, Field (2013) claims that interaction effects could be at the basis of the mixed evidence provided in the literature review about the effect of the same antecedent (SNSs use) on the same outcome (Political Participation). Yet, to the best of our knowledge, no study has investigated such interaction effect.

3.5. SUMMARISING: IDENTIFIED GAPS

Overall, the literature review on the relationship between SNSs use and Political Participation has highlighted the inconsistency of empirical findings. Several authors have argued that such contrasting findings can be related to the context in which the research is carried out (Anduiza et al., 2012; Norris, 2001). Others ascribe such differences to methodological measurement issues and particularly to the lack of differentiation between online and offline Political Participation (Boulianne, 2009; 2015; Lilleker and Koc-Michalska, 2017; Visser and Stolle, 2014). On the other hand, Dimitrova, et al. (2014), Woo-Yoo and Gil de Zuniga (2014), Skoric and Zhu (2016), Valenzuela et al. (2017) highlighted the value of "Technology Affordance" to explain the inconsistent empirical findings and observed that different SNSs may facilitate different types of participation in relation to the intrinsic characteristics and functions of the SNSs considered. Other authors blamed the lack of consideration of the intensity of use of SNSs (Valenzuela et al., 2009).

For this reason, several scholars have called for a focus on technology affordance (Boulianne, 2015; 2019; Kim and Lee, 2016; Skoric and Zhu, 2016) and intensity of use of SNSs (Valenzuela et al., 2009) when studying the effect of social media on Political Participation, also accounting for the heterogeneity of its online and offline dimensions (Boulianne, 2009; 2015; 2019). Indeed, a meta-analysis conducted by Boulianne (2019) found that extant research tends to measure social media use without identifying the specific platform. When studies have done so, the focus has mostly been on Facebook, investigating the effect of intensity of use of Facebook on Political Participation. Yet, the same is not true for Twitter. Boulianne's (2019) findings mirror the findings of the literature review reported in this Chapter. Most importantly, even the few studies that focused on intensity of use provided inconsistent findings of the effect of SNSs use on online and offline Political Participation.

Surprisingly, and to best of our knowledge, no study has investigated whether an interaction effect exists between the technology affordance and the intensity of use of the SNSs which could explain the mixed findings reported so far. Indeed, given that Facebook's affordances facilitate recreational, entertainment and social interaction activities, contrary to Twitter which facilitate information, opinions and news sharing and

diffusion (Huberman et al., 2009; Nadkarni and Hoffman, 2012; Petrocchi et al., 2015; Valenzuela et al., 2017), the amount of time spent by users on them could affect citizens' levels of Political Participation differently. Memon et al. (2019) claim that interaction effects could be at the basis of the mixed evidence provided in the literature review about the effect of the same antecedent (SNSs use) on the same outcome (Political Participation).
4. LITERATURE REVIEW ON SOCIAL CAPITAL

As highlighted in the previous Chapters, the concept of Social Capital can be employed for merging different disciplines, under one unique common interest, that is mobilising people for increasing civic and Political Participation and, thus, improve democracy and social well-being, at "a benefit for society at large". Such common interest is in line with the definition of Political Marketing provided in Chapter 2 and it emphasises the multidisciplinary nature of the concept of Social Capital and its relevance to Politics (Fukuyama,1995; Ostrom, 2000; Putnam, 1993, 1995, 2000; Skoric et al., 2009; Valenzuela et al., 2009), Economics (Page et al., 2005; Hayami, 2009), Marketing (Moore et al., 2005), Management and Organisational studies (Maurer et al., 2011), Sociology (Putnam, 2000), Ethics (Ayios et al., 2014) and many other disciplines (Kwon and Adler, 2014) (see Figure 4.1).

Such multidisciplinary use is due to the fact that Social Capital promises to explain outcomes based on social structures and interactions within social networks common to different contexts to the point that its recurrence "*in the title, abstract and key words of journal articles has multiplied rapidly over the past several decades*" (Kwon and Adler, 2014: p. 412), (Figure 4.2).



Figure 4.1. Social capital application fields.

Source: Created by author.



Figure 4.2. Research trends in Social Capital.

Source: Kwon and Adler (2014): p. 413.

However, such variety of uses leads to a wide range of conceptualisations, which implies that there is not a single universal definition of Social Capital (Adler and Kwon, 2002; Kwon and Adler, 2014; Woolcock, 2010). Rather, its conceptualisation varies and depends on the discipline, the contexts and types of investigation (Robinson, 2002). Indeed, Kwon and Adler (2014) argued that a cross-disciplinary universal definition is not required in so far as a clear conceptualisation and operationalisation of Social Capital are provided in each specific study.

This research follows this approach and tries to define Social Capital based on factors relevant to the current study, such as the individual or collective level of analysis (Section 4.1), the cultural and structural perspective of investigation (Section 4.2), the strong or weak nature of the ties in social networks (Section 4.3), and its offline and online nature (Section 4.4). Then, an account of the existing literature regarding Social Capital and its effect on Political Participation and on how the Internet is involved in this relation is provided (Section 4.5). Finally, gaps in the extant literature are discussed (Section 4.6). Figure 4.3 provides a conceptual map of this Chapter.

To accomplish the above, the following literature review refers to several seminal works and reviews of Social Capital like those of Adler and Kwon (2002), Borgatti (2003), Bourdieu (1986), Burt (1992), Coleman (1988, 1990), Fulkerson and Thompson (2008), Kwon and Adler (2014), Lin (2001, 2002), Putnam (1993, 1995, 2000), Williams (2006) and Woolcock (2010).



Figure 4.3. Conceptual Map Chapter 4.

4.1. SOCIAL CAPITAL LEVEL OF ANALYSIS: COLLECTIVE VERSUS INDIVIDUAL PERSPECTIVE

A first important debate over the concept of Social Capital is built around the level of its analysis, that is whether it must be considered as a collective/public good or a private/individual asset.

The *collective perspective* links Social Capital to the social relations embedded in social structures and communities (Knack and Keefer, 1995; Hanifan, 1913; Putnam, 2000; Skocpol, 1995). Such Communitarian approach stems from the work of Tocqueville (1835; cited in Putnam, 2000) who linked high levels of associational life with a well-functioning democracy. Putnam (2000) reinforced such position, arguing that social interactions within networks enrich individuals with democratic and cooperative values based on norms of generalised trust, reciprocity and tolerance. If a person develops such qualities and attitudes, he/she will be tolerant and will trust others. This approach clearly places the attitude of individuals (trust, reciprocity and tolerance) at the centre of the concept of Social Capital and it is based on a bottom-up mechanism, which means that those norms of reciprocity, trust and cooperation among people create good democracies and push economic and institutional development of a country benefiting society at large. Therefore, Social Capital is seen as property of a collective.

The *individual perspective* looks at Social Capital as a resource that benefits individuals through the relationships they establish among themselves. This approach stems from the rational utilitarian perspective of neo-structuralist sociologists like Bourdieu (1986), Coleman (1988, 1990), Flap (2004) and Lin (1999, 2001) for whom human beings are self-interested agents who seek to maximise utility and profit through access and use of scare resources which can help in achieving a better life condition. Accordingly, acquiring resources is the leading desire of the human behaviour.

Bourdieu (1986) considered Social Capital as an individual resource close to the symbolic, cultural, economic and political capitals and defined it as "*the aggregate of the actual or potential resources which are linked to possession of a durable network of more or less institutionalized relationships of mutual acquaintance or recognition*" (p. 248), "*made up of social obligations ('connections'), which is convertible, in certain*

conditions, into economic capital" (p. 243). The position of an individual in a hierarchical society is determined by the amount of economic, cultural and Social Capital he/she possesses. Then, the individuals constitute groups with the aim of gaining more resources or converting the capital they already own into different types of capital (mostly economic), to climb the different socio-economic stratifications of society. Just the individuals engaged in social relations benefit from the forms of capital brought in by others. Later, Flap (2004) and Lin (1999) further elaborated and simplified Bourdieu's conceptualisation of Social Capital as the accessible resources embedded in social networks, which an individual can directly or indirectly have access to. For them, such resources are information, social credentials, influence, and reinforcements which represent all individual assets. While Bourdieu (1986) considers the use of Social Capital for instrumental purposes, like a rise in economic, social or political position, Flap (2004) and Lin (2001) are more interested in the expressive purposes, that is, the use of resources for emotional support. However, regardless of the purpose for using Social Capital, the individual perspective considers it as an investment in social relations which allows individual to access and use resources embedded in social networks (Burt, 1992; Lin, 2001). The focus is on networks, resources embedded in such network and on whom people know (Halpern, 2005).

However, it seems that what is true for individuals also holds for groups. For instance, people with more individual Social Capital are more civically and politically engaged (Putnam, 2000). Similarly, it seems that societies with higher level of Social Capital show higher levels of civic and Political Participation (Putnam, 1993, 1995, 2000; Woolcook and Narayan, 2000). Hence, individual and collective levels are connected to some extent. Specifically, Glenville and Bienenstock (2009) claim that the individual level Social Capital contributes to create Social Capital at a more aggregate level, but at the same time collective Social Capital determines the type of investment in relationships made at the individual level.

This reasoning is mirrored also by some proponents of the individual tradition like Coleman (1988) and Lin (2001). They attempted to link the *sociological perspective* of social interactions with the more *atomistic perspective* of the economic theory. The former leaves little space for individuals' decisions since people actions are shaped by society. The latter has traditionally overlooked the social context in which people live and define their action on the basis of utilitarian self-interested actions.

In particular, Coleman (1990) who defined Social Capital as "some aspects of social structure [...] which facilitate certain actions of individuals who are within the structure" (p. 302) claims that there are norms and sanctions in a community (some aspects of social structure) which make it possible to act against one self-interest at a benefit of society at large. Such norms are based on trust and reciprocity, which Coleman (1988: p.102) introduced with the system of "credit slips" on moral obligations. Indeed, by doing a favour to an individual and trusting him/her (trust) to reciprocate in the future (reciprocity), an obligation is established. Sanctions are then applied in the case of not respected obligations. Accordingly, norms can be either internalised in the actor (private good) or externally enforced by a community through sanctions (collective good). In the latter case, because norms are enforced through relations among individuals embedded in social structures, Social Capital is considered as an inalienable aspect of the collective. However, Coleman adopted a more solid micro-sociological view, embedding the concept of Social Capital in the context of Rational Choice Theory, arguing that social interdependencies exist and arise among actors because they want to have access to resources owned and/or controlled by other individuals, in order to maximise their utility.

Similarly, Lin (2001) largely based his theory of Social Capital on the rational choice perspective of maximisation of profit and minimisation of loss, claiming that rational interactions lead to the creation of social structures that generate resources that are different from those owned by the single individual and that belong to the community. Although Lin (2001) concludes that a definition of Social Capital must include both, individual and structural elements, like Coleman (1990) he takes a more individualistic view claiming that the desire to maximise the amount of resources and to keep those already owned foster social interaction and therefore the individual Social Capital generate collective Social Capital.

Nevertheless, Social Capital is not located only in the actor but in the relationships among the actors (Coleman, 1988). Therefore, Social Capital "*constitutes both, and aid in accounting for different outcomes at the level of individual actions and an aid towards making the micro-to-macro transition*" (Coleman, 1988: p.101).

Therefore, as the economics Nobel laureate Ostrom claimed in his work with Ahn (2001), the individual and the collective views are not theoretically and analytically exclusive. In fact, even if is not clear what mechanism and process is behind the transition

from one level to another or which dimension causes the other, most academics agree that Social Capital exists at both levels (Coleman, 1990; Halpern, 2005; Fine, 2001; Nahapiet and Ghoshal, 1998; Ostrom and Ahn, 2003; Putnam, 2000) and benefit societies as well as individuals. For instance, a charity association and its network provide help for others (benefit for society), but also grants its members with friendship and valuables connections (benefit for individual).

Therefore, this research acknowledges Social Capital as being both, a public and an individual asset. However, this study will be focused on Social Capital at the individual level since it is interested in the individual levels of Political Participation with the awareness that individual Social Capital will ultimately benefit society at large.

4.2. SOCIAL CAPITAL QUALITY OF ANALYSIS: CULTURAL VS STRUCTURAL VIEW

Another central debate over the conceptualisation of Social Capital is related to the "Cultural" (Bourdieu, 1986; Coleman, 1988; Putnam, 1993, 1995, 2000) and "Structural" (Burt, 1995; Lin, 2001; Portes, 1998) perspectives.

Sociologists in the *cultural tradition* argue that social inequalities cannot be explained by the unequal distribution of economic capital alone, but that other forms of capital must be considered (Bourdieu, 1986). One form of capital is "Social Capital", that Bourdieu (1986) defines as a property of durable and helpful relationships that an individual may exploit to gain certain benefits or other forms of capital (Cultural or Economic) through membership in a group. Accordingly, the accumulation and internalisation of Economic and Cultural Capital require socialisation. Through socialisation, people can acquire new skills and/or cultural resources (books, paintings, etc.) increasing the chance to get better titles (Institutionalised Capital) and therefore better jobs, ultimately, maximising their utility. Socialisation creates a sense of belonginess which would push people to provide safety and mutual exchange of resources on the basis of norms of solidarity, reciprocity and trust.

Similarly, Coleman (1988, 1990) and Granovetter (1985) put emphasis on the usefulness of social relationships for creating and enforcing a cooperative environment based on trust and social norms which would facilitate the use and exchange of resources for maximising individual utility. Indeed, acting cooperatively and for the mutual benefit

require people to trust each other on norms of reciprocity, that is, I will do something for you, but then, if I need it, I trust that you will return the favour (Putnam et al., 1993). Therefore, Social Capital can be defined as "*features of social organisations such as trust, norms and networks that can improve the efficiency of society by facilitating coordinated actions*" (Putnam et al., 1993: p. 167). However, above all, *trust* is considered the lubricant of associational life and the same Social Capital may exist only if trust prevails in social networks (Putnam 2000, p. 20-21). Yet, relationships in such social networks are regulated also by *norms* of reciprocity and tolerance which are built to prevent bad externalities and decrease the cost of transactions among members of the network. Those norms are part of the cultural layer of the society an individual belongs to and specify what actions are rightfully acceptable (Coleman, 1988, 1990). Accordingly, Social Capital stems from cultural and social values.

On the other hand, the *structural tradition* considers the structures of networks existing among people relevant to the concept of Social Capital. In particular, the focus is on elements like the position occupied in networks, network dimension, structural holes, density and resources embedded in social relations.

Portes (1998) is among the first supporters of this perspective, stating: "Social Capital is the ability of actors to secure benefits by virtue of membership in social network or other social structures" (p. 6). However, Lin (2001) is considered as the main representative of the structural standpoint. He proposed a theory of Social Capital based on the neo-classical theory of capital, focused on "investment in social relations with expected returns in the marketplace" (Lin, 2001: p. 19). Accordingly, the core element of social interaction is profit and "individuals engage in interaction in order to produce profits" (Lin, 2001: p. 19). Therefore, like scholars in the cultural tradition, Lin (2001) sees individuals' connections and access to resources in such connections at the basis of Social Capital. However, Lin (2001) refuses the collectivistic proposition of Social Capital provided by the Cultural proponents, eliminating culture, norms, rules and trust from the set of dimensions which constitute Social Capital. Rather, Social Capital is an individual asset and because it is "capital", like all other forms of capital it implies investments. For Lin (2001) investments in trust and reciprocity do not exist, rather those are attitudes that are difficult to measure. For this reason, "interactions usually occur among actors with similar characteristics of resources and lifestyles. The greater the similarity the less effort is required in interaction" (Lin, 2001: p.75). Thus, interactions

are mostly "homophilus" (Lin, 2001: p. 39), meaning that those occur between actors that are similar (e.g., similar lifestyle, social status or economic characteristics) (see also Lazarsfeld and Merton, 1954; Laumann, 1966). Hence, individuals with similar resources interact more. This implies that "*maintaining resources is the primary motivation for action*" (Lin 2001: p. 75), since interacting with people that have our same resources would not grant access to other diverse forms of resources. This in turn means to interact with those who are closer in the social structure to us. Accordingly, "*The stronger the tie, the more likely the Social Capital accessed will positively affect the success of… an action*" (Lin, 2001: p. 75). Such homophily principle explains disparities in society since it causes inequalities in Social Capital and thus in ownership of and access to valuable resources.

On the other hand, actions directed at people who own different and more valuable resources (Heterophilus interaction) are very rare, as they require greater effort and higher levels of trust. Accordingly, *"The weaker the tie, the more likely ego will have access to better Social Capital"* (Lin, 2001: p. 76). Particularly, Lin refers to Granovetter (1974) who showed that, in order to have access to a different variety of information, an individual might need to go out of his/her social circle and connect to outside networks. Such kind of connections between the social circles are called "Bridges". The closer individuals are to a bridge in a network, the better is the Social Capital they can access to. Accordingly, resources are linked to social positions rather than to individuals. Indeed, whilst the occupant of a specific position may change, the resources remain in that position. In this sense, Lin (2001) distinguished between resources embedded in social structures and resources owned by individuals. Specifically, personal resources like education or wealth are fully owned by a person who can freely dispose of them. In contrast, social resources are accessible only through social connections and these are mainly power, reputation, information and credentials.

From these considerations, Lin (2001) derived his definition of Social Capital as "*all resources embedded in social structure which are accessed and mobilised in purposive actions*" (p. 29) and created his own model of Social Capital (Figure 4.4).



Figure 4.4. Lin's Model of Social Capital.

Source: Lin (2001): p. 76.

Notably, Lin (2001) approach is focused on structural constraints, hierarchical positions and structure of relationships (strong/weak ties) and his approach is a clear representation of the Structural perspective over Social Capital.

A third author who contributed to the Structural view of Social Capital is Ronald Burt (1992, 2000). He embedded the concept of Social Capital in the Structural Theory of Actions, according to which actors' goal is to maximise their utility through their personal financial (money) and human (e.g., intelligence and skills) resources. These personal resources can be expanded to maximise individual utility only through social resources (e.g., information, knowledge, contacts, etc.) which are the content of relationships among at least two actors and therefore are mutually owned. However, the maximisation of the utility is affected by the position of the actor in the social structure and by his/her interests. Therefore, interests and position in social structure determine social actions, which themselves affect the social structure (see Figure 4.5).

Therefore, Burt (1992) defines Social Capital as "friends, colleagues, and more general contacts through whom you receive opportunities to use your financial and human capital" (p. 9), or simply as "the brokerage opportunities in a network" (Burt, 1997: p. 355).



Figure 4.5. Structural Theory of Social Action Components and Interactions

Source: Burt (1982): p. 3.

According to Burt (1992), the access to Social Capital is determined by the position occupied in the network. Specifically, in any markets actors need to realise an optimal exchange, but in order to do so they need information, which is usually incomplete in the market. Therefore, actors will use their network structure to gather the required information. Accordingly, the structure of the network is a determinant of the success in a market. Structurally speaking, for Burt (1992) size, closure or openness of a network and redundancy of the contacts can have a relevant impact on the successful completion of the exchanges. Closure is given if no actor of the network has a relation with actors in other networks. Vice versa we will have an open network. Moreover, also the number of *non-redundant* contacts is relevant. A relationship is redundant if it leads to the same resources or in this case to the same information. If a relation is not redundant, then we have what Burt (1992) calls "Structural Hole", or "a relationship of nonredundancy between two contacts. [...] As a result of the hole between them, the two contacts provide network benefits that are in some degree additive rather than overlapping" (p. 18). To be additive, resources should come from heterogenous networks rather than from homogenous ones. Accordingly, structural holes represent weak ties. Therefore, in contrast to Bourdieu (1986) and Coleman (1988) and similarly to Lin (2001), for Burt (1992) valuable Social Capital exists in weak ties rather than in strong closed relations. Furthermore, Burt (1992) found that the number of structural holes is positively correlated to the size of the network. The bigger the size, the more structural holes are present in the network. Hence, when analysing Social Capital researchers should take into consideration structural aspects of social networks like size, contacts redundancy, closure and openness of the networks.

The structural view of Lin (2001) and Burt (1992) embedded in the rational choice theory has been widely criticised as they failed to pay attention to the feelings of people who actually love, like or hate one another and "*therefore associate together or avoid each other for reason that lie outside the domain of rational calculation*" (Field, 2008: p. 31). This is to say that the establishment of social relations and thus the resources embedded in them are not given just by the desire of people to maximise their utility. Hence, rational reasoning is not always a constant of social relations which instead are characterised also by irrational elements. Therefore, the "*ceteris paribus*" (maximization of utility) does not exist since each relation is the outcome of a combination of different aspects, including norms, trust, reciprocity, tolerance, etc. (Small, 2009).

Moreover, the structuralists approach mainly measure Social Capital through social networks' size and density which do not indicate anything about resources' availability and accessibility in a specific network, nor whether people are willing to share such resources, which in turn should imply mechanisms of trust and reciprocity (Finsveen and Van Oorschot 2008; Van Der Gaag and Snijders, 2005). Furthermore, two individuals may have the same network size and show the same frequency of contact but still the quality of the resources available in the network could be different as well as the quality of the relationships and therefore needs and goals cannot be met or satisfied in the same way (Finsveen and Van Oorschot 2008). Arguably, size and contact frequency are not enough for measuring the amount of Social Capital in social networks. Therefore, the *"Structural Perspective"* alone cannot provide a comprehensive and exhaustive representation of Social Capital.

However, even the "*Cultural Perspective*" by itself does not appear to be sufficient to provide a clear conceptualisation of Social Capital. For instance, the concept of group closure plays a critical role in both Bourdieu (1986) and Coleman (1990) conceptualisations of Social Capital. A social structure is closed if all the individuals in a group are connected to each other by some kind of relations. The level of closure has a positive correlation with the level of Social Capital possessed within the groups. Closed groups facilitate access to information and the establishment of reputation, trust and

reciprocity that maximise Social Capital. Such closed groups have strong relations. Accordingly, only strong ties can generate Social Capital (Coleman, 1988; Bourdieu, 1986). Notably, the cultural perspective neglects and overlooks the importance that weak ties may have in providing access to resources (Lin, 2001; Portes, 1998). In fact, closed structures prevent the entry of valuables resources like heterogeneous information and innovation into groups that can be generated by network characterised by weak ties. Moreover, Lin (2001) pointed out that by focusing only on norms of trust, solidarity and reciprocity, the cultural perspective neglects the structural assets of networks that can affect the creation and maintenance of such norms. Additionally, trust and reciprocity might be correlated also to Cultural and Human capital, but they are not considered as dimensions of those forms of Capital (Fine, 2001: p. 25). However, it has been argued that although Social Capital shares some similarities with those forms of capital, namely the non-monetary aspect (Bourdieu, 1986; Grootaert and Bastelaer, 2001), it is quite different from them. Firstly, it is "social", that means it implies social aspects like trust and reciprocity. Secondly, most of the definitions of Social Capital contain the concept of Social Networks as a "set of relations" (Knoke and Yang, 2008: p. 8) and in order for these relations to work and allow accessibility to and mobilisation of resources, people have to trust each other on norms of reciprocity (Bourdieu, 1986; Coleman 1988; Putnam, 2000). This holds true even if Social Capital is conceptualised in terms of returns over investments of money, time and energies in social relationships like in the more structural tradition (e.g., Lin, 2001). Indeed, such investments of scarce resources would not take place in absence of trust (Bourdieu, 1986; Coleman, 1988). Hence, it is difficult to overlook the importance of trust and reciprocity for Social Capital.

However, some proponents of the cultural tradition acknowledge the importance of social structure to generate norms of trust and reciprocity and take a more nuanced approach to the conceptualisation of Social Capital. For instance, Putnam (2000) introduced a differentiation between Bonding Social Capital and Bridging Social Capital which mirror the strong and week categorisation of social ties structures of Burt (1992) and Lin (2001). His example was followed by other authors who proposed a multidimensional conceptualisation of Social Capital merging the cultural and the structural perspectives. For instance, Nahapiet and Goshal (1998) proposed a threedimensional model (Figure 4.6), which has been widely adopted (e.g., Chang and Chuang, 2011; Lin, 2011; Zhao et al., 2016) and which presents:

- 1) A *Structural* dimension (network's ties, network's configuration and organisation).
- 2) A *Cognitive* dimension (shared codes, languages and culture).
- 3) A *Relational* dimension (trust, norms and obligations).

In Nahapiet and Goshal's (1998) model, the structural dimension correlates the structural view, while the cognitive and relational dimensions are related to the cultural view.



Figure 4.6. Nahapiet and Goshal (1998) Model of Social Capital.

Source: adapted from Nahapiet and Goshal (1998).

Subsequently, Uphoff (1999) also proposed a multidimensional conceptualisation of Social Capital based on the distinction between:

- 1) *A Structural* dimension, concerned with people's behaviours across different types of interpersonal interactions (like informal meetings with friends or formal participation through membership in various organisations);
- 2) A Cognitive dimension, which stems from people's perception of trust, shared values and beliefs.

Following the multidimensional tradition, this research acknowledges that Social Capital cannot be defined neither on the basis of the cultural view alone nor on principles of the structural perspective alone. Rather, the two views should be merged, and Social Capital should be considered as an individual as well as a public asset including both cultural and structural elements. The next section details and conceptualises the approach taken in this thesis.

4.3. CONCEPTUALISATION OF SOCIAL CAPITAL

In order to provide a clear and comprehensive definition of Social Capital which might include the elements common to all the perspective discussed so far (Cultural/Structural, Individual/Collective and Multidimensional) the conceptualisations of the major exponents of Social Capital belonging to those different views have been taken into consideration (Table 4.1). The final aim is to identify common points and contrasting elements to arrive at a comprehensive definition of Social Capital.

Common points to almost all the definitions and opposing perspectives appear to be the concepts of "*resources*" and "*social networks*". This observation is further validated by the meta-analysis on Social Capital conducted by Fulkerson and Thompson (2008) who found that the most frequently used terms in the definition of Social Capital are indeed "networks" and "resources" (Figure 4.7).

Networks and resources resonate with both the structural and cultural traditions of the individualistic perspective.

The importance of networks to Social Capital is given by the fact that they may facilitate or constrain the access and mobilisation of resources embedded at both individual and collective level by affecting perceptions, beliefs and actions of their members (Knoke and Yang, 2008). Examples are the network of a local Roman Catholic Church or of a workplace, which may be shaped by different values and whilst the former would include only people of the Roman Catholic faith, the latter would incorporate a variety of beliefs and faiths. Hence, they would affect access to resources by members of the network in different ways.

Perspective	Cultural		Structural		Multidimensional	
	Author/s	Definition	Author/s	Definition	Author/s	Definition
Individual	Bourdie	"the aggregate of the actual or potential	Portes,	"the ability of actors to secure	Nahapiet	"the sum of the actual and
	u, 1986:	resources which are linked to possession	1998:	benefits by virtue of membership	and	potential resources
	p. 248	of a durable network of more or less	p. 6	in social networks or other social	Ghoshal,	embedded within, available
		institutionalized relationships of mutual		structures".	1998:	through, and derived from
		acquaintance or recognition".			p. 243	the network of relationships
			Lin,	"all resources embedded in a		possessed by an individual
	Bourdie	" social obligationsconvertible, in	2001:	social structure which are		or social unit. Social capital
	u, 1986:	certain conditions, into economic capital	p. 29	accessed and/or mobilized in		thus comprises both the
	p. 243	and may be institutionalized in the form of		purposive actions".		network and the assets that
		a title of nobility".				may be mobilized through
			Burt,	"friends, colleagues, and more		that network".
	Colema	"Social capital is defined by its function. It	1992:	general contacts through whom		
	n, 1990:	is not a single entity, but a variety of	p. 9	you receive opportunities to use		
	p. 302	different entities having two		your financial and human		
		characteristics in common: They all		capital".		
		consist of some aspect of social structure,	Devet			
		and they facilitate certain actions of	Burt,	"the brokerage opportunities in a		
		individuals who are within the structure".	1997:	network".		
Collective	Dutnom	"features of social organization such as	p. 555	"maturally occurring social		
Conective	1005.	networks norms and social trust that	1002.	relationships among persons		
	n 67	facilitate coordination and cooperation for	n 100	which promote or assist the		
	p. 07	mutual honofit"	p. 100	acquisition of skills and traits		
		maraar benega .		valued in the marketplace an		
	Woolco	"the information, trust, and norms of		asset which may be as significant		
	ck.	reciprocity inhering in one's social		as financial bequests in		
	1998:	networks".		accounting for the maintenance of		
	p. 153	•		inequality in our society".		

Table 4.1. Conceptualisation of the Major Exponents of Social Capital.

Source: see references in Table.

Concept	Mainstream journals	Specialty journals	
Networks	29 (46%)	87 (36%)	
Resources	24 (38%)	68 (28%)	
Relationships	15 (24%)	67 (28%)	
Trust	13 (21%)	67 (28%)	
Reciprocity	11 (18%)	46 (19%)	
Individuals	25 (40%)	57 (24%)	
Norms	6 (10%)	48 (20%)	

Figure 4.7. Most Free	quently Used Socia	al Capital	Concepts ¹ .
0			

Note: Stratified random sample (n = 240) and mainstream journal articles (n = 70). Results are based on the *Social Science Citation Index*.

Source: Fulkerson and Thompson (2008).

Even the types of resources that could be accessed would be different. This consideration is relevant to the Technology Affordance Principle discussed in Chapter 3, and later in Section 4.4.3 for which different SNSs affordances may affect levels and types of Social Capital within social networks. Network is hereby considered in terms of structural forms of social relations (Flap, 2002; Lin, 2001), characterised by the quality of ties embedded in it (Flap, 2002; Putnam, 2000). One of the general laws of Social Networks Theory is that individuals with similar opinions and beliefs (Value homophily), cultural backgrounds (Status homophily) and who live in the same geographic area have larger and better opportunities of knowing each other (Lazzarsfeld and Merton, 1954; Mcpherson et al., 2001). This behaviour has been documented in several domains of life like friendship, religious affiliation, political membership, etc. and it is based on the "Social Focus Theory" (Homans, 1950). According to this theory people choose to focus on a specific activity on the basis of their own interests and cultural background. However, because people's time is limited, they must choose or give priority to one focus over another. Therefore, the focus is regarded as a constraining factor in developing social

¹ Mainstream Journals considered were the American Sociological Review, the American Journal of Sociology, the British Journal of Sociology, Social Forces, and Social Problems. *Specialty Journals* included in the analysis were Social Science and Medicine, Journal of Socio-Economics, World Development, Rural Sociology, Sociology of Education, Policy Sciences and American Behavioural Scientist.

relationships. Moreover, sociologically speaking similar people usually share common interests and attend same events (focus). This explains the high amount of similarity in networks. The Social Capital generated in such homogeneous and strong networks in often coined "Bonding Social Capital" and is characterised by strong ties providing high levels of moral and psychological support based on norms of trust and reciprocity reinforced by a close sense of belonginess to a group or network (Gittel and Vidal, 1998; Putnam, 2000).

Next to strong ties there are weak ties. The concept of weak ties and their beneficial effects are grounded on the seminal work of Granovetter (1973) (cited in Hofer and Aubert, 2013) who argue that the strength of ties is inversely related to the homogeneity of the parts involved in a relation. Wider contact diversity means diversity of information and resources that can be accessed through those networks. Therefore, establishing relations with people who belong to different closed groups, that is bridging different closed networks (Burt, 1992), can help to reach a different type of Social Capital that could be extremely beneficial for finding new information otherwise not available in closed structures. Particularly, weak ties have demonstrated to be useful for searching jobs (Williams, 2006), disseminate information (Granovetter, 1973; Williams, 2006) and enhancing tolerance towards others (Pettigrew and Topp, 2006). Because of their bridging function, weak ties generate Bridging Social Capital which is regarded as inclusive (Putnam, 2000) since it bridges across gender, race, geographical space and ethnicity. However, compared to strong ties, weak ties are characterised by lower levels of emotional connections. This can hinder their efficiency in cooperative behaviour for accessing scarce resource. Arguably, different ties can provide different types of Social Capital (Putnam, 2000). Hence, the introduction of network structures in the definition is essential to account for different types of Social Capital.

The second element that is almost always included in the conceptualisation of Social Capital is "resources" (Figure 4.7). Indeed, scholars agree that to assess the value of a network aspects like size, frequency of contact and quality of relationships existing in such network (weak/strong ties) are not enough (Van Der Gaag et al., 2008). In fact, possessing an equal network to that of another individual in terms of size or quality does not mean possessing the same amount and type of resources. Moreover, it does not indicate whether the resources are available and/or can be mobilised in the same measure. Like the term "networks", the word "resources" recurs in the definitions provided by authors belonging to opposing perspectives like Bourdieu (1986) and Lin (2001). Hence, its merging function with respect to those opposing perspectives comes to help in supporting the conceptualisation of Social Capital as a multidimensional concept. According to Bourdieu (1986) and Lin (2001) resources cover a wide set of life domains and are aimed at satisfying the needs of an "*average person in modern industrial society*" (Van der Gaag and Snijders, 2005). Following Lin (2001) both personal and social resources fall within the term "resources". Personal resources are those already possessed by an individual and can be further divided into symbolic (e.g., institutional accreditation) and material resources (e.g., house, pc, TV, money, etc.). They are also "actual resources" as individuals already own them. Rather, social resources are those acquired through social networks, for instance, having a friend of a friend recommending you or providing a good word when looking for a job. Those can be considered "potential resources" as they are not owned by individuals but are accessed only through successful social interactions.

Furthermore, different types of resources can be used for different purposes, namely, instrumental or expressive (Lin, 2001). Indeed, "*actors are motivated to either maintain or gain their resources in social actions – purposive actions. Action to maintain resources can be called expressive action, and action to gain resources can be called instrumental action*" (Lin 2001: p. 75). The expressive dimension refers to Social Capital as an end as well as a mean to get emotional support (e.g., sharing sentiments, providing psychological support, etc..). Returns of expressive actions concern physical health, mental health and life satisfaction. The instrumental actions involve the use of Social Capital as a means of reaching a specific goal, like looking for job, finding someone to babysit or getting a loan, and result in social (e.g., the favourable or unfavourable opinions over an individual in a social network, namely reputation), economic (e.g., disposition of goods) and political (e.g., decisional power) returns. This research considers both these type of actions as connected to personal and social resources.

Rarer seems the use of dimensions like trust and reciprocity, stemming from the cultural perspective. However, as it has been argued in Section 4.2, a social network is a *"set of relations"* (Knoke and Yang, 2008: p. 8) and for these relations to work and allow accessibility to, and mobilisation of resources, people have to trust each other on norms of reciprocity (Bourdieu, 1986; Coleman, 1988; Putnam, 2000). Indeed, social networks generate Social Capital if those people who do favours to other individuals trust

(expectation) that they will return the favour when they need it (obligation). Hence, trust requires reciprocity in forms of obligations (Coleman, 1988). However, following Putnam (2000) and Perugini et al. (2003) this research adopts a more virtuous definition of reciprocity which is regarded as an individual's tendency to reciprocate others regardless of any beneficial, instrumental outcomes. In case of strategic behaviour, reciprocity is rooted in the "self-interest maximization assumption" of the rational choice theory, while in the case of personal individual tendency, reciprocity is embedded in the concept of "internalised social norms" (Dohmen et al., 2009; Perugini et al., 2003). The main distinctive feature of the internalised social norms is that people who hold an internalised norm are likely to conform to the directives of a social rule (in this case reciprocity) even when they are not observed, externally sanctioned or cannot gain any personal instrumental benefit. This is in line with the concept of Virtue and Civic Virtue discussed earlier and which guides this study (see Chapter 3). Hence, Social Capital is here understood as:

"The amount of actual and potential resources embedded in social network structures and which can be accessed and mobilised in purposive actions on the basis of internalised norms of trust and reciprocity".

The proposed definition stems from those of Bourdieu (1986), Bourdieu and Wacquant (1992), Lin (1999), Nahapiet and Goshal (1998), Woolcock (1998) and Putnam (1995) and merges the different perspectives of Social Capital discussed in the previous two Sections following a long-established tradition in the field of Social Capital.

4.4. SOCIAL CAPITAL AND THE INTERNET

Early studies on Social Capital are focused on the relevance of face-to-face interactions. However, much of the relational interaction that people engage in nowadays is in the online environment (Gainous et al., 2013). This raises the issue of whether the Internet may provide the same benefits in terms of Social Capital creation and/or maintenance of offline interactions. Indeed, different views exist on this issue, which according to Quan-Haase and Wellman (2004) and Neves (2013) can be summarised as follow:

- 1. Internet use increases Social Capital.
- 2. Internet use decreases Social Capital.

3. Internet use transforms Social Capital.

An extensive review of literature identified two additional viewpoints (Cole et al., 2000; Franzen, 2003; Helliwell and Putnam, 2004; Uslaner 2004):

- 4. Internet use does not affect Social Capital.
- 5. Internet use produces negative Social Capital.

According to Resnick (2002), several technological affordances make the Internet a promoting agent of Social Capital. Firstly, it removes barriers to interaction allowing for communication across time and space and, as we have seen, interaction is a necessary condition for building Social Capital. Secondly, the Internet might make it possible to interact in much larger social networks. Larger social networks allow for more social ties and more paths to follow to access scarce resources (Bourdieu, 1986; Burt, 1992; Fisher, 1982). Thirdly, it may increase information flows: particularly interactions which maintain a sort of anonymity may allow people to transcend their stereotypes. This, together with the suppression of certain sensory information (voice, smell, facial expression, etc.) may allow individuals to overcome emotional reactions that would interfere with their relationships. Fourthly, the Internet may help to maintain records of previous interactions and feedback which can create accountability and help to promote collective identity and a sense of trust (Skoric et al., 2009).

Indeed, several studies suggest that through its various forms of expressions like blogs, email and so forth, the Internet is a fundamental mean through which people may maintain and create new social relationships, thus developing Social Capital (Ellison et al., 2007; Ho et al., 2003; Skoric et al., 2009; Valenzuela et al., 2009; Gil De Zuniga et al., 2012). Specifically, studies which found a positive association between Social Capital and the Internet may be further divided in three categories (Neves, 2013): those which show that the Internet reinforce offline Social Capital (Kavanaugh et al., 2005; Kraut et al., 2002; Quan-Haase and Wellman, 2004; Robinson and Martin, 2010), those which show just a positive relationship among the two variables (Alessandrini, 2006; Bauernschuster et al., 2011; Wang and Wellman, 2010) and those which claim that the Internet creates and maintains Social Capital offline (Boase, 2006; Hampton et al., 2011; Miyata et al., 2008).

However, other authors suggest that the Internet decreases Social Capital. In particular, Putnam (2000) and Norris (2001) claim that the Internet is used mainly for

recreational activities and entertainment and it occupies a large proportion of people spare time and thus it prevents people from spending time in socially desirable activities, in a logic of "*time displacement*" (Nie et al., 2002; Zhang et al., 2010). Similarly, Kraut et al. (1998) found that Internet usage not only decreased Social Capital, social interaction, levels of individual trust and participation in civic activities, but also it increased loneliness. Nie and Hillygus (2002) also identified the detrimental effect of Internet use on Social Capital, as it reduces the time spent with family, friends and on meaningful social interactions. Moreover, Nie et al. (2002) argued that even if the Internet may expand social networks size, the impact on Social Capital is ambiguous: in fact, when weak ties within a larger social network displaces strong ties within a smaller network, it is rather questionable whether this represents loss or gain of Social Capital.

Other authors suggested that the Internet transforms Social Capital. In particular, Williams (2001) acknowledged the fact that the Internet may displace offline Social Capital, but he argued that it may also create new forms of Social Capital in the cyber space. Williams's (2001) findings show that although the Internet is not a good environment for developing Bonding Social Capital, it promotes the creation of online weaker ties which may offer more knowledge on public issues with the possibility of fostering social engagement.

Another stream of the literature suggests that the Internet does not significantly affect Social Capital. For instance, Uslaner (2004), found that the use of the Internet is irrelevant for both, trust and sociability. Trust was not related to online frequency and vice versa. The Internet from this point of view was a neutral instrument (Robinson and De Haan, 2006) or, in Uslaner's (2004) words, "*an additional outlet*" (p. 13) of interaction for those people who were already connected with each other in the offline world. Several authors supported such view, providing empirical evidence. For instance, Franzen (2003) found that Internet use was irrelevant to both network size and time spent socialising (two of the variables used to measure Social Capital; Cole et al., 2001). Similarly, Wellman et al. (2001) found that online interaction supplements face-to-face communication without increasing or decreasing it. However, they also highlighted the positive effect on social involvement for heavy Internet users somehow contrasting the time displacement logic. Indeed, Resnik (2002) argued that a time displacement logic would hold true if the Internet occupies time that would otherwise be spent in face-to-face interactions or socially desirable activities. Similarly, Robinson and Martin (2010) did not find any

evidence of time displacement of Internet use and activities related to Social Capital, such as church attendance and face to face meetings. However, if the displaced activities are driving or watching television, then the effect of the Internet will be still positive, even in a logic of time displacement as it would allow for some forms of interaction with others that driving or watching television would not.

Finally, Helliwell and Putnam (2004) argued that the Internet may produce negative Social Capital. For instance, the Internet has been used to "*mobilise extremists and help terrorist fund and conduct their activities*" (Skoric et al., 2009: p. 417). An emblematic example is represented by ISIS's propaganda. However, it can be argued that, even if under a negative light, the Internet has still produced Social Capital. Such negative nature of the Social Capital produced through the Internet is reinforced by studies in organizational behaviour. Indeed, it has been argued that although the Internet may lead to higher connectivity and new forms of working (e.g., instant messaging, skype, zoom, Microsoft Teams etc..) it creates at the same time higher social distance because it is used by employees to create a sort of shield with respect to their superior (Quan-Hase et al., 2005).

Notably, empirical research on the effects of the use of the Internet on Social Capital has not revealed any clear pattern and such trend is observable also in SNSs studies which are examined in the next Section.

4.4.2. THE INCONSISTENT SNSs EFFECT

Because SNSs are centred around the building and maintenance of multiple social networks for interaction purposes, scholars in the field of Social Capital started to investigate their contribution to the field (see Chapter 3). However, similarly to what the literature highlighted for the use of the Internet in general, research has produced contrasting evidence of the impact of SNSs use on Social Capital. Indeed, three contrasting effects have been identified: positive, neutral and negative.

Most studies on SNSs use and Social Capital found a positive relationship between the two. For instance, Ellison et al. (2007) found that U.S. undergraduate students who use Facebook display higher levels of Bridging and Bonding Social Capital. However, given the cross-sectional nature of their study, a clear causal direction could not be assessed. Hence, Steinfield et al. (2008) conducted a longitudinal analysis examining the same sample one year later. Their findings indicated that the intensity of Facebook use is a strong predictor of Bridging Social Capital. Similarly, Barkhuus and Tashiro (2010), Brandtzaeg (2012), Brandtzaeg et al. (2010), Burke et al. (2010), Burke et al. (2011), Ellison et al. (2011), Hofer and Aubert (2013), Johnston et al. (2013), kaigo (2012), Lampe et al. (2013), Lee et al. (2014), Sajuria et al. (2015), Stutzman et al. (2012) and Toriumi and Kamiko (2017) found positive associations between the use of SNSs and levels of Social Capital. Scholars identified the ability of the SNSs to considerably increase the size of the social networks, as main reasons for such positive relationship. Participants can access more resources, especially information, and are able to keep track of previous interactions and feedback (likes, posts, tweets, comments, etc.) which can create accountability and may help to promote collective identity and a sense of trust within the network, consequently increasing Social Capital. This optimistic perspective is shared also by recent studies conducted during the COVID-19 pandemic, arguing that SNSs must play a larger role in generating Social Capital by creating and nurturing social connections that ultimately would allow to mitigate many social issues caused by this peculiar context where face-to-face interactions are necessarily limited (Pitas and Ehmer, 2020). Indeed, scholars called for further investigation of the effects of SNSs based Social Capital on education (Giavrimis and Nikolaou, 2020) mental health (Rodela et al., 2020) and on many other fields overall sharing an optimistic view of the effects of SNSs on people social life (Beldad, 2021; Daly et al., 2020).

Other authors found that SNSs use do not significantly affect Social Capital. For instance, Hofer and Aubert (2013) found that Twitter use does not influence Bonding Social Capital. Also, Papacharissi and Mendelson (2011) found no relation between either the number of times that people check Facebook or the amount of time they spent on it and Social Capital. Similarly, although Vitak et al. (2011) found a positive relationship between Facebook use and Bridging Social Capital, they found no significant relationship with the Bonding dimension. Brooks et al. (2014) observed that social network size and the number of visits on Facebook are not related to Bridging or Bonding Social Capital.

Finally, some studies provided empirical evidence that SNSs negatively impact Social Capital. Indeed, Bohn et al. (2014) found that Bridging Social Capital decreased with Facebook network size and that an intensive use of SNSs could lead to decreasing levels of Social Capital. Also, Li et al., (2018) found that the use SNSs for specific actions (e.g., informational, participatory, or recreational) contributes to online but not to offline Bonding and Bridging Social Capital. This finding resonates with studies showing that social interaction on Facebook primarily contributes to Social Capital online but not offline (Li et al., 2015).

Overall, empirical studies investigating the relationship between Internet and SNSs use and Social Capital have generated mixed results (see Appendix F) for a full list of the studies consulted). Scholars attributed those inconsistent empirical findings to several reasons, as discussed in the next Section.

4.4.3. REASONS FOR INCONSISTENT EMPIRICAL FINDINGS

The literature review on the effect of the Internet and SNSs use over Social Capital has highlighted the extant opposing and inconsistent findings in empirical studies. Scholars in the field ascribed such differences to five main reasons.

Firstly, the *context* in which the research is undertaken. Different countries and even different social contexts within the same country may facilitate the creation and maintenance of different types of Social Capital depending on economic and social development. The latter, in turn, can affect also the intrapersonal characterises of the population under study, for instance literacy, self-esteem, well-being etc., that can hinder or facilitate social interaction (Putnam, 1995, 2000).

Secondly, Social Capital is a *multidimensional and multidisciplinary concept* (see Section 4.1 and Section 4.2) and as such is a complicated construct that is conceptualised in several different ways (Appel et al., 2014; Engbers et al., 2017). The lack of consistency in the unit of analysis (individual or collective, see Section 4.1), the cultural or structural operationalisation (Section 4.2) and the differentiation between the Bridging and Bonding dimensions of Social Capital are emblematic of a general uncertainty of what to measure and how. Arguably, this may lead to the contrasting empirical findings highlighted by the literature.

Thirdly, Williams (2006, 2007) showed that although the *time spent online* displaced offline activities and therefore was negatively related to offline Social Capital, it was positively related to online interaction and online Social Capital. However, these differential implications have been largely overlooked by subsequent studies which failed to differentiate between consequences of SNSs use on offline or online Social Capital and have rather merged the two, yielding contrasting results (Li et al., 2018). Indeed, those

studies that have differentiated between online and offline Social Capital (e.g., Brooks et al., 2014; Collins and Freeman, 2013; Hofer and Aubert, 2013) generated a more nuanced picture of the dynamic relationships between Internet use and various forms of Social Capital and showed that SNSs use may enhance both online and offline Social Capital, but that the strong connection is with the Bridging online dimension of Social Capital.

Fourthly, SNSs use is often measured through time spent online focusing on one single digital platform (e.g., Facebook or Twitter), but the results of its effects on Social Capital are generalised to the entire Social Media category even though people use multiple digital media technologies for diverse purposes (Hampton, Lee, and Her, 2011; Li et al., 2018; Valenzuela et al., 2017). Indeed, different SNSs have different functions, settings and algorithms which affect users' perceptions of the needs that each SNS can satisfy (Brandtzaeg, 2012; Petrocchi et al, 2015; Phua et al., 2017; Valenzuela et al., 2017). For instance, Facebook is more focused on reciprocal social interaction and entertainment compared to Twitter which instead is more centred around information and opinions sharing (Petrocchi et al., 2015; Huberman et al., 2009). Hence, the former is better at satisfying the needs for belongingness, social connectedness, interaction and self-presentation (Nadkarni and Hofmann, 2012), while the latter is better at fulfil informational needs (Petrocchi et al., 2015). At the same time research has shown that different types of needs are better satisfied by different types of Social Capital (Putnam, 2000). Whereas Bonding Social Capital satisfy the need for social and psychological support through mechanisms of trust, reciprocity and solidarity embedded in closed homogenous groups characterised by strong ties, Bridging Social Capital bridges across different heterogeneous social contexts and as such it is better at spreading information through weak ties. Therefore, Facebook should affect mostly Bonding Social Capital, meanwhile Twitter should influence levels of Bridging Social Capital because of their respective technological affordances (an in-depth discussion of the Technology Affordance principle and differences between Facebook and Twitter was provided in Chapter 3). Indeed, the literature review highlighted that empirical research on online Social Capital found that Facebook use predicts and increases both its dimensions, namely Bridging and Bonding (Ahn, 2012; Donath and Boyd 2004; Ellison et al., 2007; Sckoric et al., 2009; Steinfield et al., 2008), whilst MySpace use was associated with Bonding Social Capital only (Ahn 2012; Greenhow, 2011). In contrast, Twitter use has been associated with high levels of Bridging Social Capital online, but unlike Facebook it does not seem to influence the Bonding dimension of Social Capital (Hofer and Aubert, 2013). Notably, the types of SNSs used have a significant bearing on the amount of Bonding and Bridging Social Capital accrued by their users (Skoric et al., 2016). Therefore, Social Capital Online, like its counterpart offline, relies on the context of its acquisition (Coleman, 1998; Putnam, 2000). Indeed, in the real world, the Social Capital acquired at school or in the workplace is not the same as the Social Capital acquired at church (Brady et al., 1995; Coleman, 1998; Putnam, 2000), because of differences in ties strengths and social networks' levels of heterogeneity present in those different contexts. Arguably, in the virtual world the Social Capital gained using Facebook should differ from the Social Capital acquired on other SNSs because of differences in their thematic functions and intrinsic characteristics which make for differences in technological affordances. However, most of the studies identified through the literature review, analyse one SNS and tend to generalize the results to all platforms (e.g., Ellison et al., 2007). Arguably, this is problematic, because each SNS has unique characteristics which may uniquely affect different types of social ties (strong/weak) and consequently facilitate the creation and maintenance of different types of Social Capital (Bonding/Bridging). Only recently some authors have adopted an affordance approach to investigate the specific uses of SNSs and their implications for online Social Capital (Ellison and Vitak, 2015; Pearce et al., 2015; Su and Chan, 2017; Valenzuela et al., 2017). Yet, most of those studies make no comparisons between SNSs (technology inter-affordance) but rather compare effects of different affordances within the same platform (technology intra-affordance), usually Facebook (Su and Chan, 2017). An example is the comparison between the "Liking" and the "Comment" function. The use of the "liking" function, which is considered an example of social grooming, together with the activity to post comments on Facebook, were found to be predictive of Bridging but not of Bonding Social Capital (Ellison et al., 2014; Burke et al., 2011). However, Lee et al. (2014) found the "liking function" to be positively related to Bonding Social Capital, whereas the frequent use of "comment" was negatively associated with it. Similarly, Su and Chan (2017) found that opinion expression on Facebook was negatively, and significantly, related to Bridging Social Capital, but positively related to its Bonding dimension. Notably, studies focusing on SNSs intra-affordance have reported inconsistent findings. Arguably, such pattern may be true also for different SNSs provided their differences in technological affordances.

The above discussion suggests that "technological affordances and user participation on different SNSs have an influence on Social Capital creation, raising further questions regarding possible ways to achieve an effective user engagement on these platforms" (Skoric et al., 2015: p. 1821). Yet, extant studies tend to measure SNSs use and generalise their effects without taking into consideration technological affordances (Boulianne, 2019).

Lastly, many studies investigated how the intensity of use of SNSs may be related to Social Capital (Section 4.4.2). Theoretically, given that Facebook's affordances facilitate the creation and strengthen of strong ties, more time spent on it may lead to the dominance of Bonding Social Capital. On the other hand, Twitter should create and reinforce strong ties and, therefore, the time spent on it should lead to a dominance of Bridging Social Capital. Yet, also in this case, findings are inconsistent (see Appendix G for a list of all the studies consulted).

Provided the logical link between SNSs' affordances, the time spent on them and their effects on Social Capital and that empirical evidence focusing solely on technology affordances or intensity of use of SNSs yielded contrasting results of their relationship with Social Capital, it can be argued that an interaction effect between the two could explain such opposing findings. Indeed, interaction effects could be at the basis of the mixed evidence provided in the literature review about the effect of the same antecedent (SNSs use) on the same outcome (Social Capital) (Field, 2013). Yet, to the best of our knowledge, none of the extant studies have investigated such interaction effect and analysed how the amount of different types of Social Capital (Bonding/Bridging) owned by users of different SNSs (Facebook/Twitter) varies across levels of intensity of use of those digital platforms.

4.5. SOCIAL CAPITAL, POLITICAL PARTICIPATION AND THE INTERNET

Previous research has generally established a strong connection between Social Capital and Political Participation (Klesner, 2004; Steinfield et al., 2008; Skoric et al., 2009; Teorell, 2003). Such consensus has a long tradition, from the classical work on Political Participation of Almond and Verba (1963) about Civic Culture, to later studies like those of Parry et al. (1992), Putnam (1993, 1995, 2000), Shah et al. (2001b), Verba e Nie (1972), Verba et al. (1995) and Zhang and Chia (2006).

However, this line of research was developed mostly in the offline context, utilising measures of face-to-face interpersonal relations; hence there is no consensus on whether the positive effect of high levels of Social Capital over Political Participation does hold true also in the online context. In fact, several studies have investigated the effects of the Internet and its forms (Blogs, SNS, etc.) on Political Participation, but with contrasting findings.

With the development of the Internet and SNSs, social interaction started to take place also online and many people have increasingly embraced SNSs to fulfil their informational (Gil de Zuniga et al., 2017; Xenos et al., 2014), emotional, and social needs (Gan and Wang, 2015; Hołyst, 2017; Zhan et al., 2016). This is because SNSs provide people with the chance to easily access multiple social resources like information, help or financial capital at any point in time and seemingly at no cost, when compared with the offline social networks (Phua et al., 2017; Resnick, 2002). Amongst all the resources which can be accessed on social networks, there is also Social Capital (Ellison et al., 2007; Gil De Zuniga et al., 2012; Valenzuela et al., 2009). Given the strong arguments in favour of a link between offline Social Capital and offline Political Participation (e.g., Lake and Huckfeldt, 1998), scholars started to investigate whether Social Capital Online could be linked to participatory behaviour online and offline (Gil de Zuniga et al., 2012; Gil de Zuniga et al., 2015; Valenzuela et al., 2015; Valenzuela et al., 2017).

Indeed, online and offline political activities tend to be closely interrelated (Bakker and de Vreese, 2011). It seems that political and civic organisations (and people in general) draw on their online social networks to implement offline action (Bennett and Segerberg, 2012; Rojas and Puig-i-Abril, 2009). Recent examples include the "Arab

Spring", the "Indignados" in Spain, the "Gilets Jaunes" protest movement in France and the "Black Lives Matter" movement. Even political parties, like the Five Star Movement in Italy, have recently mobilised their online networks for on-the-ground political action. Therefore, scholars have hypothesised that the Social Capital embedded in online social networks would foster not only online but also offline political actions (Gil de Zuniga et al., 2016).

However, extant research on this topic is very limited (Tsatsou and Zhao, 2016) and the few existing empirical studies have showed that this relationship is ambiguous and not always direct or positive. For instance, Gil de Zuniga et al. (2012) argued that Social Capital influences offline participation both indirectly and directly, whereas it only has an indirect effect on online participation through the use of the digital platforms for getting news. In a later study, Gil de Zuniga et al. (2016) found that online Social Capital was a significant predictor of online Political Participation. However, they found a negative relationship with offline activities like voting. Yet, Valenzuela et al. (2017) found that Facebook and Twitter are indirectly related to offline Political Participation through its association to information received from strong and weak ties. Other authors have shown that online Social Capital predicts only online participation (Valenzuela, 2013; Yamamoto et al., 2013). In contrast, Carr (2008) and Lohr (2012) found that associations and relations in online social networks have no impact on levels of participation. Valenzuela et al. (2009) found that Online Bridging Social Capital significantly predicts only Online Political Participation and that Online Bonding Social Capital has no effects on Political Participation whatsoever. Instead, Skoric et al. (2009) found that also the Bonding dimension of Social Capital significantly and positively affects Political Participation, but only in the offline environment. Meanwhile, Gibson and McAllister (2013) showed that only Online Bonding Social Capital is significantly and positively related to Political Participation and that Online Bridging Social Capital is not correlated with any political activities. However, those studies were focused on the usage of only one SNS, mainly Facebook (e.g., Valenzuela et al., 2009), and Internet use in general (Gibson and McAllister, 2013; Skoric et al., 2009). On the other hand, research on Twitter showed that it can positively affect both online and offline Political Participation (Graham et al., 2013; Jungherr, 2016).

The above discussion suggests that technology affordance could be a differentiating element of analysis, given that different characteristics and intrinsic

functions of different SNSs (e.g., Facebook and Twitter) could explain the different and opposing findings of empirical studies (see Chapter 3). Arguably, contrasting and inconsistent empirical findings regarding the effect of an independent variable (Online Social Capital) on the same dependent variable (Political Participation) raise the need to investigate whether a moderating effect can explain such differences (Memon et al., 2019). Yet to the best of our knowledge, none of the extant studies have considered the moderating effect that the use of different SNSs may exert on the relationship between Online Social Capital and Political Participation. This is surprising, given that Facebook technological affordances are found to facilitate the creation, maintenance and reinforcement of strong ties which lead to the creation of Online Bonding Social Capital, whereas Twitter can foster the reinforcement of weak ties which may enhance Online Bridging Social Capital (Valenzuela et al., 2014; Valenzuela et al., 2017). In turn, Online Bonding and Bridging Social Capital may be linked to different types of Political Participation. Indeed, Bonding Social Capital provides emotional connections reinforced by trust on closer individuals within the family and friends' networks. This provides the social weight and contagion effect needed for influencing individuals' behaviours and take collective actions which need greater moral support like protesting, marching etc. (Centola, 2011; Putnam, 2000). Bridging Social Capital is rather characterised by weak contacts and heterogeneous networks which allow for more novel and diverse information diffusion and political contacting (Somma, 2009). However, the level of trust and reciprocity embedded in Bridging Social Capital is much lower compared with Bonding Social Capital. Hence, its influence on political behaviour is limited to actions which do not require high moral support, like voting (Gladwell, 2010; Morozov, 2011). Arguably, the effect of Online Bonding and Bridging Social Capital on Political Participation may be moderated by the type of SNSs used, based on the principle of technological affordance. This argument is supported by recent meta-analyses in the field (e.g., Boulianne, 2009, 2015, 2019; Skoric et al., 2016) which found that not only there are very few attempts to investigate the effects of online networking on political engagement, but that extant research has neglected to investigate whether differences among SNSs exists with respect to such relationship.

Furthermore, nowadays a large majority of people use two or more SNSs simultaneously (Duggan, 2015; Halpern et al., 2017; Petrocch et al., 2015). Arguably, this is because each and any of those SNSs has its unique features and affordances and

therefore may satisfy different needs (Brandtzæg, 2012; Huberman et al., 2009; Petrocchi et al., 2015; Phua et al., 2017). Given that different SNSs have different affordances that can facilitate the creation and maintenance of different types of Social Capital (Section 4.4.2 and Section 4.4.3) and that can foster different types of participatory behaviour (Chapter 3) it is conceivable that the combined and not combined use of two different SNSs can moderate the relationship between Online Social Capital and Political Participation. Yet, to the best of our knowledge, no study has investigated such effect. However, it might be that the contrasting empirical evidence discussed earlier over the effects of the Online Social Capital on Political Participation depends on the combined use of different types of SNSs together which have different technological affordances and therefore can moderate the relationship between Social Capital and Political Participation.

Moreover, many studies investigated how the intensity of use of SNSs may be related to Social Capital (Section 4.4.3) and Political Participation (Chapter 3). Arguably, given that Facebook's affordances facilitate the maintenance and reinforcement of offline strong ties (Petrocchi et al., 2015; Nadkarni and Hofmann, 2012), the time spent on it may moderate the relationship between Online Bonding Social Capital and participatory behaviour that need high personal psychological and moral support to be undertaken mostly in the offline environment like protesting, marching, etc. (Valenzuela et al., 2017).

In contrast, Twitter is centred around information and opinions sharing and is better at fulfilling informational needs through weak ties which can create social support and develop a sense of reciprocity (Huberman et al., 2009; Petrocchi et al., 2015; Valenzuela et al., 2017). Thus, the time spent on Twitter may affect the relationship between Online Bridging Social Capital and participatory behaviour that need social support (creating petitions, join civic movements, etc.).

However, to the best of our knowledge, no extant study has investigated the possible moderating effect that the time spent on SNSs can have on the relationship between Online Social Capital and Political Participation.

4.6. SUMMARISING: IDENTIFIED GAPS

A systematic review of the literature on Social Capital highlighted several gaps concerning its relationship with the use of SNSs and with Political Participation.

Specifically, the literature review on the relationship between SNSs use and Social Capital revealed the contrasting findings yielded by empirical research. Indeed, while several studies found a positive relationship (e.g., Barkhuus and Tashiro, 2010; Brandtzaeg, 2012; Brandtzaeg et al., 2010; Burke et al, 2010; Burke et al., 2011; Ellison et al., 2007; Ellison et al., 2011; Hofer and Aubert, 2013; Johnston et al., 2013; kaigo, 2012; Lampe et al., 2013; Lee et al., 2014; Sajuria et al., 2015; Steinfield et al., 2008; Stutzman et al., 2012; Toriumi and Kamiko, 2017) others highlighted no effects (e.g., Brooks et al., 2014; Hofer and Aubert, 2013; Papacharissi and Mendelson, 2011; Vitak et al., 2011) or even a negative relationship (Bohn et al., 2014; Li et al., 2015; Li et al., 2018).

Scholars have ascribed such conflicting findings to several reasons but mostly to: (1) differences in technological affordances of different SNSs and to (2) levels of Intensity of use of those SNSs (Bohn et al., 2014; Boulianne, 2019; Hampton et al., 2011; Li et al., 2018; Su and Chan, 2017; Valenzuela et al., 2017).

Indeed, the literature review highlighted how studies have focused on one SNS or on all of them together and then have generalised the results to the whole category of Social Media. Yet, each SNS has unique features and affordances and therefore may satisfy different needs (Boulianne, 2019; Brandtzæg, 2012; Petrocchi et al., 2015; Phua et al., 2017). For instance, Facebook is more focused on reciprocal social interaction compared with Twitter which instead is more centred around information and opinions sharing (Petrocchi et al., 2015; Huberman et al., 2009). Hence, the former is better at satisfying the needs for belongingness, social connectedness, interaction and selfpresentation (Nadkarni and Hofmann, 2012) while the latter is better at fulfilling informational needs (Petrocchi et al., 2015). At the same time research has shown that different types of needs are better satisfied by different types of Social Capital. Indeed, according to Putnam (2000) there are two types of Social Capital, namely Bridging and Bonding Social Capital. The latter is suitable for satisfying the need for social and psychological support through means of trust and solidarity embedded in mechanisms of social interactions. In contrast, Bridging Social Capital, is better for linkage to external assets and for information diffusion" (Putnam, 2000: p. 22). Considering that different types of SNSs satisfy different needs and that different needs are met through different types of Social Capital, researchers inferred that different types of SNSs affect different types of Social Capital. Indeed, studies on online Social Capital identified through the literature review found that Facebook use predicts and increases both Bridging and Bonding online Social Capital but that the strongest connection is with the Bridging dimension (Donath and Boyd 2004; Ellison et al., 2007; Sckoric et al., 2009; Steinfield et al., 2008). Meanwhile, Twitter use has been associated with high levels of Bridging Social Capital online, but unlike Facebook it does not seem to influence the Bonding dimension of Social Capital (Hofer and Aubert, 2013). Accordingly, the time spent on different SNSs may lead to the dominance of different types of Social Capital (Bonding and Bridging) based on the specific affordance of the SNSs included in the analysis. Provided the logical link between SNSs' affordances and the time spent on them and their effect on Social Capital and that empirical evidence focusing solely on technology affordances or intensity of use of SNSs yielded contrasting results of their relationship with Social Capital, it can be argued that an interaction effect between the two could explain such opposing findings. Indeed, interaction effects could be at the basis of the mixed evidence provided in the literature review about the effect of the same antecedent (SNSs use) on the same outcome (Social Capital), (Field, 2013). Yet, to the best of our knowledge, no study has investigated such interaction effect and analysed how the number of different types of Social Capital (Bonding/Bridging) owned by users of different SNSs (Facebook/Twitter) varies across levels of intensity of use of those digital platforms. The identified gaps are supported by scholars in the field who recently called for research related to platforms effects following the affordance principle (Boulianne, 2019; Halpern et al., 2017; Valenzuela et al., 2017)

With respect to the relationship between Online Social Capital and Political Participation, the systematic literature review revealed contrasting empirical evidence too. For instance, Gil de Zuniga et al. (2016) found that Online Social Capital was a significant predictor of online Political Participation. However, they found a negative relationship with offline activities like voting. Yet, Valenzuela et al. (2017) found that Facebook and Twitter are indirectly related to offline Political Participation through their association to information received from strong and weak ties. Other authors have shown that online Social Capital predicts only online participation (Valenzuela, 2013;

Yamamoto et al., 2013). However, Carr (2008) and Lohr (2012) found that associations and relations in online social networks have no impact on levels of participation. Instead, research on Twitter showed that it can positively affect both the online and offline Political Participation (Graham et al., 2013; Jungherr, 2016). The above discussion suggests that technology affordance could be a differentiating element of analysis provided that different characteristics and intrinsic functions of different SNSs (e.g., Facebook and Twitter) could somehow explain the different and opposing findings of empirical studies. Arguably, inconsistent empirical findings regarding the effect of an independent variable (Online Social Capital) on the same dependent variable (Political Participation) call for an investigation of whether a moderating effect can explain such differences (Memon et al., 2019).

Yet to the best of our knowledge, no extant study has considered the moderating effect that the use of different SNSs may exert on the relationship between Online Social Capital and Political Participation.

Furthermore, nowadays a large majority of people use two or more SNSs simultaneously (Duggan, 2015; Hampton et al., 2011) to satisfy different needs (Brandtzæg, 2012; Huberman et al., 2009; Petrocchi et al., 2015; Phua et al., 2017). Provided that different SNSs have different affordances that can facilitate the creation and maintenance of different types of Social Capital and that can foster different types of participatory behaviour it is conceivable that the combined and not combined use of two different SNSs can moderate the relationship between Online Social Capital and Political Participation. Yet, to the best of our knowledge, no study has investigated such effect.

Also, many studies investigated how the intensity of use of SNSs may be related to Social Capital (Section 4.4.3) and Political Participation (Chapter 3). Arguably, given that Facebook's affordances facilitate the maintenance and reinforcement of offline strong ties (Petrocchi et al., 2015; Nadkarni and Hofmann, 2012), the time spent on it may moderate the relationship between Online Bonding Social Capital and participatory behaviour that need high personal psychological and moral support to be undertaken mostly in the offline environment like protesting, marching etc. (Valenzuela et al., 2017).

In contrast, Twitter is centred around information and opinions sharing and is better at fulfil informational needs through weak ties which can create social support and develop a sense of reciprocity (Huberman et al., 2009; Petrocchi et al., 2015; Valenzuela
et al., 2017). Thus, the time spent on Twitter may affect the relationship between Online Bridging Social Capital and participatory behaviour that need social support (creating petitions, join civic movements, etc).

However, to the best of our knowledge, no extant study has investigated the possible moderating effect of the time spent on SNSs on the relationship between Online Social Capital and Political Participation.

5. RESEARCH AIM, QUESTIONS AND THEORETICAL FRAMEWORK

The extensive review of the Political Marketing, Political Participation and Social Capital literature related to the use of SNSs has identified several gaps (Table 5.1).

STREAM OF	
LITERATURE	GAPS
Political Marketing	 Political Marketing as a discipline: Has not engaged with theories of democracy, ethical questions and theories of participation attracting criticisms which have hindered the research in the field. Has been focusing on the welfare of political parties and relevant stakeholders and for this reason it has been the object of criticisms. Politics has a higher purpose than commercial profitability and should benefit society at large, rather than just parties and lobbies.
Social Capital and Political Participation	 Empirical research on the effects of Online Social Capital on Political Participation both online and offline has yielded contrasting findings (positive/negative/no effect). Arguably a moderation effect on this relationship can be provided by: 3. The specific technological affordance of each SNS. 4. The intensity of use of the SNS. 5. The combined or not combined use of the SNS.
SNSs use and Social Capital	 Empirical studies on the effects of SNS use on Online Social Capital have yielded contrasting evidence (positive/ negative/no effect). Arguably, the amount of Online Social Capital owned by individuals may differ with respect to: 6. An interaction effect between the type of SNS employed and its intensity of use.
SNSs use and Political Participation	 Empirical studies on the effects of SNS use on Political Participation online and offline focusing solely on one SNSs or on its intensity of use have yielded contrasting evidence (positive/ negative/no effect). Arguably, the amount of Political activities performed by citizens online and offline may differ with respect to: 7. An interaction effect between the type of SNS employed and its intensity of use.

Table 5.1. Gaps in extant literature.

The two gaps concerning Political Marketing are related to theoretical aspects and empirical process of research. Hence, to address those gaps, this study will investigate citizens' political participatory behaviours, providing a research framework based on Social Capital at the individual level (Chapter 4). Notably, Social Capital originates from Sociology but is linked also to the Participatory Theories and the Communitarianism perspective of Politics and to Virtue Ethic (Chapter 3), because it embeds concepts like civic virtue, trust, reciprocity, social relationships and intimacy.

The remaining gaps are addressed through the following research questions:

- **Q1.** Does the type of SNSs used moderate the relationships between Online Social Capital and Political Participation? To what extent?
- **Q2.** Does the intensity of use of SNSs moderate the relationships between Online Social Capital and Political Participation? To what extent?
- **Q3.** Does the combined use of SNSs moderate the relationships between Online Social Capital and Political Participation? To what extent?
- **Q4.** To what extent do levels of Online Social Capital differ across different types of SNSs employed at different levels of their intensity of use?
- **Q5.** To what extent do levels of Political Participation differ across different types of SNSs employed at different levels of their intensity of use?

To fill the gaps identified through the literature review and answer the proposed research questions, this study aims to investigate the impact of the affordances of different types of Social Network Sites (Facebook and Twitter) and their intensity of use (time spent online) on sharing cooperative and mobilising behaviours of people in the context of Social Capital and Political Participation. The broad setting is the growth of *"mediatisation"* and *"digitalisation"* of social interactions through social media, under the assumption of *"technological determinism"* and *"technology affordances"* (Chapter 3).

To investigate the above, this study employs three main concepts. Firstly, **Social Capital** which mirrors cooperative and sharing behaviour in social interactions based on norms of trust, solidarity and reciprocity (Chapter 4). Specifically, this research focuses on social interactions online. Hence, the focus is on Online Social Capital (Williams, 2006). The second concept is **Political Participation**, representing mobilised behaviour in the form of all the voluntary and purposeful activities, individually or collectively carried out by citizens, which target and affect the political and civic spheres, through both legal or illegal praxis online and offline (Ekman and Amna, 2012; Fox, 2014; Van

Deth, 2014), (Chapter 3). The third concept is **Online Social Networks** representing the "*mediatisation*" and "*digitalisation*" of social interactions through SNSs differentiated according to their affordance and their intensity of use (Chapter 3).

The underlying theoretical frameworks of this research which bring together the above concepts are: the *Resource Mobilisation Theory* (Zald and McCarthy, 2002) which is employed to answer research questions Q1, Q2 and Q3; the *Social Capital Theory* (Bourdieu, 1986; Coleman, 1988; Putnam, 2000) which is used to answer research question Q4; and the *Social Network Theory* to answer research question Q5. These theories are expanded with the "*Technology Affordance Principle*" (Norman, 1988) to test for interaction and moderating effects of the different types of SNSs employed and their intensity of use.

The Resource Mobilisation Theory refers to the "instrumental rationality model" and infers that resources such as money, education, skills, etc., affect Political Participation. According to this theory, Political Participation is a rational behaviour given by a desire for change. The resources available to people represent the link between that desire for change and the ability to mobilise around that specific desire (Zald and McCarthy, 2002). Therefore, those with more resources are more likely to participate in political activities. Following Bourdieu (1986), Bourdieu and Wacquant, (1992) and Lin (2001), Social Capital is hereby considered a resource like all other types of capitals (human, financial, etc.). Indeed, activated by human relationships, Social Capital fulfils productive activities when people use the relationships as resources to achieve their goals (Coleman, 1988). Specifically, in this study Social Capital is understood as a resource made of the sum of all types of actual and potential resources embedded in social networks and which can be accessed and mobilised on the basis of norms of trust and reciprocity (Bourdieu and Wacquant, 1992; Putnam, 2000). Trust and reciprocity facilitate cooperative behaviour among people of the network (Putnam, 2000; Norris, 2002). In turn, cooperative behaviour will facilitate the mobilisation of the resources present in the network which are needed to perform a specific action. Those resources can take the form of useful information, psychological help, advice, financial help, reference for a job, personal relationships, the capacity to organize groups etc. (Paxton, 1999). According to the literature review, it is possible to clarify the role of Social Capital as predictor of Political Participation based on three aspects:

- Trust.
- Interactions.
- Associations in Social Networks.

Trust reduces complexity for individuals while providing them with a sense of security which encourages interaction and cooperation (Warren, 1999), and make them willing to participate in the process of producing and bearing the cost of public goods through Political Participation (Park, 2004). Moreover, social networks function as a 'school' which educate people for civic virtues, encourage the spread of information and promote mobilisation. Indeed, Helliwell and Putnam (2007) and Campbell (2009) have shown that when individuals are members of a social network, they tend to participate more. Moreover, people are more likely to become involved in politics when they are asked to do so, especially if the request comes from someone they know within their social network (Campbell, 2013; McClurg, 2006).

Thus, following the Resource Mobilisation Theory, Social Capital created on SNSs and the amount of it "owned" by individuals is a predictor of Political Participation. However, Social Capital is a multidimensional concept made of two dimensions, namely Bonding and Bridging (Putnam, 2000). Such differentiation is relevant to the Resource Mobilisation Theory since several studies have demonstrated that Bridging and Bonding Social Capital give access to different resources and consequently may facilitate different kinds of political behaviours (Chapter 4). Moreover, the lack of differentiation between Bonding and Bridging dimensions was argued to represent one of the reasons for the contrasting results regarding the effects of Social Capital on Political Participation (Boulianne 2015; Gustafsson, 2012; Skoric et al., 2009; Williams, 2006). Furthermore, Political Participation is investigated in its conventional and unconventional forms in both the online and offline environments (Dalton, 2008; Dalton et al., 2010; Ekman and Amna, 2012; Norris, 2002; Parry et al., 1992; Van Deth, 2014). Such distinction between different forms of Political Participation is advocated by scholars who emphasise how the effects of the different types of Social Capital on Political Participation vary in relation to various political activities (Chapter 4, Section 4.5). Hence, the operationalization of such variable through one unique dimension could drive to misleading results and could be at the basis of the divergent findings highlighted in the literature review (Boulianne, 2009; Gibson and Cantijoch, 2013; Nisbet and Scheufele, 2004; Vaccari, 2012).

Given that Online Social Capital is conceptualised through the Bonding and Bridging dimensions and that Political Participation is examined in its Online and Offline forms each of the five main research questions has several subsidiary questions which are formulated as reported in Table 5.2.

To account for inconsistent findings in the extant research about the effect of Online Social Capital on Political Participation (Chapter 4), the "*Technology Affordance*" principle is recalled (Norman, 1988) and the type of SNSs employed with their intrinsic affordances and their level of intensity of use are tested for moderation within the Resource Mobilisation Theory framework. Indeed, the existence of contrasting and inconsistent empirical findings regarding the effect of an independent variable (Online Social Capital) on the same dependent variable (Political Participation) calls for an investigation of whether a moderating effect can explain those differences (Memon et al., 2019). Given that different SNSs have different affordances and that people, usually, employ more than one SNS (Petrocchi et al., 2015), the combined use of SNSs is argued to moderate the relationship between Online Social Capital and Political Participation.

Therefore, the types of SNSs used, their intensity of use and their combined and not combined use are tested for moderation on the relationship between online Bonding and Bridging Social Capital and Political Participation online and offline.

Main Question	Subsidiary Question
Q1. Does the type of SNSs used moderate the relationships between Online Social Capital and Political Participation? To what extent?	Q1a. Does the type of SNSs used moderate the relationships between Online Bonding Social Capital and Online Political Participation? To what extent? Q1b. Does the type of SNSs used moderate the relationships between Online Bonding Social Capital and Offline Political Participation? To what extent?
	Qlc. Does the type of SNSs used moderate the relationships between Online Bridging Social Capital and Online Political Participation? To what extent?
	Qld. Does the type of SNSs used moderate the relationships between Online Bridging Social Capital and Offline Political Participation? To what extent?
Q2. Does the intensity of use of SNSs moderate the relationships between Online Social Capital and Political Participation? To what extent?	Q2a. Does the intensity of use of SNSs moderate the relationships between Online Bonding Social Capital and Online Political Participation? To what extent? Q2b. Does the intensity of use of SNSs moderate the relationships between Online Bonding Social Capital and Offline Political Participation? To what extent?
	Q2c. Does the intensity of use of SNSs moderate the relationships between Online Bridging Social Capital and Online Political Participation? To what extent? Q2d. Does the intensity of use of SNSs moderate the relationships between Online Bridging Social Capital and Offline Political
	Participation? To what extent?
Q3. Does the combined use of SNSs moderate the relationships between Online Social Capital and Political Participation? To what extent?	Q3a. Does the combined use of SNSs moderate the relationships between Online Bonding Social Capital and Online Political Participation? To what extent? Q3b. Does the combined use of SNSs moderate the relationships between Online Bonding Social Capital and Offline Political Participation? To what extent?
	QSc. Does the combined use of SNSs moderate the relationships between Online Bridging Social Capital and Online Political Participation? To what extent?
	Q3d. Does the combined use of SNSs moderate the relationships between Online Bridging Social Capital and Offline Political Participation? To what extent?
Q4. To what extent do levels of Online Social Capital differ across different types of SNSs employed at different levels of their intensity of use?	 Q4a. To what extent do the amounts of Online Bonding Social Capital differ across different types of SNSs employed, considering different levels of intensity of use of SNSs? Q4b. To what extent do the amounts of Online Bridging Social Capital differ across different types of SNSs employed, considering different levels of intensity of use of SNSs?
Q5. To what extent do levels of Political Participation differ across different types of SNSs employed at different levels of their intensity of use?	Q5a. To what extent do levels of Offline Political Participation differ across different types of SNSs, considering different levels of Intensity of use of SNSs? Q5b. To what extent do levels of Online Political Participation differ across different types of SNSs, considering different levels of Intensity of use of SNSs?

Figure 5.1 portrays the research framework of this thesis under the Resources Mobilisation Theory, expanded through the moderating effects proposed to answer research questions Q1, Q2 and Q3 and all their subsidiary questions.





However, having access to specific resources on social networks (in this instance to Social Capital) does not imply that people will be able or willing to perform a specific political behaviour (Lin, 2001). Indeed, one of the criticisms to the Resource Mobilisation Theory and Social Capital as related to Political Participation is that, rather than using their resources to participate in politics, people "*might be much more inclined or necessitated to use their resources for more important, more relevant, more pleasant, less threatening, or less demanding areas of life*" (Van Deth, 2000: p. 119-120). Moreover, according to Krishna (2002) "*the automaticity assumed in the Social Capital argument namely, that high Social Capital leads directly to greater Political Participation—is not analytically or conceptually clear*" (p. 439). Arguably, other factors must be considered when studying Political participation. Indeed, political studies showed that people who are likely to perform political actions usually have high levels of political knowledge (Norris, 2000; Bimber, 1998), as they use SNSs to stay informed and receive political news (Gil de Zuniga et al., 2012), and also have high levels of political interest (Krueger, 2006; Melo and Stockemer, 2014; Skoric et al. 2009; Verba et al., 1995; Wolfinger and Rosenstone, 1980), education (Pattie et al., 2004) and internal political efficacy (Moeller et al., 2014; Beaumont, 2010).

Such results support the Cognitive Engagement Theory (CET) which proposes that an individual's Political Participation is the result of access to information and his levels of education, political interest and self-efficacy (Inglehart, 1977; Pattie et al., 2004). The underlying assumption is that the more people are interested (Political Interest) and informed about civic and political affair (Political knowledge), the more they believe they can succeed in performing a specific political behavior (Internal efficacy), and therefore they will participate in politics. However, citizens must possess a modicum of education and the capacity to think critically to be capable of engaging in the civic and political life (Comber, 2007; Morse, 1993). Moreover, the lower the cost of access to information, the more citizens consume information from the media and the higher the level of political knowledge and interest among citizens which further leads to increase in Political Participation. Arguably, the emergence of social media has reduced the cost of accessing political information. Therefore, given that this study is focused on the study of SNSs and on the level of Political Participation displayed by citizens, the CET theory is of particular relevance. Hence, the control variables Education (Education), Political Interest (PInt), Political Knowledge (Pknow) and Political Efficacy (PEff) are introduced in the proposed theoretical framework.

Furthermore, previous studies in the field of political and social engagement provided empirical evidence that age may affect the levels of an individual Political Participation (Eisenstadt, 1956; Franklin, 2004; Goerres, 2007; Inglehart, 1977; Marshall, 1952; Melo and Stockemer, 2012; Pattie et al, 2004; Quintelier, 2007) together with gender (Briggs, 2017; Burns et al, 2001; Childs, 2004; Karp and Banducci, 2008; Verba et al, 1997). Indeed, the literature suggests that men are more prone to participate in politics than women with respect to both conventional (i.e., party membership) and unconventional (i.e., cause-related NGOs) forms of Political Participation (Calenda and Meijer, 2009; Kaufhold et al., 2010). Moreover, several studies provide empirical

evidence that males are more eager to participate in online forms of political activities than women (Albrecht, 2006; di Gennaro and Dutton, 2006; Gibson et al., 2005). Additionally, while young individuals are usually the most disengaged, elderly people have shown higher levels of participatory behaviour in political and civic life and many studies in the field of Political Participation have controlled for it (Gil De Zuniga et al., 2012; Skoric et al., 2009; Valenzuela et al., 2009). Hence, this research also controlled for gender and age. The final model to answer research questions Q1, Q2 and Q3 and all the subsidiary questions is represented in Figure 5.2.



Figure 5.2. Model with Moderation Effects and Control Variables.

Social Capital Theory is employed to answer research question Q4 and all its subsidiary questions and ascertain whether the amount of Online Social Capital (Bonding and Bridging) owned by SNSs users differs across different types of SNSs employed and at different levels of their intensity of use. Although Social Capital Theory claims that association and interaction in social networks can foster the creation of Social Capital, the review of the literature has highlighted contrasting results with regards to the effect of online interactions. Indeed, it seems that the use of SNSs may increase, decrease or have no effect at all on Social Capital (see Chapter 4). Several reasons were identified

through the literature review to explain such conflicting findings, but this research focuses on the differences in the affordances of the type of SNSs used and their intensity of use. Given that extant studies focusing on one SNS (Facebook or Twitter) and their intensity of use yielded contrasting empirical evidence over the effect of SNSs on Social Capital, we argue that an interaction effect between the specific type of SNS employed and its intensity of use may exist. Hence, we integrate and expand the Social Capital Theory with the *Technology Affordance Principle* to contribute to the ongoing debate over the effect of online interaction on Social Capital. This reasoning is in line also with the principle of "Context Acquisition" arising from Social Capital Theory (Coleman, 1998). Indeed, as in the real world the Social Capital acquired at school is not the same as the Social Capital acquired, say, at the Church, so in the virtual world the Social Capital gained through the use of Facebook may differ from the Social Capital acquired on other SNSs. Arguably, this could be due to differences in the thematic focuses and functions of those SNSs (Petrocchi et al., 2015; Valenzuela et al., 2017).

To answer research question Q5 and all its subsidiary questions and to investigate to what extent Political Participation differs across different types of SNSs employed at different levels of their intensity of use, the Social Network Theory (Granovetter, 1973; Putnam 2000; Wellman, 1983) is expanded through the Technology Affordance **Principle** similarly to what discussed above for the Social Capital Theory. The Social Network Theory argues that direct participation in the social networks (even if online) of everyday life promotes Political Participation through several mechanisms. First, scholars found that association in online social networks may provide users with political news and mobilising information ((Dimitrova et al., 2014; Gil de Zuniga et al., 2013; Holt et al., 2013; Pasek et al., 2009; Tang and Lee, 2013; Towner, 2013; Xenos et al., 2014). This exposure to political content increases citizens' political knowledge. According to research on traditional offline media, the more knowledgeable individuals are more likely to participate in meaningful political actions (Boulianne, 2015). This holds true also for online social networks even when exposure to such information is accidental. Accidental information can still become influential because it has been filtered by trusted sources within users' networks like family members and friends (Bode, 2012). Secondly, online social networks offer the chance to develop ties with political or civic groups that seek volunteers for their respective causes (Bode et al., 2014; Musick and Wilson, 2008; Tang and Lee, 2013). Citizens who belong to civic or political associations are more likely to

be involved in politics because their memberships increase the chance of being asked to participate (Musick and Wilson, 2008; Verba et al., 1995). A third mechanism refers to the political discussion and deliberation which take place on SNSs and that has been shown to exert a positive effect on participatory political behaviour (Halpern and Gibbs, 2013; Vraga et al., 2015).

Furthermore, other scholars have focused on the processes of "social contagion," according to which a person tends to imitate social behaviour of close and trusted members of a network. Arguably, political expressions and opinions shared on online social networks may be contagious and may mobilise their users (Vitak et al., 2011). Lastly, Social Network Theory argues that, through interaction, people develop shared attitudes and norms based on social trust and reciprocity (Putnam 2000; Wellman, 1983), which ultimately promote Political Participation as a common good. Regardless of the mechanism, Social Network Theory argues that the attributes of individuals are less important than their relationships and ties with other actors within the network which can facilitate or hinder the above mechanisms (Burt, 1992; Granovetter, 1973; Lin, 2001). Therefore, different network structures can hinder or facilitate Political Participation. Yet, the network structures depend on the context on which those networks are formed (Coleman, 1988). Translated into the online environment of the SNSs that have different affordances, this is equivalent to say that different SNSs can facilitate different network structures of Political Participation.

Some scholars in the field do support such argument. Indeed, Dimitrova et al. (2014), Woo-Yoo and Gil de Zuniga (2014), Skoric et al. (2016) and Valenzuela et al. (2017) highlighted the value of "Technology Affordance" to explain how different SNSs may facilitate different types of participation in relation to the intrinsic characteristics and functions of the SNSs considered. However, also in this case the literature review on the relationship between SNSs use and Political Participation highlighted the inconsistent findings in empirical studies (Chapter 3). Many authors blamed the lack of consideration of the intensity of use of SNSs in many studies as the reasons for the opposing empirical evidence (Valenzuela et al., 2009). Most importantly, even the few studies that focused on technology affordance and carried out cross-platforms investigation or those which focused on intensity of use provided inconsistent findings of the effect of SNSs use on Political Participation both online and offline (Skoric et al., 2016). Hence, this thesis will investigate whether the type of SNS used and the intensity of use of that specific SNS

interact in affecting Political Participation. In doing so, this research adheres to the argument that, to avoid misleading results (Chapter 3), "*a distinction between online and offline political activities should be maintained*" (Vissers and Stolle, 2014, p. 950).

6. RESEARCH STRATEGY, DESIGN, METHODS, MEASURES, SAMPLING AND TECHNIQUES OF ANALYSIS.

This chapter details the research approach taken in this study following the systematic procedure proposed by Bryman (2012) and reported in Figure 6.1.



Figure 6.1. Research Process

Source: adapted from Bryman (2012).

To tackle the research questions proposed in Chapter 5, we adopted a quantitative research strategy (Section 6.1) with a cross-sectional design (Section 6.2) employing an on-line self-administered questionnaire (Section 6.3) comprising measures used in previous studies (Section 6.4). PLS-SEM analysis and two-way parametric and non-parametric ANOVAs are employed to analyse the data (Section 6.5) gathered through a web panel sample (Section 6.6). The questionnaire was pre-tested through a pilot study (Section 6.7). Justification and limitations of the chosen approach are discussed in the related Sections.

6.1. RESEARCH STRATEGY

The choice of research strategy is linked to the research paradigm underpinning the researcher's beliefs in terms of knowledge development and interpretation. According to Bryman (2012) there are two main paradigms, Positivism and Interpretivism. They are characterised by different views related to the nature of reality (Ontology) and to what can be regarded as acceptable knowledge (Epistemology).

Ontologically speaking, Positivists believe that one objective and universal reality exists which is independent of social actors and their interactions (objectivism). In contrast, Interpretivists emphasize the subjective meaning of a reality relative to specific social settings and actors (Constructionism). Hence, there is no definitive reality.

With respect to epistemological views, positivists are concerned with the deductive approach, that is "*the purpose of theory is to generate hypotheses that can be tested…allowing explanations of laws to be assessed*" (Bryman, 2012: p. 28). Acceptable knowledge must be generated through measurement and observation directed by the human senses. Interpretivists instead are more concerned with the inductive approach that is, knowledge can be obtained through a deep and contextualised understanding of social interaction with the final aim to draw theoretical inferences out of observations (Bryman, 2012).

Differences in Ontological and Epistemological views have implications for the methodological approach to research. Indeed, Positivists tend to use quantitative research strategies where knowledge is deemed acceptable only if it can be verified through measurements and observations that can be quantified and expressed in numbers (Domholt 2005). Moreover, studies must be replicable and universally testable and findings generalisable. To achieve such objectives, researchers usually employ structured instruments of data collection such as structured interviews, surveys or laboratory experiments which allow for easier replication and large sample sizes (Orlikowski and Baroudi, 1991).

Interpretivists, instead, typically uses qualitative methods of research to explain social phenomena and uncover the reasons behind people behaviours investigating social constructions such as language, consciousness and shared values (Klein and Myers 1999). To this purpose interpretivists usually employ less structured instruments of data collection like unstructured interviews, observations and focus groups which allow for deepness of information at the expense of (1) study generalisability, because of the use of small non-probability samples and (2) study reliability, due to the absence of standardized techniques of analysis (Bryman, 2012).

This study adopts a quantitative strategy underpinned by the positivism paradigm and the objectivism view of knowledge. There are three main reasons for this.

Firstly, the research follows a deductive approach. Research questions investigating the digital context of Social Media interactions have been generated following existing theories of Social Capital and Political Engagement as derived by Putman (2000) and Zald and McCarthy (2002). Hence, the study follows a confirmatory approach to theory. However, the conceptual model presented in Figure 5.2 (Chapter 5) can be considered as an extended version of the Resource Mobilisation Theory Framework as it introduces the "Time Spent on SNSs" and "Type of SNSs used" variables as moderators of the relationships between Social Capital and Political Participation online and offline. While the introduction of these two variables stems from considerations arising from the literature review, the analysis of their moderating role is exploratory. According to Bryman (2012) the distinction between confirmatory and exploratory research as linked respectively to quantitative and qualitative strategies "should not be viewed as constituting hard-and-fast distinctions.... qualitative research can be employed to test theories, while quantitative research is often a good deal more exploratory than is typically assumed" (p. 409).

Secondly the aim of this study is to test the relative importance of several different causes of political engagement. As suggested by Bryman (2012), a quantitative strategy is best suited for this type of objective as "*the assessment of cause is one of its keynotes*" (p. 41).

Thirdly, according to the literature review, previous studies on SNSs use, Social Capital and Political Participation mainly employed quantitative research strategies using large samples and survey instruments to gather data (Boulianne, 2015; Skoric et al., 2016).

Once the research strategy has been defined, the next step of the research process (Figure 6.1) is to determine the research design of a study. This is discussed in the next section.

6.2. RESEARCH DESIGN

According to Bryman (2012) a research design "*represents a structure that guides the execution of a research project*" (p. 45) in terms of data collection and analysis. This study adopts a cross-sectional design, which is preferred when the aim is to collect data on many individuals at a single point in time, for a quantitative type of study. Indeed, this study employs a large sample of 1212 respondents (Chapter 7) for which responses were collected at a single point in time between January and March 2019. Moreover, this research is underpinned by a quantitative approach. Hence, a cross-sectional design is appropriate for this research project.

Cross-sectional designs are usually characterised by high levels of replicability, provided that the researcher spells out all the procedures adopted in the study in terms of sampling, research instruments, measures of concepts used, and techniques of data analysis employed (Bryman, 2012). Accordingly, in-depth details of all the above listed procedures are provided in the following Sections.

However, cross sectional studies are typically weak in terms of internal validity, that is whether a conclusion relating to a causal relationship between two or more variables holds true (Bryman, 2012: p. 60). This limitation is common to studies of SNSs use, Social Capital and Political Participation because they mostly employ cross-sectional designs (Boulianne, 2015; Skoric et al., 2016). Following the suggestions of Bryman (2012) to reduce problems of internal validity, logical reasoning and established theoretical frameworks like the Social Capital Theory were used to inform casual directions within the theoretical model (see Chapter 5).

Once the appropriate research design is defined, Bryman (2012) suggest focusing on the research methods employed to collect the data. These are detailed in the next Section.

6.3. RESEARCH METHODS: RESEARCH INSTRUMENT DEVELOPMENT

Following Bryman (2012) a research method can be defined as the clusters of techniques employed in a study to collect data. The following sections present the details and justification of the research instrument employed in this study, namely online self-completion questionnaire.

This research employs a self-completion online questionnaire (see Appendix H) created and hosted online on Qualtrics (<u>www.qualtrics.com/uk</u>) and administrated by the data collection company Pure Profile (<u>www.pureprofile.com</u>) to its respondents' panel (see Section 6.6). The survey took around 8 minutes to complete. The questionnaire was developed according to Bryman's (2012) and Rea and Parker's (2014) recommendations and followed the stages below:

- 1. Evaluate the appropriateness of the instrument for the focus of the study and the chosen method of research (Section 6.3.1.).
- 2. Designing the survey instrument considering the following:
 - a) Mode of administration (Section 6.3.2).
 - b) Structure of the questionnaire: questions' sequence, wording, format and level of measurement (Section 6.3.3).
 - c) Length of the questionnaire and response rate (Section 6.3.4).
- 3. Minimise Common Method Bias (Section 6.3.5).

6.3.1. APPROPRIATENESS OF THE RESEARCH INSTRUMENT FOR THIS RESEARCH: RATIONALE, ADVANTAGES AND ISSUES.

The self-completion questionnaire is an instrument commonly employed in quantitative cross-sectional studies to collect data about people's perceptions, opinions, knowledge, attitudes and behaviour (Bryman, 2012). Indeed, it is generally employed in studies of SNSs, Social Capital and Political Participation (Boulianne, 2015; Skoric et al., 2016). It is preferred to other types of data collection instruments like interviews because it is cheaper and quicker to administer and convenient for respondents as they can complete the questionnaire when they want and at their own pace. Furthermore, it

lacks the "*interviewer bias effect*" minimising the chances that the interviewers may affect the respondents' answers (Bryman, 2012) and allowing higher response rates to sensitive questions like political behaviour (Rea and Parker 2014). For all these reasons, the self-completion questionnaire is an appropriate instrument of data collection for this study, although with limitations.

Firstly, in self-completion questionnaires respondents cannot be prompted or helped to answer questions they struggle with (Bryman, 2012). To minimise the "prompting" problem this study reduced to a minimum the use of open-ended questions (Rea and Parker, 2014) and tested the level of clarity of the survey through a pilot study (see Section 6.7).

Secondly, Bryman (2012) warn against the "question order effect". Indeed, if respondents can read the whole questionnaire before answering the first question, none of the questions asked is "truly independent" from the others. This is particularly important for studies like the current one, where there are sensitive questions related to political attitudes and behaviours. Indeed, those kinds of questions may negatively affect the completion rate of a survey as respondents could feel uncomfortable answering them (Bryman, 2012). Following Bryman's (2012) suggestions to minimise the order effect bias, respondents could see on the screen only one question at a time and could not progress further into the questionnaire unless they had answered all the questions provided at each stage of the survey.

The next section details reasons, advantages and disadvantages of the chosen online mode of administration.

6.3.2. DESIGN THE SURVEY INSTRUMENT: ONLINE MODE OF ADMINISTRATION: RATIONALE, ADVANTAGES AND ISSUES.

The online administration method was preferred to other forms of survey implementation like mail-out, telephone or in-person interviews because:

- 1. The main variables under investigation relate to the online environment.
- 2. The target population comprises Facebook and Twitter users.

- 3. It represents a more convenient, efficient and cost-effective way of reaching respondents (Rea and Parker, 2014).
- 4. It saved time in the preparation process of data as Qualtrics was set to automatically convert information into an excel file ready for the analysis.
- 5. It was convenient for respondents, who could take the survey at anytime and anywhere, using any device of their choice.
- 6. Extant research shows that respondents to online surveys are more prone to use all points of likert-type scale questions (employed in this study) because of the absence of an interviewer that minimise "social desirability biases" (Bryman, 2012).
- 7. Follow-up messages reminding respondents to complete the questionnaire with the aim to increase response rate are easy and fast to deliver (Van Der Stede et al., 2005; De Leeuw and Dillman, 2008).

However, online surveys also have some disadvantages. Firstly, there is a selfselection bias that in this case is exacerbated by using a web-panel sample, as only those people who are willing to enlist onto the panel of the data collection agency can be reached. This poses a limit to the generalisability of the study and therefore to its *external validity* (Bryman, 2012; Rea and Parker, 2014). Moreover, online surveys may affect the *ecological validity* of the research because the very instruments disrupt the "natural habitat" of the respondents (Bryman, 2012: p. 48). Yet, the degree of ecological validity is still superior to other techniques like experiments, focus groups or face-to-face interviews where the research heavily intervenes in natural settings creating laboratories or special rooms.

Details of each part of the questionnaire are provided in later sections; the rationale underpinning the development of the survey is presented in the next section.

6.3.3. DESIGN THE SURVEY INSTRUMENT: STRUCTURE OF THE QUESTIONNAIRE.

Following the suggestions of Rea and Parker (2014) the survey comprised 4 sections:

- 1. Information sheet.
- 2. Informed consent.
- 3. Main questionnaire body.
- 4. Thanking Sheet.

The information sheet (Appendix H) served three main objectives. First, it helped to increase the credibility and trustworthiness of the study by providing the researcher's and the University's contact details. Second, it informed respondents about the purpose and the importance of the study, aiming to increase their motivation to take part in the survey and to complete it. Third, it allayed the concerns that the respondents might have in terms of inconvenience, risks, disadvantages, confidentiality, data storage and safety. This last point was supported by the *informed consent* form which highlighted that respondents' participation was voluntary and anonymous, following the directives of the "Ethics Guidance of Kingston University" and the 2018 Data Protection Act. This reassurance aimed to reduce Common Method Bias (CMB) related to participants not speaking their true feelings because of concerns of being identified and judged according to the so-called "Social Desirability Bias" (Bryman, 2012: p. 227). We thanked respondents at the end of the survey encouraging them to get in touch if they wanted to know the results of the study. The development of this questionnaire is detailed in the next section.

6.3.3.1. MAIN QUESTIONNAIRE BODY

The development and editing of the questionnaire (Appendix H) followed the suggestions of Bryman (2012) and Rea and Parker (2014) in terms of:

- 1. Questions type sequence.
- 2. Questions wording.
- 3. Questions format and measurement.

Questions type sequence.

Following the categorisation of Bryman (2012: p. 253), this study employed "questions about attitude and beliefs", "knowledge questions" and "personal factual questions" (Table 6.1).

Attitude and Beliefs questions relating to Social Capital, Political Interest and Political Efficacy were placed towards the beginning of the survey. This is because it is believed that attitudinal and opinion questions are less affected by question order than those that touch upon knowledge or behaviour (Bryman, 2012: p.222). Accordingly, Knowledge questions (Political Knowledge) were placed towards the end of the survey but before more sensitive Personal factual questions including online and offline political behaviours, the time respondents spent on selected SNSs and demographics (Bryman, 2012). The only exception concerns the screening question related to the use of Facebook and Twitter which was placed at the start of the survey to establish the respondents' credentials for inclusion in this study. This question order setting should help to reduce drop-out and increase the response rates (Bryman, 2012).

Questions Order in survey*	Source	Format	Туре	Measurement
SNSs Use	Author	Closed-ended	Personal Factual	4 options
Bonding Social Capital	Williams (2006)	Closed-ended	Attitudinal	7 points Likert-type items
Bridging Social Capital	Williams (2006)	Closed-ended	Attitudinal	7 points Likert-type items
Political Interest	Brady et al. (1995)	Closed-ended	Attitudinal	7 points Likert-type items
Internal Political Efficacy	Niemi et al. (1991)	Closed-ended	Attitudinal	7 points Likert-type items
Political Knowledge	BES (1997) Larcinese (2007, 2009)	Closed-ended	Knowledge	Trichotomous type (True/False/Don't know)
Online Political Participation	Gil de Zuniga et al. (2015)	Closed-ended	Personal Factual	11 points Likert- type items
Offline Political Participation	Gil de Zuniga et al. (2012)	Closed-ended	Personal Factual	11 points Likert- type items
SNSs Intensity of Use	Adapted from Ellison et al. (2007)	Open-ended	Personal Factual	Self-reported hours and minutes
Demographics (Gender, Age and Education)	Several sources**	Closed-ended except Age open-ended	Personal Factual	Gender: Trichotomous Age: Respondents' years Education: 5 options.
* The scale items of each construct are reported in Section 6.4. ** Refer to Section 6.4				

Table 6.1. Survey Measures' Format, Types and Measurements.

Source: see in-table reference.

Moreover, following the suggestions of Conway and Lance (2010), all the questions measuring a specific construct were generally arranged in the same section of the questionnaire. Different sections were separated and characterised by distinct headings and introductory statements (i.e., "*Now some questions about your interactions on Facebook!*") to help respondents to "*focus and concentrate on a specific issue at a time and without distractions*" (Rea and Parker, 2014: p. 43). However, this practice could lead to patterned responses due to the "*response sets bias*" (Bryman, 2012: p. 226), that is the propensity of respondents to answer to a battery of interrelated questions on a same topic in a consistent way (i.e., marking all high or low values). Following Rea and Parker (2014) and Bryman (2012), two measures were taken to reduce the "response sets bias". First, five reliability checks questions like, "*Please, select: 'Strongly agree' for this statement*" were scattered across the questionnaire to check respondents' commitment and attention to the questionnaire. Second, respondents could see only one Likert item at a time to avoid patterned responses.

Questions' wording.

Questions' wording is bound to the measures employed in previous studies. This helps to increase the reliability of the research and to facilitate comparisons of findings through replicability of the study (Bryman, 2012). Also, it allows to establish "measurement validity" as there is already evidence that the scales used to measure the concepts under investigation in this study have already been successfully employed to capture the same domains in previous research. However, two originally reverse coded items of the Bonding Social Capital scale which included negatives (not, no) were transformed in affirmative sentences as the pilot study revealed that some respondents missed those negatives (see Section 6.7)

Questions' format and measurements

This study mostly employs closed-ended questions measured through Likert-type items (Table 6.1). The use of closed-ended Likert-type items is in line with the objective of this study and the type of analysis employed. Indeed, their use is recommended when there is the need to quantify variables related to people's attitudes, opinions and behaviour like Social Capital and Political Engagement (Bryman, 2012). Moreover, the type of analysis employed, namely PLS-SEM, (see Section 6.5) can easily handle Likert-type measures (Hair et al., 2017). The use of closed-ended questions is recommended in

self-administered surveys to increase response rates since they make the meaning of questions clearer and reduce respondents' mental fatigue (Rea and Parker, 2014: pp. 50-51).

Yet, those advantages come with limitations. Indeed, "there is always the possibility that the respondent is unsure of the best answer and may select one of the fixed responses randomly" (Rea and Parker, 2014: p. 52). To reduce this risk, all the Likert type scales were anchored at their extreme to meaningful labels (i.e., "Strongly Disagree" and "Strongly Agree").

There is also the risk of errors in terms of answers selection, meaning that a respondent could inadvertently checkmark a response next to the one that was intended. Moreover, respondents may lack a common understanding of the meaning of each response category. For instance, the "Strongly agree" option could mean different things to different people and its distance from other labels could not be consistent for all respondents. This raises problems in terms of measurement validity (Bryman, 2012; Rea and Parker, 2014). Following Knapp (1990) and Rea and Parker (2014), to minimise those risks, adequate space between the possible ratings was provided and categories were made mutually exclusive and equidistant by assigning numerical values to each of them. Moreover, following Carifio and Perla (2007; 2008), a visual analogue response (VAR) was employed as visual measurement instrument for all the Likert type items (Figure 6.2). Finally, following Bishop and Herron (2015) all the Likert-type items have a neutral point and are balanced to either side adopting odd points ranges. All those precautions should provide the appearance and the perception of equal distance among response options and should help to minimise validity problems and answers selection errors (Rea and Parker, 2014).



Figure 6.2. Visual Analogue Response of Survey Questions.

6.3.4. LENGTH OF THE QUESTIONNAIRE AND RESPONSE RATES

According to Bryman (2012) one of the problems with online self-completion questionnaires is low response rates, which can increase the risk of non-response bias. The evidence suggests that the level and pace of response rates depends on several factors like the questionnaire length, aesthetics, layout consistency, variety of response formats and incentives offered to respondents (Bryman, 2012: p. 236).

Questionnaire length

To increase response rate the questionnaire should take less than 15 minutes to complete (Rea and Parker, 2014). The average time needed to answer the questionnaire of this research was measured through the pilot study, carried out in December 2018. The respondents (n = 314) spent on average 517.31 seconds (SD = 250.89) completing the questionnaire which is equal to 8 minutes and 37 seconds. This is below Rea and Parker's (2014) rule of thumb even though the mean was inflated by extreme values which produced a positively skewed distribution (Skew = 1.537, SE = .256; Min = 152; Max = 1468).

Following Bryman's (2012: p. 675) suggestions to minimise non-response bias, the information sheet mentioned the time needed to complete the survey (Appendix H) and a progress bar indicator was shown to the respondents as they progressed through the questionnaire.

Questionnaire aesthetics

With respect to aesthetics this research followed Bryman's (2012) guidelines who recommended darker and/or larger font for questions and lighter and/or smaller font for response categories (Figure 6.2). Numbers anchored to each category were bigger to improve clarity and minimise response selection errors.

Layout consistency

Fonts and colours pattern were kept consistent across all the survey to help respondents to speed up the process of questions and answers recognition allowing them to finish the survey earlier than expected.

Response variety

As mentioned in Section 6.3.3. this study employs almost completely closedended questions to minimise response fatigue and increase response rate. However, different types of visual analogue responses were employed to make the survey more dynamic and interesting.

Incentives

Following Bryman's (2012) guidelines to increase response rate, economic incentives and personalised follow-ups were provided to respondents by the data collection agency Pure Profile.

Once the questionnaire was created, we further tested it for clarity, comprehensiveness and acceptability through a pilot study (Rea and Parker, 2014). Details are provided in Section 6.7.

6.3.5. COMMON METHOD BIAS PRECAUTIONS

When employing research instruments like surveys, it is necessary to take precautions against Common Method Bias (CMB), that is the "variance that is attributable to the measurement method rather than to the constructs the measures represent" (Podsakoff et al., 2003: p.879). CMB can artificially inflate or deflate the relationships between constructs providing misleading findings. Hence, researchers should try to minimise CMB during the process of research instrument design.

Conway and Lance (2010) suggested that studies employing surveys using selfreported measures should indicate how the researchers have taken a proactive approach to minimise CMB by:

- 1. Providing an argument for why self-reported measures are appropriate.
- 2. Showing that there is no overlap in items of different constructs.
- 3. Providing evidence of reliability and validity of the instrument of research adopted.

Appropriateness of self-reported Likert scales for this study.

This research involves studying SNSs users' perceptions of levels of trust, reciprocity, political interest and political self-efficacy. There is no alternative source for such personal information to the users' own reports. Likert scales are considered reliable and convenient to use when the aim is to translate psychometric and attitude measures into pragmatic values (Waples et al., 2010). Furthermore, the use of self-reported measures is in line with previous studies in the field of Political Engagement (e.g., Gil de Zuniga et al., 2012, 2013, 2015; Jung et al., 2011; Lin and Chiang, 2017; Saldaña et al., 2015; Weeks et al., 2015) and is dictated by the practicality and minimum effort required by the respondents to complete the survey.

Absence of overlapping.

By conducting the pilot study, we were able to empirically ascertain that no overlap was present among variables by assessing the discriminant validity of the different constructs involved in the analysis through the evaluation of the cross-loadings values, the Fornell-Larcker criterion and the Heterotrait-Monotrait Ratios (HTMT).

Evidence of reliability and validity of the instrument of research adopted.

Several measures were taken to improve reliability and validity of the study. First, the use of scales already employed in previous research helped to establish the *face validity* of the constructs being measured and to ensure their measurements reliability as they have been already tested for stability and consistency over time. Second, reliability and validity of the measurements were further improved by setting the appropriate questions order, format and instructions as already discussed in Section 6.3.3.1. Further,

a pilot study was employed to check the clarity of the questionnaire and the levels of reliability and validity of the constructs (Section 6.7).

The above actions indicate that this study has adopted a proactive approach to CMB minimisation, as suggested by Conway and Lance (2010). Appendix I provides a table that summarises the factors identified by MacKenzie and Podskoff (2012) that have the potential to increase CMB and indicates the actions implemented in this study to tackle them.

Moreover, following Kock (2015), a full collinearity assessment (FCA) test in PLS-SEM was performed to empirically test for CMB contamination a posteriori of data collection. The FCA tests for both vertical and lateral collinearity. Vertical collinearity is present "when two or more predictors measure the same underlying construct, or a facet of such construct" (Kock, 2015: p. 6). Rather, lateral collinearity occurs when "a predictor variable measures the same underlying construct, or a facet of such construct, as a variable to which it points in a model" (Kock, 2015: p. 6). The FCA computes the Variance Inflation Factor (VIF) for all latent constructs of a model. The VIF provides an index that measures how much of the variance of an estimated regression coefficient is increased because of collinearity. If all VIFs of the latent constructs are equal to or lower than 3.3, the model is considered free from CMB (Kock, 2015). This method will be employed in Chapter 7 when performing the statistical analysis of data.

6.4 OPERATIONALIZATION OF RESEARCH CONSTRUCTS

The fourth step of the systematic research process presented in Figure 6.1 consists of defining the measures for operationalizing the constructs of a proposed theoretical model. While some constructs in social science research, such as a person's age, income, or gender, may be easy to measure, because they are directly observable, other constructs, such as Social Capital and Political Interest, may be considerably harder to gauge, because they cannot be directly and uniquely observed and may have several dimensions (Bryman, 2012). Therefore, concepts indicators must be used. "*An indicator is [...] an indirect measure of a concept*" (Bryman, 2012: p. 164) usually artificially coded (i.e., Likert-type items) that can explain only part of the variance of the concept being

measured. For this reason, many theoretical constructs are measured through multiple indicators so to capture much of their variance and touch upon all their dimensions. This research employs multiple indicators in the form of Likert-type items for many of the constructs included in the proposed theoretical framework (Table 6.1). The following Sections focus on the operationalisation of the constructs included in the theoretical framework. For each construct/variable, the corresponding section presents a review of its use in the literature, a specification of its measurement indicators and its suitability for this study.

6.4.1. OPERATIONALISATION OF SOCIAL CAPITAL

This Section presents the main perspectives existing for measuring Social Capital and discusses the method that has been selected for this research.

As with the definitions of Social Capital (Chapter 4, Section 4.3), its measurements are interrelated to the goal of each study. Through an extensive literature review two methodological approaches were identified:

- 1. The first approach is focused on the structure of social networks (Burt, 1988, 2000; Lin, 2001, 2008), where size, density, quality (strong/weak ties) and frequency of social contacts are considered as the principal aspects for measuring Social Capital. This approach mirrors the theoretical arguments of the "Structural View" of Social Capital and is more concerned with an analysis at the individual rather than collective level (Chapter 4, Section 4.2).
- The second approach is more concerned with the Cultural Perspective of Social Capital (Chapter 4, Section 4.2). Authors in this tradition consider civic membership, trust, integration and reciprocity as proxies of Social Capital at both individual (Bourdieu, 1986; Coleman, 1990) and collective levels (Ostrom and Ahn, 2003; Putnam, 2000; Woolcock, 1998).

The first approach has been largely criticised since it focuses on measuring social networks' size, density and frequency of contacts, which do not indicate anything about resources' availability and accessibility in a specific network (Fiori, Smith and Antonucci, 2007), nor whether people are willing to use or share such resources

(Obukhova and Lan, 2013), which in turn should imply mechanisms of trust and reciprocity (Coleman 1988; Finsveen and Van Oorschot 2008; Putnam, 2000; Van Der Gaag and Snijders, 2005).

However, also the second approach by itself is not sufficient to provide a clear operationalisation of Social Capital. For instance, the concept of group closure plays a critical role in both Bourdieu's (1986) and Coleman's (1990) conceptualisations of Social Capital. A social structure is closed if all the individuals in a group are connected to each other by some kind of relations. The level of closure is considered to have a positive correlation with the level of Social Capital possessed within the groups. Indeed, closed groups facilitate the creation of strong relations or ties that allow to access information and establish reputation, trust and reciprocity that maximise Social Capital. Accordingly, only strong ties are included in Social Capital (Coleman, 1988; Bourdieu, 1986). However, closed structures prevent the entry of valuables resources like heterogeneous information and innovation that could derive from weak ties (Portes, 1998; Stanton-Salazar and Dornbush, 1995). Putnam (2000) acknowledged this limitation and recognised the importance of week ties by introducing the concept of Bridging Social Capital in his "cultural" analysis. Yet, as Lin (2001) pointed out, Putnam's (2000) conceptualisation is focused on norms of trust and reciprocity which lead to the identification of Social Capital mainly with cultural aspects, neglecting the structural asset of networks which can affect the creation and maintenance of such norms.

To overcome the above limitations, a possible alternative approach, suggested by Williams (2006), is that of considering networks and their structures (strong/weak ties) as agents that facilitate the creation and maintenance of trust and reciprocity that determine access to resources ultimately defining Social Capital. This means that the Social Capital owned by individuals is not the network itself (Structural Perspective). Rather, networks' structures are considered at the basis of the creation of those norms of trust and reciprocity which facilitate people in mobilising their resources (Cultural Perspective). This study follows the latter approach and considers Social Capital as the resources embedded in social relations that can be mobilised on the basis of norms of trust and reciprocity. Hence, this study employs the Internet Social Capital Scale (ISCS) devised by Williams (2006) to operationalise online Bonding and Bridging Social Capital (Figure 6.3).

Figure 6.3. ISCS	(Internet Social	Capital Scale)
------------------	------------------	------------------------

Bonding Subscale

- 1. There are several people online/offline I trust to help solve my problems.*
- There is someone online/offline I can turn to for advice about making very important decisions.*
- There is no one online/offline that I feel comfortable talking to about intimate personal problems. (reversed)*
- 4. When I feel lonely, there are several people online/offline I can talk to.
- 5. If I needed an emergency loan of \$500, I know someone online/offline I can turn to.*
- 6. The people I interact with online/offline would put their reputation on the line for me.
- 7. The people I interact with online/offline would be good job references for me.
- 8. The people I interact with online/offline would share their last dollar with me.
- I do not know people online/offline well enough to get them to do anything important. (reversed)
- 10. The people I interact with online/offline would help me fight an injustice.

Bridging Subscale

- Interacting with people online/offline makes me interested in things that happen outside of my town.
- 2. Interacting with people online/offline makes me want to try new things.
- Interacting with people online/offline makes me interested in what people unlike me are thinking.
- 4. Talking with people online/offline makes me curious about other places in the world.
- 5. Interacting with people online/offline makes me feel like part of a larger community.
- 6. Interacting with people online/offline makes me feel connected to the bigger picture.
- Interacting with people online/offline reminds me that everyone in the world is connected.
- 8. I am willing to spend time to support general online/offline community activities.
- 9. Interacting with people online/offline gives me new people to talk to.
- 10. Online/Offline, I come in contact with new people all the time.

Source: Williams (2006): p.602.

An outline of Williams' (2006) scale is presented next, to provide more context and specify the reasons for which his approach fits this study.

William (2006) distinguished between two dimensions of Social Capital, namely *Bonding* and *Bridging*, in the offline and online environments.

To measure *Bonding Social Capital*, Williams (2006) identified 5 criteria: (1) emotional support, (2) access to limited resources, (3) capacity to mobilise solidarity, (4) out-group antagonism and (5) homogeneity. As Putnam (2000) had argued, the Bonding dimension represents an exclusive type of Social Capital since it is mainly associated with networks of like-minded people bonded by kinship, ethic, religion and cultural or ideological ties. Hence, Williams (2006) considered out-group antagonism and

homogeneity as criteria of Bonding Social Capital, linking it to the *resources* accessible in the individual's "strong ties" (Hampton, 2011). This definition closely reflects the Structural Perspective of Social Capital (quality of the tie structure). By including also emotional support in Bonding Social Capital (Items 1, 2, 3 and 4), Williams (2006) attempted to measure trust in others, that is a typical component of the Cultural Perspective of Social Capital and a central element of the definition of Social Capital provided in this study. Furthermore, when considering access to limited resources, Williams (2006) considers not only material resources like money (Item 5) but also symbolic resources like job references and reputation (Items 6 and 7). This fits the discussion over the conceptualisation of Social Capital in this study, which considers both types of resources (see Chapter 4). The remaining items of the Bonding scale (8, 9 and 10) measure the ability of individuals to mobilise solidarity, which is at the basis of the Cultural Perspective of Social Capital.

To measure *Bridging Social Capital*, Williams (2006) considered 5 criteria: (1) outward looking, (2) contact with a broader range of people, (3) view of oneself as part of a broader group, (4) diffuse reciprocity within a broader community and (5) heterogeneity (as meeting new people). Indeed, Bridging Social Capital is regarded as "inclusive" Social Capital as it bridges across gender, race, geographical space, and ethnicity. It is related to more heterogeneous and diverse groups when compared to the Bonding dimension (Putnam, 2000) and it is characterised by "week ties". For these reasons, in his scale of Bridging Social Capital, Williams (2006) included 'outward looking' items, that is the predisposition to look outside one's group, being more open minded (Items 1, 2, 3 and 4). He also devised items tapping on contacts with a broader range of people (Items 9 and 10) and items measuring heterogeneity which indicates whether people view themselves as part of a broader group (Items 5, 6 and 7). Also, by looking at diffuse reciprocity (Item 8), Williams (2006) addresses the idea stemming from the cultural perspective of Social Capital that a general sense of "givingness" (p. 600) would help the exchange of resources within a network. However, this sense of "givingness" must be based on reciprocity without immediate gain which means that people must be willing to give to others without expecting something back from them. This concept is in line with the definition of reciprocity provided in this study and with the framework of civic virtue upon which this research is based.

Based on the above discussion, the ISCS measure developed by Williams (2006) fits this study particularly well because (1) the items were created specifically for measuring online Social Capital and (2) it is in line with the definition of Social Capital adopted in this study. Indeed, the ISCS measures resources, trust, and reciprocity, also accounting for the quality of the relationships. These are concepts relevant to both the Structural and the Cultural perspectives of Social Capital.

All the items of Williams' (2006) scale were employed verbatim in this study, except for questions 3 and 9 of the Bonding section (Table 6.2) which were rephrased to eliminate the need for reverse coding and reduce response errors (see Section 6.7).

Labels Indicators*	Items**
Bond_1	There is someone on Facebook I can turn to for advice about making very important decisions.
Bond_2	There is someone on Facebook that I feel comfortable talking to about intimate personal problems.
Bond_3	When I feel lonely, there are several people on Facebook I can talk to.
Bond_4	If I needed an emergency loan of £ 500, I know someone on Facebook I can turn to.
Bond_5	The people I interact with on Facebook would put their reputation on the line for me.
Bond_6	The people I interact with on Facebook would provide good job references for me.
Bond_7	The people I interact with on Facebook would share their last pound (f) with me.
Bond_8	I know people on Facebook well enough to get them to do something important.
Bond_9	The people I interact with on Facebook would help me fight an injustice.
Bond_10	There are several people on Facebook I trust to help solve my problems.
* Used to identify indicators in Statistical Analysis.	

Table 6.2. Online Bonding Social Capital Indicators and Respective Labels.

** The items were also adapted to tap on Twitter and Facebook and Twitter combined users.

Source: adapted from Williams (2006).

Labels Indicators*	Items**
Brid_1	Interacting with people on Facebook makes me interested in things that happen outside of my town.
Brid_2	Interacting with people on Facebook makes me want to try new things.
Brid_3	Interacting with people on Facebook makes me interested in what people unlike me are thinking.
Brid_4	Talking with people on Facebook makes me curious about other places in the world.
Brid_5	Interacting with people on Facebook makes me feel like part of a larger community.
Brid_6	Interacting with people on Facebook makes me feel connected to the bigger picture.
Brid_7	Interacting with people on Facebook reminds me that everyone in the world is connected.
Brid_8	I am willing to spend time to support general Facebook community activities.
Brid_9	Interacting with people on Facebook gives me new people to talk to.
Brid_10	On Facebook, I come in contact with new people all the time.
* Used to identify indicators in Statistical Analysis.	

 Table 6.3. Online Bridging Social Capital Indicators and Respective Labels.

** The items were also adapted to tap on Twitter and Facebook and Twitter combined users.

Source: adapted from Williams (2006).

6.4.2. OPERATIONALISATION OF OFFLINE AND ONLINE POLITICAL PARTICIPATION

Consistently with the definition of Political Participation provided in Chapter 3, this research investigates Political Participation in both the offline and online environments in its conventional and unconventional forms (Ekman and Amna, 2012; Van Deth, 2014).

Offline Political Participation (Off_PP) is operationalised through a scale developed by Gil de Zuniga et al. (2012) for Structural Equation Modeling (SEM) analysis and further used in the SEM context by Gil de Zuniga et al. (2013) and Gil de Zuniga et al. (2014b). In all those studies, the scale yielded good reliability values (respectively Cronbach's α =.82, Cronbach's α =.82 and Cronbach's α =.87). The scale showed sufficient reliability as well as discriminant and convergent validity also in the pilot study conducted for the current research (Section 6.7). Gil de Zuniga et al.'s 11-point scale (Table 6.4) consists of nine items anchored from 0 = "Never" to 10 = "Every

time I could have a chance". Each item concerns a specific offline political activity engaged by the respondents in the past 12 months.

Labels Indicators*	Items
Off_PP_1	Attended a public hearing, town hall meeting, or city council meeting.
Off_PP_2	Called or sent a letter to an elected public official.
Off_PP_3	Spoken to a public official in person.
Off_PP_4	Posted a political sign, banner, button or bumper sticker.
Off_PP_5	Attended a political rally.
Off_PP_6	Participated in any demonstrations, protests, or marches.
Off_PP_7	Written a letter to a news organisation.
Off_PP_8	Participated in groups that took local actions for social or political reforms.
Off_PP_9	Been involved in public interest groups, political action groups, political clubs, or party committees.
* Used to ident	tify indicators in Statistical Analysis.

 Table 6.4. Offline Political Participation Indicators and Respective Labels.

Source: adapted from Gil de Zuniga et al., 2012.

Consistently with Van Deth's (2014) categorisation of political activities, the scale items tap on both conventional (items 1, 2, 3, 4, and 5) and unconventional (items 6, 7, 8 and 9) forms of Political Participation relating to voluntary activities conducted by citizens, individually or collectively, with the aim to affect the political sphere. Hence, the scale covers all the main theoretical domains of Political Participation provided in the definition.

Because of its employability in SEM analysis, the good level of reliability shown in previous studies and for its alignment to the conceptualisation of Political Participation adopted in this study, the scale of Gil de Zuniga et al. (2012) is considered an appropriate operationalisation of the offline Political Participation variable.

Online Political Participation (On_PP) is operationalised through a scale of six items, concerning how often respondents engage in a series of political activities on the Internet (Table 6.5). The items are measured on eleven points ranging from 0 to 10 where 0 = "Never" and 10 = "Every time I could have a chance". The scale, developed by Gil de Zuniga et al., (2015), was used in a SEM context and showed high reliability (Cronbach's α =.81). Among others, the scale was also employed by Lin and Chiang

(2017) in SEM analysis, showing high reliability (Cronbach's α =.97). Furthermore, the scale was pre-tested for this research through a pilot study and showed adequate internal consistency reliability, and discriminant and convergent validity (Section 6.7). Item On_PP_5 of the original scale was re-worded from "*used a mobile phone to donate money to a campaign or political cause*" to "*donated money online to a campaign or political cause*" to "*donated money online to a campaign or political cause*", reflecting the focus of this study on online Political Participation. This amendment is consistent with other studies employing the same scale in a similar context (e.g., Lin and Chiang, 2017).

Labels Indicators*	Items
On_PP_1	Created an online petition
On_PP_2	Signed a petition online.
On_PP_3	Participated in an online question-and-answer session with a politician or public official.
On_PP_4	Signed up online to volunteer to help with a political cause.
On_PP_5	Donated money online to a campaign or political cause.
On_PP_6	Started a political or cause-related group online.
* Used to identify indicators in Statistical Analysis.	

 Table 6.5. Online Political Participation Indicators and Respective Labels.

Source: adapted from Gil de Zuniga, Garcia-Perdomo and McGregor (2015)

The items of the On_PP scale include both conventional (items 3, 5 and 6) and unconventional (items 1, 2 and 4) online forms of voluntary political actions as identified by Barrett and Brunton-Smith (2014), Ekman and Amna, 2012 and Van Deth (2014). Hence, the scale is in line with the definition of Political Participation adopted in the current study.

Due to its comprehensiveness, its suitability to SEM analysis, the good level of reliability showed in previous studies and in the pilot study of this thesis, the scale of Online Political Participation devised by Gil de Zuniga et al. (2015) is deemed appropriate for the current study.

6.4.3. OPERATIONALISATION OF MODERATOR VARIABLES

One of the aims of this study is to ascertain the moderating effects of time spent on SNSs, type of SNSs used and combined and not combined use of SNSs on the
relationships between online Bonding and Bridging Social Capital and online and offline Political Participation.

For type of SNSs used and combined and not combined use of SNSs, the operationalisation process is straightforward. Indeed, they are categorical variables, operationalised by asking respondents whether they were Facebook users or Twitter users or whether they used both Facebook and Twitter.

With regards to "time spent on SNSs", one of the operationalisation methods commonly employed by researchers is based on self-reported measures of time spent on the Internet in hours or minutes per day (Becker et al., 2012; Carrier et al., 2009; Ellison et al., 2007; Junco, 2013; Kimbrough, et al., 2014; Padilla-Walker and Coyne, 2011; Pea et al., 2012; Reich et al., 2012; Rosen et al. 2013; Turner and Croucher, 2013). This research employs the same method. However, to help respondents in the process of recalling the amount of time spent online, and consistently with other studies in the field (i.e., Ellison et al., 2007), before asking respondents to indicate the time in hours and minutes spent on the selected SNS, we provided them with categories (Figure 6.4). Indeed, Araujo et al. (2017) argued that it is of crucial importance to help respondents to overcome recall and estimation problems related to media use, for instance, by developing anchors and categories that are targeted at those problems.

Figure 6.4. Fixed Choice Option for Time Spent on SNSs for this Study (Copy)



Source: adapted from Ellison et al. (2007)

6.4.4. OPERATIONALISATION OF CONTROL VARIABLES

According to the literature, a control variable is a variable that has an impact, typically over the dependent variable, that cannot be ignored (Becker, 2005; Becker et al., 2016; Breaugh and Arnold, 2007). However, its theoretical background is usually not explored as a part of the main relationships under investigation in a study. Indeed, the results for control variables are usually not further interpreted (Hair et al. 2017). Yet, their inclusion is believed to strengthen the explanatory power of a conceptual model.

As outlined in Chapter 5, the CET framework proposes that an individual's Political Participation is a result of his/her education, political knowledge, political interest and internal political efficacy. Moreover, empirical studies in the field of political and social engagement provided evidence that age may affect individual levels of Political Participation, together with gender (see Chapter 5). Hence, all the above variables have been controlled in this study. The following sections deal with the operationalisation of those control variables.

6.4.4.1. OPERATIONALISATION OF POLITICAL INTEREST

Extant literature defined and operationalised Political Interest (PInt) in different ways (see Sigel and Hoskin, 1981; Bennett, 1984; Gabriel, 1986; Van Deth, 1990; Zaller, 1992). In the current study, political interest is defined as the "*degree to which politics arouses a citizen's curiosity*" (Van Deth 1990: p. 278). This definition fits this study particularly well. Indeed, according to Van Deth (1990) political interest is not a form of political behaviour. This is consistent with the CET assumption that Political Interest is part of the Cognitive dimension of people and cannot represent a behaviour, rather it is an individual intrinsic factor which can shape a behavioural pattern.

In line with this definition, is the scale devised by Brady et al. (1995) who operationalised Political Interest through 3 averaged items anchored from 1 = "Strongly disagree" to 7 = "Strongly agree" (Table 6.6).

The scale has been widely used in previous research (e.g., Brady et al., 1999; Dawes et al., 2011; Oliver, 2000 and Scheufele, 2002) and has always shown Cronbach's alpha above 0.7. The scale was pre-tested through the pilot study and displayed good levels of internal reliability, discriminant and convergent validity (Section 6.7). It was also used in the SEM analysis context by Bell et al. (2009), Carter (2006), Fowler et al. (2008) and Gastil and Xenos (2010). For all these reasons, the scale of Brady et al. (1995) was deemed appropriate for the current study.

Labels Indicators*	Items	
PInt_1	In regards to my local community, I am interested in local community politics and local community affairs.	
PInt_2	I am interested in national politics and national affairs.	
PInt_3	In general, I am interested in politics.	
* Used to identify indicators in Statistical Analysis.		

Table 6.6. Political Interest Indicators and Respective Labels.

Source: adapted from Brady et al. (1995).

6.4.4.2. OPERATIONALISATION OF INTERNAL POLITICAL EFFICACY

Because of the CET's emphasis on individual intrinsic factors, many studies have paid attention to the notion of Internal Political Self-Efficacy, which is here defined as: "*individuals' self-perceptions that they are capable of understanding politics and competent enough to participate in political acts*" (Miller et al., 1980: p. 253). It consists of a self-appraisal of political competence and capabilities linked to the individual's belief that he/she can be successful in performing a specific political behaviour (Gil de Zuniga et al., 2012; Jung et al., 2011; Velenzuela et al., 2012).

Internal Political Efficacy (PEff) is operationalised with Niemi et al.'s (1991) four-items seven-point scale anchored from 1 = "Strongly disagree" to 7 = "Strongly Agree". (Table 6.7). The scale has been widely employed in previous studies using SEM analysis techniques either in its original form or slightly adapted to the researchers' aims (e.g., Anim et al., 2019; Ardèvol-Abreu et al., 2019; Park and Karan 2014; Ritter, 2008; Warner, 2018; Yang and DeHart, 2016). In all those studies, the scale showed good reliability and validity. The scale was pre-tested through the pilot study and displayed good levels of internal consistency reliability, discriminant and convergent validity (Section 6.7). Moreover, as Niemi et al. (1991) argued, the items of the scale touch upon an individual's self-perceptions of being capable of understanding politics and feeling competent enough to participate. Hence, the scale is a good operationalisation of the definition provided by Miller et al. (1980).

Labels Indicators*	Items
PEffic_1	I consider myself to be well qualified to participate in politics.
PEffic_2	I feel I have a pretty good understanding of the important political issues facing our country.
PEffic_3	I feel that I could do as good a job in public office as most other people.
PEffic_4	I think that I am better informed about politics and government than most people
* Used to identify	indicators in Statistical Analysis.

 Table 6.7. Political Efficacy Scale Indicators and Respective Labels.

Source: adapted from Niemi et al. (1991).

6.4.4.3. OPERATIONALISATION OF POLITICAL KNOWLEDGE

According to the CET framework, politically knowledgeable citizens are more likely to participate in politics (Delli Carpini and Keeter, 1996; Galston, 2001). Hence, Political Knowledge must be controlled for, in studies focusing on Political Participation.

In this study, Political Knowledge is defined as "*the range of factual information about politics that is stored in long-term memory*" (Delli Carpini and Keeter, 1996, p. 10). In line with this definition, the Political Knowledge construct (Pknow) was operationalised as the additive score of the responses to a battery of seven *factual* political questions to which respondents answered either "True", or "False" or "Do not Know" (Table 6.8). Respondents scored one point for each correct answer and zero for each wrong or "do not know" answer. Points were added up to create an index of Political Knowledge ranging from zero to seven (Larcinese, 2007, 2009). Notably, PLS supports single-item constructs (Hair et al., 2011). Also, additive scoring methods as part of an index or scale of multiple items are considered to yield good content validity and measures reliability (Rossiter, 2002, 2008).

This measure has been employed in several studies concerned with "*factual*" political knowledge (i.e., British Election Study, 1997; Chan and Clayton, 2006; Larcinese, 2007, 2009; Sturgis et al., 2008). The items of the scale touch upon the most important categories identified by Delli Carpini and Keeter (1996: p. 65) as the governing principles of what people should know about politics and that are "the rules of the game" (questions 2, 3, 4, 5, 6, and 7) and "people and parties" (question 1).

Labels Indicators*	Items	
Quest_1	Margaret Thatcher was a Conservative prime minister.	
Quest_2	The number of members of parliament is about 100.	
Quest_3	The longest time allowed between general elections is four years.	
Quest_4	Britain's electoral system is based on proportional representation.	
Quest_5	MPs from different parties are on parliamentary committees.	
Quest_6 Britain has separate elections for the European Parliament and the British Parliament.		
Quest_7	No one may stand for parliament unless they pay a deposit.	
* Used to identify indicators in Statistical Analysis.		

 Table 6.8. Political Knowledge Questions Battery.

Source: Adapted from British Election Study (1997) and Larcinese (2007, 2009).

Given that the above scale is in line with the conceptualisation provided in this research and since it has been employed in several studies concerned with the *factual* level of political knowledge, the scale is deemed appropriate for this study.

6.4.4.4. OPERATIONALISATION OF SOCIO-DEMOGRAPHIC VARIABLES: AGE, EDUCATION AND GENDER

As mentioned in Chapter 5, socio-demographic variables like age, education and gender may affect levels of Political Participation (Gil De Zuniga et al., 2012; Melo and Stockemer, 2014; Quintelier, 2007; Skoric et al., 2009; Velenzuela et al., 2009).

In this study, age is measured with an open-ended question asking: "Please, indicate what was your age on your last birthday (in years)". This formulation has been widely employed in previous studies related to Political Participation and media use (see among the others Gil de Zuniga et al., 2012; Jung et al., 2011; McCluskey et al., 2004).

The operationalisation of educational level included five options: [1] "Less than high school", [2] "High School", [3] "Bachelor's degree", [4] "Master's degree", [5] "Doctoral degree". To facilitate the analysis and interpretation of results (Field, 2013), the variable Education was dummy coded with 2 categories where 0 = "Low Education" comprised options one and two and 1 = "High Education" comprised options three to five. To operationalise Gender, respondents had to indicate whether they were Males, Females or whether they preferred not to say. No respondents selected the last option. A dummy coded variable was created, where 0 = "Males" and 1 = "Females".

6.5. STATISTICAL METHODS OF ANALYSIS

Following the systematic approach to the research process reported in Figure 6.1, once the appropriate research strategy and design, instrument and related measures have been discussed, researchers should address the appropriateness of the statistical method of analysis chosen for answering the research questions.

This study adopts the Structural Equation Model (SEM) with the Variance Structure Analysis technique (PLS) to answer research questions Q1, Q2 and Q3. A series of parametric and non-parametric ANOVAs analysis (Table 6.9) address research questions Q4 and Q5, respectively. Reasons and justifications for the chosen techniques of analysis are discussed in the following Sections.

Research questions	Statistical aims	Statistical	Software	
Research questions	Statistical anns	techniques		
Q1 . Does the type of SNSs	Reveal the significance	Multi-Group	SmartPLS	
used (Type_SNSs_Used)	of the moderating	analysis	version	
moderate the relationships	effects of the type of	(MGA) in	3.2.8	
between Online Bridging	SNSs employed on the	SEM-PLS		
(Brid_SC) and Online	relationship between	through		
Bonding (Bond_SC) Social	Online	pairwise		
Capital with Online and	Bonding/Bridging	comparison		
Offline Political	Social Capital and	of		
Participation (On_PP and	Online/Offline	Permutation		
Off_PP)? To what extent?	Political Participation.	tests		
Q2. Does the intensity of	Reveal the significance	Moderation	SmartPLS	
Use of SNSs (Int_SNS_Use)	of the moderating	Analysis	version	
moderate the relationships	effect of intensity of	through the	3.2.8	
between Online Bridging	SNSs used on the	Two-stage		
(Brid_SC) and Online	relationship between	Approach in		
Bonding (Bond_SC) Social	Online	SEM-PLS		
Capital with Online and	Bonding/Bridging			
Offline Political	Social Capital and			
Participation (On_PP and	Online/Offline			
Off_PP)? To what extent?	Political Participation.			
Q3. Does the combined and	Reveal the significance	Multi-Group	SmartPLS	
not combined use of SNSs	of the moderating	analysis	version	
(Comb_SNS_Use) moderate	effects of the	(MGA) in	3.2.8	
the relationships between	combined use of SNSs	SEM-PLS		
Online Bridging (Brid_SC)	employed on the	through		
and Online Bonding	relationship between	pairwise		
(Bond_SC) Social Capital	Online	comparison		
with Online and Offline	Bonding/Bridging	of		
Political Participation	Social Capital and	Permutation		
(On_PP and Off_PP)? To	Online/Offline	tests		
what extent?	Political Participation.			
Q4. To what extent do the	Ascertain whether	Two-way	SPSS	
amounts of Online Bridging	there are any	ANOVA	version 24	
(Brid_SC) and Bonding	statistically significant			
(Bond_SC) Social Capital	differences in the			
differ across different types	amount of Social			
of SNSs employed	Capital owned by			
(Type_SNSs_Used) and	different users of			
across different levels of	different SNSs, who			
Intensity of SNSs used	show different levels			
(Int_SNS_Use)?	of intensity of SNSs			
	use.			

Table 6.9. Research Questions and Statistical Techniques of Analysis.

Research questions	Statistical aims	Statistical techniques	Software
Q5 . To what extent do the	Ascertain whether	Two-way	R software
amounts of Online and	there are any	non-	
Offline Political	statistically significant	parametric	
Participation (On_PP and	differences in the	ANOVA with	
Off_PP) differ across	amount of Political	Aligned Rank	
different types of SNSs	Participation reported	Test (ART)	
employed	by different users of		
(Type_SNSs_Used) and	different SNSs, who		
across different levels of	show different levels		
Intensity of SNSs used	of intensity of SNSs		
(Int_SNS_Use)?	use.		

6.5.1. RESEARCH QUESTIONS Q1, Q2 AND Q3: THE USE OF PLS-SEM

Research questions Q1, Q2 and Q3 will be addressed through Structural Equation Modeling (SEM) with the Variance Structure Analysis technique using the SmartPLS software version 3.2.8. There are three main reasons why SEM analysis is considered appropriate for the purpose of this research in comparison to other multivariate techniques, such as multiple regression analysis.

First, SEM analysis allows to estimate the impact of both categorical and continuous moderating variables within a single model (Hair et al., 2017) and on multiple relationships at the same time, allowing for more complex Modeling like moderation which is one of the aims of this study (Hair et al., 2018). This provides a more realistic reflection of theory but at the same time allows for simplified testing procedures.

Second, SEM analysis allows to test models which include latent constructs and their relationships using multiple indicators (MacKenzie et al., 2005) which allows for more precise specification of results, as in this way reliance is not placed on a single response (Hair et al., 2017). Additionally, SEM technique enable the establishment of the convergent and discriminant validity of measures.

Thirdly, SEM allows to test the theoretical structure of constructs for non-random measurement error that is the difference between a measured quantity and its true value associated with factors that systematically affect the measurement of a variable across

the sample (Bollen, 1989; Hair et al., 2006; Steenkamp and van Trijp, 1991; Schumaker and Lomax, 2004). This systematic error could be due to: (1) items characteristics, like when a measure of a construct is omitted by error or to (2) respondents' characteristics, like when respondents are more inclined to agree with a specific question item because of personal biases (Byrne, 2012).

Considering the advantages discussed above, SEM is widely regarded as the key methodological approach to theory development and testing (Anderson and Gerbing, 1988; Deshon, 1997; Hair et al., 2017; Steenkamp and Trijp, 1991) and fits the purpose of this research.

There are two main approaches to SEM analysis: Covariance Based analysis (CB) and Variance Based approach, known as Partial Least Squares (PLS), (Gefen et al., 2000; Hair et al., 2012). CB-SEM techniques estimate path coefficients and loadings by minimizing the difference between observed and predicted variance–covariance matrices. The observed variance–covariance matrix is calculated by the covariance structure of the Manifest variables. PLS is a component-based SEM technique which estimates parameters similar to the principal component analysis with a multiple regression approach. Specifically, PLS analysis operates a series of ordinal least squares (OLS) interdependent regressions (Fornell and Bookstein, 1982). In this research, the proposed model will be tested using PLS analysis. Following Hair et al. (2017), this choice was made with reference to some aspects of this study that can be better handled by PLS-SEM. These include:

- 1. The complexity of the proposed model.
- 2. The small sample size.
- 3. The type of data used.
- 4. The non-normal distribution of data.
- 5. The need to use latent variable scores in subsequent analyses.

Extant literature suggests that the PLS technique yields more reliable results compared to CB when dealing with complex models which include also indirect and moderating effects (Hair et al., 2018). A model can be considered complex when it has 10 or more constructs and/or 50 or more items (Akter et al., 2017; Chin 2010). The model

of this study has 10 constructs that increase to 14 when testing for the moderation of "Intensity of SNSs Use" (research question Q2), because of the inclusion of the interaction terms created for the moderation analysis though the two-stage approach (see Section 6.5.1.5). Hence, following Hair et al. (2017), PLS must be preferred to CB in this study.

Moreover, PLS-SEM was shown to reliably estimate a model with many latent variables and indicators even with a small sample size, for all those situations where CB-SEM provides inadmissible solutions (Chin et al., 2008; Henseler et al., 2014). As shown in more details in Section 6.6.3 and following the rule of thumb to determine the adequate sample size for PLS-SEM analysis, this research employs a smaller sample of the one required for CB analysis due to limited amount of financial resources.

Concerning the type of data, PLS has been shown to be more reliable than CB when Likert type measures and single items constructs are employed (Fornell and Bookstein 1982; Haenlein and Kaplan 2004; Reinartz et al., 2009). In this study, most of the indicators of the constructs use Likert-type items (Section 6.4) and some of the constructs (i.e., Political Knowledge and the interaction term created for the moderation analysis through the two-stage approach - Section 6.4.4.3) have just one indicator.

Moreover, when the data is not normally distributed, PLS-SEM must be preferred to CB-SEM analysis because it employs non-parametric techniques of analysis like bootstrapping or permutation procedures (Hair et al., 2017; Vinzi at al., 2010; Wong, 2010). As reported in Chapter 7, most of the indicators and variables of this study are non-normally distributed. Hence, PLS-SEM should be preferred to CB-SEM as method of analysis.

Moreover, when one of the aims of the study is to use latent variable scores in subsequent analyses, PLS should be preferred to CB analysis. Indeed, a final aspect of this study is to check whether the Social Capital and the Political Participation displayed by respondents differ on the basis of the type of SNSs employed and their usage level. This can be checked by analysing the different (unstandardised) Latent Variable Scores of each group of SNSs user (i.e., Facebook and Twitter) based on the time spent on those SNSs. The latent variable scores are automatically created when using the PLS algorithm but not in CB analysis. For all the reasons stated above, PLS-SEM was chosen as the analytical technique to answer the proposed research questions of this study. Following best practice guidelines on how to conduct PLS-SEM (e.g., Anderson and Gerbing, 1988; Bagozzi and Yi, 2012; Chin, 1998; Hair et al., 2017), the analysis was performed in two stages. The first step involves defining and analysing the measurement model to establish the reliability and validity of the measures, while the second step involves testing the structural model, which allows to delineate relationships among latent constructs. The following Sections discuss the two-steps approach and provide the metrics employed to assess and validate the proposed model of this study at each stage.

6.5.1.1. DEFINING AND ASSESSING THE MEASUREMENT MODEL

The first stage of the PLS-SEM two-step approach consists of defining and assessing the measurement model. A measurement model specifies the relationship between the observable indicators and each latent construct (Hair et al., 2017). The literature distinguishes between reflective and formative measurement models (Figure 6.5)

In reflective measurement models, the indicators are considered the effects of a latent construct (Chin, 1998), meaning that the causality is from the construct to its measures. This implies that all indicators of a constructs should be highly and positively correlated because they stem from the same domain and therefore are interchangeable (Hair et al., 2017).

With respect to formative measurement models, the causality of the relationship goes from the indicators to the latent constructs that they measure. Therefore, the indicators *form* the constructs and represent different aspects of it. This implies that the formative indicators may not correlate or may even be inversely related and are not interchangeable (Bollen, 1984).

Following Hair et al. (2017), the decisions on how to model the constructs and its indicators was taken considering how the constructs were modelled in previous studies and by looking at their conceptualisation in this thesis. In addition, a pre-test of the instruments through a pilot study to check validity and reliability of the indicators and their level of correlation was performed (Section 6.7). Moreover, a Confirmatory Tetrad Analysis in PLS-SEM (CTA-PLS) was conducted to empirically substantiate the

specification of measurement models (Hair et al., 2017). The CTA-PLS "*relies on the concept of tetrads that describe the difference of the product of one pair of covariances and the product of another pair of covariances*" (Hair et al., 2019: p. 14). The CTA-PLS analyses only "Non-Redundant Tetrads".

Figure 6.5. Reflective Measurement Models Vs Formative Measurement Models Operationalisation.



Source: adapted from Hair et al. (2017).

Redundancy exists when a tetrad can be represented by two other tetrads. In a reflective measurement model, tetrads are expected to have a value of zero and vanish because reflective indicators should represent one specific construct equally well, since they are assumed to stem from the same domain (Nunnally and Bernstein, 1994). The CTA-PLS draws on the bootstrapping procedure and calculates the bias-corrected Bonferroni-adjusted confidence intervals. If the confidence interval includes zero, then the tetrad is not significantly different from zero (the tetrad vanishes) and the model can be specified as reflective. Conversely, the model should be modelled as formative. However, the CTA-PLS can be performed only for constructs that have four or more indicators. Hence, the Political Interest construct that has three indicators has not been included in this kind of empirical analysis. However, whenever the above procedure cannot be implemented, Hair et al. (2017) provide some general guidelines summarised in Table 6.10 to help the researcher decide whether to measure a construct in a reflective or formative mode.

This study adopts reflective measurements for all the latent constructs measured through multiple indicators. In particular, the ISCS devised by Williams (2006) and used

in this study to operationalise Bond_SC and Brid_SC has been employed in previous research in a reflective manner (see among the others Alves et al., 2016 and Liu et al., 2014). Williams (2006) himself conceptualises Social Capital as the element which causes and increases trust, reciprocity and emotional support. These are all aspects measured through the indicators of the scale, implying a causal direction from the construct to its indicators. Moreover, the scale was pre-tested through a pilot study and a CTA-PLS was carried out which confirmed the reflective measurement specification of Bond_SC and Brid_SC. Indeed, all the tetrad reported in Table 6.11 vanished since all the 90% bias-corrected Bonferroni-adjusted confidence intervals of the non-redundant tetrad contain the 0. To carry out the CTA-PLS, procedure and settings suggested by Hair et al. (2018) with the SmartPLS software 3.2.8 have been applied (see Appendix K). All the other CTA-PLSs performed during the pilot study for the remaining constructs are provided in Appendix J.

Criterion	Decision	Reference	
Causal priority between the indicator and the construct	 From the construct to the indicators: reflective From the indicators to the construct: formative 	Diamantopoulos and Winklhofer (2001)	
Is the construct a trait explaining the indicators or rather a combination of the indicators?	 If trait: reflective If combination: formative 	Fornell and Bookstein (1982)	
Do the indicators represent consequences or causes of the construct?	 If consequences: reflective If causes: formative 	Rossiter (2002)	
Is it necessarily true that if the assessment of the trait changes, all items will change in a similar manner (assuming they are equally coded)?	 If yes: reflective If no: formative	Chin (1998)	
Are the items mutually interchangeable?	 If yes: reflective If no: formative	Jarvis, MacKenzie, and Podsakoff (2003)	

 Table 6.10. General Guidelines to Choose the Measurement Model Mode.

Source: Hair et al. (2017: p 52).

CTA - Bond_SC			Tetrad
Non-Redundant Tetrad	CI Low adj.	CI Up adj.	Vanish
1: Bond_1,Bond_10,Bond_2,Bond_3	-2.671	0.744	Yes
2: Bond_1,Bond_10,Bond_3,Bond_2	-1.506	1.643	Yes
4: Bond_1,Bond_10,Bond_2,Bond_4	-2.476	0.695	Yes
6: Bond_1,Bond_2,Bond_4,Bond_10	-1.922	1.698	Yes
7: Bond_1,Bond_10,Bond_2,Bond_5	-2.044	0.111	Yes
10: Bond_1,Bond_10,Bond_2,Bond_6	-2.280	0.244	Yes
13: Bond_1,Bond_10,Bond_2,Bond_7	-2.460	0.370	Yes
17: Bond_1,Bond_10,Bond_8,Bond_2	-2.290	0.937	Yes
20: Bond_1,Bond_10,Bond_9,Bond_2	-1.514	0.581	Yes
29: Bond_1,Bond_10,Bond_6,Bond_3	-1.678	1.447	Yes
31: Bond_1,Bond_10,Bond_3,Bond_7	-1.533	1.238	Yes
35: Bond_1,Bond_10,Bond_8,Bond_3	-2.459	1.310	Yes
41: Bond_1,Bond_10,Bond_5,Bond_4	-1.146	1.323	Yes
43: Bond_1,Bond_10,Bond_4,Bond_6	-1.253	2.086	Yes
47: Bond_1,Bond_10,Bond_7,Bond_4	-0.754	2.367	Yes
50: Bond_1,Bond_10,Bond_8,Bond_4	-0.953	1.941	Yes
60: Bond_1,Bond_5,Bond_7,Bond_10	-1.394	1.207	Yes
64: Bond_1,Bond_10,Bond_5,Bond_9	-1.208	1.594	Yes
66: Bond_1,Bond_5,Bond_9,Bond_10	-0.404	1.781	Yes
71: Bond_1,Bond_10,Bond_8,Bond_6	-1.086	2.373	Yes
80: Bond_1,Bond_10,Bond_9,Bond_7	-0.136	2.324	Yes
91: Bond_1,Bond_2,Bond_3,Bond_6	-0.504	3.044	Yes
120: Bond_1,Bond_5,Bond_6,Bond_2	-0.652	1.187	Yes
169: Bond_1,Bond_3,Bond_5,Bond_8	-0.896	1.826	Yes
182: Bond_1,Bond_3,Bond_9,Bond_6	-0.407	2.368	Yes
205: Bond_1,Bond_4,Bond_6,Bond_7	-0.995	2.139	Yes
233: Bond_1,Bond_5,Bond_8,Bond_7	-1.737	0.671	Yes
236: Bond_1,Bond_5,Bond_9,Bond_7	-0.833	1.573	Yes
248: Bond_1,Bond_6,Bond_9,Bond_8	-1.133	1.544	Yes
281: Bond_10,Bond_2,Bond_8,Bond_4	-0.866	2.823	Yes
324: Bond_10,Bond_4,Bond_7,Bond_3	-1.570	1.147	Yes
358: Bond_10,Bond_3,Bond_8,Bond_9	-0.502	2.192	Yes
395: Bond_10,Bond_5,Bond_8,Bond_6	-1.123	1.631	Yes
434: Bond 2,Bond 3,Bond 9,Bond 4	-0.941	2.012	Yes
526: Bond 3.Bond 4.Bond 5.Bond 6	-1.589	1.497	Yes

Table 6.11. Bias-Corrected and Bonferroni-Adjusted Confidence Intervals of theCTA-PLS analysis for Bond_SC of the Facebook Users Only Sample (n = 101).

The six-item Online Political Participation scale, adopted from Gil de Zuniga et al. (2015) was originally devised for the SEM context and has been employed (among others) by Lin and Chiang (2017), with a reflective measurement specification. Offline Political Participation was operationalised through the nine-item scale devised by Gil de Zuniga et al. (2012) for SEM analysis and further used in the SEM context by Gil de Zuniga et al. (2013) and Gil de Zuniga et al. (2014b). Both scales were pre-tested through a pilot study and CTA-PLS was performed which confirmed the reflective measurement specification of the two constructs (Appendix J).

Political Interest (PInt)was measured through the scale devised by Brady et al. (1995) and employed in CB-SEM analysis by Bell et al. (2009), Carter (2006), Fowler et al. (2008) and Gastil and Xenos (2010), always in a reflective manner. The scale was tested through a CTA-PLS during the pilot study which substantiated the reflective measurement specification of PInt.

Niemi et al.'s (1991) Political Efficacy four-item scale adopted in this study has been employed in numerous previous studies using SEM (e.g., Anim et al., 2019; Ardèvol-Abreu et al., 2019; Park and Karan 2014; Ritter, 2008; Warner, 2018; Yang and DeHart, 2016). Park and Karan (2014), who used PLS technique and SmartPLS software, modelled the scale as reflective. Moreover, this study's conceptualisation of Political Efficacy as the "*individuals' self-perceptions that they are capable of understanding politics and competent enough to participate in political acts*" (Miller et al., 1980; p. 253) implies that the construct is a trait explaining the indicators, in line with reflective measurement specifications. A CTA-PLS analysis performed for the pilot study empirically confirmed the reflective specification of PEff (Appendix J).

All other variables included in the model are measured either at the categorical level or through a single item and therefore do not require any reflective/formative specification. The general reflective measurement model is reported as displayed in SmartPLS in Figure 6.6.

Once the measurement model has been specified it must be assessed and validated in terms of reliability and validity. The next Section deals with this issue.



Figure 6.6. General Measurement Model.

Note: the moderator variables "Type of SNSs Used" and "Combined/not-Combined Use of SNSs" are not included in the model as they will be used as grouping variables to allow for the comparison of the proposed model across different groups to answer research questions Q1 and Q3. Also, "Intensity of SNS Used" will be successively included as interaction term in moderation analysis to answer research question Q2

ASSESSING THE MEASUREMENT MODEL

Following Hair et al. (2017), reflective measurement models must be assessed with respect to their internal consistency reliability, convergent and discriminant validity.

Internal consistency reliability refers the degree to which all the indicators appointed to measure the same construct are interrelated (Hair et al., 2017). Following best practices in PLS-SEM, the internal consistency reliability is assessed using Cronbach's Alpha (CA) and Composite Reliability (CR) (Hair et al., 2017; Nunally and Bernstein 1994). CA yields an estimate of the reliability based on the inter-correlations of the indicator variables of a construct. However, CA tends to inflate as the number of scale items increases (Henseler et al., 2009). For this reason, researchers called for the use of a more robust indicator like the CR which provides the shared variance among a set of indicators measuring an underlying construct (Hair et al. 2017). Hair et al. (2017: p.112) suggest that values of CA and CR above 0.7 are satisfactory.

<u>Convergent validity</u> is the extent to which an indicator correlates positively with other indicators and share a high proportion of the variance of a construct. To evaluate convergent validity the outer loadings (OLs), the communality of the indicators and the average variance extracted (AVE) are employed (Hair et al., 2017).

The OLs refer to the correlation between an indicator variable and the theoretical construct it represents (Chin 1998). A common rule of thumb is that the standardised OLs should be 0.6 or higher (Hair et al., 2017). Following Hair et al. (2017), the communality of the indicators is computed as the square of a standardized indicator's outer loading which specifies how much of the variation in an item is explained by the construct. An established rule of thumb is that a latent variable should explain at least 50% of each indicator's variance (Hair et al., 2017: p. 113). The AVE measures the amount of explained variance that a construct obtains from its indicators relative to the amount of variance due to measurement errors (Fornell and Larcker, 1981). As a rule of thumb, AVE should be higher than 0.5 (Henseler et al., 2009).

<u>Discriminant Validity</u> is "the extent to which a construct is truly distinct from other constructs by empirical standards" (Hair et al., 2017: p. 115). Following best practices in PLS-SEM, this research employs several tests to empirically assess discriminant validity (Henseler et al., 2015).

First, the Cross Loadings are assessed to ascertain whether each indicator's loading with its associated latent construct is higher than its loadings with all the other constructs (Chin 1998). In that case, discriminant validity is established at the indicator level. Second, the Fornell–Larcker criterion (Fornell and Larcker, 1981) is employed to ascertain if a latent construct shares more variance with its assigned indicators than with another construct. Accordingly, the AVE of each latent construct should be greater than the latent construct's highest squared correlation with any other latent variable. Lastly, Henseler et al. (2015) suggest using the heterotrait-monotrait ratio (HTMT) of indicators correlations to estimate what the true correlation between two constructs would be if they were perfectly measured. Henseler et al. (2015: p. 121) suggest that if the HTMT value is below 0.90, discriminant validity has been established between a given pair of reflective constructs. Furthermore, the HTMT can serve as the basis of a statistical discriminant validity test carried out through the bootstrapping procedure to derive a

distribution of the HTMT statistic assuming a two-tailed 95% confidence interval level and generated using 5000 subsamples (Hair et al., 2017). As a rule of thumb, discriminant validity is established, if the HTMT confidence intervals do not contain the value 1.

All the above metrics will be provided by the SmartPLS software 3.2.8 when running the PLS-SEM algorithm. If the measurement model does not meet the indicated thresholds, some modifications can be made to fit the guidelines provided as reported in Table 6.12.

Problem	Possible Solutions	Source
Outer loading between .40 and .70	Remove indicator only if deletion leads to an increase of either the CR or the AVE.	Hair et al. (2017)
Outer loadings < .40	Always remove item from the model unless deletion jeopardise the content validity of the construct.	Hair et al. (2017, 2018)
Indicators Communality < .5	Accepted if removing indicator does not lead to an increase of CR and AVE or if deletion compromise content validity.	Hair et al. (2017)
AVE < .5	Option 1: if CR > 0.6, the convergent validity can be still considered adequate.	Fornell and Larcker, 1981. Lam, 2012. Malhotra and Dash, 2011.
	Option 2: check VIF to see if multicollinearity is an issue of the model and consider combining high correlated constructs into second order constructs.	Hair et al. (2010)
	Option 3: remove the problematic construct or the items with low loading values if theory support it and if face validity is not jeopardised.	
HTMT > .90	Option 1: delete items that shows low correlations with other items of the same construct.	Hair et al. (2017)
	Option 2: remove those items that are strongly correlated with other items in different constructs.	
	Option 3: merge the constructs that are strongly correlated if measurement theory support this step.	

Table 6.12. Measurement Model Assessment Problems and Solutions.

Source: see references in table.

Once the measurement model is assessed, the next step of PLS-SEM analysis involves defining and assessing the structural model (Hair et al., 2017). The next two Sections deal with this second step.

6.5.1.2. DEFINING AND ASSESSING THE STRUCTURAL MODEL

The structural (inner) model of this research stems from the theoretical frameworks of the Resource Mobilisation Theory applied within the context of Social Capital and Political Participation as discussed in Chapter 5 and which is represented in Figure 6.7. However, it must be highlighted that for each research question Q1, Q2 and Q3 separate structural models will be provided, according to the type of moderation analysis performed (see Chapter 7). Indeed, when multiple moderators are included in the structural model, Hair et al. (2013) suggest considering one moderator at a time to maintain interpretability of the results.



Figure 6.7. General Structural Model.

ASSESSING THE STRUCTURAL MODEL

Following Hair et al. (2017), assessing the structural model means to determine the predictive capability and accuracy of the model with respect to the endogenous variables and ascertain the magnitude and significance of the relationships between constructs. Accordingly, several criteria were employed to assess the inner models of this study.

Firstly, the Variance Inflation Factor (VIF) which indicates whether collinearity exists between variables (multicollinearity) was assessed. Collinearity would increase variables' standard errors affecting the ability to spot any significant effects of the path coefficients. As a rule of thumb, if VIF coefficients are below 5.0, the model is considered free of multi-collinearity issues.

Then, the standardised values of the path coefficient between the latent variables were evaluated with respect to their: (1) algebraic sign (positive/negative), (2) magnitude (the closer to +1 or -1 the stronger the relationship), and (3) significance. To ascertain the significance of the path coefficients, this study relies on the bootstrapping procedure suggested by Hair et al. (2017) that generates confidence intervals (CI) for each path estimate through 5000 random samples assuming a 95% level of confidence. If a path coefficient CI does not include zero, we can assume that a significant effect exists. Following Hair et al.'s (2017) recommendations, this study employs the Bias-Corrected and accelerated Confidence Intervals (BCa) because the model is relatively complex and the data is critically skewed (see Chapter 7).

To further strengthen confidence in the results and control for type II error, we ascertained whether any non-significant effects were due to the lack of statistical power (false negative) conducting post-hoc power tests (1- β) as a function of significance level α , sample size, and variables' effect size f^2 using the G*Power 3.1.9 software (Faul et al., 2007, 2009). The recommended statistical power value is 0.80 or higher (Cohen, 1988).

Furthermore, the R^2 or coefficient of determination was assessed to estimate a latent variable explained variance to its total variance (Hair et al., 2017: p. 198). The R^2 value ranges from 0 to 1, with higher levels indicating greater predictive accuracy. Hair et al. (2017) suggest referring to the appropriate research discipline and practices to establish which R^2 values can be considered acceptable in any study. In particular, Sarstedt and Mooi (2019) pinpointed that in cross-sectional designs like the one of this thesis, values of 0.30 are common and therefore " R^2 values of 0.50, 0.30, and 0.10 can, as a rough rule of thumb, be respectively described as substantial, moderate, and weak"

(p. 229). However, Rovai et al. (2013) and Tokunaga (2018) claim that Cohen's (1988) R^2 thresholds of 0.02 (weak) 0.13 (moderate) and 0.26 (substantial) are an accepted rule of thumb, in social and behavioural research. Nonetheless, there is apparent consensus among researchers that the values of \mathbb{R}^2 should be sufficiently high (at least 0.10) for the model to have a minimum level of explanatory power (Falk and Miller, 1992). Provided that this threshold is above the minimum value of R^2 suggested by Cohen (1988), this research adopted the rule of thumb suggested by Sarstedt and Mooi (2019) and supported the R^2 with a measure of precision coming from Ohtani (2000) and Streukens and Leroi-Werelds (2016). They proposed to construct BCa CI around the estimated R^2 value to formally assess whether an endogenous construct's R² value is significantly different from zero. If the p value of the bootstrap distribution of the observed value of R^2 is below 0.05 level of significance, then the obtained R^2 is significantly different from 0 and the model has significant explanatory power. The bootstrap procedure is preferred to other test of significance like the "F-test of significant deviation from zero" because it does not rely on distributional assumption of normality. Thus, it is considered adequate for this research. BCa Bootstrap confidence intervals are obtained in SmartPLS through the bootstrapping procedure suggested by Hair et al. (2017).

However, according to Hair et al. (2017), the R² is not a reliable measure when the aim is to compare and select different models according to their predictive accuracy (in this thesis we aim to compare moderated and non-moderated relationships). Indeed, the R² coefficient increases with the number of variables used. Therefore, "*it has an inherent bias toward selecting models with many predictor variables, including those that may be only slightly related to the endogenous constructs*" (Hair et al., 2017: p. 199). Rather, for comparisons purposes it is recommended to use the Adjusted Coefficient of Determination (R²_{adj}) which recalculates the R² values accounting for the number of exogenous variables of a model and the relative sample size.

To properly assess the structural model, we also tested for the size of the impact of the predictor variables on the endogenous constructs by means of Cohen's f^2 effect size (Cohen 1988) which shows changes in the R² value of an endogenous construct when an exogenous variable is removed from the analysis (Hair et al., 2017). As a rule of thumb, values of .02, .15 and .35 represent respectively "small", "medium" and "large" effects (Cohen, 1988). Moreover, to evaluate the predictive ability of the structural model, the Stone-Geisser's Q^2 value was obtained using the blindfolding resampling technique. The blindfolding procedure deletes every d^{th} data points in the endogenous construct's indicators and estimates the parameters with the remaining data points. The omission distance d must be chosen between the values of 5 and 10 so that the number of data points divided by d is not an integer (Hair et al., 2017). The deleted data points are then predicted through the estimates obtained from the blindfolding procedure. The difference between the original (deleted) and the predicted value is called prediction error and serves as input of the Q^2 measure. As a rule of thumb, Q^2 values larger than zero indicate good predictive relevance for the specified endogenous constructs.

Linked to the Q^2 value is the q^2 effect size, which measure the relative impact of the predictive relevance of each exogenous variable on the endogenous constructs (Hair et al., 2017) through the following formula:

$$q^{2} = \frac{Q_{\text{included}}^{2} - Q_{\text{excluded}}^{2}}{1 - Q_{\text{included}}^{2}}$$
Equation 6.1

where Q^2_{included} is the computation of the Q^2 value with all the constructs of model included and Q^2_{excluded} is the predictive relevance value obtained when the construct of which we want to detect the relative predictive relevance is excluded. As a rule of thumb, values of .02, .15, and .35 indicate that an exogenous construct has a small, medium, or large predictive relevance, respectively (Hair et al., 2017).

However, one of the objectives of this thesis is to compare different models (moderation/non-moderation) and ascertain which one shows higher predictive accuracy meanwhile retaining acceptable "fit" levels. Prior research in PLS-SEM has demonstrated that the R^2 , Q^2 and related statistics are not reliable enough when the objective is model's comparison and selection according to their predictive relevance (Berk, 2008; Evermann and Tate, 2010; Henseler et al., 2014; Rönkkö and Evermann, 2013; Sharma et al., 2019b). Hence, the criteria employed to compare models are presented in the following Section.

MODEL SELECTION CRITERIA

Although PLS-SEM was initially devised for prediction purposes, in recent years academics extended its application to theory testing by developing model fit measures (Hair et al., 2017) to be used especially when the path model includes reflectively measured constructs like the one proposed in this study (Dijkstra and Henseler, 2015; Hair et al., 2011, Henseler et al., 2014). These measures provide an indication of how well a model fits the empirical data and therefore allow to identify any model misspecifications.

Specifically, Henseler et al. (2014) found that the Standardised Root Mean Square Residual (SRMR) index can be used to successfully avoid model misspecification in PLS-SEM and to make the right selection among competing models. Indeed, researchers have adopted this model fit index for PLS-SEM (e.g., Gaskin et al., 2018). The SRMR index allows the assessment of the average magnitude of the discrepancies between observed and expected correlations as an absolute measure of model fit criterion. As a rule of thumb, values below 0.10 are generally considered good fit (Hair et al., 2017).

Also, the Root Mean Square Residual Covariance (RMS_{theta}) performs well with reflective measurement models. This index assesses the degree to which the residuals of the measurement model correlate. As such, the RMS_{theta} should be close to zero to indicate good model fit, because it would imply that the correlations between the outer model residuals are very small. Henseler et al. (2014) suggest that values below 0.12 indicate a well-fitting model.

However, researchers should try to strike a balance between predictive power and model fit when analysing different competing models (Sharma et al. 2019a,b). Accordingly, Shmueli et al. (2016) and Sharma et al. (2019a, b) proposed a series of *insample* and *out-of-sample* criteria to analyse the predictive capability of structural models that must be used in combination with the model fit measures discussed and with the R^2 , Q^2 and R^2_{adj} metrics.

<u>In-sample prediction measures</u> are helpful in the explanatory modelling context. To generate in-sample prediction measures, the parameters of a PLS model are estimated using a given dataset and then the same hypothesised model is used to predict values for cases from the same sample (Shmueli et al., 2016). Typical in-sample measures are R^2 and R^2 adj (Hair et al., 2017). However, through a Monte Carlo study, Sharma et al. (2019b) found that other criteria like the Akaike's (1973) information criterion (AIC) corrected by Sugiura's (1978) (AICc), the McQuarrie and Tsai's (1998) unbiased AIC (AICu) and the Bayesian information criterion (BIC) achieved significantly better model selection success rates than R^2 and R^2 adj. The AIC helps to estimate the relative amount of information lost when a given model estimated from data is compared to a "true" but unknown data generating process. The BIC metric instead "*provides an estimate of the posterior probability of a model being true and chooses the model that maximizes this probability on a given data set*" (Sharma et al., 2019a: p. 4). As a rule of thumb, researchers should select the models which show lower values of AICu, AICc and BIC as those models have superior explanatory power.

However, as Shmueli et al. (2016) stated: "fundamental to a proper predictive procedure is the ability to predict measurable information on new cases" (p. 4553). For this purpose, <u>out-of-sample criteria</u> are helpful as they involve "estimating the model on a training sample that excludes the cases to be predicted. The model is then used to predict new cases" (Shmueli et al., 2016: p. 4555).

In particular, the use of the Root Mean Squared Error (RMSE), the Mean Absolute Deviation (MAD) and the Q^2 values are encouraged. The RMSE is the standard deviation of the residuals and indicates the prediction errors that is how spread out those residuals are. The MAD is the average of the absolute differences between prediction and actual observation where all individual differences have equal weight. The Q^2 metric has been already discussed. As a rule of thumb researchers should select the model, which minimises RMSE and MAD values in the latent variable scores (Sharma et al., 2019a).

In this thesis, the "PLS predict procedure" was used in the SmartPLS software to obtain RMSE and MAD (Shmueli et al., 2016). Specifically, out-of-sample predictive validity measures are obtained using the k-fold cross-validation procedure with holdout samples which splits the sample into k equally sized subsamples of data. The PLS algorithm then predicts each fold (hold-out sample) with the remaining k-1 subsets, which, in combination, become the training sample. To obtain more stable results of the

predictive relevance of the model, the k-fold cross-validation procedure is repeated several times with different random data partitions. Then, the average across the repetitions is computed. Following Shmueli et al. (2016), k and the number of repetitions are set at 10 when running the PLS Predict procedure in SmartPLS.

Also, a significance test ascertaining whether the change in \mathbb{R}^2 due to the introduction of new variables in the model (control variables or moderators) is significant will be provided. The test is commonly known as F-test (Field, 2013) and is computed through the following formula:

$$F = \frac{(R_L^2 - R_S^2) / (df_L - df_s)}{(1 - R_L^2) / (N - df_L - 1)}$$
 Equation 6.2

where R^2_L is the R-squared of the endogenous variable from the model with more variables, and R^2_S is the R-squared of the predicted variable from the model with fewer variables. The quantity df_L - df_S is the difference in the number of variables between the two models. N is the sample size.

However, some precautionary notes must be made. Firstly, the in-sample and outof-sample selection criteria discussed should not be considered as substitute of the commonly used PLS criteria (R^2 , R^2 adj and Q^2) but rather as complementary measures useful for models' comparisons (Sharma et al., 2019a). Secondly, Sharma et al. (2019b) advise that "model selection criteria are not meant to automate model selection with minimal thought. Rather, [...] selecting one model over another should proceed primarily on the basis of theoretical arguments aided by empirical evidence" (p. 379). Even the priority that researchers should give to all the in-sample and out-of-sample criteria whether they provide discordant results is disputed (Sharma et al., 2019a, b). Moreover, the strike of a balance between adequate model fit and good predictive relevance is left to the researcher who should always take into consideration the main objective of the study and whether it focuses on explanatory power or predictive relevance or both like in the case of this thesis. Hence, in the process of model selection when comparing structural models with and without control variables and with and without moderators, we prioritised the model that showed good predictive relevance while retaining acceptable "fit" levels.

6.5.1.3. ANSWERING RESEARCH QUESTIONS Q1, Q2, AND Q3 AND TESTING FOR MODERATION

A moderator variable can be defined as a "variable that affects the direction and/or strength of the relation between an independent variable and a dependent variable" (Baron and Kenny, 1986: p. 1174). Following Hair et al. (2017) two approaches can be employed to examine moderating effects in PLS-SEM depending on the purpose of the moderation and the type of moderator variables to test:

- 1. The Interaction Term Moderation approach (ITM).
- 2. The Multi-Group Analysis approach (MGA).

Both approaches are employed in this thesis to answer different research questions as discussed in the following Sections (See also Table 6.9, Section 6.5).

<u>The ITM approach</u> should be employed when very few relationships within a structural model are moderated by a third variable that is a metric or quasi-metric continuous or scaled variable (Hair et al., 2017). Given an exogenous (Y_1) and an endogenous (Y_2) variable, the ITM (Figure 6.8) creates a third variable (interaction term) computed as a mathematical product of the moderator (M) and Y₁ that is theorised to affect the path between Y₁ and Y₂.



Figure 6.8. Interaction Term Moderation Approach.

Source: Hair et al. (2017, p. 248).

Researchers have proposed several statistical techniques for creating the interaction term (Table 6.13) like the product indicator approach, the orthogonalising approach, and the two-stage approach (Hair et al., 2017).

Approaches	Interaction Term Computational Technique	Advantages	Disadvantages
Product Indicator	Multiplies each indicator of the exogenous latent variable with each indicator of the moderator variable. The product indicators become the indicators of the interaction term.	– Easy to implement.	 Works only with reflective constructs. Produces collinearity yielding inflated standard errors and biased path coefficient estimates. Variables standardisation prevent from easily compare the main effect with the simple effect.
Two-Stage	Runs the PLS algorithm for the model with no moderation to obtain the latent scores of the exogenous and moderating variables (stage 1) and then multiply those scores to create a single item measure that is the interaction term.	 Works with reflective and formative constructs. Higher levels of statistical power compared to the other two approaches maximising chances to find moderation effects. 	– Does not maximise prediction.
Orthogonalizing Approach	Creates all product indicators of the interaction term like the product indicator approach and then regresses each product indicator on all indicators of the exogenous construct and the moderator variable. Then, uses the standardized residuals " <i>e</i> " as indicators for the interaction term.	 Prevent any collinearity issues. Maximise prediction. Facilitates the interpretation of the moderating effect's strength compared to product indicator approach. 	 Lower statistical power than other approaches.

 Table 6.13. Interaction Term Approaches Characteristics.

Source: adapted form Hair et al. (2017).

This thesis employs the two-stage approach proposed by Chin et al. (2003) to answer research question Q2. The moderator variable "Int_SNS_Use" is a metric measure operationalised through the amount of time spent on the SNSs gauged in minutes. Accordingly, an interaction term approach must be preferred to MGA techniques. Moreover, in a Monte Carlo study comparing the statistical power, predictive accuracy and reliability of the above interaction term approaches, Henseler and Chin (2010) suggest using the two-stage approach when the objective of the analysis is to ascertain whether a moderating effect is significant or not because it yields higher levels of statistical power. <u>The MGA approach</u> should be preferred when the researcher needs to test all the relationships of a model (or many of them) for the moderating effects of a categorical moderator variable (Vinzi et al., 2010). Categorical variables (like "Type of SNSs Used" and "Combined/not Combined Use of SNSs" in this study) are employed as grouping variable which split the data into subsamples to be compared to investigate any significant differences in the path coefficients of the relationships of the structural model. Following Hair et al. (2017) several types of non-parametric MGA can be used to test for moderating effects depending on the number of groups that must be compared (Figure 6.9).



Figure 6.9. MGA Approaches in PLS-SEM

Source: Hair et al. (2017, p. 293).

In particular, when the aim is to test difference across two groups and data are non-normally distributed like in this study (see Chapter 7), either the Permutation test or the Partial Least Square Multigroup Analysis (PLS-MGA) can be employed. The former works by permuting "observations between the groups and re-estimating the model for each permutation" for both groups (Hair et al., 2017, p. 294). Then, the differences for the path coefficients of the two models are computed for each permutation and a twotailed 95% CI of the differences is created. If the original difference of a specific path coefficient does not fall within the CI range, then the moderation effect is statistically significant. The PLS-MGA works on the bootstrapping procedure by comparing "each bootstrap estimate of one group with all other bootstrap estimates of the same parameter in the other group. By counting the number of occurrences where the bootstrap estimate of the first group is larger than those of the second group, the approach derives a probability value for a one-tailed test" (Hair et al. 2017, p. 294).

Extant PLS-SEM literature (e.g., Ernst, 2004; Hair et al., 2018; Sarstetd et al., 2011) provides evidence that the permutation test controls very well for type I error, performs well across a wide range of conditions, and must be preferred when two-tailed tests are performed. Provided the categorical nature of the moderator variable "Type of SNSs Used" and "Combine and not Combined Use of SNSs", given the non-directional nature of the hypothesis to be tested and provided the advantages in controlling for type I error of the permutation test over PLS-MGA, this research employed the permutation procedure to answer research questions Q1 and Q3.

However, before conducting a Permutation test, the <u>measurement invariance</u> among the constructs of the different groups to be compared must be established to ensure the validity of outcomes and conclusions (Hair et al., 2017). *Measurement invariance* indicates that the same constructs which are being measured across different groups have the same meaning for all the groups compared (Hair et al., 2017). Hence, "failure to establish data equivalence is a potential source of measurement error" (Hult et al., 2008: p. 1028) which could lead to misleading interpretation of the results. To test measurement invariance, Henseler et al. (2016) proposed the Measurement Invariance of Composite Models (MICOM) procedure consisting of the three hierarchical stages:

- 1. **Establish configural invariance,** that is constructs must be equally parameterised (identical measures) and estimated (identical PLS algorithm setting) across groups.
- 2. Establish compositional invariance, meaning that the composite scores of the latent variables are the same across groups.
- 3. Ensuring equality of composite mean values and variances across groups.

Following Henseler et al. (2016), configural invariance was assessed through a qualitative inspection of the composites' specifications and algorithm settings across all

the groups. Compositional invariance was assessed through a permutation test examining whether the correlation (*c*) between composite score of the same constructs across different groups was equal to 1. Otherwise, "*if c is significantly different from one, we must* [...] conclude that there is no compositional invariance" (Henseler et al., 2016: p. 414). To run the test in SmartPLS 300 iterations, with stop criterion of $1*10^{-7}$ with a number of 5000 permutations and a significance level of 5% as suggested by Sarstedt et al. (2011) were employed.

When configural and compositional invariance are established, then partial measurement invariance is confirmed. Accordingly, MGA and the comparison of path coefficients is meaningful. If the composites have also equal means and variances across the groups, full measurement invariance is confirmed, and researchers can pool the data of different groups and carry out an aggregate analysis benefiting from the increase in statistical power. However, given that the main concern of this study is not to carry out an analysis on the pooled level of the data of the groups but rather account for heterogeneity of respondents, the measurement invariance analysis will be conducted till step 2 of the MICOM procedure.

For reasons of convenience and ease of consultation all the metrics and procedures to assess the measurement and the structural model and the related SmartPLS algorithm settings are summarised in tables reported in Appendix K.

6.5.2. RESEARCH QUESTION Q4: TWO-WAY ANOVA

Research question Q4 investigates significance differences in the amount of Bond_SC and Brid_SC owned by users of different SNSs (Facebook and Twitter) according to the intensity of use of those SNSs (light, mild and heavy users). This means testing for the "interaction effect" of those categorical variables (Field, 2013). For this purpose, a two-way (factorial) analysis of variances (ANOVA) is recommended (Field, 2013). Following best practice guidelines in research, separate ANOVAs for each dependent variable were performed (Field, 2013; George and Mallery, 2020). Although this practice ignores any relationship between the dependent variables and can increase the Type I error because multiple tests are carried out on the same data and groups, the literature legitimate it in the name of clarity, ease of interpretation of results and parsimony (George and Mullary, 2020: p. 300). Researchers suggested to control for

Type I error inflation (familywise error rate) using procedures like the Sidak correction (Hair et al., 2018) which tests each individual hypothesis at a significance level of 1-(1-alpha)^{1/m} where m is the number of comparisons and alpha is the initial established level of significance. The Sidak correction was preferred to other procedures like the Bonferroni correction as it achieves a higher statistical power (Hair et al., 2018).

Yet, some limitations must be highlighted. The variable "Time Spent Online" is a ratio variable, therefore it had to be rearranged into three categories (light, mild and heavy users) through the quartile split method, so that it could be included as a categorical variable in the analysis and meet assumption 2 of the ANOVA. This entailed that part of its variance was lost (Henseler and Fassott, 2010). However, transforming a ratio variable to a categorical one is a common approach in research, especially when the use of more complicated statistical techniques of analysis like the Analysis of Covariances (ANCOVA) is not feasible. Indeed, although ANCOVA allows to test for interaction effect of ratio variables it was not employed in this study because its many assumptions were not met. Also, as a rule of thumb the quartiles employed to split the variable "Time Spent Online" should not be included into any of the groups created because they are used as splitting points (Field, 2013). This means that information is lost. However, this is a problem common to any other type of splitting procedure, like the Median or the Mean split methods.

Before carrying out the two-way ANOVA analysis the following assumptions must be met (Field, 2013):

1. Assumption 1: the dependent variable should be measured at the continuous or discrete level and should be either interval or ratio variables. Specifically, Bond_SC and Brid_SC are measured at the interval level using the unstandardized variable scores obtained from the analysis carried out in PLS-SEM. The use of Unstandardized Scores was preferred to the sum score procedure because it accounts for differences in the individual item weights (Henseler et al., 2014) that if ignored can produce "substantial biases in the parameter estimates" (Hair et al., 2017: p. 30) as the researcher would not account for measurement error (Section 6.5.1).

- Assumption 2: the independent variables should each consist of two or more categorical, independent groups. In this study, the variable "Type of SNSs Used" has two categories (Facebook users and Twitter Users) and the variable "Intensity of SNSs Used" is split in three categories (Light, Mild and Heavy users).
- 3. Assumption 3: the groups of analysis must be independent. Also in this case the assumption is met; Facebook users are not in the group of Twitter users and vice versa. Also, light, mild and heavy users' samples of Facebook and Twitter include different respondents.
- 4. **Assumption 4**: There should be no significant outliers in the data. Outliers will be investigated through scatter plot diagrams. As Chapter 7 demonstrates, this assumption is also met for both dependent variables (Bond_SC and Brid_SC).
- 5. Assumption 5: the dependent variable should be approximately normally distributed for each combination of the groups of the two independent factorial variables. However, the two-way ANOVA is quite "robust" to violations of normality (Field, 2013; George and Mallery, 2020). This is particularly true with studies that employ samples with more than 30 respondents (like this study) (Field, 2013). The normality of variables is assessed by looking at the values of Skewness and Kurtosis. To be normal, a distribution should have values of Skewness between ±1 (Hair et al., 2017) and Kurtosis between ±2 (George and Mallery, 2020). The observation of those metrics was preferred to normality tests like the Kolmogorov–Smirnoff test, the Shapiro–Wilk test and the significance tests of skewness and kurtosis because in large samples like that of Mild Facebook users (n = 162) "they can be significant even for small and unimportant effects" (Field, 2013; p. 184).
- 6. Assumption 6: there must be homogeneity of variances of the dependent variables (Bond_SC and Brod_SC) for each combination of the groups of the two independent variables (Time Spent Online and Type of SNSs Used). This assumption was tested through the Levene's test (Field, 2013). If the Levene's test is significant (p < .05) then the assumption of homogeneity of variances is violated. However, when the sample size is large, small differences in group variances can produce a Levene's test that is significant (Field, 2013). Moreover, the two-way ANOVA is robust against violations of the assumption of

homogeneity of variances if sample sizes are roughly equal (Field, 2013). In the case of this study, samples sizes of groups are not equal. Hence, violation of this assumption could provide unreliable results.

Notably, assumptions 1, 2 and 3 have been met. Assumptions 4, 5 and 6 are tested using the SPSS 24 software and the outcomes are reported in Chapter 7, Section 7.4.

6.5.3. RESEARCH QUESTION Q5: NON-PARAMETRIC TWO-WAY ANOVA

Research question Q5 investigates whether there are any statistically significant differences in the amount of offline and online Political Participation reported by different users of different SNSs (Facebook and Twitter) who show different levels of intensity of SNSs use (Light, Mild and Heavy). Following the rationale provided in the previous Section, for reasons of clarity of interpretation of the results and parsimony (Field, 2013; George and Mallery, 2020), the Online and Offline Political Participation dependent variables will be analysed separately.

However, in this case a non-parametric version of the two-way ANOVA is employed. This is because the two dependent variables did not meet the assumptions of normality and homogeneity of variances across groups (Chapter 7, Section 7.5). Although the two-way ANOVA is robust against small variation from normality or when homogeneity of variances is not met, empirical evidence suggests that when sample sizes of the groups to be compared are not similar (like in the case of this study) then ANOVA can yield unreliable results (Field, 2018). Hence, a non-parametric alternative must be considered, unless data transformation and deletion of outliers are used.

"A very popular method for a nonparametric analysis of variance judging from the number of publications...is the Aligned Rank Transformation (ART)" (Luepsen, 2017: p. 6924). This procedure was devised by Hodges and Lehmann (1962) and was extend to the factorial design (two-way) by Higgins and Tashtoush (1994). The ART procedure transforms the dependent variable into ranks and reorders them by giving the smallest observation the rank 1, the second smallest the rank 2, etc., but only after an alignment of the dependent variable is performed according to the effect of primary interest. Then, a two-way ANOVA can be performed (Salter and Fawcett, 1993).

The ART procedure is run through the "R" software employing the ARTool package (Wobbrock et al., 2011; Kay and Wobbrock, 2016) which is recommended for between x between factorial designs like the one of this study (Feys, 2016). Notability, being a non-parametric statistical method of analysis, the ART test does not require to meet any distributional assumption.

6.6. SAMPLING

Once the appropriate statistical techniques of analysis have been defined, the next step is the decision of the sampling technique to be employed and the sample size required (Bryman, 2012). Accordingly, the target population is defined in Section 6.6.1., then the advantages, limitations and source of errors of using a web panel sample are discussed (Section 6.6.2). Finally, the appropriate sample size is determined, considering the type of analysis carried out in this study (Section 6.6.3).

6.6.1. TARGET POPULATION

A clear identification of the target population of any research is important to reduce inaccurate response and minimise the *response error* bias (Bryman, 2012). Following Malothra et al. (2017, p. 414) the target population of this study (Table 6.14) was defined accordingly to four criteria:

- 1. The element from which information is sought.
- 2. The sampling unit or element that can be selected for the purpose of the research.
- 3. The geographical extent of interest to the research.
- 4. The timeframe within which the sampling units are relevant for the research.

Target Population of the Study		
Element	Citizens	
Sampling Unit	Citizens aged over 18, who use Facebook, Twitter or both.	
Geographical Extent	Living in the UK	
Timeframe	Jan-19	

 Table 6.14. Target Population of the Study

6.6.2. WEB PANEL SAMPLE ADVANTAGES, LIMITATIONS AND SOURCES OF ERROR

This study employed a web panel sample, that is as a bank of potential survey respondents who are recruited to participate in various online surveys (Rea and Parker, 2014). For the collection of the data, the web agency Pure Profile (www.pureprofile.com) was selected after an extensive evaluative process of due diligence, whereby costs and benefits of multiple options were considered. Pure Profile was chosen for its large panel sample including more than 700,000 UK respondents and its lower cost per complete response, compared with other UK-based data collection agencies at the time of this study. Other aspects were considered, like compliance with data protection legislation, panel representativeness of the target population, and quality assurance process of respondents' reliability.

The use of a web panel sample was deemed suitable, because the target population and most of the variables investigated in this study pertain to the online world. Moreover, web panel samples are easy to access, promptly available and reduce the resources needed for participants screening. The socio-demographic information provided by respondents before joining the panel were used to screen respondents who qualified to participate in this study (Facebook and Twitter users).

Yet these advantages come with some limitations. For instance, Rea and Parker (2014) and Williams (2012) highlight that:

1. <u>Participant self-selection negates pure random selection</u>. Facebook and Twitter accounts holders who do not feel comfortable answering online survey or who

are not aware of Pure Profile and have not been recruited by the agency will not be included in the panel sample. The absence of such individuals creates a systematic bias in the sample population or what is called *error of nonobservation*, which may increase *sampling error*. Hence, the sample is usually not representative of the target population. This may limit the generalisation of the findings to the entire population of Facebook and Twitter UK users. However, in a study comparing results from surveys adopting online web panels and random sampling techniques, Williams (2012) found that the correlations and relationships between survey estimates were similar, whether drawn from the online panel or a probability sample. Accordingly, online panels can be considered as a reliable option for generating statistics very close to outputs provided by random sampling techniques.

- 2. <u>Non-response error is difficult to control</u>. This problem arises when respondents either do not agree to participate in the survey or leave the questionnaire incomplete. As with all surveys (whether online or via traditional methods) it is difficult to rule out *non-response bias*, since we cannot be sure that those who respond are not in some unknown way different, to those who choose not to (Bryman, 2012). However, some steps were taken in this study to minimise this bias (Section 6.3.4).
- <u>Respondent identity is difficult to establish</u>. The researcher is not able to control who responds (i.e., the person at whom the questionnaire is targeted is the person who answers). This may affect the reliability and validity of the findings.

6.6.3. SAMPLE SIZE

Prior to gathering data, the appropriate sample size was calculated according to the type of analysis performed (Field, 2013). As mentioned in Section 6.5, this research employs PLS-SEM to answer research questions Q1, Q2 and Q3, and a series of two-way parametric and non-parametric ANOVAs to answer research questions Q4 and Q5. The following Sections discuss the appropriate sample size needed to generate adequate statistical power for each type of analysis.
6.6.3.1. SAMPLE SIZE IN PLS-SEM

To determine the appropriate sample size in PLS-SEM two approaches were followed:

- 1. Rule of thumb.
- 2. A priori statistical power analysis.

Regarding the <u>rule of thumb</u> in PLS-SEM, several authors (e.g., Bentler and Chou, 1987; Bollen, 1989; Nunnally, 1967) have tried to identify the approximate sample size required in PLS-SEM based on model complexity. The general advice is that models with more parameters, variables and indicators require larger samples (Hair et al., 2011). Accordingly, Nunnally (1967) and Barclay et al. (1995) suggest that the sample size should be equal to 10 times the largest number of structural paths directed at a particular construct in the structural model. The largest number of structural paths in the proposed model are directed at both On_PP and_Off_PP when testing for the moderation of "Int_SNS_Use" (Chapter 7, Section 7.2). They both have 11 structural paths directed at them (9 paths of the basic model plus 2 interaction terms. See Figure 6.10). Accordingly, the minimum sample size for the model should be 110 (10x11). This minimum sample size is considered adequate also for the MGA conducted through permutation tests employed to answer research questions Q1 (Chapter 7, Section 7.1) and Q3 (Chapter 7, Section 7.3) which does not require the creation of any interaction terms.



Figure 6.10. Structural Model of the Two-Stage Moderation Analysis of Int_SNS_Use.

Note: The green circles represent the interactions terms created between Bond_SC and Brid_SC with Int_SNS_Use as displayed in SmartPLS.

Further, a *priori statistical power analysis* was conducted through G*Power (Faul et al., 2007) which is a popular software employed in several scholarly articles for sample size computation (Hair et al., 2017). The level of significance set in the study, the effect size, the statistical power sought, and the numbers of parameters and observations included in the model were considered when computing the adequate sample size in PLS-SEM. While the α level of significance sets the threshold, which determines whether an effect exists, the effect size reveals the importance of that effect and the statistical power represents the probability that the effect is found (Field, 2013). Following best practice in PLS-SEM, a statistical power of 80% and a significance level of 0.05 were set (Cohen, 1992; Hair, 2017). With respect to the minimum effect size f^2 , Cohen (1988) suggests

that small effect sizes are 0.02, medium effect sizes are 0.15, and large effect sizes are 0.35. Since we are testing for moderation which is usually linked to small/medium effect sizes (Hair et al., 2017; 2018) we set $f^2 = 0.085$ as a value which is in-between small and medium effect. To compute sample size the option "Linear multiple regression: Fixed model, single regression coefficient" is selected (Faul et al., 2007) and the "a priori" type of power analysis is requested. Accordingly, G*Power calculated a target sample sizes of 95 respondents (Figure 6.11).

Yet, when conducting moderation analysis by means of permutation test (Section 6.5.1.4) the aim is to test differences in the path coefficients of different groups. Hence, the minimum sample size should be reached for each group. For this study we have three groups of respondents categorised according to the type of SNS used, namely Facebook, Twitter, and Facebook and Twitter combined. Then each of those groups is further divided according to the intensity of use in three subgroups, namely light, mild, and heavy users. Therefore, we have 9 groups as reported in Table 6.15. To categorise the variable "Intensity of SNSs Use" into three categories, we employed the "Quartile Split Method". Accordingly, all observations below the first quartile would be considered light users and all those above the third quartile would be considered heavy users. All those in between are considered mild users. Following the rule of thumb discussed above to have at least 110 respondents in both the light and heavy users, 440 respondents for each of the three types of SNSs users considered are needed. Hence, the final sample size should be 1320 (440x3) respondents. Following to G*Power's computation we would need a final sample size of 1140 respondents. Due to financial constraints and considering that PLS-SEM provides reliable results also with small sample sizes (Hair et al., 2017), we accepted G*Power's threshold.



Figure 6.11. Minimum Sample Size Computed in G*Power for PLS-SEM Moderation Analysis.

Table 6.15. Number of Groups for MGA analysis.

SNS_Used	Heavy Users	Mild Users	Light Users	Grand Total
Facebook and Twitter combined	x	х	x	3
Facebook	х	x	х	3
Twitter	x	х	х	3
Grand Total	3	3	3	9

6.6.3.2. SAMPLE SIZE FOR THE TWO-WAY ANOVA ANALYSIS

To compute sample size for the two-way ANOVA, this study follows the recommendations of Field (2013) and Faul et al. (2007) according to whom the level of significance set for the analysis, the effect size, the statistical power required, the degree of freedoms of the numerator and total amount of groups to include in the analysis are

the parameters to consider when computing sample size. Accordingly, the F-test family is selected from the G*Power main menu and the option "ANOVA: fixed effects, special, main effects and interactions" is chosen. The significance level is set at 5% and the desired level of power at 80%. The effect size is here set at 0.17 which according to Cohen (1988; 1992) mirrors a small to medium effect size. This is in line with previous research conducted in the field of Social Capital as related to SNSs studies (Boulianne, 2009; 2015; Skoric et al., 2016). The degree of freedom is set to be 2 (degree of freedom of the interaction term). Finally, we specified six as the total number of groups involved in the analysis (Facebook and Twitter users divided in light, mild and heavy). Hence, the sample size was computed (Figure 6.12). The power analysis indicates that at a 5% level of significance there is an 80% chance of finding an effect size as low as 0.17 if we recruit a total of 337 respondents. Considering the threshold set earlier for the PLS-SEM analysis of 1140 the minimum sample size required for the two-way ANOVA analysis is met (1140/3 = 380).



Figure 6.12. Sample Size Computation for the Two-Way ANOVA analysis.

6.6.3.3. SAMPLE SIZE FOR THE ART TEST

As discussed in Section 6.5.3, to answer research questions Q5 a non-parametric two-way ANOVA through the ART test is performed. The ART test performs well when sample sizes are small (lower than 50). The general rule of thumb is that the ART test can provide reliable results even with smaller samples than those required for the equivalent parametric test (Fagerland, 2012; Kelley and Sawilowsky, 1997; Sawilowsky, 1990). Hence, the sample size computed earlier for the two-way ANOVA is considered adequate.

6.7. PILOT STUDY: TESTING FOR QUESTIONNAIRE CLARITY. COMPREHENSIVENESS, ACCEPTABILITY AND FOR PSYCHOMETRIC PROPERTIES OF MEASURES

Before launching any online survey, Bryman (2012) and Rea and Parker (2014) suggest conducting a pilot study to:

- 1. Ensure that the questionnaire is free from any mistakes which can compromise its clarity, comprehensiveness and acceptability.
- 2. Confirm the psychometric properties of the measures employed in this thesis and conduct a preliminary assessment of the structural models.

The pilot study of this research was launched in December 2018. A total of 314 respondents aged over 18 and belonging to the panel sample of the data collection agency Pure Profile took part in the pilot study. The respondents were randomly selected from the population of individuals using Facebook and/or Twitter. The validity of completes was checked against time employed to complete the survey (see Section 6.3.4), response set biases and suspicious response patterns. Accordingly, 9 respondents were eliminated and a total of 305 respondents were retained, divided according to the different groups of users as shown in Table 6.16.

Sample Size Pilot Study				
Groups	Ν			
Facebook Users (F_U)	102			
Twitter Users (T_U)	103			
Facebook and Twitter Combined (F+T_U)	100			
Total	305			

 Table 6.16. Sample Size Pilot Study.

The size of the sample for the pilot study was set following best practice guidelines with respect to: (1) the type of instrument of data collection employed and (2) the type of analysis conducted in this study. A sample size in the range of twenty to forty respondents was considered adequate to pre-test a questionnaire for clarity and acceptability (Rea and Parker, 2014; Cocks and Torgerson, 2012). With respect to the type of analysis employed, to compute the adequate sample size we followed the recommendations discussed in Section 6.6.3. Accordingly, a total sample size of 300 respondents was deemed appropriate (100 per each group of Facebook, Twitter and Facebook and Twitter Combined). Notably, for the preliminary analysis of the moderating effect of "Int_SNS_Use", only the two-stage interaction approach was employed, and the groups were not split into light, mild and heavy users. This was done only successively in the main statistical analysis to get a deeper insight on our findings (see Chapter 7). Splitting the users groups according to their level of intensity of use of the SNSs would have required a much larger sample size, which for the purpose of the pilot was not considered necessary.

Ensure questions clarity, comprehensiveness and acceptability.

Regarding *question clarity*, the pilot study highlighted no significant problems. Indeed, all questions were answered by all respondents. However, upon completion of the questionnaire, respondents were asked to provide qualitative feedback on the questionnaire, reporting any difficult or ambiguous questions. Although respondents did not report any issues, we identified some problems with respect to comprehensiveness and reverse coding.

With respect to comprehensiveness, researchers must ensure that, for each question, the response choices cover a reasonably complete and mutually exclusive range of alternatives. In the pilot study, when asking respondents how much time they spent on the SNSs used, the fixed choices options of Ellison et al. (2007) were provided (Figure 6.13).

Figure 6.13. Fixed Choice Response Options for Time Respondents Spent on SNS in Ellison et al. (2007).

In the past week, on average, approximately how many minutes per day have you spent on Facebook? 0 = less than 10, 1 = 10-30, 2 = 31-60, 3 = 1-2 hours, 4 = 2-3 hours, 5 = more than 3 hours

Source: Ellison et al. (2007, p. 1150).

The pilot study revealed that some respondents reporting to spend 1 hour on the selected SNS placed themselves in the 31-60 minutes category meanwhile others selected in the 1-2 hours category. The same happened for those who reported to spend 2 hours. This problem was spotted thanks to a follow-up open question asking respondents to specify hours and minutes spent on the selected SNS, on a typical day. This question was asked to facilitate the moderation analysis in PLS-SEM using "time spent on SNSs" as a ratio continuous moderator rather than as a categorical one. Indeed, the use of continuous moderators is encouraged because allows the researcher to carry out the analysis just once on the entire sample. Meanwhile, using categorical moderators would force the researcher to use group comparison techniques, which require more time and effort in terms of analysis. Also, by dividing the sample in different groups the variable variance is lost. Accordingly, the options provided were changed, resulting in mutually exclusive categories (Figure 6.14).

Figure 6.14. Fixed Choice Option for Time Spent on SNSs for This Study



These amendments to the response categories are in line also with changes reported in other studies (i.e., AlAteeq et al., 2016; Fullwood et al., 2016; Lou et al., 2012) which adopted the scale of Ellison et al. (2007).

Another problem became evident with the reverse coded questions of the Bonding Scale of Williams (2006) employed to operationalise online Bonding Social Capital (Figure 6.15).

Figure 6.15. Online Bonding Scale

Bonding Subscale

- 1. There are several people online/offline I trust to help solve my problems.*
- There is someone online/offline I can turn to for advice about making very important decisions.*
- There is no one online/offline that I feel comfortable talking to about intimate personal problems. (reversed)*
- 4. When I feel lonely, there are several people online/offline I can talk to.
- 5. If I needed an emergency loan of \$500, I know someone online/offline I can turn to.*
- 6. The people I interact with online/offline would put their reputation on the line for me.
- 7. The people I interact with online/offline would be good job references for me.
- 8. The people I interact with online/offline would share their last dollar with me.
- I do not know people online/offline well enough to get them to do anything important. (reversed)
- 10. The people I interact with online/offline would help me fight an injustice.

Source: Williams (2006, p.602).

Question 3 and 9 are reversed coded questions and when analysing the pilot data, it was evident that some of the respondents (10%) who scored high values on all the other questions had low values in those questions (once they were reversed coded). This could be due to the presence of negatives like "no" and "not" which are sometimes easy for respondents to miss in a self-completion questionnaire (Bryman, 2012). On the other hand, these respondents answered correctly all the attention checks questions and the time they spent completing the survey was close to the average. Hence, after consulting with experts within the Faculty and following the suggestions of Bryman (2012) and Rea and Parker (2014) the words "not" and "no" were dropped, to avoid any misinterpretation and minimise any problems related to the reliability and validity of the study. Hence, questions 3 and 9 were rephrased using a positive formulation as follow:

- Question 3 → When I feel lonely, there are several people on Facebook I can talk to.
- Question 9 → I know people on Facebook well enough to get them to do something important.

With respect to questionnaire *acceptability*, Rea and Parker (2014) claims that it can decrease when: (1) the questionnaire is too long, or (2) it contains questions that are perceived as too invasive and violating ethical and moral standards, or (3) when

respondents are concerned with anonymity and protection of the data provided. Considering that all the respondents completed the survey, it could be safe to assume that they were not particularly concerned about those aspects and that the instructions provided in the information sheet concerning the above issues were clear and exhaustive.

Confirm the psychometric properties of the measures employed and conduct a preliminary assessment of the structural models.

Another important aim of a pilot study is to test the reliability and validity of the questionnaire items (Bryman, 2012). Such assessment followed established guidelines in conducting PLS-SEM analysis for reflective measurement models employing the metrics discussed earlier. The main output of the analysis in reported in Appendix J. Here, it is worthy to mention that for all the groups in Table 6.16, the constructs demonstrated adequate internal consistency reliability, convergent and discriminant validity. Also, we carried out CTA-PLSs (see Section 6.5.1.1) which empirically ascertained that the constructs were correctly defined as Reflective. Further, FCAs (see Section 6.3.5) confirmed that the study was free from CMB. All models showed between moderate and substantial predictive accuracy and adequate predictive relevance.

6.8. SUMMARISING

This study adopted a quantitative research strategy to address the research questions, employing a cross-sectional design where data was collected through an online survey hosted on Qualtrics and administered by the data collection agency Pure Profile to its web panel. The minimum sample size (n = 1140) required to achieve sufficient statistical power was determined considering rule of thumbs for the type of statistical analysis employed and performing a priori statistical power analysis. PLS-SEM analysis, two-way ANOVA and the ART test were chosen to answer the research questions. Assumptions, metrics and algorithm setting for the validation process of the chosen statistical approach were discussed. The questionnaire was piloted and designed to minimise CMB, social desirability and acquiescence biases. The measures employed in this research have all been adopted in previous academic peer-reviewed articles. This was meant to increase the reliability of the research, facilitate study replicability and establish face and measurement validity. In the next chapter, the results of the analysis carried out to answer the proposed research questions are provided.

7. DATA ANALYSIS

This chapter provides the results of the data analysis carried out to investigate the research questions presented in Chapter 5. Different techniques of analysis are employed for each research question as discussed in Chapter 6, Section 6.5 and summarised in Table 6.9. To recapitulate, research questions Q1, Q2 and Q3 will be investigated using PLS-SEM through the SmartPLS software 3.2.8 version following a multistage procedure (Figure 7.1), as suggested by Hair et al. (2017). Questions 1 to 3 aim to test the significance of different moderating effects on the relationships between Bond_SC and Brid_SC with Off_PP and On_PP. In particular, to answer research question Q1 (Section 7.1) a Multi-Group Analysis (MGA) by mean of a permutation test is employed because the moderator variable "Type of SNSs Used" (Type_SNSs_Used) is operationalised as a categorical grouping variable [Group 1 = Facebook Users only (\mathbf{F}_{U}) and Group 2 = Twitter Users only (\mathbf{T}_{U})]. Research question Q2 (Section 7.2) is investigated using the two-stages approach for the moderation analysis of "Intensity of SNSs Use" (Int SNS Use) which is a ratio measure operationalised through the length of time spent on the SNSs gauged in minutes. Research question Q3 (Section 7.3) involves testing for the significance of the moderating effect of the combined and not Combined use of Facebook and Twitter (Comb_Use_SNSs) which is a categorical variable. Hence, a MGA by mean of permutation test is employed. Questions Q4 (Section 7.4) and Q5 (Section 7.5) are investigated through a series of parametric two-way ANOVAs using SPSS 24 and non-parametric two ways ANOVAs employing the "R" software version 3.6.0. Validation, interpretation and discussion of the results are based on the metrics and procedures discussed in the previous Chapter.

For convenience and ease of consultation, the criteria discussed in Section 6.5.1.1 and Section 6.5.1.2 to assess respectively the measurement and the structural models to answer research questions Q1, Q2 and Q3 have been schematically summarised in tables and reported in Appendix K together with the algorithm settings employed in SmartPLS software to get all those assessment metrics.



Figure 7.1. A Systematic Procedure for Applying PLS-SEM.

Source: Adapted from Hair et al., 2017, p. 30.

7.1. INVESTIGATING THE MODERATING EFFECT OF TYPE OF SNS USED

This Section and its sub-sections present the results obtained through the empirical analysis of the moderating effect of the Type of SNSs used (**Type_SNSs_Used**) on the relationships between Bond_SC and Brid_SC with On_PP and Off_PP. This section and its sub-sections are arranged in a sequential order that follows the systematic procedure to apply PLS-SEM analysis reported in Figure 7.1.

The structural model to answer research question Q1 (Figure 7.2) stems from the theoretical frameworks of the Resource Mobilisation Theory, within the context of Social Capital and Political Participation as discussed and presented in Chapter 5.

Following the scheme proposed by Hair et al. (2017) in Figure 7.1, the second stage of the procedure for applying PLS-SEM analysis consists of determining the Measurement (or outer) model which illustrates the relationships between the latent constructs and their corresponding indicator variables. As discussed in Section 6.5 this study adopts reflective measurements for all the latent constructs (Figure 7.3).



Figure 7.2. Structural Model Research Question Q1.

Figure 7.3. Proposed Measurement Model for Research Question Q1.



Note: the moderator variable "Type_SNS_Used" is not represented in the measurement model as it is employed as grouping variable for the Permutation test.

The next few Sections (and relating sub-sections) are organised as follows. Section 7.1.1 and its sub-sections deal with data screening. Section 7.1.2. discusses the technique of analysis, the options and the parameter settings required to run the PLS-SEM algorithm. Section 7.1.3 and its sub-sections assess the internal consistency, reliability, convergent validity and the discriminant validity of the measurement model. Section 7.1.4 assesses the structural model predictive capability and accuracy and provides the results of the MGA by mean of permutation test employed to ascertain the moderating effect of "Type_SNS_Used". Also, CMB empirical test and post-hoc Power analysis are run.

7.1.1. DATACOLLECTION AND SCREENING

The third stage of the systematic procedure for applying PLS-SEM consist of data collection and screening (Hair et al., 2017). Details about instrument and methods of data collection were provided in Chapter 6. Hence, this Section and its sub-sections outline the procedures for screening the data and for dealing with any missing values (Section 7.1.1.1), suspicious response patterns (Section 7.1.1.2) descriptive statistics and outliers (Section 7.1.1.3) (Tabachnick and Fidell 2007, Hair, et al., 2010).

7.1.1.1. MISSING DATA

The collected data, especially when using survey instruments, may present some missing values due to errors in data-entering, data-collection or simply due to the purposely or inadvertently omission made by the respondents (Hair et al., 2010). This might affect the validity and the reliability of the study depending on the amount of missing data values and their patterns.

This study presents no missing data because it took a forced answer approach to the questions. Although this approach may spur some people to leave the survey, research suggest that it will help the respondents to complete the questionnaire "*because more often than not the reason for skipping questions is inadvertence*" (Hair et al., 2017: p. 57). However, a forced-answer approach may produce inaccurate data because people may lie to go ahead with the survey and skip sensitive questions. Nonetheless, this study employs a panel from the data collection agency Pure Profile and a forced answer approach should not represent a problem. Indeed, potential panel recruits must provide sensitive information like their demographics (gender, income, age), their political beliefs or their use of the internet when enlisting. Therefore, most of the information required for this survey had already been made available by the respondents willingly. Moreover, following Rea and Parker (2014) the questionnaire included some reliability checks to minimise any problems with the forced-answer approach (see Chapter 6, Section 6.3.3.1).

7.1.1.2. SUSPICIOUS RESPONSE PATTERNS

A further step involved in data screening is the analysis of suspicious response patterns in the data collected. Firstly, the data was screened to eliminate all respondents who failed to answer the reliability check questions like "Please, select strongly agree for this statement (this is a system check)". Next, all respondents who showed inconsistent response patterns were highlighted. For instance, in the Political Interest scale (Table 6.6, Chapter 6) those who answered that they were strongly interested in local community affairs (item 1) and also in national politics (item 2) but that in general they were not interested in politics (item 3) were considered not reliable and considered for removal. The literature review highlighted that high levels of Political Participation are positively correlated with political interest (Krueger, 2006; Melo and Stockemer, 2014; Pattie et al., 2004; Skoric et al. 2009; Verba et al., 1995; Wolfinger and Rosenstone, 1980). Therefore, all the respondents displaying high levels of participation in politics but very low interest in politics raised a red flag and were considered for removal. Overall, only 6 respondents exhibited such inconsistent patterns and were discarded.

Following Hair et al. (2017) and Field (2013), the dataset was then visually examined in excel to identify straight lining, diagonal lining and alternating extreme pole responses. Although several respondents (~10% of the sample) showed some straight lining in some scales of the questionnaire, no one was dropped from the study. This is because their response patterns were consistent and because they passed all other reliability checks meaning that their answers were likely to be genuine. For instance, respondents who reported low levels of Offline Political Participation answering all zeros or one to many of the items of the scale but showed also low levels of political interest, political knowledge and political self-efficacy were considered reliable. So, although

some respondents showed straight lining, as long as their answers were consistent with previous research findings and they passed all reliability check questions they were retained in the study. Hence, the final number of respondents was 1212.

7.1.1.3. DESCRIPTIVE STATISTICS, DATA DISTRIBUTION AND OUTLIERS

The demographics of the entire sample (E_S), Facebook users only (F_U), Twitter users only (T_U), Facebook and Twitter combined users (F+T_U) are showed in Table 7.1, Table 7.2, Table 7.3 and Table 7.4.

SNS_Used	Count	% of Total Count
F_U	432	35
T_U	383	32
F+T_U	397	33
Total (E_S)	1212	100

 Table 7.1. Sample Count.

					% of Total
SNS_Used	Gender	Education_label	Avg. Age	Count	Count
		High education	38	268	22
	Female	Low education	42	380	31
E_S		Total	40	648	53
		High education	44	271	23
	Male	Low education	44	293	24
		Total	44	564	47
	Total	Whole Sample	42	1,212	100
		High education	41	74	17
F_U	Female	Low education	46	169	39
		Total	44	243	56
	Male	High education	49	87	20
		Low education	51	102	24
		Total	50	189	44
	Total	Whole Sample	47	432	100
	Female	High education	38	99	26
		Low education	38	99	26
		Total	38	198	52
T_U		Low education	38	91	24
	Male	High education	39	94	24
		Total	39	185	48
	Total	Whole Sample	38	383	100
		High education	35	95	24
	Female	Low education	41	112	28
		Total	38	207	52
F+T_U		High education	43	90	23
	Male	Low education	43	100	25
		Total	43	190	48
	Total	Whole Sample	40	397	100

Table 7.2. Demographic Characteristics of E_S, F_U, T_U and F+T_U Samples.

In order to establish the representativeness of Pure Profile panel, the composition of the sample was checked against the characteristics of the general UK population. Overall, the percentages of females (53%) and males (47%) in our sample are very close to the proportions of the general UK population (51% of females and 49% of males) (Statista, 2021b). Such proportion is fairly constant across all the SNSs groups analysed in this study and is line with the general trend of Facebook users in UK where females constitute a higher percentage of Facebook users (53%) compared to males (47%) (Statista, 2021a). These proportions are very close to those of our sample (56% Females and 44% males Facebook users). However, our Twitter sample does not mirror the gender proportion of the UK population where males represent the 56% of users (Ipsos, 2018) against the 44% of our sample. Also, the average age of the sample (42 years) is similar to the age reported by the ONS (Office for National Statistics) in 2019 which found an average age of 40.3 years for the UK population (ONS, 2019). Notably, in our sample Facebook users are the oldest (M = 47) and Twitter users the youngest (M = 38). This seems to mirror once again the trend of the UK population which sees a higher percentage of elderly people using Facebook rather than Twitter (Ipsos, 2018). However, the level of education of respondents of our sample with 45% of high educated people (university degree) and 55% of low educated people does not mirror the percentages provided by the ONS (2013)² which is closer to 38% of people who actually obtained at least a bachelor's degree. Hence, any generalization of findings should be made with caution considering the high and positive influence of education levels on Political Participation (see Chapter 5).

Also, demographics are provided for the samples obtained from the quartile split procedure. Indeed, Facebook users, Twitter users and Facebook and Twitter users combined were split into three subgroups to allow comparisons for the moderation effect of the variable "Type_SNS_Used" and "Comb_Use_SNSs" when controlling for "Int_SNS_Use". The quartiles split method on the "Int_SNS_Use" variable allowed for a trichotomous categorisation of users in Heavy (all those users above the 3rd quartile) Mild (all those users between the 1st and the 3rd quartile) and Light (all those users below the 1st quartile). As a rule of thumb, the quartiles values should not be included into any of the subgroups created (Field, 2013; Henseler and Fassott, 2010). Accordingly, the number of respondents dropped to 369 for both F_U and T_U and to 164 for F+T_U obtaining a sample of 902 respondents split into nine groups (Table 7.3).

Sample	Heavy Users	Mild Users	Light Users	Grand Total
F_U	108	162	99	369
T_U	88	186	95	369
F+T_U	41	78	45	164

Table 7.3. Sample Count of Heavy, Mild and Light users of F_U, T_U and F+T_U.

 $^{^{2}}$ This is the most up-to-date reliable source we could find for education levels in the UK for the general population.

SNS_Used	Heavy Users %	Mild Users %	Light Users %	Grand Total %
F_U	29	44	27	100
T_U	24	50	26	100
F+T_U	25	48	27	100

Table 7.4. Sample Percentage of Heavy, Mild and Light users of F_U, T_U and F+T_U.

Demographics on age, gender and level of education of F_U, T_U and F+T_U, divided in heavy, mild and light users are provided in Appendix L.

Although to investigate research question Q1, Q2 and Q3 this study employs PLS-SEM which is a non-parametric statistical method of analysis, Hair et al. (2017) suggest identifying any outliers. This is because some outliers could be due to typing errors, data collection errors or are simply observations that are not part of the intended population (Hair et al., 2017).

"An outlier is an extreme response to a particular question, or extreme responses to many or all questions" (Hair et al., 2017: p. 59). In the first case outliers are called univariate, meanwhile in the latter case they are known as multivariate outliers (Tabachnick and Fidell, 2007). Following Field (2013), Hair et al., (2017) and Tabachnick and Fidell, (2007) this study employed graphical techniques such as boxplots to identify univariate outliers and Mahalanobis distance to identify multivariate outliers. Also, descriptive statistics like mean (*M*), median (*Mdn*), mode (*Mode*), minimum (*Min*), maximum (*Max*), standard deviation (*SD*), Skewness (*Skew*) and Kurtosis (*Kurt*) are provided for each item of the constructs of the structural model and for the composite scales to facilitate the identification of the variables that may contain any outliers (Clason and Dormody, 1994; Boone and Boone, 2012; Brown, 2011; Sullivan and Artino, 2013). Although a thorough examination of all the groups involved in the analysis (Table 7.1 and Table 7.3) was conducted, descriptive statistics are provided in Appendix L only for the groups reported in Table 7.1 for reasons of convenience and space. Those statistics are obtained using SPSS 24.0.

To determine the normality of the data the values of kurtosis and skewness were used. The acceptable range of skewness and kurtosis are respectively ± 1 and ± 2 (George

and Mallery, 2020). Reasons for inspecting the normality of data distribution through those values rather than through significance tests of normality were discussed in Chapter, 6, Section 6.5.2. Accordingly, many of the indicators were found to be non-normally distributed. Similarly, Those results were further validated by the visual inspection of the boxplots obtained in SPSS.

We further investigated outliers with respect to their multivariate nature. Multivariate outliers were identified based on the Mahalanobis distance using IBM SPSS 24. The Mahalanobis distance "*is the distance of a case from the centroid of the remaining cases where the centroid is the point created at the intersection of the means of all the variables*" (Tabachnick and Fidell, 2007, p. 74). These distances follow a chi-square distribution. To establish whether an observation is cause for concern the critical value of chi-square for the desired alpha level (i.e., 0.001) and degrees of freedom equal to the number of predictors must be found. Hair et al. (2009) and proposed that Mahalanobis distance should not surpass the critical chi-squared value with degrees of freedom equal to the number of predictors and alpha = 0.001. In this study, the Mahalanobis distance was calculated based on 42 items. The criterion of p < 0.001 and critical value of $\chi 2 = 76.084$ were adopted. Accordingly, several multivariate outliers were identified.

However, all the respondents left in the sample at this stage could be considered reliable as they had been already screened for the reliability check questions and for suspicious response patterns. Moreover, further inspection revealed that those extreme observations were not due to any measurement, typing or entry errors (Beckman and Cook, 1983; Hair et al., 2017). Also, following Hair et al. (2017) we checked for any patterns and communality in the outliers' origin related to gender age, etc. which can lead to the formation of subgroups of the sample and explain the outliers' origin. Yet, no common patterns or trends were spotted. For all the reasons reported above the outliers identified in this study were retained as suggested by Hair et al. (2017). Moreover, this research employs the PLS-SEM technique which is a non-parametric analysis based on bootstrapping procedure which can adequately handle the non-normality of the data distributions (Hair, Sarstedt et al. 2014).

7.1.2. ASSESSING PLS-SEM RESULTS OF THE REFLECTIVE MEASUREMENT MODEL FOR Q1

The fourth stage of the systematic procedure to apply PLS-SEM analysis (Figure 7.1) concerns the specification of the appropriate algorithm settings required to estimate the measurement and the structural model. Those were provided at the beginning of this Chapter.

Once the PLS algorithm runs and produces path models estimates, the next step in the systematic procedure proposed by Hair et al. (2017) to apply PLS-SEM consists of assessing the quality of the results of the measurement model (Figure 7.1). The assessment of the measurement model is conducted for all the groups that must be analysed to answer research question Q1. Following best practices in PLS-SEM, as we needed to test the moderating effect of "Type_SNS_Used" through Permutation test the full sample was split into two groups (F_U, n = 432; T_U, n = 383) and the related measurement model were assessed without and with control variables (Figure 7.4 and Figure 7.5 respectively). The aim was to check if the model with the control variables had higher predictive accuracy and relevance than the model with no control variables and determine which one to employ in the Permutation analysis. This process allowed also to establish configural invariance (See Chapter 6, Section 6.5.1.3), while accounting for the principle of parsimony (Field, 2013). Accordingly, if the model with control variables shows no significant improvements in terms of predictive accuracy and relevance, then the model with fewer variables will be further employed in the analysis.





Figure 7.5. Proposed Measurement Model with Control Variables.



Table 7.5 and Table 7.6 show the metrics for the measurement model assessment of F_U with no control variables, while Table 7.7 and Table 7.8 show the metrics related to the T_U sample with no control variables. Notably, for both models all the reliability coefficients like CA and CR are above the threshold of 0.70 indicating that the measurements are reliable. Also, both models show sufficient discriminant validity. Indeed, all the values of the square roots of the AVE (Fornell-Larcker Criterion) for each item are greater than the inter-construct correlations with other items (Chin 1998). Moreover, each indicator's loading with its construct is higher than its cross-loadings with other constructs (Hair et al., 2017). Also, the values of the HTMT are below 0.90 and the related confidence intervals do not include 1. A full account of the Fornell-Larcker Criterion, the cross-loadings and the HTMT analysis is provided in Appendix M. With respect to the assessment of convergent validity, all the AVEs of the Twitter users' model were above 0.50 indicating that the latent constructs account for at least 50 percent of the variance in the items (Hair et al., 2017). Moreover, all the outer loadings were in an acceptable range (> 0.05) (Chin, 1998; Hair et al., 2010). Also, the communality values were all above the threshold of 0.50 but those of the indicators Off_PP_2 (0.312) and Off_PP_7 (0.417). However, although the communalities of the two indicators were low, they did not affect the AVE of Offline Political Participation (0.588) and therefore the two indicators were retained (Hair et al., 2017). Rather, the measurement model of F_U showed a problem with the AVE of Off_PP which had a value of 0.488.

LV Indicators		Convergent Validity		Internal Reliability		
LV	Indicators	Loadings	Communality	AVE	CR	CA
sc	Bond_1	0.817	0.667			
	Bond 2	0.781	0.610			
	Bond_3	0.787	0.619			
	Bond_4	0.749	0.561]		
^d S	Bond_5	0.824	0.679	0.654	0.050	0.041
son	Bond_6	0.740	0.548	0.054	0.950	0.541
щ	Bond_7	0.804	0.646			
	Bond_8	0.843	0.711			
	Bond_9	0.837	0.701			
	Bond 10	0.892	0.796			
	Brid_1	0.865	0.748			
	Brid_2	0.854	0.729		0.959	0.953
	Brid_3	0.863	0.745			
υ	Brid_4	0.844	0.712			
S	Brid_5	0.866	0.750	0.703		
Brid	Brid_6	0.879	0.773	0.705		0.955
	Brid_7	0.807	0.651			
	Brid 8	0.791	0.626			
	Brid_9	0.846	0.716			
	Brid_10	0.766	0.587			
	Off_PP_1	0.749	0.561			
	Off_PP_2	0.627	0.393			
	Off_PP_3	0.598	0.358			
문	Off_PP_4	0.657	0.432			
Ē	Off_PP_5	0.700	0.490	0.488	0.895	0.872
0	Off PP 6	0.651	0.424			
	Off_PP_7	0.670	0.449			
	Off_PP_8	0.813	0.661			
	Off_PP_9	0.791	0.626			
	On PP 1	0.748	0.560			
	On PP 2	0.648	0.420			
E.	On_PP_3	0.703	0.494	0.500	0.961	0 000
8	On_PP_4	0.765	0.585	0.508	0.801	0.000
	On_PP_5	0.725	0.526			
	On_PP_6	0.682	0.465			

Table 7.5. Summary Convergent Validity and Internal Consistency ReliabilityAssessment for Facebook Users Sample (n = 432).

Table 7.6. Discriminant Validity Summary Measurement Model F_U with no

Discriminant Validity Summary						
Constructs	Fornell-Larcker	Cross-Loadings	HTMT Ratio	HTMT CI		
Bond_SC Brid_SC Off_PP On_PP	Condition met	Condition met	< 0.90	∌1		

Control Variables (n = 432).

Table 7.7. Summary Convergent Validity and Internal Consistency Reliability

Assessment for T_U (n= 383).

	T	Convergent Validity		Internal Reliabilit		
LV	Indicators	Loadings	Communality	AVE	CR	CA
	Bond_1	0.898	0.806			
	Bond_2	0.890	0.792			
Bond_SC	Bond_3	0.837	0.701			
	Bond_4	0.826	0.682			
	Bond_5	0.874	0.764	0.720	0.064	0.058
	Bond_6	0.811	0.658	0.723	0.904	0.950
	Bond_7	0.831	0.691			
	Bond_8	0.817	0.667			
	Bond_9	0.812	0.659			
	Bond_10	0.935	0.874			
	Brid_1	0.848	0.719			0.957
	Brid_2	0.847	0.717		0.963	
	Brid_3	0.821	0.674			
U	Brid_4	0.866	0.750			
S.	Brid_5	0.831	0.691	0.72		
Brid	Brid_6	0.850	0.723	0.72		0.957
"	Brid_7	0.852	0.726			
	Brid_8	0.813	0.661			
	Brid_9	0.897	0.805			
	Brid_10	0.857	0.734			
	Off_PP_1	0.836	0.699			
	Off_PP_2	0.559	0.312			
	Off_PP_3	0.815	0.664			
£	Off_PP_4	0.765	0.585			
E	Off_PP_5	0.824	0.679	0.588	0.927	0.912
Ö	Off_PP_6	0.763	0.582			
	Off_PP_7	0.646	0.417			
	Off_PP_8	0.832	0.692			
	Off_PP_9	0.816	0.666			
	On_PP_1	0.829	0.687			
	On_PP_2	0.763	0.582			
	On_PP_3	0.828	0.686	0.600	0.029	0.006
9 9	On_PP_4	0.896	0.803	0.082	0.928	0.900
	On_PP_5	0.859	0.738			
	On_PP_6	0.771	0.594			

Discriminant Validity Summary					
Constructs	Fornell-Larcker	Cross-Loadings	HTMT Ratio	HTMT CI	
Bond_SC Brid_SC Off_PP On_PP	Condition met	Condition met	< 0.90	∌ 1	

Table 7.8. Discriminant Validity Summary Measurement Model T_U with noControl Variables (n = 383).

Following Hair et al. (2017) to improve the fitness of F_U measurement model we considered removing the indicators with low loadings. We started with the item that had the lowest outer loading (Hair et al., 2010) that is Off_PP_3. In fact, in reflective measurement models removing an item should not be problematic as by definition, indicators are interchangeable. However, following, Hair et al. (2018) and Latan (2018) suggestions, before removing the indicator Off_PP_3 a CTA-PLS was performed to empirically substantiate the correct specification of Off_PP as reflective for both F_U (Table 7.9) and T_U (Table 7.10). A CTA-PLS was conducted also for OFF_PP of T_U because to carry out multi-group comparison the configural invariance of the constructs across groups must be established (see Chapter 6, Section 6.5.1.3). Accordingly, all models must be measured exactly through the same indicators.

Following the CTA-PLS interpretation guidelines provide on Section 6.5.1.1 all the 90% bias-corrected and Bonferroni-adjusted Confidence Intervals include zero, meaning that the tetrads vanish and that the Off_PP's measurement model is indeed reflective for Both F_U and T_U samples. Hence, the Off_PP_3 indicator was dropped from the analysis and the modified measurement model of F_U was re-estimated to check if the removal had a positive impact on the AVE of Off_PP while retaining sufficient internal reliability (Table 7.11) and discriminant reliability.

CTA-Off_PP	CI Low	CI Up	Tetrad
Non-Redundant Tetrads	adj.	adj.	Vanish
1: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_4	-2.698	1.645	Yes
2: Off_PP_1,Off_PP_2,Off_PP_4,Off_PP_3	-2.349	0.779	Yes
4: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_5	-1.485	1.044	Yes
6: Off_PP_1,Off_PP_3,Off_PP_5,Off_PP_2	-3.044	1.306	Yes
9: Off_PP_1,Off_PP_3,Off_PP_6,Off_PP_2	-2.744	2.508	Yes
10: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_7	-3.989	1.047	Yes
13: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_8	-3.320	1.540	Yes
17: Off_PP_1,Off_PP_2,Off_PP_9,Off_PP_3	-3.368	2.194	Yes
20: Off_PP_1,Off_PP_2,Off_PP_5,Off_PP_4	-1.271	0.978	Yes
26: Off_PP_1,Off_PP_2,Off_PP_7,Off_PP_4	-0.823	1.190	Yes
29: Off_PP_1,Off_PP_2,Off_PP_8,Off_PP_4	-0.691	2.355	Yes
33: Off_PP_1,Off_PP_4,Off_PP_9,Off_PP_2	-1.340	1.925	Yes
41: Off_PP_1,Off_PP_2,Off_PP_8,Off_PP_5	-0.967	1.944	Yes
47: Off_PP_1,Off_PP_2,Off_PP_7,Off_PP_6	-1.728	1.033	Yes
49: Off_PP_1,Off_PP_2,Off_PP_6,Off_PP_8	-1.425	2.059	Yes
51: Off_PP_1,Off_PP_6,Off_PP_8,Off_PP_2	-1.902	2.004	Yes
57: Off_PP_1,Off_PP_7,Off_PP_8,Off_PP_2	-2.523	1.201	Yes
109: Off_PP_1,Off_PP_4,Off_PP_5,Off_PP_6	-0.429	1.495	Yes
113: Off_PP_1,Off_PP_4,Off_PP_7,Off_PP_5	-0.469	0.741	Yes
133: Off_PP_1,Off_PP_4,Off_PP_7,Off_PP_9	-1.125	1.219	Yes
137: Off_PP_1,Off_PP_4,Off_PP_9,Off_PP_8	-1.270	2.134	Yes
149: Off_PP_1,Off_PP_5,Off_PP_8,Off_PP_7	-1.065	0.824	Yes
151: Off_PP_1,Off_PP_5,Off_PP_7,Off_PP_9	-0.627	1.078	Yes
161: Off_PP_1,Off_PP_6,Off_PP_9,Off_PP_7	-0.630	1.536	Yes
165: Off_PP_1,Off_PP_8,Off_PP_9,Off_PP_6	-1.417	1.917	Yes
174: Off_PP_2,Off_PP_4,Off_PP_6,Off_PP_3	-1.158	1.518	Yes
231: Off_PP_2,Off_PP_6,Off_PP_8,Off_PP_4	-0.833	1.927	Yes

 Table 7.9. Bias-Corrected and Bonferroni-Adjusted Confidence Intervals of the

CTA-PLS analysis for Off_PP of F_U (n = 432).

CTA-Off_PP	CI Low	CI Up	Tetrad
Non-Redundant Tetrads	adj.	adj.	Vanish
1: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_4	-4.803	2.286	Yes
2: Off_PP_1,Off_PP_2,Off_PP_4,Off_PP_3	-4.535	1.501	Yes
4: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_5	-2.494	3.055	Yes
6: Off_PP_1,Off_PP_3,Off_PP_5,Off_PP_2	-8.269	1.151	Yes
9: Off_PP_1,Off_PP_3,Off_PP_6,Off_PP_2	-9.587	0.893	Yes
10: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_7	-7.378	0.616	Yes
13: Off_PP_1,Off_PP_2,Off_PP_3,Off_PP_8	-5.647	3.350	Yes
17: Off_PP_1,Off_PP_2,Off_PP_9,Off_PP_3	-3.408	2.511	Yes
20: Off_PP_1,Off_PP_2,Off_PP_5,Off_PP_4	-3.000	5.410	Yes
26: Off_PP_1,Off_PP_2,Off_PP_7,Off_PP_4	-1.173	2.544	Yes
29: Off_PP_1,Off_PP_2,Off_PP_8,Off_PP_4	-2.358	3.916	Yes
33: Off_PP_1,Off_PP_4,Off_PP_9,Off_PP_2	-1.169	6.173	Yes
41: Off_PP_1,Off_PP_2,Off_PP_8,Off_PP_5	-0.953	4.254	Yes
47: Off_PP_1,Off_PP_2,Off_PP_7,Off_PP_6	-0.895	3.253	Yes
49: Off_PP_1,Off_PP_2,Off_PP_6,Off_PP_8	-4.374	5.326	Yes
51: Off_PP_1,Off_PP_6,Off_PP_8,Off_PP_2	-0.493	7.671	Yes
57: Off_PP_1,Off_PP_7,Off_PP_8,Off_PP_2	-6.811	0.973	Yes
109: Off_PP_1,Off_PP_4,Off_PP_5,Off_PP_6	-4.441	0.939	Yes
113: Off_PP_1,Off_PP_4,Off_PP_7,Off_PP_5	-3.288	0.281	Yes
133: Off_PP_1,Off_PP_4,Off_PP_7,Off_PP_9	-2.453	1.231	Yes
137: Off_PP_1,Off_PP_4,Off_PP_9,Off_PP_8	-2.276	4.676	Yes
149: Off_PP_1,Off_PP_5,Off_PP_8,Off_PP_7	-1.471	2.924	Yes
151: Off_PP_1,Off_PP_5,Off_PP_7,Off_PP_9	-2.372	3.049	Yes
161: Off_PP_1,Off_PP_6,Off_PP_9,Off_PP_7	-1.625	2.919	Yes
165: Off_PP_1,Off_PP_8,Off_PP_9,Off_PP_6	-4.192	2.291	Yes
174: Off_PP_2,Off_PP_4,Off_PP_6,Off_PP_3	-0.786	4.462	Yes
231: Off_PP_2,Off_PP_6,Off_PP_8,Off_PP_4	-7.245	0.673	Yes

Table 7.10. Bias-Corrected and Bonferroni-Adjusted Confidence Intervals of theCTA-PLS analysis for Off_PP of T_U (n = 383).

IV	Indicators	Convergent Validity			Internal Reliability	
LV	Indicators	Loadings	Communality	AVE	CR	CA
0	Bond_1	0.817	0.667			
	Bond_2	0.781	0.610]		
	Bond_3	0.787	0.619			
	Bond_4	0.749	0.561	1		
- S(Bond_5	0.824	0.679	0.654	0.050	0.041
ond	Bond_6	0.739	0.546	0.054	0.950	0.941
щ	Bond_7	0.804	0.646	1		
	Bond_8	0.843	0.711	1		
	Bond_9	0.837	0.701	1		
	Bond_10	0.892	0.796	1		
	Brid_1	0.865	0.748			0.053
	Brid_2	0.853	0.728	1	0.050	
	Brid_3	0.863	0.745	1		
Brid_SC	Brid_4	0.844	0.712	0.703 0.959		
	Brid_5	0.866	0.750			
	Brid_6	0.879	0.773		0.955	
	Brid_7	0.807	0.651	1		
	Brid_8	0.791	0.626	1		
	Brid_9	0.846	0.716	1		
	Brid_10	0.766	0.587	1		
	Off_PP_1	0.742	0.551			1
	Off_PP_2	0.622	0.387	1		
	Off_PP_4	0.661	0.437	1		
FP -	Off_PP_5	0.703	0.494	0.504	0.000	0.001
ΟĒ	Off_PP_6	0.651	0.424	0.504	0.890	0.861
	Off_PP_7	0.668	0.446			
	Off_PP_8	0.817	0.667]		
	Off_PP_9	0.792	0.627			
	On_PP_1	0.748	0.560			
	On_PP_2	0.648	0.420			
£,	On_PP_3	0.703	0.494	0.500	0.054	0.000
8	On_PP_4	0.765	0.585	0.508	0.861	0.808
	On_PP_5	0.725	0.526			
	On PP 6	0.682	0.465			

Table 7.11. Summary Measurement Model Assessment Metrics for F_U Samplewith no Control Variables and Off_PP_3 Indicator Removed (n= 432).

Discriminant Validity Summary					
Constructs	Fornell-Larcker	Cross-Loadings	HTMT Ratio	HTMT CI	
Bond_SC					
Brid_SC	Condition met	Condition met	< 0.90	∌ 1	
Off_PP					
On_PP					

Table 7.12. Discriminant Validity Summary Measurement Model F_U with noControl Variables and Off_PP_3 Indicator Removed (n= 432).

Notably, the AVE of the construct Off_PP increased to 0.504 which is above the recommended threshold of 0.50. Also, CR (0.890) and CA (0.861) values remained within acceptable limits. Similarly, the measurement model of T_U was re-estimated without the problematic indicator (Table 7.13). Results showed good internal consistency reliability, convergent validity and discriminant validity. The Fornell-Larcker Criterion, the cross-loadings and the HTMT tables with related confidence intervals of the new models are provided in Appendix M.

CTA-PLS were run also for all the other constructs of the model (Appendix N). The results supported the reflective specifications of the measurements models of all constructs.

IV	Indiastawa	Convergent Validity			Internal I	Internal Reliability	
LV	indicators	Loadings	Communality	AVE	CR	CA	
	Bond_1	0.898	0.806			0.958	
	Bond_2	0.890	0.792]			
	Bond_3	0.837	0.701				
Ð	Bond_4	0.826	0.682]			
S	Bond_5	0.874	0.764	0.720	0.964		
ono	Bond_6	0.810	0.656	0.729			
щ	Bond_7	0.831	0.691]			
	Bond_8	0.817	0.667]			
	Bond_9	0.812	0.659]			
	Bond_10	0.935	0.874				
	Brid_1	0.849	0.721		0.963	0.957	
	Brid_2	0.847	0.717	1			
	Brid_3	0.821	0.674	1			
	Brid_4	0.866	0.750	0.720			
SC	Brid_5	0.831	0.691				
3rid	Brid_6	0.850	0.723				
ш	Brid_7	0.852	0.726				
	Brid_8	0.813	0.661				
	Brid_9	0.897	0.805	1			
	Brid_10	0.857	0.734	1			
	Off_PP_1	0.820	0.672			0.007	
	Off_PP_2	0.553	0.306]			
	Off_PP_4	0.780	0.608	1			
L L	Off_PP_5	0.833	0.694	0.502	0.017		
ŌĤ	Off_PP_6	0.782	0.612	0.585	0.917	0.897	
	Off_PP_7	0.644	0.415	1			
	Off_PP_8	0.831	0.691	1			
	Off_PP_9	0.815	0.664	1			
	On_PP_1	0.829	0.687				
	On_PP_2	0.763	0.582]			
E.	On_PP_3	0.828	0.686	0.600	0.000	0.006	
8	On_PP_4	0.896	0.803	0.082	0.928	0.900	
_	On_PP_5	0.859	0.738	1			
	On_PP_6	0.771	0.594	1			

Table 7.13. Summary Measurement Model Assessment Metrics for T_U Samplewith no Control Variables and Off_PP_3 Indicator Removed (n = 383).

Discriminant Validity Summary					
Constructs	Fornell-Larcker	Cross-Loadings	HTMT Ratio	HTMT CI	
Bond_SC					
Brid_SC	Condition mot	Condition met	< 0.90	⊉ 1	
Off_PP	Condition met				
On_PP					

Table 7.14. Discriminant Validity Summary Measurement Model T_U with no
Control Variables and Off_PP_3 Indicator Removed (n= 383).

Once the measurement models of F_U and T_U samples were assessed with no control variables, an assessment of the measurement models with control variables (Figure 7.5) was performed following the same rationale. According to Hair et al. (2017) control variables that can account for some of the target construct's variation must be included in the analysis to check their effects on the main relationships being tested but usually are not further interpreted. A summary is reported in Table 7.15, Table 7.16 and Table 7.17. Notably, all measurement models of F_U and T_U showed good internal consistency reliability, convergent and discriminant validity.

 Table 7.15. Internal Consistency Reliability Summary F_U and T_U with control

 variables and Off_PP_3 removed.

Internal Consistency Reliability Summary				
	F_U		T_U	
Constructs	CA	CR	CA	CR
Bond_SC	0.941	0.950	0.958	0.964
Brid_SC	0.953	0.959	0.957	0.963
Off_PP	0.861	0.891	0.897	0.918
On_PP	0.808	0.862	0.906	0.928
PEff	0.889	0.923	0.906	0.934
PInt	0.891	0.933	0.919	0.949

Convergent Validity Summary					
Generation	F_U	F_U	T_U		
Constructs	Loadings	Communality	AVE	AVE	
Bond_SC			0.654	0.729	
Brid_SC	Adaguata	Adequate	0.703	0.720	
Off_PP			0.508	0.586	
On_PP	Adequate		0.511	0.682	
PEff			0.750	0.780	
PInt			0.823	0.861	

Table 7.16. Convergent Validity Summary F_U and T_U with control variables
and Off PP 3 removed.

 Table 7.17. Discriminant Validity Summary F_U and T_U with control variables

 and Off_PP_3 removed.

Discriminant Validity Summary				
	F _	U and T_U		
Constructs	Fornell-Larcker	Cross-Loadings	HTMT	HTMT CI
Bond_SC				
Brid_SC				
Off_PP	Condition mot	Condition met	< 0.90	⊉ 1
On_PP	Condition met			
PEff				
PInt				

Having ascertained that the measurement models with and without control variables are reliable and show good convergent and discriminant validity, in the next Section we assess and compare the related structural models.

7.1.3. ASSESSING PLS-SEM RESULTS OF THE STRUCTURAL MODEL FOR Q1

Following Hair et al. (2017) the Structural (inner) model (Figure 7.2) was assessed for predictive relevance, accuracy and paths' significance through the criteria reported in Appendix K. Also, to finalise the analysis and control for type II error we performed post-hoc statistical power analysis (see Chapter 6, Section 6.5.1.2) as a function of the significance level α , sample size, and variables' effect size f^2 using the G*Power 3.1.9 software (Faul et al., 2007, 2009). Moreover, for the purpose of comparing different models' predictive accuracy and relevance while retaining

acceptable levels of explanatory power and chose the one (with or without control variables) to further employ in the Permutation analysis, the selection criteria discussed in Chapter 6, Section 6.5.1.2 and summarised in Appendix K were employed. Further, a FCA was conducted to empirically test for CMB (See section 6.3.5).

We first assessed the structural model without control variables of F_U sample. The first step to assess the structural model is to check for the presence of any collinearity issues, by examining the VIF of all sets of predictor constructs. Table 7.18 reports the VIF values of all combinations of endogenous constructs (represented by the columns) and corresponding exogenous variables (represented by the rows). Notably, the VIF value is below the threshold of 5 (Hair et al., 2017). Therefore, collinearity among the exogenous variables is not a critical issue.

Table 7.18. VIF Values of the Structural Model of F_U with no Control Variables (n = 432).

VIF Sample F_U no Control Variables			
	Off_PP/On_PP		
Bond_SC/Brid_SC	1.444		

Then, the R^2 and the R^2 adj values are assessed (Table 7.19). Following the rule of thumb provided in Appendix K the R^2 values of Off_PP (0.107) and On_PP (0.130) can be considered weak. R^2 adj is employed later for model comparison.

Table 7.19. Coefficients of Determination of the Structural Model of F_U with noControl Variables (n = 432).

Coefficient of Determination R ² Sample F_U no Control Variables				
	R Square	R Square Adjusted		
Off_PP	0.107	0.102		
On_PP	0.130	0.126		

Following Ohtani (2000) and Streukens and Leroi-Werelds (2016), we further investigated the R^2 with a measure of precision to assess whether the R^2 values were significantly different from zero through BCa bootstrap CI. Table 7.20 shows that the R^2
for both On_PP (p < 0.001) and Off_PP (p = 0.001) are significantly different from 0. Hence, the model has significant explanatory capacity for both endogenous variables.

Table 7.20. BCa Confidence Intervals of the R^2 of F_U with no Control Variables (n = 432).

BCa Confidence Intervals R ² Significance								
	Original	Sample				Std.	Т	Р
	Sample (O)	Mean (M)	Bias	2.5%	97.5%	Dev.	Stat.	Values
Off_PP	0.107	0.115	0.009	0.052	0.166	0.032	3.362	0.001
On_PP	0.130	0.140	0.010	0.069	0.187	0.031	4.164	0.000

The next step to assess the structural model consists of examining the f^2 effect size of the exogenous constructs on the endogenous variables (Table 7.21). Following Cohen (1988) guidelines, the impact of Bond_SC on Off_PP ($f^2 = 0.004$) and On_PP ($f^2 = 0.002$) can be considered negligible meanwhile Brid_SC has between a small and a medium effect size on both Off_PP ($f^2 = 0.101$) and On_PP ($f^2 = 0.088$). Then the path coefficients in terms of their sign, magnitude and significance are assessed by mean of BCa CI (Table 7.22 and Table 7.23).

Table 7.21. Effect Size f^2 of the Structural Model of the Sample of F_U with noControl Variables (n = 432).

Effect Size <i>f</i> ² Sample Facebook Users no Control Variables					
	Off_PP	On_PP			
Bond_SC	0.004	0.002			
Brid_SC	0.101	0.088			

Table 7.22. Path Coefficients of the Structural Model of the Sample of F_U withno Control Variables (n = 432).

Paths Coefficients Sample Facebook Users no Control Variables						
	Off_PP	On_PP				
Bond_SC	-0.074	0.045				
Brid_SC	0.362	0.333				

Path Coefficients BCa CI Sample Facebook Users no Control Variables								
	Original Sample							
	Sample (O)	Mean (M)	Bias	2.50%	97.50%			
Bond_SC -> Off_PP	-0.074	-0.068	0.006	-0.190	0.025			
Bond_SC -> On_PP	0.045	0.051	0.006	-0.057	0.127			
Brid_SC -> Off_PP	0.362	0.366	0.005	0.240	0.464			
Brid_SC -> On_PP	0.333	0.339	0.006	0.229	0.421			

Table 7.23. (Confidence Intervals BCa of the Path Coefficients of the Structural
	Model of F U with no Control Variables ($n = 432$).

All the exogenous constructs have a positive effect on Off_PP and On_PP except for Bond_SC on Off_PP ($\beta = -0.074$). The relationships of the structural model are all significant at 5% significant level but Bond_SC \rightarrow Off_PP, $\beta = -0.074$, 95% BCa CI [-0.190, 0.025] and Bond_SC \rightarrow On_PP, $\beta = 0.045$, 95% BCa CI [-0.057, 0.127]. However, to control for Type II error a post-hoc statistical power test is performed to compute achieved power and minimise the chances of getting false negatives. To this purpose, the G*power software 3.1.9. is employed. The "linear multiple regression: fixed model, Single regression coefficient" test is selected to ascertain the true significance of the single effects of Bond_SC on Off_PP and Bond_SC on On_PP (Faul et al., 2009). Inserting the respective effect size ($f^2 = 0.004$ and $f^2 = 0.002$) the alpha value = 0.05, n = 432 the software yields a power (1- β) of 26% and 15% respectively. According to Cohen (1988) we do not have enough power to claim that no effect would be present in the population.

Next, the blindfolding procedure is run to assess the predictive relevance of the path model through the Q^2 index (Hair et al., 2017). An omission distance (D) of seven (7) is specified (Hair et al., 2017) Table 7.24 shows the results of the blindfolding procedure. Notably all Q^2 values are above the threshold of zero. and provide support for the model predictive relevance with respect to both Off_PP and On_PP.

Blindfolding procedure Facebook Users no Control Variables - Distance 7						
	SSO	SSE	Q ² (=1-SSE/SSO)			
Off_PP	3,456.00	3,297.73	0.046			
On_PP	2,592.00	2,446.06	0.056			

Table 7.24. Predictive Relevance Q2 of the Structural Model of the Sample ofFacebook Users with no Control Variables (n = 432).

Finally, Table 7.25 summarises the results of the q^2 effect size that is computed manually following the formula (Equation 6.1) provided in Section 6.5.1.2. Following the rule of thumb the q^2 effect size for the relationships Bond_SC \rightarrow Off_PP (q^2 = 0.002) and Bond_SC- \rightarrow On_PP (q^2 = 0.000) can be considered negligible. Rather, Brid_SC shows a small predictive relevance on both Off_PP (q^2 = 0.042) and On_PP (q^2 = 0.037).

Table 7.25. q2 Effect Size of the Sample of Facebook Users with no ControlVariables (n = 432).

q ² Effect Size Facebook Users no Control Variables					
	OFF_PP	On_PP			
Bond_SC	0.002	0.000			
Brid_SC	0.042	0.037			

Moreover, following the suggestions of Hair et al. (2017) and Henseler et al. (2014) model fit measures (SRMR and RMS_{theta}) are provided in Table 7.26. According to their guidelines we can conclude that the model shows sufficient model fit with (SRMR = 0.099 and RMS_{theta} = 0.105). Following Shmueli et al. (2016) and Sharma et al. (2019a) in-sample and out-of-sample predictive accuracy measures are reported in Table 7.26. As noted earlier, these criteria cannot be interpreted as stand-alone statistics as they are relative rather than absolute measures and therefore will be used only as a mean for model comparison.

	Model Selection Criteria							
Endogenous Variables	In-Sample Criteria		Out-of-Sample Criteria		Model fit measures			
	AICu	AICc	BIC	RMSE	MAD	RMS _{Theta}	SRMR	
Off_PP	-40.672	390.411	-31.477	0.976	0.635	0.105	0.000	
On_PP	-51.990	379.093	-42.795	0.971	0.640	0.105	0.099	

Table 7.26. In-Sample and Out-of-Sample Model Selection Criteria and Model Fitmeasures of the Sample of Facebook Users with no Control Variables (n = 432).

Finally, following Cock (2015) a FCA is provided in Table 7.27 to empirically check for the presence of CMB. Notably, all the VIF values are below the 3.3 threshold. Hence, CMB is not an issue for the analysis.

Table 7.27. CMB Assessment Through Multicollinearity for the Sample ofFacebook Users with no Control Variables (n = 432).

CMB Full VIF Assessment							
	Bond_SC	Brid_SC	Off_PP	On_PP			
Bond_SC		1.055	1.471	1.441			
Brid_SC	1.126		1.590	1.569			
Off_PP	1.393	1.722		1.108			
On_PP	1.504	1.787	1.139				

Following the above procedures, we further assessed the structural model for T_U (n = 383) sample. Overall, the analysis reveals that there are no problems of Collinearity (all VIF < 5). The model has between a weak and a moderate predictive accuracy and significant explanatory power for both Off_PP [($R^2 = 0.163$), 95% BCa CI (0.097, 0.219), p < 0.05] and On_PP [($R^2 = 0.210$), 95% BCa CI (0.143, 0.269), p < 0.05]. All the exogenous constructs have a positive effect on both Off_PP and On_PP and their relationships are all significant at 5% significance level. Specifically:

- Bond_SC \rightarrow Off_PP, $\beta = -0.222, 95\%$ BCa CI [-0.119, 0.315], $f^2 = 0.053$.
- Bond_SC \rightarrow On_PP, $\beta = 0.246$), 95% BCa CI [-0.145, 0.342], $f^2 = 0.069$.

- Brid_SC \rightarrow Off_PP [($\beta = 0.272$), 95% BCa CI [0.167, 0.280], $f^2 = 0.079$.
- Brid_SC \rightarrow On_PP [($\beta = 0.314$), 95% BCa CI [0.214, 0.405], $f^2 = 0.112$.

The model shows predictive relevance for both Off_PP ($Q^2 = 0.084$) and On_PP ($Q^2 = 0.136$). However, the relative impact of predictive relevance of Bond_SC is small for Off_PP ($q^2 = 0.025$) and On_PP ($q^2 = 0.042$). Also, Brid_SC has a small effect size in term of predictive relevance on Off_PP ($q^2 = 0.037$) but shows an effect size between small and medium for On_PP ($q^2 = 0.065$). Furthermore, the model shows sufficient model fit (SRMR = 0.098 and RMS_{theta} = 0.108). There is no empirical evidence of issues concerning the CMB as all the VIF values of the latent constructs are below 3.3.

The structural models with control variables for F_U and T_U samples were assessed using the above rationale. Noteworthy, all the models with control variables of F_U and T_U samples did not show any problems of Multicollinearity and were free from CMB. Also, all the control variables were retained even if their influence was non-significant. This is due to the fact that their inclusion stemmed from theoretical grounds based on Political Participation theories and their inclusion increase the validity of the study. Also, following Hair et al. (2013) "*regardless of whether those control variables are significant or not, the results for control variables are usually not further interpreted*" (p. 3).

For the purpose of this analysis, a summary of statistics of all the models (with control and no control variables) like R^2 , R^2_{adj} , Q^2 , model fit measures and in-sample and out-of-sample measures of predictive relevance are reported in Table 7.28 and Table 7.30 for F_U and T_U respectively. Following Hair et al. (2017), Sharma et al. (2019a,b) and Shmueli et al. (2016) the aim is to compare and select those models that show higher predictive accuracy and relevance while retaining good model fit and further employ them to analyse the moderation effects of "Type_SNS_Used" by means of Permutation test. Evaluating predictive performance is recognized as a useful practice for theory building and validation purposes (Shmueli et al., 2016).

Notably, for both On_PP and Off_PP in the F_U sample, the model that includes the control variables shows better in-sample and out-of-sample predictive relevance since it has lower AICu, AICc, BIC, RMSE and MAD. Also, the Q^2 index of predictive power, R^2 and R^2_{adj} are higher for the model with control variables. Moreover, the change in R^2 due to the introduction of the control variables is significant for both Off_PP ($F_{6,423} = 13.892$, p < 0.05) and On_PP ($F_{6,423} = 7.733$, p < 0.05) (Table 7.29). Only the value of the RMS_{Theta} (0.105) indicates that the model with no control variables is a better fit of the data. However, the difference with the RMS_{Theta} of the model with control variables is 0.001. In light of all the considerations made above the model with control variables included is considered superior in terms of predictive performance and fit of the data when compared to the model with no control variables for the F_U sample.

Table 7.28. Model Comparison with and without Controls Variables for F_U (n = 432).

	Model Comparison of the Sample Facebook Users With and Without Control Variables											
	genous ables	ाn-Sample Measures Out-of-Sample Measures M		Model fit 1	Model fit measures							
Tudor	Endoge Varia Mod	Mo	R ²	R ² adj	AICu	AICc	BIC	RMSE	MAD	Q ²	RMS _{Theta}	SRMR
	PP	No Controls	0.107	0.102	-40.672	390.411	-31.477	0.976	0.635	0.046	0.105	0.099
	Off	Controls	0.254	0.240	-100.682	324.746	-73.161	0.907	0.620	0.120	0.106	0.076
	PP	No Controls	0.130	0.126	-51.990	379.093	-42.795	0.971	0.640	0.056	0.105	0.099
	Ō	Controls	0.216	0.201	-79.025	346.402	-51.504	0.931	0.621	0.098	0.106	0.076

Significance Test R ²						
LV	Model	F test	F critical	Sig.		
Off PP	No Controls	13 892	2 1 2 0	Ves		
011	Controls	15:072	2.120	105		
On_PP	No Controls	7 722	2 1 2 0	Yes		
	Controls	1.135	2.120			

Table 7.29. F test for Significance of the Change in R2 Between the ModelsWithout and With Control Variables of F_U (n = 432).

Finally, Table 7.30 provides the values of the selection criteria of the model with and without control variables for the T_U sample. Notably, the model with control variables shows higher in-sample and out-of-sample predictive accuracy and relevance for Off_PP while retaining a better fit of the data compared to the model with no control variables. Moreover, the change in R^2 (Table 7.31) due to the introduction of the control variables is significant for Off_PP (F_{6,374} = 16.478, p < 0.05).

However, for the endogenous variable On_PP, although the model fit measures show that the model with control variables ($RMS_{Theta} = 0.100$, SRMR = 0.077) is a better fit than the one with no control variables ($RMS_{Theta} = 0.108$, SRMR = 0.098), only the out-of-sample criteria and the R^2 (0.210) indicate that the model has higher predictive accuracy and relevance compared to the model with no control variables. The remaining measures of in-sample predictive relevance show better values for the model with no control variables. Also, the F test yield not significant results for On_{PP} (F_{6,374} = 0.880, p > 0.05). The priority that researcher should give to all the in-sample and out-of-sample criteria whether they provide discordant results should be based on theoretical and/or empirical aims of the research projects (Sharma et al., 2019a,b). Provided that to compare models by means of Permutation test we need to establish configural invariance which implies that measurement models across groups must have identical indicators and latent variables (Hair et al., 2017) we retain the model with control variables also for On_PP in T_U. Indeed, the T_U model without control variables shows poor fit compared to the model with control variables included. Also, according to Shmueli et al. (2016) a good model must be able to predict new cases and out-of-sample prediction criteria should be preferred to the in-sample metrics.

Table 7.30. Model Comparison with and without Controls Variables for T_U (n =
383).

	Model Comparison of the Sample Twitter Users With and Without Control Variables											
Endogenous Variables	del	In-Sample Measures					Out N	-of-Samj Ieasures	ple	Model fit measures		
	Mo	\mathbf{R}^2	R ² adj	AICu	AICc	BIC	RMSE	MAD	Q^2	RMS _{Theta}	SRMR	
ЪР	No Controls	0.163	0.158	-60.011	322.083	-51.179	0.932	0.687	0.084	0.108	0.098	
Off	Controls	0.338	0.324	-131.910	244.574	-105.485	0.845	0.606	0.186	0.100	0.077	
ЪР	No Controls	0.210	0.206	-82.140	299.954	-73.308	0.915	0.694	0.136	0.108	0.098	
On	Controls	0.221	0.204	-69.550	306.933	-43.126	0.901	0.681	0.142	0.100	0.077	

Table 7.31. F test for Significance of the Change in R2 Between the Models Without and With Control Variables of T_U (n = 383).

Significance Test R ²						
LV	Model	F test	F critical	Sig.		
Off_PP	No Controls Controls	16.478	2.122	Yes		
On_PP	No Controls Controls	0.880	2.122	No		

In the next sections the model of F_U and T_U users will be tested for measurement invariance and then compared through permutation test.

7.3.1.1. ASSESSMENT OF MODERATION EFFECTS THROUGH PERMUTATION TEST.

This Section provides the results of the MGA Permutation test carried out to ascertain any moderating effects of the categorical variable "Type_SNS_Used" (two categories, Facebook and Twitter) over the relationships between Bond_SC and Brid_SC with On_PP and Off_PP.

Following the MGA Permutation procedure discussed in Chapter 6, Section 6.5.1.3, measurement invariance among the constructs of the groups must be established through the MICOM procedure before comparing groups parameters (Hair et al., 2017; Henseler et al., 2016). This involves:

- 1. Establishing configural invariance.
- 2. Establishing compositional invariance.
- 3. Ensuring equality of composite mean values and variances.

Consistently with Henseler et al. (2016) *configural invariance* was assessed through a qualitative inspection of the model composite, ensuring that measurement model specifications and algorithm computation settings for their assessment were the same across F_U and T_U samples.

To test for *compositional invariance*, we run a permutation test proposed by Henseler et al. (2016) and Hair et al. (2017) to examine whether the correlation between composite score of the same constructs across F_U and T_U samples is equal to one. To confirm compositional invariance the permutation test should yield a *p* value larger than 0.05. However, because at a later stage to answer research question Q3 the two groups of F_U and T_U are compared against the sample of $F+T_U$, we adopted the Sidak correction discussed in Chapter 6 in Section 6.5.2 to control for Type I error inflation due to multiple comparisons. Hence, since we have three groups (m), at a starting level of significance of 0.05 we computed the Sidak correction through the formula 1-(1-alpha)^{1/m} which yields a significance level of 0.017. Table 7.32 reports the results of the compositional invariance test. Notably, all the 1.7% quantiles of the empirical distribution of the correlations between the composite scores of F_U and T_U samples

are smaller than or equal to the original correlations and all p values are not significant. Therefore, compositional invariance is established.

Compos	Compositional Invariance Between Facebook Users and Twitter Users								
	Original	Correlation	1 70/	Permutation					
	Correlation	Permutation Mean	1./%	p-Values					
Bond_SC	0.997	0.998	0.996	0.676					
Brid_SC	0.999	0.999	0.999	0.350					
Off_PP	0.998	0.997	0.996	0.232					
On_PP	0.999	0.999	0.997	0.604					
PEff	0.998	0.998	0.998	0.250					
PInt	1.000	0.999	0.999	0.244					

Table 7.32. Compositional Invariance Constructs of Facebook Users (n = 432) andTwitter Users (n = 383).

Age and Pknow are not included in the analysis as they are single items constructs and by definition single items constructs have single outer relationship of 1 (Hair et al., 2017).

Once configural and compositional invariance are established partial measurement invariance is confirmed and a Permutation test is feasible (Hair et al., 2017).

However, the MICOM procedure includes a further step to test whether the means and the variances of the latent constructs across groups are equal. In that case the researcher can confirm full measurement invariance and a pooled data analysis can be conducted on the entire dataset. Table 7.33 provides the lower (0.85%) and the upper (99.15%) boundaries of the 98.3% CIs of the scores means differences. If the CIs do not include the original difference in mean values, then it can be concluded that there is significant difference in the composites' mean scores across the two groups. To further validate the results, the relative p values can be examined . If the p value is below 0.017 then the difference of the composite means values of different groups is significant. Notably, for the variables Age, Bond_SC, Off_PP, On_PP, PEff, and PInt the CIs do not include the original mean difference and the relative p values are below the 0.017 threshold.

	Equality of Composite Mean Values for Facebook and Twitter Samples						
	Mean - Original Difference [Groups_F_U_(2) - T_U_(3)]	Mean - Permutation Difference [Groups_F_U_(2) - T_U_(3)]	0.85%	99.15%	Permutation p-Values	Equality of Means Established?	
Age	0.604	-0.001	-0.138	0.135	0.000	No	
Bond_SC	0.236	0.001	-0.139	0.143	0.001	No	
Brid_SC	-0.033	0.000	-0.138	0.138	0.632	Yes	
Off_PP	-0.389	-0.001	-0.137	0.137	0.000	No	
On_PP	-0.401	0.000	-0.139	0.138	0.000	No	
PEff	-0.319	0.001	-0.140	0.141	0.000	No	
PInt	-0.211	0.001	-0.139	0.139	0.003	No	
Pknow	0.098	-0.002	-0.139	0.133	0.162	Yes	

Table 7.33. Equality of Composite Mean Values for Facebook (n = 432) andTwitter (n = 383) Samples.

Similarly, Table 7.34 shows the results for the composites' variances. Also, in this case for the variables Age, Off_PP, PEff, PInt, and Pknow the variances CIs do not include the original variance difference and the relative p values are below 0.017. Accordingly, compositional invariance cannot be confirmed.

Table 7.34. Equality of Composite Variances Values for Facebook (n = 432) andTwitter (n = 383) Samples.

	Equality of Composite Variances Values for Facebook and Twitter Samples						
	Variance - Original Difference [Groups_F_U_(2) - T_U_(3)]	Variance - Permutation Mean Difference [Groups_F_U_(2) - T_U_(3)]	2.50%	97.50%	Permutation p-Values	Equality of Variances Established?	
Age	-0.272	0.002	-0.156	0.162	0.001	No	
Bond_SC	0.031	0.000	-0.137	0.138	0.667	Yes	
Brid_SC	-0.121	0.000	-0.161	0.154	0.134	Yes	
Off_PP	-0.067	0.000	-0.035	0.039	0.001	No	
On_PP	-0.015	0.001	-0.021	0.026	0.235	Yes	
PEff	-0.686	-0.003	-0.400	0.399	0.001	No	
PInt	-0.867	-0.002	-0.356	0.349	0.000	No	
Pknow	-0.170	-0.001	-0.144	0.142	0.017	No	

In summary, partial measurement invariance is established and we can proceed further with the analysis of differences of the path coefficients across groups by means of permutation tests. The Algorithm settings are specified in Appendix K. Table 7.35 shows the output of the permutation test reporting the two tailed 98.3% permutationbased Cis for the relations of interest (Sidak's correction applied).

	Permutation Test F_U (432) Vs T_U (383) Output							
	Std.β	Std.β	Std.β (d)	Std. β Perm.			Permutation	
	[F_U]	[T_U]		Mean (d)	0.85%	99.15%	p-Values	
Bond_SC -> Off_PP	-0.081	0.152	-0.233	0.000	-0.133	0.135	0.001	
Bond_SC -> On_PP	0.027	0.230	-0.203	0.001	-0.147	0.149	0.005	
Brid_SC -> Off_PP	0.312	0.180	0.132	-0.002	-0.125	0.105	0.014	
Brid SC -> On PP	0.285	0.297	-0.011	-0.002	-0.155	0.150	0.883	

Table 7.35. Permutation Test Results for Significant Differences betweenFacebook (n = 432) and Twitter (n = 383) users.

The cells representing significant difference are highlighted in green. Notably, the relationships that differ significantly at a 1.7% level of significance are:

- Bond_SC \rightarrow Off_PP, $d_{(\beta_i \beta_j)} = -0.233, 98.3\%$ CI [-0.133, 0.135], p = 0.001.
- Bond_SC \rightarrow On_PP, $d_{(\beta i \beta j)} = -0.203, 98.3\%$ CI [-0.147, 0.149], p = 0.005.
- Brid_SC \rightarrow Off_PP, $d_{(\beta i \beta j)} = 0.132, 98.3\%$ CI [-0.125, 0.105], p = 0.014.

Therefore, "Type_SNS_Used" significantly moderates the relationships between Bond_SC and On_PP and Off_PP and between Brid_SC and Off_PP.

However, in Section 7.2 when testing for the moderating effect of the variable "Int_SNS_Use" to answer research question Q2, we found a significant moderating effect in the sample of Twitter users over the relationships Brid_SC \rightarrow Off_PP and Brid_SC \rightarrow On_PP. Hence, to provide a more comprehensive assessment of the moderation effect of "Type_SNS_Used" we controlled for the "Int_SNS_Use" by splitting each group of users (Facebook and Twitter) into three subgroups through the quartile split method (see Section 7.1.1.3).

Permutation tests were carried out across Facebook and Twitter users according to their level of intensity of use. All the comparisons are reported in Table 7.36. However,

later, to answer research question Q3 (Section 7.3) also the F+T_U sample is compared to the F_U and T_U samples across each category of intensity of use. Hence for each set of three pairwise comparison the Sidak correction was applied ($\alpha = 0.017$).

Table 7.36. Comparisons of Facebook and Twitter Users Across Different Levelsof Intensity of SNSs Use.

Comparison of Facebook and Twitter Samples Across Different Levels of Intensity of Use of SNSs					
Levels of Intensity of SNSs Use	s Use Type of SNS Used				
	Facebook	\mathbf{Vs}	Twitter		
Heavy Users	HF_U		HT_U		
Mild Users	MF_U		MT_U		
Light Users	LF_U		LT_U		

Before proceeding with the group comparisons, we assessed the measurement and the structural models of the obtained subgroups and tested for the measurement invariance of constructs following the same procedures adopted earlier. All models were tested with the control variables included and the item Off_PP_3 removed to maintain consistency and facilitate comparisons of the results with the analysis conducted earlier by establishing configural invariance. However, before removing Off_PP_3, CTA-PLSs were performed on the Off_PP constructs of all the subgroups listed in Table 7.36 following the same rationale provided in Section 7.1.2. Notably, all the Off_PP constructs were correctly specified as reflective and we confidently proceeded to the removal of the items.

Regarding the measurement models, all the groups showed good internal consistency reliability (Table 7.37) and adequate discriminant validity.

Internal Con	Internal Consistency Reliability Summary						
	HF	<u>U</u>	НТ	<u>_</u> U			
Constructs	CA	CR	CA	CR			
Bond_SC	0.950	0.957	0.952	0.958			
Brid_SC	0.955	0.961	0.950	0.958			
Off_PP	0.927	0.942	0.921	0.936			
On_PP	0.775	0.844	0.899	0.923			
PEff	0.882	0.918	0.941	0.958			
PInt	0.905	0.941	0.954	0.970			
	MI	F_U	MT	ſ_U			
Constructs	CA	CR	CA	CR			
Bond_SC	0.932	0.929	0.955	0.961			
Brid_SC	0.924	0.937	0.960	0.965			
Off_PP	0.797	0.850	0.888	0.909			
On_PP	0.827	0.875	0.855	0.892			
PEff	0.902	0.932	0.904	0.932			
PInt	0.890	0.932	0.898	0.935			
	LF	_ U	LT	_ U			
Constructs	CA	CR	CA	CR			
Bond_SC	0.934	0.939	0.962	0.967			
Brid_SC	0.928	0.939	0.951	0.958			
Off_PP	0.795	0.843	0.839	0.879			
On_PP	0.849	0.890	0.855	0.890			
PEff	0.888	0.921	0.869	0.910			
PInt	0.885	0.929	0.906	0.941			

Table 7.37. Internal Consistency Reliability Summary of F_U and T_U withControl Variables and Off_PP_3 Removed.

Also, all constructs had good convergent validity and displayed adequate AVE values (above 0.5) but those reported in Table 7.38. Hair et al. (2017) suggest eliminating indicators with the lowest outer loadings to increase the AVE of a construct and also look at the effect of the removal on its CA, CR and discriminant validity. For this reason, in Table 7.38 the CR and the CA values together with a summary of the assessment of all the discriminant validity metrics like Fornell-Larker Criterion (F-L), Cross Loadings (C-L), and HTMT are reported. Moreover, the table provides an indication of the lowest outer loadings (L_O_L) of the reflective indicators of the problematic constructs and their respective values (V_L_O_L).

Cor	Constructs with AVE < 0.5 in the Heavy, Mild and Light Users of Facebook and Twitter Samples								
LV	Sub- Group	AVE	CR	CA	F-L	C-L	нтмт	L_O_L	V_L_O_L
Ч	LF_U	0.407	0.843	0.795	ok	ok	ok	Off_PP_4	0.436
ff_P	MF_U	0.426	0.850	0.797	ok	ok	ok	Off_PP_5	0.404
0	LT_U	0.482	0.879	0.839	ok	ok	ok	Off_PP_2	0.481
On_PP	HF_U	0.476	0.844	0.775	ok	ok	ok	On_PP_2	0.556

Table 7.38. Problematic Constructs with AVE < 0.5 in the Heavy, Mild and Light</th>Sub-groups of Facebook and Twitter Users.

We started by eliminating the indicator with the lowest outer loading of the construct with the lowest AVE in a group. So, the indicator Off_PP_4 was at first eliminated from the Off_PP construct of LF_U. Items were removed until an AVE above the 0.5 threshold was obtained for all problematic constructs. The outcome of the elimination process is provided in Table 7.39.

Co	Constructs with AVE < 0.5 in the Heavy, Mild and Light Users of Facebook and Twitter Samples after removing Several Indicators							
LV	Sub- Groups	AVE	CR	CA	F-L	C-L	нтмт	Indicators Removed
ff_PP	LF_U	0.563	0.837	0.743	ok	ok	ok	Off_PP_4
	MF_U	0.542	0.818	0.696	ok	ok	ok	Off_PP_1 Off_PP_2
0	LT_U	0.536	0.821	0.711	ok	ok	ok	Off_PP_5
On_PP	HF_U	0.565	0.795	0.612	ok	ok	ok	On_PP_6 On_PP_3 On_PP_2

 Table 7.39. Outcome of Indicators Elimination Process.

Notably, all the problematic constructs now meet the 0.5 threshold for the AVE. However, this came at the expenses of the CA that for two constructs is below the recommended threshold of 0.7. Also, the loss of information is high (50%) provided the three indicators out of six needed to be removed from the On_PP construct and four out of eight were dropped from the Off_PP variable. Eliminating so many indicators from the measurement models of those constructs may raise concerns about their content validity and makes any comparison with results obtained from previous analysis irrelevant. In this case, Little et al. (1999) recommend to maintain indicators in the model even when the AVE is below the threshold of 0.5. Also, Fornell and Larcker (1981) argued that if AVE is below 0.5 but CR is above 0.6 then the construct presents still good convergent validity. Also, if discriminant validity is not an issue then problematic indicators can be retained. Looking at Table 7.38 we can see how all the problematic constructs show CR above 0.6 and have sufficient discriminant validity. Accordingly, we kept all the indicators.

The structural model was assessed for all groups reported in Table 7.36. Overall, the models showed no problems of multicollinearity and display good levels of predictive relevance and accuracy (Table 7.40).

	Q ² and R ² Significance Across Heavy, Mild and Light F_U and T_U								
		Bootstrap				Т	Р		
		\mathbf{Q}^2	\mathbb{R}^2	Sample Mean	SD	Statistics	Value		
	HF_U	0.217	0.368	0.419	0.063	5.860	0.000		
•	MF_U	0.084	0.247	0.305	0.061	4.066	0.000		
E.	LF_U	0.078	0.302	0.375	0.066	4.601	0.000		
ĴĴĴ	HT_U	0.348	0.592	0.635	0.062	9.595	0.000		
	MT_U	0.165	0.345	0.373	0.052	6.676	0.000		
	LT_U	0.117	0.344	0.418	0.065	5.286	0.000		
	HF_U	0.167	0.437	0.492	0.057	7.714	0.000		
•	MF_U	0.069	0.189	0.255	0.062	3.066	0.002		
E E	LF_U	0.074	0.191	0.293	0.078	2.459	0.014		
ő	HT_U	0.322	0.519	0.571	0.071	7.282	0.000		
	MT_U	0.160	0.320	0.350	0.055	5.817	0.000		
	LT_U	0.117	0.266	0.344	0.066	4.008	0.000		

Table 7.40. Q² and R² Values and their Significance for Facebook and TwitterUsers Across Different Levels of Intensity of SNSs Use.

However, as the main aim of assessing the structural models is to prepare the data for the multiple pairwise comparison analysis, Table 7.41 reports the paths coefficients values and their BCa CI to ascertain the behaviour of the variables to be tested for moderation in different group categories (significant values highlighted in green). Notably, the path coefficients inspection show that the model does not behaves very differently across F_U and T_U when controlling for the Int_SNS_Use. Also, post-hoc tests for the non-significant relationships are provided. Notably, the achieved power is below the recommended threshold of 80% (Cohen, 1988). Therefore, we cannot be confident that significant effects indeed do not exist.

	Heavy Users Significance 5%					
	HF U	HT U				
Bond_SC -> Off_PP	$\beta = -0.057$ $f^{e} = 0.003$ BCa CI = [-0.314, 0.168] Power (1- β) = 0.087	$\beta = 0.058$ $f^{e} = 0.007$ BCa CI = [-0.116, 0.201] Power (1- β) = 0.121				
Bond_SC -> On_PP	$\beta = 0.079$ f = 0.006 BCa CI = [-0.13, 0.271] Power (1- β) = 0.125	$\beta = 0.070$ $f^{e} = 0.009$ BCa CI = [-0.121, 0.247] Power (1- β) = 0.142				
Brid_SC -> Off_PP	$\beta = 0.397$ f = 0.117 BCa CI = [0.164, 0.629]	β = 0.439 f ⁴ = 0.413 BCa CI = [0.282, 0.572]				
Brid_SC -> On_PP	$\beta = 0.372$ f = 0.116 BCa CI = [0.154, 0.578]	$\beta = 0.712$ f = 0.922 BCa CI = [0.541, 0.846]				
	Mild Users Si	gnificance 5%				
	MF U	MT U				
Bond_SC -> Off_PP	$\beta = -0.016$ $f^{e} < 0.001$ BCa CI = [-0.440, 0.197] Power (1- β) = 0.050	$\beta = 0.138$ $f^{e} = 0.020$ BCa CI = [-0.023, 0.283] Power (1- β) = 0.483				
Bond_SC -> On_PP	$\beta = -0.017$ $f^{4} < 0.001$ BCa CI = [-0.449, 0.141] Power (1- β) = 0.050	$\beta = 0.199$ $f^{4} = 0.040$ BCa CI = [0.056, 0.346]				
Brid_SC -> Off_PP	$\beta = 0.249$ f = 0.057 BCa CI = [0.111, 0.401]	$\beta = 0.237$ f = 0.048 BCa CI = [0.100, 0.364]				
Brid_SC -> On_PP	$\beta = 0.244$ $f^{4} = 0.050$ BCa CI = [0.120, 0.400]	$\beta = 0.289$ f = 0.069 BCa CI = [0.144, 0.419]				
	Light Users Si	gnificance 5%				
Bond_SC -> Off_PP	$\beta = -0.093$ $f^{2} = 0.009$ BCa CI = [-0.346, 0.203] Power (1- β) = 0.154	$\beta = 0.112$ $f^{2} = 0.014$ BCa CI = [-0.102, 0.329] Power (1- β) = 0.207				
Bond_SC -> On_PP	$\beta = 0.062$ f = 0.003 BCa CI = [-0.313, 0.257] Power (1- β) = 0.083	$\beta = 0.235$ f = 0.055 BCa CI = [-0.042, 0.472] Power (1- β) = 0.618				
Brid_SC -> Off_PP	$\beta = 0.100$ $f^{4} = 0.010$ BCa CI = [-0.204, 0.369] Power (1- β) = 0.166	$\beta = 0.041$ f = 0.002 BCa CI = [-0.213, 0.238] Power (1- β) = 0.071				
Brid_SC -> On_PP	$\beta = 0.181$ $f^{4} = 0.027$ BCa CI = [-0.156, 0.383] Power (1- β) = 0.366	$\beta = 0.125$ f = 0.015 BCa CI = [-0.153, 0.361] Power (1- β) = 0.218				

Table 7.41. Path Coefficients and their Significance Across Levels of Intensity ofSNSs Use per Type of SNSs used.

Furthermore, CMB was investigated for all groups through FCA tests (Kock, 2015). Overall, all VIF values were below the conservative threshold of 3.3 but those reported in Table 7.42 for the sample of Facebook light users and in Table 7.43 for Twitter mild users. Yet, as argued by Kock (2015) VIFs tend to increase with model complexity and when algorithms that incorporate measurement errors are used. According to Akter et al. (2017) and Chin (2010) a model can be considered complex when it has 10 or more constructs. The proposed model to answer research question Q1 comprises 10 constructs and therefore can be considered complex. Moreover, this study employs the PLS-SEM analysis that incorporates measurement errors in the computation of the algorithm. Arguably, in these circumstances the VIF threshold used in CMB tests should be set to 5 (Kock and Lynn, 2012). Accordingly, CMB is not an issue at this stage.

Table 7.42. Problematic VIF values found through CMB Assessment of FacebookLight Users Sample with Control Variables and Item Off_PP_3 Removed (n = 99).

CMB Full VIF Assessment							
	Off_PP	Pknow					
PEff	3.364	3.407					

Table 7.43. Problematic VIF values found through CMB Assessment of TwitterMild Users Sample with Control Variables and Item Off_PP_3 Removed (n = 95).

CMB Full VIF Assessment										
	Bond_SC	Brid_SC	PEff	PInt						
Off_PP	3.811	3.681	3.476	3.531						
On_PP	3.473	3.399	3.489	3.535						

Moreover, before proceeding with the comparisons of the groups, measurement invariance was assessed following the procedure employed earlier. Notably, partial measurement invariance was established. across F_U and T_U groups at each level of their intensity of use (Appendix O). Hence, we proceeded with the pairwise permutation tests to compare groups as reported in Table 7.36.

The results of the Permutation outputs showed that the type of SNSs used moderates only the relationship Brid_SC \rightarrow On_PP, $d_{(\beta F_U-\beta T_U)} = -0.340$, 98.3% CI [-0.325, 0.339], p = 0.014 in the heavy users category, when controlling for Int_SNS_Use.

Notably, Int_SNS_Use seems to cancel out the moderating effect of the Type_SNS_Used which could be arguably due to differences in the time spent on those social networks by respondents. To further confirm this assumption, we tested Facebook and Twitter samples for significant differences in Int_SNS_Use using a Mann-Whitney U test. We used a non-parametric test because the variable "Int_SNS_Use" is not normally distributed in both samples. Indeed, Facebook users spent an average of 44 minutes online (SD = 71.81) and their time was non-normally distributed, with skewness of 4.567 (SE = 0.117) and kurtosis of 7.124 (SE = 0.234). Similarly, Twitter users spent an average of 32 minutes online (SD = 40.18) and the variable was non-normally distributed, with skewness of 2.24 (SE = 0.125) and kurtosis of 5.30 (SE = 0.429). The Mann-Whitney test (Table 7.45) indicated that the time spent online was significantly higher for Facebook users (Mdn = 20) than for Twitter users (Mdn = 20), U = 72903.5, Z = -2.939, p = 0.003, r = -0.103.

Table 7.44. Mann-Whitney Ranks Comparing Total Time Spent on Facebook (n =432) and Twitter (n = 383).

Mann-Whitney Ranks for Time Spent Online F UVs T U										
SNS Group	Ν	Mean Rank	Sum of Ranks							
FU	432	430.74	186080.50							
ΤU	383	382.35	146439.50							
Total	815									

Table 7.45. Mann-Whitney Test Statistics Comparing Total Time Spent on Facebook (n = 432) and Twitter (n = 383).

Mann-Whitney Test for Time Spent Online F UVs T U ^a							
	Totime						
Mann-Whitney U	72903.500						
Wilcoxon W	146439.500						
Z	-2.939						
Asymp. Sig. (2-tailed)	0.003						
a. Grouping Variable: SNS	Group						

Following the above rationale, it can be concluded that the type of SNS used moderates the relationships between Bond_SC and Brid_SC with On_PP and Off_PP only if the time spent on those SNS is not controlled for, with the exception of the relationship Brid_SC \rightarrow On_PP in the heavy category of users which is significantly stronger for T_U than for F_U.

7.2. INVESTIGATING THE MODERATING EFFECT OF INTENSITY OF USE OF SNSs

This Section presents the results of the analysis conducted to answer research question Q2 by testing the moderating effect of "Int_SNS_Use" on the relationships between Bond_SC and Brid_SC with Off_PP and On_PP of Facebook and Twitter users' samples.

Following the systematic approach to PLS-SEM analysis suggested by Hair et al. (2017), the Structural model and the Measurement model to answer research question Q2 and all its sub-questions are specified in Figure 7.6 and Figure 7.7. The structural model will be separately tested for F_U and T_U since the MICOM procedure conducted in Section 7.1.3.1 showed that an analysis on the pooled data of the two samples combined is meaningless because those groups do not show equality of means and variances. Notably, the model with the control variables is proposed as it has already been proved to yield higher predictive accuracy and relevance and better model fit compared to the model with no control variables for both the samples (see Section 7.1.3.1).



Figure 7.6. Structural Model Research Question Q2.

Following, Hair et al. (2017) and (Henseler and Chin, 2010) the two-stage approach was employed to ascertain whether Int_SNS_Use exerts a significance effect on the hypothesised relationships. Accordingly, the measurement model (Figure 7.7) includes four interaction terms (green constructs) as single-item measures created by multiplying the latent variable scores of the exogenous latent variables with those of the moderator variable (Hair et al., 2017).





Note: item Off_PP_3 was removed from the analysis as it was found to negatively affect the convergent validity of the Off_PP construct in Section 7.1.5.

The criteria and procedures employed to assess the measurement models of Facebook and Twitter users and the algorithm setting employed for the parameters' estimation are the same used for the analysis carried out in Section 7.1 and reported in Appendix K. Overall, the measurement models of Facebook and Twitter users showed good internal consistency reliability (CA and CR > 0.70) and adequate convergent

validity (AVEs > .50 and outer loadings in acceptable range). Yet, some of the indicators showed communality values below the 0.5 threshold (Table 7.46). However, they were retained as they did not affect neither the AVE nor the reliability of the related constructs. Also, all constructs show sufficient discriminant validity.

	Indicators' Communality Values Below 0.5											
Sample	LV	Indicators	Communality	AVE	СА	CR						
		Off_PP_2	0.456									
	Off_PP	Off_PP_4	0.399									
		Off_PP_5	0.497	.497 0.508		0.891						
F_U		Off_PP_6	0.413									
		Off_PP_7 0.496										
	On DD	On_PP_2	0.397	0.511	0 000	0.862						
	OIL_PP	P On_PP_6 0.475		0.311	0.808	0.862						
тц	Off DD	Off_PP_2	0.352	0.586	0.807	0.018						
1_0	OII_PP	Off_PP_7	0.458	0.380	0.897	0.918						

Table 7.46. Indicators with Communality Values Below 0.5 in Facebook andTwitter Users Samples.

Once the measurement models were assessed, the structural model validity was evaluated for both F_U and T_U samples employing the criteria summarised in Appendix K. Notably, collinearity was not an issue in any of the structural models (VIF < 5). Also, the model displayed moderate and substantial levels of predictive accuracy respectively for F_U and T_U samples and good levels of predictive relevance (Table 7.47).

Table 7.47. Summary Statistics of Predictive Relevance and Accuracy for F_U andT_U Samples.

	Summary Statistics of Predictive Relevance and Accuracy with R ² Significance											
	Sample	Q ²	R ²	Bootstrap Mean R ²	Bias	2.5%	97.5%	SD	T Stat	P Value		
PP	F_U	0.115	0.264	0.292	0.028	0.184	0.314	0.040	6.666	0.000		
Οf	T_U	0.226	0.409	0.433	0.024	0.307	0.468	0.043	9.431	0.000		
PP	F_U	0.099	0.224	0.248	0.024	0.139	0.270	0.036	6.187	0.000		
On	T_U	0.302	0.453	0.476	0.023	0.357	0.517	0.044	10.238	0.000		

Following Hair et al. (2017), when the main objective of the analysis is to ascertain significant moderation effects, the next step involves analysing the size and the significance of the moderation path and its effect size. According to Kenny (2016) values of 0.005, 0.01, and 0.025 constitute small, medium, and large effect sizes, respectively.

Considering the Facebook users sample when assuming a 95% level of confidence:

- the Int_SNS_Use moderator had a *positive but not significant moderation effect* on the relationships:
 - Bond_SC → On_PP, β = 0.009, BCa CI [-0.077, 0.114], $f^2 < 0.001$,
 - Brid_SC → Off_PP, β = 0.092, BCa CI [-0.101, 0.343], f^2 = 0.006,
 - Brid_SC → On_PP, β = 0.039, BCa CI [-0.062, 0.161], f^2 = 0.001.
- the Int_SNS_Use showed a *negative not significant moderation effect* on the relationship:
 - Bond_SC → Off_PP, β = -0.046, BCa CI [-0.232, 0.127], f^2 = 0.002.

For the Twitter users' sample, assuming a 95% level of confidence:

- the Int_SNS_Use moderator had a *negative but not significant moderation effect* on the relationships:
 - Bond_SC → On_PP, β = -0.031, BCa CI [-0.180, 0.104], f^2 = 0.001,
 - Bond_SC → Off_PP, β = -0.003, BCa CI [-0.145, 0.149], $f^2 < 0.001$.
- the Int_SNS_Use exerted a *positive significant and large moderating effect* on the relationships:

◦ Brid_SC → Off_PP, β = 0.166, BCa CI [0.011, 0.286], f^2 = 0.030,

◦ Brid_SC → On_PP, β = 0.186, BCa CI [0.040, 0.301], f^2 = 0.041.

Yet, it must be noted that for all non-significant effects we achieved a statistical power $(1-\beta)$ below the recommended threshold of 80%.

To aid with the interpretation of the moderation analysis results, a slope plot is used to represent the relationships found to be significantly moderated (Hair et al., 2017). Figure 7.8 shows the results of the moderation effect of Int_SNS_Use on the relationship Brid_SC (x axis) \rightarrow Off_PP (y axis) for T_U. The middle line (red) represents the relationship for an average level of the moderator variable Int_SNS_Use also called simple effect. Specifically:

- For an average level of the moderator variable the simple effect Brid_SC → Off_PP is 0.171 (red line).
- For higher levels (mean value of Int_SNS_Use plus one standard deviation) of Int_SNS_Use, Brid_SC → Off_PP is 0.337 (green line).
- For lower levels (mean value of Int_SNS_Use minus one standard deviation unit) of Int_SNS_Use, Brid_SC → Off_PP is 0.005 (blue line).

Notably, the relationship between Brid_SC and Off_PP is positive as all three lines have a positive (upwards) slope. This relationship is significantly moderated by the time users spend online, given that, for higher levels of intensity of use, the relationship Brid_SC \rightarrow Off_PP is stronger than the simple effect, while for lower levels of Int_SNS_Use the relationship is still positive but weak.

Figure 7.8. Simple Slope Plot Brid_SC → Off_PP when Moderated by Int_SNS_Use (Twitter Sample).



Similarly, Figure 7.9 shows the results of the moderation effect of Int_SNS_Use on the relationship Brid_SC \rightarrow On_PP for T_U. Specifically:

- For an average level of the moderator variable the simple effect Brid_SC → On_PP is 0.270 (red line).
- For higher levels of Int_SNS_Use (mean value of Int_SNS_Use plus one standard deviation), Brid_SC → On_PP is 0.456 (green line).
- For lower levels of Int_SNS_Use (mean value of Int_SNS_Use minus one standard deviation unit) Brid_SC → On_PP is 0.084 (blue line).

Notably, the relationship between Brid_SC and On_PP is positive for all three lines. Hence, higher levels of Brid_SC provide higher levels of On_PP. However, their relationship is significantly moderated by the time users spend online.



Figure 7.9. Simple Slope Plot Brid_SC → On_PP when Moderated by Int_SNS_Use (Twitter Sample).

To corroborate the above results, we conducted a further assessment of the models with and without the moderation term with respect to their in-sample and out-of-sample metrics and the model fit measures summarised in Appendix K. Notably, in the Facebook sample, for the endogenous variables Off_PP the model without the moderator shows better in-sample and out-of-sample predictive relevance (Table 7.48). The same holds true for On_PP. In fact, although the RMSE and the Q^2 index suggest that the model

with the moderator variable should be chosen in terms of predictive accuracy all the remaining in-sample and out-of-sample measures point in favour of the model with no moderator except for the R^2 and the R^2_{adj} which are higher for both Off_PP and On_PP in the moderation model. Yet, those metrics are "*largely uninformative in the context of model selection as they always improve with model complexity*" (Sharma et al., 2019b: p. 348). Hence, they are usually used as second-best choice for model comparison. Therefore, we focus primarily on the other criteria and, based on the principle of parsimony, select the model with no moderator as best alternative for both endogenous variables. Finally, an *F*-test showed non-significant increase in R^2 in the moderation model for both Off_PP, $F_{(2,421)} = 2.86$, p = 0.058, and On_PP, $F_{(2,421)} = 2.17$, p = 0.115, further validating our choice.

Similarly, we assessed the models with and without the moderation term with respect to their in-sample and out-of-sample metrics and model fit measures for T_U (Table 7.49). Notably, for both Off_PP and On_PP the model with the moderator shows better in-sample and out-of-sample predictive relevance while retaining good model fit. Also, the increase in R^2 in the model with the moderator is significant for both, Off_PP, $F_{(2,372)} = 22.345$, p < 0.001) and On_PP, $F_{(2,372)} = 78.888$, p < 0.001.

Overall, the model selection process further confirms and supports the results of the moderation analysis providing confidence in our findings. Hence, the model with the moderator variable Int_SNS_Use should be preferred to the one with no moderator for both endogenous variables Off_PP and On_PP in the Twitter users sample although the moderation is significant only for the relationships Brid_SC \rightarrow Off_PP and Brid_SC \rightarrow On_PP.

	Model Comparison of the Sample of Facebook Users with and without the Moderator Int SNS Use												
T 37	Model		In-Sample Measures					ample M	Model fit measures				
		R ²	R²adj	AICu	AICc	BIC	RMSE	MAD	Q^2	RMS _{Theta}	SRMR		
ſ_PP	No Moderation	0.254	0.240	-100.682	324.746	-73.161	0.907	0.620	0.120	0.106	0.076		
Off	Moderation	0.264	0.245	-97.452	325.249	-60.800	0.913	0.622	0.115	0.115	0.074		
ЪР	No Moderation	0.216	0.201	-79.025	346.402	-51.504	0.931	0.621	0.098	0.106	0.076		
On	Moderation	0.224	0.203	-74.208	348.493	-37.556	0.928	0.620	0.099	0.115	0.074		

Table 7.48. Model Con	aparison with and v	without the Moderator	Int_SNS	Use for F	_U (n =	= 432).
	1			-	- `	

Table 7.49. Model Comparison with and without the Moderator Int_SNS_Use for T_U (n = 383).

	Model Comparison of the Sample of Twitter Users with and without the Moderator Int SNS Use											
T X 7	Model		In-Sample Measures					ample M	Model fit measures			
LV		R ²	R ² adj	AICu	AICc	BIC	RMSE	MAD	Q^2	RMS _{Theta}	SRMR	
[PP	No Moderation	0.338	0.324	-131.910	244.574	-105.485	0.845	0.606	0.186	0.100	0.077	
Off	Moderation	0.409	0.391	-166.250	207.545	-131.065	0.814	0.586	0.226	0.107	0.065	
PP	No Moderation	0.221	0.204	-69.550	306.933	-43.126	0.915	0.681	0.142	0.100	0.077	
On	Moderation	0.453	0.437	-196.057	177.738	-160.872	0.780	0.555	0.302	0.107	0.065	

7.3. INVESTIGATING THE MODERATING EFFECT OF THE COMBINED USE OF SNSs

This Section presents results of the analysis carried out to answer research question Q3 involving testing for the significance of the moderating effect of Comb_Use_SNSs on the relationships between Bond_SC and Brid_SC with Off_PP and On_PP, by means of Permutation test following the procedure applied earlier in Section 7.1.3.1.

First, the structural model (Figure 7.10) and the measurement model (Figure 7.13) were specified. Notably, the model for $F+T_U$ with control variables is endorsed, given that it achieves higher predictive accuracy and relevance and better fit compared with the model with no control variables (Appendix P). In Section 7.2. we found that Int_SNS_Use significantly moderates the relationships between Brid_SC and Off_PP and On_PP in T_U. This implies that comparing T_U with other groups without accounting for heterogeneity in the amount of time people spend online would lead to misleading results in terms of the true moderation effect of Comb_Use_SNSs. Therefore, for a more exhaustive and comprehensive analysis, the proposed structural model was compared across different types of SNSs users accounting for different levels of intensity of use of SNSs (Figure 7.11). Yet, the analysis is performed at first across the three full samples of F+T_U, F_U and T_U (Figure 7.12). This is meant to account for the loss of information and for the biases related to the arbitrariness of the assignation of observations and selection of procedures related to the quartile split method employed for the variable "Int SNS Use" (see Section 7.1.3.1).



Figure 7.10. Structural Model Research Question Q3.

Figure 7.11. Pairwise Comparisons of Combined and Non-Combined Use of SNSs Across Different Levels of Intensity of SNSs use.



Figure 7.12. Pairwise Comparisons of Combined and Non-Combined Use of SNSs Across Different Types of SNSs used.



Groups for Comparison to Test Moderating Effect of Combined Use of SNSs										
Catagory of Haava	Type of SNS Used									
Category of Users	Facebook	Twitter	Facebook + Twitter							
Full Samples	n = 432	n = 383	n = 397							
Heavy Users	n = 108	n = 88	n = 41							
Mild Users	n = 162	n = 186	n = 78							
Light Users	n = 99	n = 95	n = 45							

Table 7.50. Sample sizes for Facebook, Twitter and, Facebook and Twittercombined users when accounting for Intensity of use of SNSs.

Overall, 2 pairwise comparisons across each category of Int_SNS_Use and the full samples are involved in the analysis. For each set of pairwise comparisons the Sidak correction procedure was applied (see Section 6.5.2) to counteract the increase of the familywise error rate. Notably, in Section 7.1.3 F_U and T_U were already compared to answer research question Q1 across all different levels of Int_SNS_Use. Therefore, when we computed the Sidak correction, we accounted for three pairwise comparisons at each level of Int_SNS_Use. Accordingly, the Sidak correction yielded an $\alpha = 0.017$. Also, consistently with the analysis carried out in Section 7.1.3.1, the Item Off_PP_3 was removed to allow for configural invariance to be established across groups to be compared. However, before removing the item a CTA-PLS was performed which supported the reflective measurement level of Off_PP for all subgroups of Facebook and Twitter combined users (heavy, mild and light).



Figure 7.13. Proposed Measurement Model for Research Question Q3.

Note: the moderator variable "Comb_Use_SNSs" is not represented in the measurement model as it used as grouping variable for the Permutation test. Also, the item Off_PP_3 is removed to allow for configural invariance and meaningful results comparison with previous analysis conducted (Section 7.1.3.1).

The values of assessment criteria of the measurement models of the Facebook and Twitter users combined sample and all its subgroups (Heavy, mild and light) are summarised below. Overall, all constructs of all samples show good internal consistency reliability (Table 7.51) and sufficient discriminant validity as all the conditions for the Fornell-Larker criterion, Cross-loadings, HTMT Ratio and related HTMT CIs are met. With respect to convergent validity all constructs of all samples have AVE > 5 but Off_PP and On_PP of the group of mild users. Table 7.52 reports a summary of the problematic constructs (interpretation guidelines were provided in Section 7.1.3.1).

	Summary Internal Consistency Reliability of F+T_U											
	EF+T_U		HF+	T_U	MF+	T_U	LF+T_U					
	CA	CR	CA	CR	CA	CR	CA	CR				
0	.930	0.940	0.911	0.925	0.920	0.931	0.910	0.924				
0	.950	0.956	0.944	0.952	0.932	0.942	0.884	0.905				
0	.906	0.924	0.933	0.945	0.840	0.875	0.904	0.929				
0	.882	0.911	0.889	0.916	0.706	0.804	0.851	0.890				
0	.900	0.930	0.908	0.936	0.905	0.934	0.844	0.895				
0	.929	0.955	0.961	0.975	0.910	0.944	0.914	0.945				

Table 7.51. Summary Internal Consistency reliability for all F+T_U Samples.

Table 7.52. Problematics Constructs with AVE < 0.5 in the Mild Facebook and</th>Twitter Combined Users Sample (n = 78).

Con	Constructs with AVE < 0.5 in the Mild Facebook and Twitter Combined Users Sample										
LV	Sub- Groups	AVE	CR	СА	F-L	C-L	нтмт	L_O_L	V_L_O_L		
Off_PP	MF+T_U	0.471	0.875	0.840	ok	ok	ok	Off_PP_7	0.518		
On_PP	MF+T_U	0.414	0.804	0.706	ok	ok	ok	On_PP_6	0.544		

Yet, following the reasoning provided in Section 7.1.3.1 to avoid problems of content validity (Little et al., 1999) and allow for configural invariance and meaningful comparisons with results obtained from previous analysis, the problematic indicators were retained as CR was above 0.6 and the problematic constructs showed adequate discriminant validity (Fornell and Larcker, 1981).

Once, the measurement models were assessed, the structural models were evaluated through the criteria reported in Appendix K. Overall, the results of the entire sample of F+T_U confirmed that there were no problems of Collinearity (all VIF < 5). The model had a moderate predictive accuracy for both Off_PP ($R^2 = 0.262$) and On_PP ($R^2 = 0.215$) and significant explanatory power for Off_PP with 95% BCa CI R^2 [0.232, 0.329], p < 0.001 and for On_PP with 95% BCa CI R^2 [0.132, 0.276], p < 0.001. The impact of Bond_SC on Off_PP ($f^2 = 0.010$) and On_PP ($f^2 = 0.030$), Brid_SC on Off_PP ($f^2 = 0.005$) could be considered very small. Among the relationships of interest that needed to be tested for moderation effect only Bond_SC \rightarrow

On_PP, $\beta = 0.170$, 95% BCa CI [0.060, 0.272] was significant at 5% significance level. However, the achieved statistical power of the test is below the recommended threshold (80%) for all other relationships. The model showed predictive relevance for both Off_PP ($Q^2 = 0.146$) and On_PP ($Q^2 = 0.126$), but the relative impact of predictive relevance was negligible for both Bond_SC and Brid_SC. Also, the model showed sufficient model fit with values of SRMR (0.072) and RMS_{theta} (0.106) below the suggested thresholds (see Appendix P for more details). CMB is not an issue as the full VIF test provided values below the recommended threshold of 3.3.

Furthermore, as mentioned earlier, since we needed to control for Int_SNS_Use, we split each sample of users into three categories, namely Heavy, Mild and Light (see Table 7.50). Overall, all the structural models of these sub-groups showed no issues of multicollinearity (VIFs < 5). Also, predictive relevance and accuracy were good, except for MF+T_U which showed a low non-significant R^2 (Table 7.53).

Summary Predictive Accuracy and Relevance Heavy, Mild and									
			Lign	<u>t F+1_U</u> Bootstrap		Т	P		
LV	Groups	\mathbf{Q}^2	\mathbb{R}^2	Mean R ²	SD	Stat	Value		
P	HF+T_U	0.351	0.602	0.689	0.073	8.210	0.000		
Ē	MF+T_U	0.187	0.498	0.563	0.071	7.007	0.000		
õ	LF+T_U	0.184	0.428	0.551	0.120	3.562	0.000		
P	HF+T_U	0.259	0.489	0.597	0.090	5.449	0.000		
	MF+T_U	0.020	0.099	0.269	0.079	1.256	0.209		
Õ	LF+T_U	0.129	0.390	0.546	0.117	3.341	0.001		

Table 7.53. Summary predictive Accuracy and Relevance for all F+T_U Samples.

As the main aim of assessing the structural models is to prepare the data for the multiple pairwise comparison analysis, Table 7.54 reports the paths coefficients values, with their f^2 effects and the 95% BCa CI to ascertain the behaviour of the variables in different group categories. Notably, the path coefficients of the relationships that need to be tested for moderation of Comb_Use_SNS differ not only in terms of their signs but also in terms of their significance (significant values highlighted in green). This indicates that the model behaves differently according to the type of SNSs considered and according to the Intensity of use of the SNSs analysed. Notably, for all the non-significant relationships the achieved statistical power is below the rule of thumb of 80% (Cohen,

1988). Therefore, we cannot be confident that significant effects do not exist for those relationships.

Deletionshine	Heavy Users Significance 5%							
Relationships	HF+T U	HF U	HT U					
Bond SC -> Off_PP	$\beta = 0.313$ $f^{4} = 0.152$ BCa CI = [-0.018, 0.517]	$\beta = -0.057$ f = 0.003 BCa CI = [-0.314, 0.168] Power (1- β) = 0.087	$\beta = 0.058$ f = 0.007 BCa CI = [-0.116, 0.201] Power (1- β) = 0.121					
Bond SC -> On_PP	$\beta = 0.048$ $f^{4} = 0.003$ BCa CI = [-0.355, 0.356] Power (1- β) = 0.063	$\beta = 0.079$ $f^{e} = 0.006$ BCa CI = [-0.13, 0.271] Power (1- β) = 0.125	$\beta = 0.07$ $f^{e} = 0.009$ BCa CI = [-0.121, 0.247] Power (1- β) = 0.142					
Brid SC -> Off_PP	$\beta = -0.270$ f = 0.104 BCa CI = [-0.510, -0.008]	$\beta = 0.397$ f = 0.117 BCa CI = [0.164, 0.629]	$\beta = 0.439$ f = 0.413 BCa CI = [0.282, 0.572]					
Brid SC -> On_PP	β = 0.459 f ⁴ = 0.234 BCa CI = [0.072, 0.799]	$\beta = 0.372$ $f^{4} = 0.116$ BCa CI = [0.154, 0.578]	β = 0.712 f ^e = 0.922 BCa CI = [0.541, 0.846]					
		C11 II	t.					
Relationships	MET U	Mild Users Significance 5%) MT U					
	MF+1 U 8 = 0.212	MF U 8 = 0.016	MI U 8 - 0 129					
Bond SC -> Off_PP	f = 0.213 f = 0.051 BCa CI = [-0.242, 0.420] Power $(1-\beta) = 0.502$	f < 0.010 f < 0.001 BCa CI = [-0.440, 0.197] Power $(1-\beta) = 0.050$	f = 0.138 f = 0.020 BCa CI = [-0.023, 0.283] Power $(1-\beta) = 0.483$					
Bond SC -> On_PP	$\beta = -0.215$ $f^{e} = 0.029$ BCa CI = [-0.531, 0.321] Power (1- β) = 0.317	β = -0.017 f ⁴ < 0.001 BCa CI = [-0.449, 0.141] Power (1-β) = 0.050	$\beta = 0.199$ $f^{4} = 0.040$ BCa CI = [0.056, 0.346]					
Brid SC -> Off_PP	β = 0.053 f ⁴ = 0.003 BCa CI = [-0.192, 0.298] Power (1-β) = 0.076	$\beta = 0.249$ $f^{e} = 0.057$ BCa CI = [0.111, 0.401]	$\beta = 0.237$ $f^2 = 0.048$ BCa CI = [0.100, 0.364]					
Brid SC -> On_PP	$\beta = 0.323$ f = 0.056 BCa CI = [-0.382, 0.622] Power (1- β) = 0.540	$\beta = 0.244$ $f^{a} = 0.050$ BCa CI = [0.120, 0.400]	$\beta = 0.289$ $f^{a} = 0.069$ BCa CI = [0.144, 0.419]					
		-14TI	L					
Relationships	I E+T II	Light Users Significance 5%						
Bond SC -> Off_PP	$\beta = -0.178$ $f^{e} = 0.046$ BCa CI = [-0.547, 0.389] Power (1- β) = 0.290	$\beta = -0.093$ $f^{e} = 0.009$ BCa CI = [-0.346, 0.203] Power (1- β) = 0.154	$\beta = 0.112$ $f^{e} = 0.014$ BCa CI = [-0.102, 0.329] Power (1- β) = 0.207					
Bond SC -> On_PP	$\beta = 0.389$ f = 0.205 BCa CI = [0.089, 0.622]	$\beta = 0.062$ f = 0.003 BCa CI = [-0.313, 0.257] Power (1- β) = 0.083	$\beta = 0.235$ f = 0.055 BCa CI = [-0.042, 0.472] Power (1- β) = 0.618					
Brid SC -> Off_PP	$\beta = -0.036$ $f^{2} = 0.002$ BCa CI = [-0.418, 0.434]	$\beta = 0.100$ $f^{4} = 0.010$ BCa CI = [-0.204, 0.369]	$\beta = 0.041$ f' = 0.002 BCa CI = [-0.213, 0.238]					

Table 7.54. Path Coefficients and their Significance Across Levels of Intensity ofSNSs Use per Type of SNSs used.

Before carrying out any group comparison, partial measurement invariance (Appendix O) was established following the same procedure employed in Section 7.1.3.1. Then, permutation analysis for all sets of pairwise comparisons as showed in Figure 7.11 and Figure 7.12 were performed at a 0.017 significant level. Table 7.55 and Table 7.56 show the output of the pairwise permutation tests between $F+T_U$ and F_U and T_U respectively for the relationships of interest.

Table 7.55. Permutation Test Results for Significant Differences between $F+T_U$ (n = 397) and F_U (n = 432).

Permutation Test F+T U ($n = 397$) Vs F U ($n = 432$) Output								
	Std. β	Std. β	Std. β	Std. β Perm.				
	[F+T_U]	[F_U]	(d)	Mean (d)	0.85%	99.15%	Р	
Bond_SC -> Off_PP	0.093	-0.081	0.174	0.001	-0.171	0.176	0.029	
Bond_SC -> On_PP	0.170	0.027	0.143	0.000	-0.179	0.159	0.043	
Brid_SC -> Off_PP	0.058	0.311	-0.253	0.004	-0.195	0.186	0.001	
Brid_SC -> On_PP	0.070	0.284	-0.214	0.001	-0.177	0.194	0.006	

Table 7.56. Permutation Test Results for Significant Differences between $F+T_U$ (n = 397) and T_U (n = 383).

Permutation Test F+T_U (n = 397) Vs T_U (n = 383) Output								
	Std. β	Std. β	Std. β	Std. β Perm.				
	[F+T_U]	[F_U]	(d)	Mean (d)	0.85%	99.15%	Р	
Bond_SC -> Off_PP	0.093	0.150	-0.057	-0.003	-0.177	0.163	0.407	
Bond_SC -> On_PP	0.170	0.230	-0.060	0.001	-0.170	0.162	0.407	
Brid_SC -> Off_PP	0.058	0.180	-0.122	0.000	-0.152	0.154	0.070	
Brid_SC -> On_PP	0.070	0.296	-0.227	0.002	-0.164	0.168	0.002	

The cells representing significant difference are highlighted in green in both tables. Notably, Comb_Use_SNSs significantly and negatively moderate the relationships Brid_SC \rightarrow On_PP when F+T_U is compared to both F_U, $d_{(\beta i-\beta j)} = -0.214$, 98.3% CI [-0.177, 0.194], p = 0.006 and T_U, $d_{(\beta i-\beta j)} = -0.227$, 98.3% CI [-0.164, 0.168], p = 0.002 at the 1.7% level of significance. However, a significant moderating effect of Comb_Use_SNSs was found also on the relationship Brid_SC \rightarrow Off_PP but only in the comparison between F+T_U and F_U, $d_{(\beta i-\beta j)} = -0.253$, 98.3% CI [-0.195, 0.186], p = 0.001.
Table 7.57 summarises the results of the permutation tests conducted across combined and not combined users of SNSs at different levels of intensity of use (significant results are highlighted in green).

Table 7.57. Summary Pairwise Permutation Tests Across Different Levels of
Intensity of Use.

Summary Pairwise Permutation Test Across Different Levels of Intensity of Use								
Relationships			Pair	wise Co	mparisor	15		
	I	IF+T_U	Vs HF_U]	HF+T_U	Vs HT_U	
				Sig.				Sig.
	β(d)	0.85%	99.15%	1.7%	β(d)	0.85%	99.15%	1.7%
Bond_SC -> Off_PP	0.369	-0.616	0.494	NO	0.254	-0.375	0.390	NO
Bond_SC -> On_PP	-0.031	-0.475	0.462	NO	-0.022	-0.404	0.393	NO
Brid_SC -> Off_PP	-0.668	-0.545	0.657	YES	-0.709	-0.437	0.369	YES
Brid_SC -> On_PP	0.087	-0.524	0.532	NO	-0.253	-0.399	0.344	NO
	N	IF+T_U`	Vs MF_U		1	MF+T_U	Vs MT_U	
				Sig.				Sig.
	β(d)	0.85%	99.15%	1.7%	β(d)	0.85%	99.15%	1.7%
Bond_SC -> Off_PP	0.229	-0.495	0.384	NO	0.075	-0.306	0.349	NO
Bond_SC -> On_PP	-0.198	-0.497	0.418	NO	-0.413	-0.364	0.379	YES
Brid_SC -> Off_PP	-0.196	-0.334	0.275	NO	-0.185	-0.358	0.308	NO
Brid_SC -> On_PP	0.079	-0.348	0.355	NO	0.034	-0.365	0.331	NO
]	LF+T_U`	Vs LF_U			LF+T_U	Vs LT_U	
				Sig.				Sig.
	β(d)	0.85%	99.15%	1.7%	β(d)	0.85%	99.15%	1.7%
Bond_SC -> Off_PP	-0.085	-0.432	0.470	NO	-0.290	-0.461	0.394	NO
Bond_SC -> On_PP	0.327	-0.544	0.441	NO	0.154	-0.516	0.439	NO
Brid_SC -> Off_PP	-0.135	-0.585	0.553	NO	-0.076	-0.439	0.432	NO
Brid_SC -> On_PP	0.038	-0.525	0.365	NO	0.093	-0.507	0.498	NO

The pairwise Permutation tests output shows that when controlling for Int_SNS_Use, Comb_Use_SNSs moderates only the relationship Brid_SC \rightarrow Off_PP when F+T_U is compared to both F_U, $d_{(\beta_i - \beta_j)} = -0.668$, 98.3% CI [-0.545, 0.657], p = 0.008 and T_U, $d_{(\beta_i - \beta_j)} = -0.709$, 98.3% CI [-0.437, 0.369], p < 0.001 in the heavy category. Also, MF+T_U showed a significantly lower path coefficient of the relationship Bond_SC \rightarrow On_PP when compared to T_U, $d_{(\beta_i - \beta_j)} = -0.413$, 98.3% CI [-0.364, 0.379], p = 0.010.

However, given that the Permutation test can yield misleading results when samples sizes have large differences (Hair et al., 2018, p. 154) like in the case of this research (Table 7.50), a sensitivity analysis was conducted to assess the robustness of the Permutation tests' results and examine the extent to which the outcomes of the analysis are affected by any limitations of the statistical method employed (Field, 2013). In particular, Hair et al. (2018) suggest using the PLS-MGA test when groups sizes are very different. Moreover, also the Non-parametric Confidence Set Approach proposed by Sarstedt et al. (2011) was employed as it was shown to be more conservative than the PLS-MGA test and less prone to Type II error. For both procedures, the Sidak correction was applied and the analysis was carried out at the 1.7% level of significance.

Table 7.58 reports the results of the PLS-MGA analysis conducted using the setting suggested by Hair et al. (2018) indicating a maximum of 300 iterations, using the path weighting scheme, a stop criterion of $1*10^{-7}$ with a number of 5000 bootstrapped samples. Only the results of the variables of interest necessary to answer research question Q3 are reported here. The first column, $\beta(d)$, in Table 7.70 indicates the difference in the path coefficients of the group analysed. The second column shows the *p* values of the PLS-MGA analysis. As the PLS-MGA is a one-tailed test, these *p* values indicate whether the path coefficient is significantly larger in the first group than in the second one. Hence, the third column (1-*p* value) shows if there is a significant difference in the other direction (Hair et al., 2018). Lastly, the fourth column indicates whether the difference is significant at the 1.7% level of significance (results highlighted in green). Notably, only the path coefficients of the relationship Brid_SC \rightarrow Off_PP differ across the heavy category of F+T_U, F_U and T_U.

Summary Pairwise PLS-MGA Comparisons Across Different Levels of Intensity of Use								
Relationships		HF+T_U	U Vs HF_U			HF+T_U	Vs HT_U	
	β(d)	p value	1-p value	Sig. 1.7%	β(d)	p value	1-p value	Sig. 1.7%
Bond_SC -> Off_PP	0.369	0.020	0.980	NO	0.254	0.043	0.957	NO
Bond_SC -> On_PP	-0.031	0.551	0.449	NO	-0.022	0.533	0.467	NO
Brid_SC -> Off_PP	-0.668	1.000	0.000	YES	-0.709	1.000	0.000	YES
Brid_SC -> On_PP	0.087	0.329	0.671	NO	-0.253	0.908	0.092	NO
	-							
Relationships		MF+T_U	U Vs MF_U		MF+T_U Vs MT_U			
	β(d)	p value	1-p value	Sig. 1.7%	β(d)	p value	1-p value	Sig. 1.7%
Bond_SC -> Off_PP	0.229	0.115	0.885	NO	0.075	0.960	0.040	NO
Bond_SC -> On_PP	-0.198	0.802	0.198	NO	-0.413	0.908	0.092	NO
Brid_SC -> Off_PP	-0.196	0.915	0.085	NO	-0.185	0.383	0.617	NO
Brid_SC -> On_PP	0.079	0.318	0.682	NO	0.034	0.044	0.956	NO
Relationships		LF+T_U	U Vs LF_U			LF+T_U	Vs LT_U	
	β(d)	p value	1-p value	Sig. 1.7%	β(d)	p value	1-p value	Sig. 1.7%
Bond_SC -> Off_PP	-0.085	0.647	0.353	NO	-0.290	-0.461	0.394	NO
Bond_SC -> On_PP	0.327	0.043	0.957	NO	0.154	-0.516	0.439	NO
Brid_SC -> Off_PP	-0.135	0.709	0.291	NO	-0.076	-0.439	0.432	NO
Brid_SC -> On_PP	0.038	0.418	0.582	NO	0.093	-0.507	0.498	NO

Table 7.58. Summary Pairwise PLS-MGA Comparisons for the Variable ofInterest Across Different Levels of Intensity of Use.

Table 7.59 presents the results of the non-parametric Confidence Set Approach proposed by Sarstedt et al. (2011) which involves constructing the BCa CI of the path estimates of the two groups to be compared and check whether the path coefficients of the first group fall within the corresponding BCa CI of the second group and vice versa. For two path coefficients to be statistically different, both should not fall within the corresponding BCa CI of the other group. Table 7.59 reports the results of the comparison between the samples of HF+T_U and HF_U. The BCa CI are obtained running the PLS algorithm indicating a maximum of 300 iterations, using the path weighting scheme, a stop criterion of: $1*10^{-7}$ with a number of 5000 bootstrapped samples (Hair et al., 2017). Notably, only the relationship Brid_SC \rightarrow Off_PP is significantly different in the two groups compared, $\beta_{(HF+T_U)} = -0.270 \notin 98.3\%$ BCa CI_(HF_U) [0.104, 0.669], and $\beta_{(HF_U)} =$ -0.397 \notin 98.3% BCa CI_(HF+T_U) [-0.590, 0.059].

Table 7.59. Non-Parametric Confidence Set Approach for the Comparison
between the Samples of Facebook and Twitter Combined Heavy Users $(n = 41)$
and Facebook Heavy Users (n = 108).

Non - Parametric Confidence Set Approach										
		BCA CI				BCA	CI			
		98.3% of				98.3 9	% of			
		HF_U				HF+	T_U			
	β	ТР	ΠD	Sign.	β	ТР	ΠD	Sign.		
Relationships	(HF+T_U)	L-D	0-В	1.7%	(HF_U)	L-D	0-В	1.7%		
Bond_SC -> Off_PP	0.313	-0.369	0.233	YES	-0.057	-0.083	0.581	NO		
Bond_SC -> On_PP	0.048	-0.185	0.323	NO	0.079	-0.439	0.409	NO		
Brid_SC -> Off_PP	-0.270	0.104	0.669	YES	0.397	-0.590	0.059	YES		
Brid_SC -> On_PP	0.459	0.083	0.633	NO	0.372	-0.071	0.878	NO		

We conducted all the necessary pairwise comparisons also for T_U and we found that the Confidence Set Approach confirmed the results of the Permutation tests carried out earlier.

Finally, Table 7.60 summarises the results of the sensitivity analysis. Notably, only the relationship Brid_SC \rightarrow Off_PP in the heavy category of users has been found to be significantly different across all the groups of different SNSs users in the heavy category. Rather, the path coefficients of the relationship Bond_SC \rightarrow Off_PP were found to differ between the mild users of Facebook and Twitter combined and the mild users of Twitter only in the Permutation test. Yet, provided that the Permutation test may not yield reliable results when samples have large differences like in this case, we accepted the results of the PLS-MGA and of the non-parametric Confidence Set Approach and accept the non-significant difference of the path coefficients of the relationship Bond_SC \rightarrow Off_PP. The insight of the sensitivity analysis provided additional confidence in the final results obtained.

PLS Multigroup Pairwise Comparisons Results Across Different Statistical Methods							
	HF+T UVs HF U			HF+T UVsHT U			
		PLS-	Confidence		PLS-	Confidence	
Relationship	Permutation	MGA	Set	Permutation	MGA	Set	
Bond_SC -> Off_PP							
Bond_SC -> On_PP							
Brid_SC -> Off_PP	X	Х	X	Х	X	X	
Brid_SC -> On_PP							
	MF+7	ſ_U Vs I	MF_U	MF+T	_U Vs M	IT_U	
		PLS-	Confidence		PLS-	Confidence	
Relationship	Permutation	MGA	Set	Permutation	MGA	Set	
Bond_SC -> Off_PP							
Bond_SC -> On_PP				Х			
Brid_SC -> Off_PP							
Brid_SC -> On_PP							
	LF+7	[_U Vs I	LF_U	LF+T_U Vs LT_U			
		PLS-	Confidence		PLS-	Confidence	
Relationship	Permutation	MGA	Set	Permutation	MGA	Set	
Bond_SC -> Off_PP							
Bond_SC -> On_PP							
Brid_SC -> Off_PP							
Brid_SC -> On_PP							
Note: X indicates signi	ficant differenc	es of pat	h coefficients a	icross groups at	1.7% le	vel of	
significance							

Table 7.60. Sensitivity Analysis of the Multigroup Pairwise Comparisons ResultsAcross Different Statistical Methods of Analysis.

7.4. ASSESSING RESULTS OF TWO-WAY FACTORIAL ANOVA TO ANSWER RESEARCH QUESTION Q4.

This section presents the results of the analysis related to research question Q4, namely whether there are any significant differences in the amount of Bond_SC and Brid_SC displayed by F_U and T_U groups, according to Int_SNSs_Use. The aim is to test for what is called the "interaction effect" of "Type_SNSs_Used" and "Int_SNSs_Use (Field, 2013). To this purpose, two separate two-way factorial ANOVAs with a 2x3 design are carried out for Bond_SC and Brid_SC. (reasons and justifications were discussed in Chapter 6, Section 6.5.2). The next section analyses the assumptions that must be met in order to carry out the two-way factorial ANOVA (Section 7.4.1). Successively, the output of the analysis will be provided for Bond_SC (Section 7.4.2) and Brid_SC (Section 7.4.3).

7.4.1. MEETING THE TWO-WAY FACTORIAL ANOVA ASSUMPTIONS.

To carry out the two-way ANOVA a total of 6 assumptions must be met as discussed in Section 6.5.2. All the assumptions except homogeneity of variance for Brid_SC were met. Indeed:

- The dependent variables Bond_SC and Brid_SC are measured at the interval level using the unstandardized composites scores obtained from the analysis carried out in PLS-SEM (*Assumption 1*).
- Both independent variables (Type_SNS_Used and Int_SNS_Use) are categorical (*Assumption 2*).
- Moreover, the groups analysed are independent (Assumption 3).
- Furthermore, an inspection of the data through boxplots for each combination of the independent variables (Figure 7.14 and Figure 7.15) reveals that the data are free from outliers (*Assumption 4*).
- Bond_SC and Brid_SC are approximately normally distributed for each combination of the groups of the two independent variables (*Assumption 5*). Indeed, all the values of Skewness and Kurtosis fall within the acceptable ranges

of ± 1 for Skewness (Field, 2013) and ± 2 for Kurtosis George and Mallery (2020) (Table 7.61).

Figure 7.14. Outliers Investigation for Bonding Social Capital for each combination of Type of SNS Used and Intensity of Use of the SNSs considered.



Figure 7.15. Outliers Investigation for Bridging Social Capital for each combination of Type of SNS Used and Intensity of Use of the SNSs considered.



Normality Statistics Summary								
Type_SNS_Use	Int SNS Use	Variables	Skewness	Kurtosis				
		Bond_SC	0.692	-0.489				
	Light (n = 99)	SE	0.243	0.481				
		Brid SC	0.654	-0.875				
		SE	0.243	0.481				
		Bond_SC	0.084	-0.989				
Facabook	M(14) (n = 162)	SE	0.191	0.379				
racebook	Milia (n - 102)	Brid_SC	-0.354	-0.527				
		SE	0.191	0.379				
		Bond_SC	-0.131	-1.054				
	$H_{0,0,m}$ (n = 100)	SE	0.233	0.461				
	Heavy (11 – 108)	Brid_SC	-0.172	-0.486				
		SE	0.233	0.461				
		Bond_SC	0.704	-0.317				
	$\mathbf{I}_{iabt}(n=05)$	SE	0.247	0.490				
	Light (n – 95)	Brid_SC	0.314	-1.011				
		SE	0.247	0.490				
		Bond_SC	0.741	-0.520				
Twitton	M(14) = 106	SE	0.178	0.355				
Iwitter	Milia (li – 180)	Brid_SC	0.244	-0.764				
		SE	0.178	0.355				
		Bond_SC	-0.047	-0.893				
	$\mathbf{H}_{0,0,1,1}$ $(n - 90)$	SE	0.257	0.508				
	meavy (n – 88)	Brid_SC	0.317	-0.894				
		SE	0.257	0.508				

Table 7.61. Skewness and Kurtosis of Groups Split According to Type of SNSsUsed and Intensity of Use.

• Finally, we tested for the homogeneity of variances (*Assumption 6*) of Bond_SC and Brid_SC for each combination of the groups of the two independent variables through the Levene's test based on both the mean and the median. This is because the Levene's test based on the means is considered too sensitive especially when sample sizes are large like in the case of this study (Field, 2013). Therefore, George and Mallery (2020) suggest assessing the test based on the median which is less sensitive to sample size. The results of the Levene's tests for homogeneity of variances of Bond_SC and Brid_SC are reported in Table 7.62.

Levene's Test of Equality of Error Variances ^a									
		Levene Statistic	dfl	df2	Sig.				
Bond_SC	Based on Mean	0.356	5	732	0.879				
	Based on Median	0.416	5	732	0.838				
Brid_SC	Based on Mean	5.366	5	732	0.000				
	Based on Median	5.472	5	732	0.000				
a. Design: Intercept + Type_SNS_Use + Int_SNS_Use +									
Type SNS U	Use * Int SNS Use.		_						

Table 7.62. Levene's Test for Homogeneity of Variances of Online Bonding andBridging Social Capital for Type of SNSs Used Across different Levels of Intensity

of Use.

Notably, the assumption of homogeneity of variance is met for Bond_SC ($F_{(5,732)}$ = 0.416, p = 0.838) but not for Brid_SC ($F_{(5,732)} = 5.472$, p < 0.001). However, when the sample size is large (more than 30 respondents per group according to Field, 2013), small differences in group variances can produce a Levene's test that is significant (Field, 2013). Hence, any decisions regarding the assumption of homogeneity of variances based only on the Lavene's test will be confounded by the sample size. Indeed, even Gearge and Mallery (2020) argued that "measures of homogeneity of variance act more as a warning than as a disqualifier" (p. 166) provided that when sample sizes are big the twoway ANOVA is robust to heteroskedasticity especially when the dependent variables have met the assumption of normality like in the case of this study and when sample sizes of the groups to be compared are equal. However, Games and Howell (1976) showed that when the samples sizes are unequal (like in the case of this study see Table 7.63) and heteroskedasticity is confirmed, the p-value yielded by the ANOVA tends to be conservative when the larger variance is associated with the larger sample size. This would increase the type II error. Conversely, Field (2013) argued that when the groups with larger sample sizes have smaller variances than the groups with smaller samples sizes, the resulting F-ratio tends to be liberal increasing the Type I error rate. However, the heteroskedasticity does not affect the ANOVA analysis when variances are proportional to the means (i.e., they increase in unison) even when sample sizes are different. Consistent with the above rationale, we checked the variances for each group of SNSs users involved in the analysis sorting the samples from the smallest to the largest (Table 7.63). Notably, the MT_U sample is not only the largest (n = 186) but has also the largest variance ($\sigma^2 = 2.452$). Hence, if anything, when interpreting results we must be cautious and mention that the two-way ANOVA would report a conservative p-value and will probably reach a lower statistical power compared to an analysis with equal sample sizes.

Table 7.63. Means and Related Variances of Online Bridging Social Capital	for
each Combination of the Two Independent Variables.	

Means and Variances of Online Bridging Social Capital for all SNSs Users Samples							
Groups	M	σ^2	Ν				
HT_U	3.149	1.959	88				
LT U	2.759	1.636	95				
LF_U	2.120	1.103	99				
HF_U	4.085	2.147	108				
MF U	3.491	1.371	162				
MT U	3.598	2.452	186				

Following Field's (2013) guidelines for when heteroscedasticity may be present, we conducted a sensitivity analysis by (1) running the HC4 robust tests to heteroskedasticity and by (2) bootstrapping the post-hoc tests of the simple effects.

The next two sections provide the relevant output of the two-way ANOVA for Bond_SC and Brid_SC, respectively.

7.4.2. TWO-WAY ANOVA FOR ONLINE BONDING SOCIAL CAPITAL

This section reports the results of the two-way ANOVA performed on Bond_SC. The test is run at 5% level of significance. Table 7.64 shows descriptive statistics for each group that need to be compared and Table 7.65 reports a summary of the two-way ANOVA metrics. Following Field (2018), two effect size were provided, namely the partial eta squared (η_p^2) and the omega squared (ω^2). This is because η_p^2 is more liberal than ω^2 in rendering effect size. Yet, researchers still report η_p^2 for the purposes of comparison with similar studies. The ω^2 effect size is not provided by SPSS and has been computed in excel following the formula provided by Field (2013) below:

$$\omega_{\text{effect}}^2 = \frac{\hat{\sigma}_{\text{effect}}^2}{\hat{\sigma}_{\text{total}}^2}$$
 Equation 7.1

where $\hat{\sigma}^{2}_{effect}$ is the variance component of the effect for which we want to compute the omega squared and $\hat{\sigma}^{2}_{total}$ is the sum of all the variance components of the variables included in the analysis (Field, 2013: p. 537). Field (2013) suggests that values of 0.01, 0.06 and 0.14 represent small, medium and large effects, respectively, for both effect sizes.

Descriptive Statistics Bond_SC									
Type_SNS_Use	Int_SNS_Use	M	SD	Ν					
F_U	Light	2.779	1.568	99					
	Mild	3.591	1.555	162					
	Heavy	3.905	1.689	108					
	Total	3.465	1.653	369					
T_U	Light	2.787	1.543	95					
	Mild	2.649	1.537	186					
	Heavy	3.892	1.514	88					
	Total	2.981	1.613	369					
Total	Light	2.783	1.552	194					
	Mild	3.088	1.613	348					
	Heavy	3.899	1.609	196					
	Total	3.223	1.650	738					

Table 7.64. Descriptive Statistics for Online Bonding Social Capital Scores byType of SNSs Used and Intensity of Use of SNSs.

Table 7.65. Summary of Two-Way ANOVA for Online Bonding Social Capital.

Tests of Between-Subjects Effects									
Source	Type III SS	df	MS	F	Sig ^b .	η_P^2	ω^2		
Corrected Model	210.207ª	5	42.041	17.137	0.000	0.105	0.099		
Intercept	7278.023	1	7278.023	2966.650	0.000	0.802	0.801		
Type_SNS_Use	16.986	1	16.986	6.924	0.009	0.009	0.008		
Int_SNS_Use	129.508	2	64.754	26.395	0.000	0.067	0.065		
Type_SNS_Use* Int_SNS_Use	40.282	2	20.141	8.210	0.000	0.022	0.019		
Error	1795.801	732	2.453						
Total	9672.306	738							
Corrected Total	2006.007	737							
a. R Squared = .105 (Adjusted R Squared = .099)									
b. Computed using a	alpha = .05								

Notably, the interaction effect of type of SNSs used by the intensity of use of the SNS on Online Bond_SC is significant with $F_{(2,732)} = 8.21$, p < .001 and has a small effect size ($\eta_p^2 = .022$, $\omega^2 = .019$). This means that the effect of the intensity of use of SNSs on Online Bond_SC changes when considering different types of SNSs such as Facebook and Twitter and vice versa.

When the interaction effect is significant, Field (2018) advises against interpreting the main effects (p. 814). Rather an analysis of the simple effects must be conducted to break down the interaction effect that is, researchers must look at the effect of one independent variable at all different levels of the second independent variable and vice versa. This would involve conducting a series of t-tests between Facebook and Twitter users for each level of intensity of use (light, mild and heavy) and a series of one-way ANOVA comparing light, mild and heavy users across Facebook and Twitter. The simple effect analysis was carried out in SPSS through the ANOVA framework by modifying the syntax as suggested by Field (2013: p. 531).

Table 7.66 provides the univariate test for differences for Bond_SC between Facebook and Twitter users across each level of intensity of SNS use. Notably, only for the mild category of intensity of use there is a significant difference between Facebook and Twitter users, $F_{(1,732)} = 31.266$, p < 0.001, and this effect is small r = 0.202. Field (2013) suggests reporting *r* as an indication of the effect size for the simple effect analysis when this is conducted across two groups (Facebook and Twitter) calculated through the following formula:

$$\sqrt{\frac{F(df_M, df_R)}{F(1, df_R) + df_R}}$$
 Equation 2

where $df_{\rm M}$ is the degree of freedom of the model and $df_{\rm R}$ is the degree of freedom of the residuals. Following Cohen (1988), values between 0.1 and 0.3, between 0.3 and 0.5 and above 0.5 represent small, medium and large effect size respectively.

	Univariate Tests								
Int_SN	S_Use	<i>SS</i>	df	MS	F	Sig.	η_P^2	r	Power ^a
Light	Contrast	0.003	1	0.003	0.001	0.974	0.000	0.001	0.05
	Error	1795.801	732	2.453					
Mild	Contrast	76.703	1	76.703	31.266	0.000	0.041	0.202	
	Error	1795.801	732	2.453					
Heavy	Contrast	0.009	1	0.009	0.003	0.953	0.000	0.002	0.05
	Error	1795.801	732	2.453					

Table 7.66. Univariate Test for Differences in Online Bonding Social CapitalBetween Facebook and Twitter Users Across Each Level of Intensity of SNS Use.

Each F tests the simple effects of Type_SNS_Use within each level combination of the other effects shown. These tests are based on the linearly independent pairwise comparisons among the estimated marginal means.

a. Computed using alpha = .05

For light and heavy intensity of use, the differences in mean values of Online Bond_SC between Facebook and Twitter users are not significant and the achieved statistical power of the tests for either level of intensity of use was very low (5%). Analysis of the differences in mean values and their significance together with their 95% CI provides an insight of these results (Table 7.67).

Table 7.67. Pairwise Comparisons of Facebook and Twitter Users Across EachLevel of Intensity of Use of SNSs.

			Pairwise Con	iparisons			
						95%	CI for
			Mean			Diffe	rence
			Difference			Lower	Upper
Int_SNS_Use	Type_S	NS_Use	(I-J)	SE	Sig.ª	Bound	Bound
Light	F_U	T_U	-0.007	0.225	0.974	-0.449	0.434
Mild	F_U	T_U	.941*	0.168	0.000	0.611	1.272
Heavy	F_U	T_U	0.013	0.225	0.953	-0.428	0.455
Based on estima	ated margin	nal means					
*The mean diffe	erence is si	gnificant	at the .05 level.				

a. Adjustment for multiple comparisons: Least Significant Difference (equivalent to no

adjustments).

Notably, only in the mild category we have a significant difference. Indeed, mild Facebook Users show, on average, higher levels of Bond_SC (M = 3.59, SD = 1.55)

compared to mild Twitter users (M = 2.65, SD = 1.54) and this difference, $d_{(Mi-Mj)} = 0.94$, 95% CI [0.611, 1.272] was statistically significant (p < 0.001).

Similarly, the simple effect analysis of the intensity of use of SNSs across Facebook and Twitter was carried out (Table 7.68). Notably, there are significant differences between light, mild and heavy users of both Facebook, $F_{(2,732)} = 14.266$, p < 0.001, $\eta_p^2 = 0.038$, $\omega^2 = 0.048$, and Twitter, $F_{(2,732)} = 14.266$, p < 0.001, $\eta_p^2 = 0.051$, $\omega^2 = 0.048$.

	Univariate Tests							
Type_SNS_Use		SS	df	MS	F	Sig.	η_P^2	ω^2
ЕU	Contrast	69.998	2	34.999	14.266	0.000	0.038	0.035
F_U	Error	1795.801	732	2.453				
TU	Contrast	97.005	2	48.502	19.770	0.000	0.051	0.048
1_0	Error	1795.801	732	2.453				
			'					

Table 7.68. Univariate Test for Differences in Online Bonding Social CapitalBetween Light, Mild and Heavy Users Across Each SNS.

Each F tests the simple effects of Int_SNS_Use within each level combination of the other effects shown. These tests are based on the linearly independent pairwise comparisons among the estimated marginal means. a. Computed using alpha = .05

To ascertain which group (light, mild or heavy users) is different from the others, in both SNSs we run pairwise comparisons with level of significance adjusted through the Sidak correction to control for the familywise error rate (Table 7.69).

The results in Table 7.69 suggest that the intensity of use of a specific SNS affects levels of Bond_SC in both Facebook and Twitter but at different levels of intensity of use. Specifically, the level of Bond_SC is similar for Mild (M = 3.59, SD = 1.55) and Heavy (M = 3.90, SD = 1.69) users of Facebook with a difference $d_{(Mi-Mj)} = -0.314$, 95% CI [-0.780, 0.151], p = .287. However, light Facebook users (M = 2.78, SD = 1.57) show significantly lower levels of Bond_SC when compared to both mild and heavy Facebook users with differences in means respectively of $d_{(Mi-Mj)} = -0.81$, 95% CI [-1.290, -0.333] p < 0.001 and $d_{(Mi-Mj)} = -1.13$, 95% CI [-1.647, -0.604], p < 0.001.

			Pairwise Comj	parisons			
			Mean		95% (Differ	CI for rence ^b	
Type_SNS_Use	Int_SP	NS_Use	Difference (I-J)	SE	Sig. ^b	Lower Bound	Upper Bound
	Light	Mild	-0.811*	0.200	0.000	-1.290	-0.333
		Heavy	-1.126*	0.218	0.000	-1.647	-0.604
F 11	Mild	Light	0.811*	0.200	0.000	0.333	1.290
r_0		Heavy	-0.314	0.195	0.287	-0.780	0.151
	Heavy	Light	1.126*	0.218	0.000	0.604	1.647
		Mild	0.314	0.195	0.287	-0.151	0.780
	Light	Mild	0.137	0.198	0.865	-0.335	0.610
		Heavy	-1.105*	0.232	0.000	-1.659	-0.550
T II	Mild	Light	-0.137	0.198	0.865	-0.610	0.335
1_0		Heavy	-1.242*	0.203	0.000	-1.727	-0.757
	Heavy	Light	1.105*	0.232	0.000	0.550	1.659
		Mild	1.242*	0.203	0.000	0.757	1.727

Table 7.69. Pairwise Comparisons Across Each Level of Intensity of Use of SNSsby type of SNSs used.

Based on estimated marginal means

*The mean difference is significant at the .05 level.

b. Adjustment for multiple comparisons: Sidak.

In contrast, Twitter heavy users (M = 3.89, SD = 1.51) showed significantly higher levels of Boind_SC when compared to both mild (M = 2.65, SD = 1.54) and light (M = 2.79, SD = 1.54) Twitter users with differences in means respectively of $d_{(Mi-Mj)} =$ 1.24, 95% CI 0.757, 1.727], p < 0.001 and $d_{(Mi-Mj)} = 1.10$, 95% CI [0.550, 1.659], p <0.001. However, light and mild users do not significantly differ in levels of Bond_SC with a difference of $d_{(Mi-Mj)} = 0.137$, 95% CI [-0.335, 610], p = 0.865.

The interpretation of the interaction effect can be further facilitated by the graphical inspection of the plot of the estimated marginal means of all the groups of SNSs users (Figure 7.16).



Figure 7.16. Plot of the Estimated Marginal Means for the Interaction Effect of Type of SNSs used and Intensity of use of SNSs on Online Bonding Social Capital.

Indeed, the graph shows that the Bond_SC is similar for both Facebook and Twitter users at light and heavy levels of intensity of use, but we can speculate that there is a difference at the mild level. Also, light users of Facebook seem to differ from mild and heavy users of Facebook as the 95% means error bars do not overlap. Yet, the same is probably not true for mild and heavy users of Facebook as their 95% error bars do overlap. For twitter users instead, only heavy users appear to significantly differ from the light and mild users. Hence, the analysis of the plot of the estimated marginal means of Bond_SC confirms the simple main effect analysis results provided in Table 7.67 and Table 7.69. Therefore, the effect for intensity of SNS use interacts with the type of SNS used, that is, the intensity of use of an SNS (or simply put the time spent online) affects Facebook users differently than Twitter users in levels of Bond_SC.

7.4.3. TWO-WAY ANOVA FOR ONLINE BRIDGING SOCIAL CAPITAL

In the following paragraphs we report the two-way ANOVA results to test for the combined effect of the type of SNSs used and their intensity of use on Brid_SC. As discussed in Section 7.4.1, given that the homogeneity of variances was not

unequivocally established across the groups, we employed the H4 robust tests to heteroscedasticity to estimate standard errors and p values for the parameter estimates and bootstrapped the pairwise comparisons of the simple effects. Table 7.70 reports descriptive statistics of the Online Brid_SC variable for each group. Table 7.71 reports a summary of the two-way ANOVA metrics.

Desci	Descriptive Statistics Brid_SC									
Type SNS Use	Int SNS Use	М	SD	Ν						
F_U	Light	2.120	1.050	99						
	Mild	3.491	1.171	162						
	Heavy	4.085	1.465	108						
	Total	3.297	1.446	369						
T_U	Light	2.759	1.279	95						
	Mild	3.598	1.566	186						
	Heavy	3.149	1.400	88						
	Total	3.275	1.496	369						
Total	Light	2.433	1.208	194						
	Mild	3.549	1.395	348						
	Heavy	3.665	1.507	196						
	Total	3.286	1.470	738						

Table 7.70. Descriptive Statistics for Online Bridging Social Capital Scores byType of SNSs Used and Intensity of Use of SNSs.

Table 7.71. Summary of Two-Way ANOVA for Online Bridging Social Capital.

	Tests o	f Betw	een-Subject	s Effects			
Source	Type III SS	df	MS	F	Sig ^b .	η_P^2	ω^2
Corrected Model	256.606ª	5	51.321	28.113	0.000	0.161	0.155
Intercept	6983.437	1	6983.437	3825.411	0.000	0.839	0.829
Type_SNS_Use	0.688	1	0.688	0.377	0.539	0.001	0.000
Int_SNS_Use	183.258	2	91.629	50.193	0.000	0.121	0.118
Type_SNS_Use* Int_SNS_Use	63.178	2	31.589	17.304	0.000	0.045	0.042
Error	1336.295	732	1.826				
Total	9561.944	738					
Corrected Total	1592.901	737					
a. R Squared = .105 (Adjusted R Squared = .099)							
b. Computed using a	alpha = .05						

Notably, the interaction effect of type of SNSs used by the intensity of use of the SNS on Brid_SC is significant with $F_{(2,732)} = 17.304$, p < 0.001 and represents between a small and a medium effect size, $\eta_p^2 = 0.045$ and $\omega^2 = 0.042$. This means that the effect

of the intensity of use of SNSs on Brid_SC changes according to the different types of SNSs considered (Facebook and Twitter). To break down the interaction effect an analysis of the simple effects was conducted following the same rationale of the previous Section.

Table 7.72 provides the univariate test results for differences in Brid_SC between Facebook and Twitter users across each level of intensity of SNS use and Table 7.73 provides the 95% BCa bootstrapped CI of means differences as results of a sensitivity analysis robust to heteroskedasticity.

Table 7.72. Univariate Test for Differences in Online Bridging Social CapitalBetween Facebook and Twitter Users Across Each Level of Intensity of SNS Use.

Univariate Tests									
Int_SN	S_Use	55	df	MS	F	Sig.	$\eta_{P}{}^{2}$	r	Power ^a
Light	Contrast	19.809	1	19.809	10.851	0.001	0.015	0.121	
	Error	1336.295	732	1.826					
Mild	Contrast	0.985	1	0.985	0.540	0.463	0.001	0.027	0.114
	Error	1336.295	732	1.826					
Heavy	Contrast	42.531	1	42.531	23.298	0.000	0.031	0.176	
	Error	1336.295	732	1.826					

Each F tests the simple effects of Type_SNS_Use within each level combination of the other effects shown. These tests are based on the linearly independent pairwise comparisons among the estimated marginal means.

a. Computed using alpha = .05

Table 7.73. Pairwise Comparisons of Facebook and Twitter Users Across EachLevel of Intensity of Use of SNSs.

			Pairwise Con	nparisons			
						95% (CI for
			Mean			Differ	rence ^a
			Difference			Lower	Upper
Int_SNS_Use	Type_S	NS_Use	(I-J)	SE	Sig. ^a	Bound	Bound
Light	F_U	T_U	-0.639*	-0.001	0.163	0.000	-0.959
Mild	F_U	T_U	-0.107	0.000	0.146	0.458	-0.397
Heavy	F_U	T_U	0.937*	0.008	0.206	0.000	0.485

Based on estimated marginal means

*The mean difference is significant at the .05 level.

a. Adjustment for multiple comparisons: Least Significant Difference (equivalent to no

adjustments).

Notably, the type of SNSs used has a significant effect on Brid_SC only at the light and heavy level of usage (Table 7.73). Specifically, light Facebook Users show on average lower levels of Online Bridging Social Capital (M = 2.12, SD = 1.05) compared to light Twitter users (M = 2.76, SD = 1.28) and this difference, $d_{(Mi-Mj)} = -0.64$, 95% BCa CI [-0.959, - 0.317] was statistically significant (p < 0.001). Also, heavy Facebook Users show on average higher levels of Online Bridging Social Capital (M = 4.09, SD = 1.47) compared to heavy Twitter users (M = 3.15, SD = 1.40) and this difference, $d_{(Mi-Mj)} = 0.94$, 95% BCa CI [0.485, 1.377] was statistically significant (p < 0.001).

Similarly, an analysis was conducted to test for the simple effect of the intensity of SNSs use (Table 7.74). Notably, there are significant differences between light, mild and heavy users of both Facebook, $F_{(2,732)} = 57.625$, p < 0.001, $\eta_p^2 = 0.136$, $\omega^2 = 0.134$, and Twitter, $F_{(2,732)} = 12.632$, p < 0.001, $\eta_p^2 = .033 \omega^2 = 0.030$. Following Field (2013), pairwise comparisons were run at a significance level adjusted through the Sidak correction and differences estimates were bootstrapped using the BCa method and 2000 samples (Table 7.75).

		Ur	ivariat	e Tests				
Type_SNS_Use		SS	df	MS	F	Sig.	η_p^2	ω^2
тц	Contrast	210.392	2	105.196	57.625	0.000	0.136	0.134
F_U	Error	1336.295	732	1.826				
T T	Contrast	46.122	2	23.061	12.632	0.000	0.033	0.030
1_0	Error	1336.295	732	1.826				

Table 7.74. Univariate Test for Differences in Online Bridging Social CapitalBetween Light, Mild and Heavy Users Across Each SNS.

Each F tests the simple effects of Int_SNS_Use within each level combination of the other effects shown. These tests are based on the linearly independent pairwise comparisons among the estimated marginal means.

a. Computed using alpha = .05

As evident from Table 7.75 below, the intensity of use of a specific SNS affects levels of Brid_SC in both Facebook and Twitter but at different levels of intensity of use. Indeed, light Facebook users showed significantly lower levels of Brid_SC (M = 2.12, SD = 1.05) than mild (M = 3.49, SD = 1.71) and heavy Facebook users (M = 4.09, SD = 1.47), with differences $d_{(Mi-Mj)} = -1.37$, 95% BCa CI [-1.635, -1.100], p < 0.001 and $d_{(Mi-Mj)} = -1.97$, 95% BCa CI [-2.302, -1.630], p < 0.001, respectively. Also, mild Facebook

users (M = 3.49, SD = 1.71) showed significantly lower levels of Brid_SC when compared to heavy Facebook users (M = 4.09, SD = 1.47), with a difference in means of $d_{(Mi-Mj)} = -0.6$, 95% BCa CI[-0.938, -0.253] p = 0.001.

For Twitter, only mild users showed significantly higher levels of Brid_SC (M = 3.60, SD = 1.57) when compared to both light (M = 2.76, SD = 1.28) and heavy (M = 3.15, SD = 1.40) Twitter users with differences in means respectively of $d_{(Mi-Mj)} = 0.84$, 95% BCa CI [0.491, 1.158] p < 0.001 and $d_{(Mi-Mj)} = 0.45$, 95% BCa CI [0.089, 0.824], p < 0.018. Light and heavy users did not significantly differ, $d_{(Mi-Mj)} = -0.39$, 95% BCa CI [-0.783, 0.040], p = 0.051.

Table 7.75. Pairwise Comparisons Across Each Level of Intensity of Use of SNSsby type of SNSs used.

			Pairwise (Comparis	ons			
			Mean				BCa 959 Differ	% CI for rence ^b
Type_SNS_Use	Int_SN	NS_Use	Difference (I-J)	Bias	SE	Sig.ª	Lower Bound	Upper Bound
	Light	Mild	-1.372*	-0.005	0.143	0.000	-1.635	-1.100
		Heavy	-1.965°	-0.003	0.176	0.000	-2.302	-1.630
EII	Mild	Light	1.372*	0.005	0.143	0.000	1.073	1.662
F_U		Heavy	-0.594*	0.002	0.167	0.001	-0.938	-0.253
	Heavy	Light	1.965*	0.003	0.176	0.000	1.619	2.316
		Mild	0.594*	-0.002	0.167	0.001	0.283	0.916
	Light	Mild	-0.839*	0.006	0.178	0.000	-1.186	-0.462
		Heavy	-0.389	0.005	0.200	0.051	-0.783	0.040
T II	Mild	Light	0.839*	-0.006	0.178	0.000	0.491	1.158
1_0		Heavy	0.450*	-0.002	0.193	0.018	0.089	0.824
	Heavy	Light	0.389	-0.005	0.200	0.051	0.010	0.754
		Mild	-0.450*	0.002	0.193	0.018	-0.833	-0.085

Based on estimated marginal means

*The mean difference is significant at the .05 level.

a. Adjustment for multiple comparisons: Sidak.

b. Unless otherwise noted, bootstrap results are based on 2000 bootstrap samples

Next, following Field (2013), we run the Games-Howell procedure for pairwise comparisons (Table 7.76), with the purpose of conducting a sensitivity analysis and test the robustness of the results to heteroskedasticity and unequal sample size.

We also run a robust analysis based on confidence intervals and p values less sensitive to heteroskedasticity of data using the HC4 method suggested by Field (2018).

The results are reported in Table 7.77. Overall, the sensitivity analysis confirms previous results improving the reliability of the analysis and increasing confidence in our findings.

Table 7.76. Games-Howell Pairwise Comparisons of Different Levels of Intensity of Use across Facebook and Twitter Users for Online Bridging Social Capital.

Games-H	Howell Pa	airwise C	omparisons o	f Online l	Bridging S	ocial Capit	al
			Mean			BCa 959 Differ	% CI for rence ^a
Type_SNS_Use	Int_SN	NS_Use	Difference (I-J)	Bias	SE	Lower Bound	Upper Bound
	Light	Mild	-1.372	-0.003	0.140	-1.654	-1.109
		Heavy	-1.965	-0.001	0.181	-2.317	-1.612
E 11	Mild	Light	1.372	0.003	0.140	1.098	1.673
F_U		Heavy	-0.594	0.002	0.172	-0.925	-0.263
	Heavy	Light	1.965	0.001	0.181	1.610	2.322
		Mild	0.594	-0.002	0.172	0.247	0.931
	Light	Mild	-0.839	-0.004	0.173	-1.178	-0.516
		Heavy	-0.389	-0.005	0.194	-0.741	0.030
T II	Mild	Light	0.839	0.004	0.173	0.487	1.207
1_0		Heavy	0.450	-0.001	0.188	0.082	0.818
-	Heavy	Light	0.389	0.005	0.194	-0.018	0.797
		Mild	-0.450	0.001	0.188	-0.831	-0.077
Based on observe	Based on observed means						

a. Unless otherwise noted, bootstrap results are based on 2000 bootstrap samples.

Table 7.77. Confidence Intervals and p Values for Online Bridging Social Capital
robust to heteroskedasticity.

Parameter Estimates with Robust Standard Errors for Bridging Social Capital									
	Robust			95% CI					
Parameter	В	SE ^a	t	Sig.	Low	Up	η_{P}^{2}	Power ^b	
Intercept	3.149	0.150	21.056	0.000	2.855	3.442	0.377		
[Type_SNS_Use=F_U]	0.937	0.206	4.556	0.000	0.533	1.340	0.028		
[Int_SNS_Use=L]	-0.389	0.199	-1.957	0.051	-0.780	0.001	0.005	0.498	
[Int_SNS_Use=M]	0.450	0.188	2.386	0.017	0.080	0.820	0.008		
[Type_SNS_Use=F] * [Int_SNS_Use=l]	-1.576	0.266	-5.926	0.000	-2.098	-1.054	0.046		
[Type_SNS_Use=F] * [Int_SNS_Use=M]	-1.043	0.253	-4.128	0.000	-1.539	-0.547	0.023		
a. HC4 method									
b. Computed using alpha	= .05								

The results of the analysis are further validated and confirmed by the graphical inspection of the plot of the estimated marginal means of Brid_SC (Figure 7.17).

Figure 7.17. Plot of the Estimated Marginal Means for the Interaction Effect of Type of SNSs used and Intensity of use of SNSs on Online Bridging Social Capital.



7.5. ASSESSING RESULTS OF TWO-WAY FACTORIAL ANOVA TO ANSWER RESEARCH QUESTION Q5.

This Section presents the results of the analysis addressing research question Q5, and tests for any significant differences in the amount of Off_PP and On_PP displayed by Facebook and Twitter users (Type_SNS_Used), according to their intensity of use. Hence, we aimed to test for "interaction effect" between "Type_SNS_Used" and Int_SNS_Use". To answer research question Q5, the non-parametric Aligned Rank Transform Test (ART) was used because the assumptions to carry out a parametric ANOVA were not met (see Section 7.5.1.) The results are provided in Section 7.5.2 for Off_PP and in Section 7.5.3 for On_PP.

7.5.1. MEETING THE TWO-WAY FACTORIAL ANOVA ASSUMPTIONS.

The aim of this section is to illustrate that not all the 6 assumptions discussed in Section 6.5.2 to carry out a parametric two-way ANOVA were met. In particular, several outliers were detected for both Off_PP and On_PP (Figure 7.18 and Figure 7.19). Since the outliers were not due to any measurement, typing or entry errors and since no patterns or communality in the outliers' origin were found with respect to demographics, they were retained (Beckman and Cook, 1983; Hair et al., 2017; Iglewicz and Hoaglin, 1993).



Figure 7.18. Outliers Investigation for Offline Political Participation for each combination of Type of SNS Used and Intensity of Use of the SNSs considered.





Consequently, the assumption of normality for Off_PP and On_PP was not met for each combination of the groups of the two independent variables (Table 7.78).

Finally, the assumption of homogeneity of variances of Off_PP and On_PP for each combination of Type_SNS_Used and Int_SNS_Use was not met (Table 7.79).

Indeed, the Levene's test based on the medians yielded a significant result for both Off_PP ($F_{(5,732)} = 11.316$, p < 0.001) and On_PP ($F_{(5,732)} = 28.258$, p < 0.001).

Normality Statistics Summary									
Type SNS Use	Int SNS Use	Variables	Skewness	Kurtosis					
		Off_PP	2.171	5.545					
	Light (n = 99)	SE	0.243	0.481					
		On_PP	2.098	4.126					
		SE	0.243	0.481					
		Off_PP	2.667	7.431					
Facabook	Mild $(n - 162)$	SE	0.191	0.379					
FACEDOOK	$\operatorname{Minu}\left(\operatorname{II}-\operatorname{IO2}\right)$	On_PP	4.005	3.146					
		SE	0.191	0.379					
		Off PP	2.419	5.480					
	Heavy (n = 108)	SE	0.233	0.461					
		On PP	2.024	5.813					
		SE	0.233	0.461					
		Off PP	1.156	0.940					
	$\mathbf{Light}(\mathbf{n} = 05)$	SE	0.247	0.490					
	Light (II – 95)	On_PP	0.522	0.218					
		SE	0.247	0.490					
		Off_PP	2.029	3.793					
Twittor	Mild (n - 186)	SE	0.178	0.355					
1 white	Willa (ll – 180)	On_PP	1.730	2.612					
		SE	0.178	0.355					
		Off_PP	1.353	1.390					
	Heavy $(n = 88)$	SE	0.257	0.508					
	псаvу (п — 00)	On PP	0.192	-1.099					
		SE	0.257	0.508					

Table 7.78. Skewness and Kurtosis of Offline and Online Political Participation ofGroups of SNSs users Split According to Type of SNSs Used and Intensity of Use.

Table 7.79. Levene's Test for Homogeneity of Variances of Offline Political
Participation for Type of SNSs Used Across different Levels of Intensity of Use

Levene's Test of Equality of Error Variances ^a								
		Levene Statistic	df1	df2	Sig.			
Dand SC	Based on Mean	18.706	5	732	0.000			
Bond_SC	Based on Median	11.316	5	732	0.000			
Drid SC	Based on Mean	36.287	5	732	0.000			
Brid_SC	Based on Median	28.258	5	732	0.000			
a. Design: Intercept + Type SNS Use + Int SNS Use +								
Type SNS	Use * Int SNS Use.							

Therefore, for both On_PP and Off_PP we conducted non-parametric two-way ANOVAs in R through the ART test using the *ARTool* package (see Section 6.5.3). To further test the robustness of the results to heteroskedasticity and unequal sample sizes, we also conducted the Games-Howell test for the post-hoc pairwise comparisons and the HC4 test in SPSS (Field, 2013).

The next two sections will provide the relevant output of the two-way ANOVA analysis for Off_PP and On_PP respectively.

7.5.2. ART TEST FOR OFFLINE POLITICAL PARTICIPATION

This Section presents the results of the ART test and the sensitive analysis of the Games-Howell procedure and the HC4 test conducted to test the combined effect of the type of SNSs used and their intensity of use on Off_PP. Table 7.80 reports descriptive statistics of the Offline Political Participation variable for each combination of the two independent variables (Type_SNS_Use and Int_SNS_Use). Table 7.81 reports the results of the ART test.

Descriptive Statistics Off_PP								
Type_SNS_Use	Int_SNS_Use	M	SD	Ν				
F_U	Light	0.611	0.966	99				
	Mild	0.534	1.052	162				
	Heavy	0.919	1.656	108				
	All F_U	0.667	1.248	369				
T_U	Light	1.092	1.136	95				
	Mild	0.925	1.514	186				
	Heavy	2.066	2.349	88				
	All T_U	1.240	1.734	369				
Total	Light	0.847	1.077	194				
	Mild	0.743	1.332	348				
	Heavy	1.434	2.072	196				
	All F_U and T_U	0.954	1.536	738				

Table 7.80. Descriptive Statistics of Offline Political Participation Scores by Typeof SNSs Used and Intensity of Use of SNSs.

Tests of Between-Subjects Effects								
Source	Type III SS	Df ^b	MS	F	Sig ^a .	η_p^2	ω^2	
Type_SNS_Use	2681245.256	1	2681245.256	68.414	0.000	0.085	0.084	
Int_SNS_Use	1656528.097	2	828264.048	20.186	0.000	0.052	0.049	
Type_SNS_Use* Int_SNS_Use	1243366.062	2	621683.031	14.926	0.000	0.039	0.036	
a. Computed	using $alpha = .0$	5						
b. Residuals	df = 732							

Table 7.81. Summary of the ART test for Offline Political Participation.

The interaction effect of type of SNSs used by the intensity of use of the SNSs on Off_PP is significant with $F_{(2,732)} = 14.926$, p < 0.001, and exert between a small and medium effect ($\eta_p^2 = 0.039$ and $\omega^2 = 0.036$).

Consistently with Field's (2013) suggestions, to break down the interaction effect an analysis of the simple effects was conducted. Table 7.82 provides the results of the pairwise comparisons for differences in Off_PP between Facebook and Twitter users across each level of Int_SNS_Use. Notably, the type of SNSs used has a significant effect on Off_PP at each level of usage.

Table 7.82. Robust Results of the ART two- way ANOVA of Offline Political Participation between Facebook and Twitter Users at each Level of Intensity of

Pairwise Comparisons								
Mean Difference								
_Int_SNS_Use	Type_	SNS_Use	(I-J)	SE	df	t-ratio	Sig.	r
Light	F_U	T_U	-97.371*	28.396	732	-3.429	0.000	0.125
Mild	F_U	T_U	-51.496*	21.248	732	-2.424	0.015	0.089
Heavy	F_U	T_U	-130.068*	28.393	732	-4.581	0.000	0.167
Based on estimated marginal means								
*The mean difference is significant at the 05 level								

Use.

Similarly, we run pairwise comparisons for both Facebook and Twitter and their different levels of Int_SNS_Use (Table 7.83), to test for the simple effect of Int_SNS_Use on Off_PP. The results suggest that Int_SNS_Use does not affect levels of Off PP for Facebook users.

For Twitter, only heavy users (M = 2.07, SD = 2.35) showed significantly higher levels of Off_PP when compared with both light (M = 1.09, SD = 1.14) and mild (M =0.93, SD = 1.51) users, with differences in means of $d_{(Mi-Mj)} = 0.98$, p = 0.003 and $d_{(Mi-Mj)}$ = 1.14, p < 0.001, respectively.

Table 7.83. Pairwise Comparisons of Levels of Offline Political Participation within the Robust ART framework.

Robust ART A	Robust ART ANOVA Pairwise Comparisons of Offline Political Participation								
			Mean						
			Difference						
Type_SNS_Use	Int_SN	S_Use	(I-J)	SE	df	t-ratio	Sig. ^a	r	
	Light	Mild	15.569	27.510	732	0.566	0.572	0.021	
		Heavy	-34.383	25.222	732	-1.363	0.173	0.050	
ЕЦ	Mild	Light	-15.569	27.510	732	-0.566	0.572	0.021	
F_0		Heavy	-49.952	24.561	732	-2.034	0.423	0.075	
	Heavy	Light	34.383	25.222	732	1.363	0.173	0.050	
	-	Mild	49.952	24.561	732	2.034	0.423	0.075	
	Light	Mild	48.267	29.253	732	1.650	0.099	0.061	
		Heavy	-80.257	24.933	732	-3.219	0.001	0.118	
тц	Mild	Light	-48.267	29.253	732	-1.650	0.099	0.061	
I_0		Heavy	-128.524	25.581	732	-5.024	0.000	0.183	
	Heavy	Light	80.257	24.933	732	3.219	0.001	0.118	
	-	Mild	128.524	25.581	732	5.024	0.000	0.183	
Based on estimate	ed margir	al means							

a. Adjustment for multiple comparisons: Sidak

We further supported and validated the above results by running a sensitivity analysis in SPSS through: (1) the Games-Howell pairwise test procedure, bootstrapping the differences between groups using the BCa method with 2000 samples at a 95% confidence interval (Table 7.84) and (2) the HC4 test (Table 7.85). These tests are robust to heteroskedasticity in presence of unequal sample sizes (Field, 2018).

Games-I	Games-Howell Pairwise Comparisons of Offline Political Participation								
						BCa 959	% CI for		
			Mean			Differ	rence ^a		
			Difference			Lower	Upper		
Type_SNS_Use	Int_SN	S_Use	(I-J)	Bias	SE	Bound	Bound		
F_U	Light	Mild	0.078	-0.007	0.125	-0.217	0.372		
		Heavy	-0.308	-0.004	0.190	-0.780	0.136		
	Mild	Light	-0.078	0.007	0.125	-0.382	0.243		
		Heavy	-0.385	0.004	0.183	-0.867	0.046		
	Heavy	Light	0.308	0.004	0.190	-0.151	0.798		
		Mild	0.385	-0.004	0.183	-0.034	0.830		
T_U	Light	Mild	0.168	0.001	0.157	-0.195	0.549		
		Heavy	-0.973	-0.003	0.278	-1.626	-0.319		
	Mild	Light	-0.168	-0.001	0.157	-0.542	0.187		
		Heavy	-1.141	-0.004	0.278	-1.855	-0.495		
	Heavy	Light	0.973	0.003	0.278	0.286	1.679		
		Mild	1.141	0.004	0.278	0.509	1.851		
D 1 1	1								

Table 7.84. Games-Howell Pairwise Comparisons of Different Levels of Intensityof Use across Facebook and Twitter Users.

Based on observed means

a. Unless otherwise noted, bootstrap results are based on 2000 bootstrap samples.

Table 7.85. Confidence Intervals and p Values for Offline Political Participationrobust to heteroskedasticity.

Parameter Estimates with Robust Standard Errors for Offline Political Participation								
		Robust		95% CI				
Parameter	В	SE ^a	t	Sig.	Low	Up	η_{P}^{2}	
Intercept	2.066	0.251	8.231	0.000	1.573	2.558	0.085	
[Type_SNS_Use=F_U]	-1.147	0.297	-3.857	0.000	-1.731	-0.563	0.020	
[Int_SNS_Use=L]	-0.973	0.277	-3.517	0.000	-1.517	-0.430	0.017	
[Int_SNS_Use=M]	-1.141	0.274	-4.158	0.000	-1.680	-0.602	0.023	
[Type_SNS_Use=F] * [Int_SNS_Use=l]	0.666	0.334	1.994	0.047	0.010	1.321	0.005	
[Type_SNS_Use=F] * [Int_SNS_Use=M]	0.756	0.328	2.305	0.021	0.112	1.400	0.007	
a. HC4 method								
b. Computed using alpha =	= .05							

Overall, the sensitive analysis confirms the ART test results and gives us confidence in our conclusions. The interpretation of the interaction effect is further facilitated by the graphical inspection of the plot of the estimated marginal means of Off_PP for all the groups of SNSs users which further confirms the results reported above (Figure 7.20).



Figure 7.20. Plot of the Estimated Marginal Means for the Interaction Effect of Type of SNSs used and Intensity of use of SNSs on Offline Political Participation.

Finally, it should be noted that since the sample sizes of the groups to be compared have a number of observations either close or above 100, the normality assumption should not be a concern because of the "*central limit theorem*" which states that regardless of the shape of the population, parameters estimates of that population will have a normal distribution provided that the samples are large enough (Field, 2013). Moreover, the two-way ANOVA is known to be robust against violations of normality and heteroskedasticity especially when sample size is above 30 (see Section 7.4.2). Accordingly, we conducted a parametric two-way ANOVA which further confirmed our findings.

7.5.3. ART TEST FOR ONLINE POLITICAL PARTICIPATION

This Section presents the results of the two-way ANOVA conducted to test the combined effect of the type of SNSs used and their intensity of use on Online Political Participation (On_PP). Consistently with the previous Section, we employed the ART test and conducted robust tests to heteroscedasticity and unequal sample sizes like the Games-Howell and the HC4 tests.

Table 7.86 reports descriptive statistics of On_PP for each combination of the two independent variables (Type_SNS_Use and Int_SNS_Use) and Table 7.87 shows

the results of the ART test. Notably, the interaction effect of type of SNSs used by the intensity of use of the SNSs on On_PP is significant with $F_{(2,732)} = 33.777$, p < 0.001, $\eta_p^2 = 0.084$ and $\omega^2 = 0.082$. This suggests that the effect of the intensity of use of SNSs on Online Political Participation changes according to the different types of SNSs considered such as Facebook and Twitter and vice versa.

Descriptive Statistics On_PP								
Type_SNS_Use Int_SNS_Use M SD								
F_U	Light	0.883	1.303	99				
	Mild	0.782	1.223	162				
	Heavy	1.399	1.429	108				
	All F_U	0.990	1.331	369				
T_U	Light	0.702	0.503	95				
	Mild	1.213	1.577	186				
	Heavy	3.432	2.489	88				
	All T_U	1.611	1.966	369				
Total	Light	0.794	0.997	194				
	Mild	1.012	1.437	348				
	Heavy	2.312	2.216	196				
	All F U and T U	1.300	1.706	738				

Table 7.86. Descriptive Statistics of Online Political Participation Scores by Typeof SNSs Used and Intensity of Use of SNSs.

Table 7.87. Summar	y of the ART	test for Online	Political F	'articipation.
	•			

Tests of Between-Subjects Effects							
Source	Type III SS	Df^b	MS	F	Sigª.	η_p^2	ω^2
Type_SNS_Use	3118333.678	1	3118333.678	76.777	0.000	0.095	0.093
Int_SNS_Use	4625612.170	2	2312806.085	61.792	0.000	0.144	0.142
Type_SNS_Use* Int_SNS_Use	2767319.444	2	1383659.722	33.777	0.000	0.084	0.082
a. Computed using alpha = .05							
b. Residuals $df = 732$							

An analysis of the simple effects of the two independent variables was conducted in R using the ARTool framework (Wobbrock et al., 2011; Kay and Wobbrock, 2016). Then pairwise comparisons were conducted.

We found that:

• the type of SNSs used has a significant effect on On_PP at each level of usage except for light users (Table 7.88).

• The intensity of use of a specific SNS affects levels of On_PP for both Facebook and Twitter users but at different levels of intensity of use.

For F_U :

- the level of On_PP was significantly higher for heavy users (M = 1.40, SD = 1.43) compared to mild (M = 0.78, SD = 1.22) and light (M = 0.88, SD = 1.30) with differences in averages of $d_{(Mi-Mj)} = 0.6$, p < 0.001 and $d_{(Mi-Mj)} = 0.52$, p < 0.001 respectively.
- light and mild Facebook users showed non-significant differences in levels of On_PP, $d_{(Mi-Mj)} = 0.10$, p = 0.999.

For T_U:

- heavy users showed significantly higher levels of On_PP (M = 3.43, SD = 2.49) compared with both mild (M = 1.21, SD = 1.58) and light (M = 0.70, SD = 0.50) users with differences in means of $d_{(Mi-Mj)} = 2.22$, p < 0.001 and $d_{(Mi-Mj)} = 2.73$, p < 0.001, respectively.
- light and mild users significantly differed in levels of On_PP, $d_{(Mi-Mj)} = -0.51$, p = 0.012.

Table 7.88. Results of the ART two- way ANOVA of Online Political Participation between Facebook and Twitter Users at each Level of Intensity of Use.

Pairwise Comparisons								
			Mean Difference					
Int_SNS_Use	Type_S	NS_Use	(I-J)	SE	df	t-ratio	Sig.	r
Light	Fυ	ΤU	39.874	28.352	732	1.406	0.160	0.052
Mild	F_U	TU	-47.561*	21.215	732	-2.242	0.025	0.083
Heavy	F_U	T_U	-131.893*	28.349	732	-4.652	0.000	0.169
Based on estimated marginal means								

*The mean difference is significant at the .05 level.

Robust ART ANOVA Pairwise Comparisons of Online Political Participation								
			Mean					
			Difference					
Type_SNS_Use	Int_SN	S_Use	(I-J)	SE	df	t-ratio	Sig. ^a	r
F_U	Light	Mild	0.571	25.183	732	0.023	0.982	0.001
		Heavy	-120.529	27.467	732	-4.388	0.000	0.160
	Mild	Light	-0.571	25.183	732	-0.023	0.982	0.001
		Heavy	-119.958	24,523	732	-4.892	0.000	0.178
	Heavy	Light	120.529	27.467	732	4.388	0.000	0.160
		Mild	119.958	24,523	732	4.892	0.000	0.178
TU	Light	Mild	-58.258	20.894	732	-2.788	0.012	0.103
_	-	Heavy	-212.548	29.207	732	-7.277	0.000	0.260
	Mild	Light	58.258	20.894	732	2.788	0.012	0.103
		Heavy	-204.290	25.541	732	-7.998	0.000	0.283
	Heavy	Light	212.548	29.207	732	7.277	0.000	0.260
		Mild	204.290	25.541	732	7.998	0.000	0.283
Based on estimated marginal means.								
a Adjustment for multiple comparisons: Sidel								

Table 7.89. Pairwise Comparisons of Levels of Online Political Participation within the ART test framework.

a. Adjustment for multiple comparisons: Sidak.

The sensitivity analysis conducted in SPSS through the Games-Howell procedure (Table 7.90) and the HC4 test further confirmed and validated our results.

Table 7.90. Games-Howell Pairwise Comparisons of Different Levels of Intensity of Use across Facebook and Twitter Users.

Games-Howell Pairwise Comparisons of Online Political Participation							
						BCa 959	% CI for
			Mean			Differ	encea
			Difference			Lower	Upper
Type_SNS_Use	Int_SN	S_Use	(I-J)	Bias	SE	Bound	Bound
F_U	Light	Mild	0.101	-0.003	0.164	-0.191	0.401
		Heavy	-0.516	-0.001	0.189	-0.898	-0.141
	Mild	Light	-0.101	0.003	0.164	-0.445	0.239
		Heavy	-0.617	0.002	0.168	-0.954	-0.284
	Heavy	Light	0.516	0.001	0.189	0.137	0.901
		Mild	0.617	-0.002	0.168	0.291	0.947
T_U	Light	Mild	-0.511	0.000	0.124	-0.776	-0.278
		Heavy	-2.731	-0.002	0.273	-3.241	-2.201
	Mild	Light	0.511	0.000	0.124	0.293	0.755
		Heavy	-2.219	-0.002	0.295	-2.789	-1.620
	Heavy	Light	2.731	0.002	0.273	2.170	3.261
		Mild	2.219	0.002	0.295	1.578	2.818
Based on observed means							

Based on observed means

a. Unless otherwise noted, bootstrap results are based on 2000 bootstrap samples.

Parameter Estimates with Robust Standard Errors for Online Political Participation							
		Robust			95%	6 CI	
Parameter	В	SE ^a	t	Sig.	Low	Up	η_p^2
Intercept	3.432	0.266	12.907	0.000	2.910	3.954	0.185
[Type_SNS_Use=F_U]	-2.033	0.299	-6.791	0.000	-2.621	-1.445	0.059
[Int_SNS_Use=L]	-2.731	0.271	-10.080	0.000	-3.262	-2.199	0.122
[Int_SNS_Use=M]	-2.219	0.290	-7.654	0.000	-2.788	-1.650	0.074
[Type_SNS_Use=F] * [Int_SNS_Use=I]	2.215	0.331	6.693	0.000	1.565	2.865	0.058
[Type_SNS_Use=F] * [Int_SNS_Use=M]	1.602	0.335	4.783	0.000	0.944	2.260	0.030
a. HC4 method							
b. Computed using alpha = .05							

Table 7.91. Confidence Intervals and p Values for Online Political Participa	tion
robust to heteroskedasticity.	

Finally, following the same rationale of the previous section we conducted a parametric two-way ANOVA which further supported our finding.

The inspection of the plot of the estimated marginal means of On_PP for all the groups of SNSs users further confirms the results reported above (Figure 7.21).

Figure 7.21. Plot of the Estimated Marginal Means for the Interaction Effect of Type of SNSs used and Intensity of use of SNSs on Online Political Participation.



8. RESULTS INTERPRETATION, DISCUSSION AND CONCLUSIONS. IMPLICATIONS FOR THEORY AND PRACTICE AND FURTHER RESEARCH.

Following Hair et al.'s (2017) systematic procedure to apply PLS-SEM analysis (Figure 7.1, Chapter 7) the final stage of any analysis should consist of interpreting the results and drawing conclusions. As part of the results' interpretation process, the research questions are answered and empirical results from the study are explained through extant knowledge in the field. Conclusions on whether empirical findings contradict or confirm prior theoretical knowledge are drawn (Section 8.1); accordingly, contributions of the research to theory and practice are established (Section 8.2). Next, the limitations of the research are presented with respect to theoretical outcomes, research methodology and empirical techniques of analysis employed (Section 8.3). Finally, directions for future research are provided (Section 8.4).

8.1. RESULTS INTERPRETATION, DISCUSSION AND CONCLUSIONS

8.1.1. RESULTS INTERPRETATION AND DISCUSSION FOR Q1

The results of the moderation analysis carried out in Section 7.1 addressed RQ1 and all its sub-questions. Findings indicate that the relationships between Online Social Capital (Bonding and Bridging) and Online and Offline Political Participation depends on (is moderated by) the type of SNS (Facebook or Twitter) used. However, this holds true only if the intensity of use of the SNSs is not controlled for. The research questions and related findings are summarised in Table 8.1.

Main Question	Subsidiary Question	Findings
Q1. Does the type of SNS used moderate the relationships between Online Social Capital and Political Participation? To what extent?	Q1a. Does the type of SNS used moderate the relationships between Online Bonding Social Capital and Online Political Participation ? To what extent?	Yes. For Twitter users, the relationship shows a significantly higher path coefficient than for Facebook users, but only if the intensity of use of those SNSs is not controlled for.
	Q1b. Does the type of SNS used moderate the relationships between Online Bonding Social Capital and <i>Offline Political</i> <i>Participation</i> ? To what extent?	Yes. For Twitter users, the relationship shows a significantly higher path coefficient than for Facebook users, but only if the intensity of use of those SNSs is not controlled for.
	Q1c. Does the type of SNS used moderate the relationships between Online Bridging Social Capital and Online Political Participation? To what extent?	Yes, but only when controlling for the intensity of use of those SNSs. Heavy Twitter users show a significantly higher path coefficient than heavy Facebook Users.
	Qld. Does the type of SNSs used moderate the relationships between Online Bridging Social Capital and <i>Offline Political</i> <i>Participation</i> ? To what extent?	Yes. For Facebook users, the relationship shows a significantly higher path coefficient than for Twitter users, but only if the intensity of use of those SNSs is not controlled for.

 Table 8.1. Findings Summary for Research Question Q1.

According to the technological determinism principle, technology per se can affect individuals' actions and interactions through its intrinsic features and characteristics (Bimber, 1994; Robins and Webster 1999). Consequently, online SNSs are valuable assets which not only represent the connection among individuals but also may affect perceptions, beliefs, and actions of their members (e.g., Online and Offline Political Participation) according to their specific affordances (Brandtzaeg, 2012; Petrocchi et al, 2015; Phua et al., 2017; Valenzuela et al., 2017). Indeed, we found that the Online Bonding and Bridging Social Capital created on SNSs may significantly predict Online and Offline Political Participation but with some differences according to the type of SNS used, as summarised in Table 8.1 above.

The results support the theoretical reasoning for which the relationships among Online Bonding and Bridging Social Capital and Online and Offline Political
Participation vary in strength depending on the context on which they are formed. This mirrored the assumption of Social Capital formed in offline networks (Coleman, 1998; Knoke and Yang, 2008; 2019). For example, the network of a local Roman Catholic Church is different from the network of a workplace. Networks are shaped by different values and whilst the former would include only Roman Catholic faithful, the latter would incorporate a variety of beliefs and faith ultimately leading to different types and levels of participation. For instance, the church network can probably push civic participation in local community activities being rooted on small local realities. The workplace network can instead widen the horizon of its members given that it can provide relationships with people of different cultural backgrounds and push them to be engaged in activities that go beyond local geographic boundaries. This discussion supports the argument that the relationship between Social Capital and Political Participation is not a homogeneous entity, rather it varies depending on the type and context of the network (offline) in which people are involved. The same reasoning appeared to hold true also in the online environment with respect to Facebook and Twitter social networks.

Furthermore, this study found that the intensity of use of the SNSs moderated the relationship between Bridging Social Capital and both Online and Offline Political Participation in the Twitter users' sample (Section 7.2.4). Hence, to properly ascertain the moderating effect of the type of SNS used and adequately answer research question Q1, we controlled for the intensity of use of Facebook and Twitter by splitting each sample into Heavy, Mild and Light users through the quartile split method. Then we tested for differences across groups using a series of Permutation tests across each level of intensity of use. The results showed that, when controlling for the intensity of use of Facebook and Twitter) moderated only the relationship Brid_SC \rightarrow On_PP, in the heavy category with Twitter Users showing significantly higher paths coefficients compared with Facebook users.

The results suggest that the differences in the strength of the relationships between Online Social Capital and Online and Offline Political Participation between Facebook and Twitter users are not due to differences in the thematic focuses, functions and intrinsic characteristics of those online networks (like it is for the offline networks) but rather to differences in the level of their intensity of use. Yet, it could be argued that the intensity of use of a SNS is a function of its specific affordance. Indeed, it is not surprising that Facebook users were found to spend, on average, 44 minutes online compared with 32 minutes on Twitter. This could be due to differences in the communication affordance of those two SNSs. Indeed, Twitter communication is centred on brief messages no longer than 280 characters (140 until 2017), as opposed to Facebook, which allows users to upload lengthier and more elaborated posts (Small, 2011). As a result, Facebook users are more likely to read the posts published by their closer contacts, to whom they feel more emotionally connected (Kaun and Stiernstedt, 2014). Hence, Facebook is more focused on reciprocal social interaction compared with Twitter (Petrocchi et al., 2015; Huberman et al., 2009) and is mostly used to satisfy the needs for belongingness, social connectedness, interaction and self-presentation (Nadkarni and Hofmann, 2012). This inevitably increases the time people spend on Facebook compared with Twitter.

However, spending more time on Facebook does not result in higher levels of Offline and Online Political Participation compared to Twitter. Indeed, our findings suggest that Twitter heavy users are more engaged in online political activities than Facebook heavy users. This could be explained again by the intrinsic characteristics and affordance of the two SNSs. Indeed, Facebook's algorithm shows news and feeds of people that a user interacts the most with, usually close friends and relatives (Bucher, 2012). Accordingly, users are exposed predominantly to like-minded information containing a limited and restricted view of political issues. This does not help to foster political debate and therefore does not facilitate political engagement (Hayat et al., 2016; Himelboim et al., 2013). Moreover, because of the lengthier and more elaborated posts allowed, Facebook users are more likely to spend more time on homogenous like-minded type of information rather than on more heterogeneous type of news. The reverse is likely to happen in Twitter. Indeed, Twitter's algorithm rewards immediacy and recurrence rather than interaction and affinity (Valenzuela et al., 2017). Twitter users are more exposed to content from sources that tweet more often, such as news companies and political groups. As a result, Twitter users are potentially exposed to more novel and timely information provided also by sources with whom they have no strong relationship. Research has shown that the exposure to diverse, opposing and heterogeneous news sources, even if accidental, is a predictor of Political Participation as it may improve deliberative democracy in many ways, such as enhancing mutual understanding, promoting political tolerance, generating political debates, critical reasoning and elaboration of news (Gil de Zuniga et al., 2012; Mutz, 2006; Valenzuela et al., 2009;

Vitak et al., 2011). Furthermore, by allowing users to connect with people they do not know personally, Twitter reinforce "weak ties" which are linked to Bridging Social Capital (Putnam, 2000). Hence, this could explain why our findings suggest that Twitter affordance moderates the relationship between Online Bridging Social Capital and Online Political participation. It must be noted that Online Bridging Social Capital was found to affect both Offline and Online Political Participation at all levels of Intensity of Use, except light users. This is in line with previous studies (e.g., Skoric et al., 2009; Zhong, 2014). However, it seems that for lower levels of intensity of use the relationship is not significant. This result contradicts the "*time displacement logic*" and supports a more optimistic view of SNSs as mobilisation sources (see discussion Chapter 3).

Finally, our results suggest that. on Twitter, the relationships between Online Bonding Social Capital and both Offline and Online Political Participation is significantly stronger than on Facebook. This finding could be due to differences in users' anonymity. Indeed, Twitter allows its users to remain anonymous as they do not need to disclose themselves to express their opinions (Huberman et al., 2009; Woo-Yoo and Gil de Zuniga, 2014), contrary to Facebook. Arguably, anonymity could encourage greater participation online because it resolves the social desirability bias allowing people to publicly take a stand on political matters that can be opposed by close friends and family whose ties form Bonding Social Capital (Collins and Butler, 2003; Towne and Herbsleb, 2012).

8.1.2. RESULTS INTERPRETATION AND DISCUSSION FOR Q2

A moderation analysis through the two-stage approach in PLS-SEM was conducted to ascertain the moderating effect of the time that respondents spent on Facebook and Twitter over the relationships between Online Bonding and Bridging Social Capital and Online and Offline Political Participation. Findings are summarised in Table 8.2.

Notably, only for Twitter Users, the time spent online was a significant moderator with a positive and large effect size on the relationships: (1) Brid_SC \rightarrow Off_PP and (2) Brid_SC \rightarrow On_PP, An analysis of the simple effect revealed that for an average level of the moderator variable the simple effect Brid_SC \rightarrow Off_PP is .171. For higher levels of time spent online (one standard deviation unit above the mean) the relationship Brid_SC \rightarrow Off_PP increase of .166. For lower levels (mean value of time spent online

minus one standard deviation unit) it decreases of the same amount. Similarly, the simple effect Brid_SC \rightarrow On_PP is β = .270. For higher levels of time spent online (one standard deviation above the mean) the relationship Brid_SC \rightarrow On_PP increases by .186 Meanwhile, for lower levels (one standard deviation below) the simple effect Brid_SC \rightarrow On_PP decreases of the same amount.

The above results were further confirmed through a sensitivity analysis, assessing the models with and without the moderation term with respect to their in-sample and outof-sample predictive metrics and model fit measures.

Our findings indicate that the time spent online moderates only the relationship between Online Bridging Social Capital and Online and Offline Political Participation and only on Twitter. From a perspective of technological determinism, such differences may be explained by the different technological affordances of Facebook and Twitter.

Indeed, Twitter is known to facilitate the creation of week ties and therefore it promotes Bridging Social Capital as it helps people to connect with more strangers than other social media (Zhang et al., 2010), especially compared with Facebook which, as argued by Woo-Yoo and Gil de Zuniga (2014), is linked to and mirrors the network people have already established offline.

Furthermore, Gil de Zuniga and Valenzuela (2010) found that conversations with weak ties online are strong predictors of civic and political behaviours and Skoric et al. (2009) found that Online Bridging Social Capital was a strong predictor of Online Political Participation. Hence, with its capacity to facilitate weak ties connections, Twitter could arguably encourage Political Participation. Yet, this relationship between Online Bridging Social Capital and Online Political Participation is moderated by the time spent online and (we can further speculate) whether individuals devote that time to online conversation with weak ties. Conversely, Facebook is more centred on recreational and entertainment activities (Petrocchi et al., 2015) which arguably can take the place of more meaningful civic and political activities conducted online, like political discussion with strong ties.

Hence the time spent on Facebook would not necessarily moderate the relationship between Online Bridging Social Capital and Online Political Participation.

Main Question	Type of SNSs	Subsidiary Question	Findings
Q2. Does the intensity of use of SNS moderate the relationships between	Facebook	Q2a Does the intensity of use of Facebook moderate the relationship between Online Bonding Social Capital and Offline Political Participation? To what extent?	No
and Political Participation? To what extent?		Q2b Does the intensity of use of Facebook moderate the relationship between Online Bonding Social Capital and Online Political Participation? To what extent?	No
		Q2c Does the intensity of use of Facebook moderate the relationship between Online Bridging Social Capital and Offline Political Participation? To what extent?	No
		Q2d Does the intensity of use of Facebook moderate the relationship between Online Bridging Social Capital and Online Political Participation? To what extent?	No
	Twitter	Q2e Does the intensity of use of Twitter moderate the relationship between Online Bonding Social Capital and Offline Political Participation? To what extent?	No
		Q2f Does the intensity of use of Twitter moderate the relationship between Online Bonding Social Capital and Online Political Participation? To what extent?	No
		Q2g Does the intensity of use of Twitter moderate the relationship between Online Bridging Social Capital and Offline Political Participation? To what extent?	Yes, and with a large positive effect
		Q2h Does the intensity of use of Twitter moderate the relationship between Online Bridging Social Capital and Online Political Participation? To what extent?	Yes, and with a large positive effect

Tε	ıbl	e 8	.2.	Fin	dings	Summar	y for	Research	Question	Q2.
									· ·	~

Regarding the relationship between Online Bridging Social Capital and Offline Political Participation, we can argue that the moderating effect of the time spent online could once again be linked to Twitter's affordances. Indeed, compared with Facebook, Twitter has been found to be more centred around information and news diffusion and seems to excel in terms of information richness and diversity to the extent that previous research indicates that Twitter resembles mass media (Kwak et al., 2010). Notably, previous studies have documented that informational use of the Internet and the exposure to political news can facilitate people engagement in offline political activities (Bakker and DeVreese, 2011; Gil de Zuniga et al., 2012). Since Twitter facilitates exposure to news in a higher and more diversified amount compared with Facebook, the relationship between Online Bridging Social Capital and Offline Political Participation could be strengthen by the time spent online, assuming that Twitter users are more exposed to political news or use the social network to retrieve political information (Goldbeck et al., 2010; Pew Research Centre, 2015).

With respect to Bonding Social Capital, the close association among strong ties like relatives and close friends limit their access to more and diversified resources like political news because the resources owned by strong ties are often overlapped (Putnam, 2000). Furthermore, usually strong ties are individuals with common backgrounds and are very likely to have similar views and social networks. Arguably, the type of political news to which people are exposed within strong ties networks are not so diverse and/or so large in terms of amount. And this probably would not change in relation to the time individuals spend online. Moreover, having the same background and vision would probably not facilitate political debate of foster political activities. Indeed, empirical evidence suggests that strong family ties are linked with disengagement from civic activities and with less interest in politics (Alesina and Giuliano, 2011).

Overall, these findings suggest that when investigating the relationship between Online Social Capital and Offline Political Participation, researchers should consider the moderating role of the time spent online on the relationships between Online Bridging Social Capital and Online and Offline Political Participation on Twitter, given the differences in its technological affordances compared with Facebook.

8.1.3. RESULTS INTERPRETATION AND DISCUSSION FOR Q3

To answer research question Q3, we tested the moderating effect of the combined use of SNSs over the relationships between Online Bonding and Bridging Social Capital and Online/Offline Political Participation through a series of Permutation tests in PLS-SEM. Purposely, the group of combined users of Facebook and Twitter was compared with the separate groups of Facebook users and Twitter users. The findings are summarised in Table 8.3.

We found that the combined use of Facebook and Twitter significantly and negatively moderates the relationships between Online Bridging Social Capital and Online Political Participation. Indeed, the combined users of Facebook and Twitter showed significantly lower path coefficients on the relationship Brid_SC \rightarrow On_PP when compared with either Facebook users or Twitter users. A significant moderating effect of the combined use of Facebook and Twitter was found also on the relationship Brid_SC \rightarrow Off_PP but only compared with Facebook users.

Main Question	Subsidiary Question	Findings
Q3. Does the combined use	Q3a. Does the combined use of	No.
of SNSs moderate the	SNSs moderate the relationships	
relationships between	between Online Bonding Social	
Online Social Capital and	Capital and Online Political	
Political Participation? To	Participation? To what extent?	
what extent?	Q3b. Does the combined use of	No
	SNSs moderate the relationships	
	between Online Bonding Social	
	Capital and Offline Political	
	Participation? To what extent?	
	Q3c. Does the combined use of	Yes, but only if the time spent
	SNSs moderate the relationships	online is not controlled for.
	between Online Bridging Social	
	Capital and Online Political	
	Participation? To what extent?	
	Q3d. Does the combined use of	Yes, but only compared with
	SNSs moderate the relationships	Facebook users, if the time
	between Online Bridging Social	spent online is not controlled
	Capital and Offline Political	for.
	Participation? To what extent?	Yes, for the heavy user category
		if the intensity of use of SNSs is controlled for.

Table 8.3. Findings Summary for Research Question Q3.

Yet, when controlling for the intensity of use, the combined use of Facebook and Twitter moderates the relationship Brid_SC \rightarrow Off_PP. Specifically, the combined users of Facebook and Twitter showed a significantly lower relationship between Online Bridging Social Capital and Offline Political Participation when compared with both Facebook users and Twitter users, but only in the heavy category.

Considering that the interpretation of the results should account for heterogeneity of the samples to avoid misleading conclusions we also examined the pairwise comparisons across different levels of Intensity of Use of different types of SNSs.

The results of this analysis indicated that for the combined heavy users of Facebook and Twitter there is a negative and significant relationship between Online Bridging Social Capital and Offline Political Participation. Conversely, for heavy Facebook users and heavy Twitter users there was a significant positive relationship between Online Bridging Social Capital and Offline Political Participation. For Twitter in particular, the effect was very large. Arguably, such difference can be due to the time that the users of Facebook and Twitter combined spend online in a logic of "time displacement", according to which the time spent on the Internet reduces the time available to carry out other meaningful social activities like civic and political actions (Nie and Erbring, 2002; Putnam, 2000).

Indeed, an analysis of variances on the intensity of use of the SNSs conducted through a Kruskal-Wallis test revealed that the time spent online was significantly different across the three type of users, namely Facebook, Twitter, Facebook and Twitter combined in the heavy category. Specifically, we found that Facebook and Twitter combined heavy users spent on average 240 minutes per day on those SNSs, meanwhile Facebook users spent 124 minutes and Twitter users 92 minutes. However, heavy Facebook users and heavy Twitter users did not significantly differ in minutes spent online in a day. Given that the more SNSs an individual use, the more time he/she spends online compared with people who use only one or few SNSs, the time displacement logic could hold true for those Internet users. Indeed, although Facebook heavy users and Twitter heavy users spend respectively around two hours and one and a half hour per day online, probably this would not prevent them from engaging in offline political activities.

This reasoning is further confirmed by the analysis of the Facebook and Twitter combined users in the mild usage category which show similar patterns to the comparisons made above to Facebook and Twitter users only. Moreover, these differences in moderation effects could be exacerbated by the intrinsic characteristics and features of Facebook and Twitter. Indeed, Gil de Zuniga et al. (2016) suggest that social media change the structure and nature of social connections, and therefore they may alter the nature of the relationship between Social Capital and Political Participation. In short, how people are connected to one another matters for Political Participation (Gil de Zuniga et al., 2016). This is particularly true if we consider the final aim and the content of the two SNSs, whereby Twitter has a more informational role, based on the spread of news and reciprocal exchange and injection of information (e.g., political news, mobilizing information, contacting, etc.) by relying on weaker, more heterogeneous, social ties (Lin and Qiu, 2013; Velenzuela et al., 2017).

Indeed, extant empirical evidence suggests that weak ties are more important for Political Participation and that the SNSs' news content is related to greater Political Participation offline (Gil de Zuniga et al., 2012). Even if exposure to news is incidental, they may still become influential as they have been shared or posted online by a trusted network (Bode, 2012). Conversely, Facebook users connect with individuals with whom they already have an existing relationship and/or shared identity offline (Ellison et al., 2007). This is because connection in this case is reciprocal and needs approval from the counterpart to be established and eventually start sharing information (Vitak et al., 2011) (meanwhile on Twitter such approval is not needed). Hence, Facebook is more helpful for the maintenance of existing social ties, including family, as well as current and old friends and as such may facilitate the connection with strong ties which constitute Bonding Social Capital. As a result, and contrary to Twitter, Facebook does not facilitate connection with people who are not known personally to the user, such as politicians, commentators, news organizations, journalists, or any opinion leader (Valenzuela et al., 2017). Arguably, drawing from strong homogenous ties, Facebook users may acquire more redundant information as close-knit networks will usually hold similar views and may provide equivalent information (Gil de Zuniga and Valenzuela, 2011).

The above discussion based on empirical evidence mirrors the results of our analysis, which indicates that the relationship between Online Bridging Social Capital and Offline Political Participation is very strong on Twitter (facilitates creation of weak ties and diffusion of more heterogenous information) and medium on Facebook (does not facilitate creation of weak ties and provide more homogenous redundant information) for heavy users. Yet, when it comes to the combined use of those SNSs the relationship is negative. Arguably this could be due to the fact that Facebook is more focused on entertainment and emotional expression (Groshek and Dimitrova, 2011; Towner, 2013) and, because of this, people tend to spend more time on it than on Twitter. Indeed, Facebook and Twitter combined users spent on average 56 minutes on Facebook and 29 on Twitter. Hence, they will be less exposed to information coming from weak ties and are more involved in strong ties activities which do not facilitate Offline Political Participation. Hence, content and intrinsic characteristics of SNSs also known as the technological affordance principle and the time displacement logic may explain the moderation effect of the combined use of SNSs on the relationship between Online Bridging Social Capital and Offline Political participation.

8.1.4. RESULTS INTERPRETATION AND DISCUSSION FOR Q4

To answer research question Q4 in Section 7.4 we presented the empirical results from two separate two-way (factorial) ANOVAs, designed to ascertain the extent to which levels of Online Bonding and Bridging Social Capital differ across different types of SNSs (Facebook and Twitter) and different levels of intensity of use. Findings are summarised in Table 8.4.

With respect to Online Bonding Social Capital, results showed a significant interaction between the amount of time spent on a SNS and the type of SNS used; levels of Online Bonding Social Capital reported on Facebook and Twitter are affected by the time spend on those SNSs.

Simple effects analysis revealed that when levels of intensity of use were kept constant, Online Bonding Social Capital was significantly higher for Facebook Users (M = 3.59, SD = 1.55) compared with Twitter users (M = 2.65, SD = 1.54), in the mild category. In the light and heavy category non-significant differences were found between Facebook and Twitter users with negligible effect sizes.

These results are partially consistent with the "Context Acquisition Principle" mentioned in the literature review (Coleman, 1998; Putnam, 2000) for which the amount and type of Social Capital differ according to the context in which it is acquired. Yet, it seems that this principle is dependent upon the level of intensity of use of the specific SNS considered.

Moreover, when analysing differences across intensity of use within the same type of SNS, that is differences in light, mild and heavy users across Facebook and then Twitter separately, light Facebook users (M = 2.78, SD = 1.57) showed significantly lower levels of Online Bonding Social Capital compared with both mild (M = 3.59, SD = 1.55) and heavy (M = 3.90, SD = 1.69) Facebook users. Mild and heavy Facebook users showed non significantly different levels of Online Bonding Social Capital.

Heavy Twitter users (M = 3.89, SD = 1.51) showed significantly higher levels of Online Bonding Social Capital when compared to both mild (M = 2.65, SD = 1.54) and light (M = 2.79, SD = 1.54) users. In contrast, light and mild Twitter users did not show significantly different levels of Online Bonding Social Capital.

Main Question	Subsidiary Question	Findings
Q4. To what extent do levels of Online Social Capital differ across different types of SNSs employed at different levels of their intensity of use?	Q4a. To what extent do the amounts of Online Bonding Social Capital differ across different types of SNSs employed, considering different levels of intensity of use of SNSs?	 A significant but small interaction effect was found between the amount of time spent on a SNS and the type of SNS used. When levels of intensity of use are kept constant, mild Facebook users show significantly <i>higher</i> levels of Online Bonding SC than mild Twitter users. Within the same type of SNS, mild and heavy Facebook users show significantly higher levels of Online Bonding SC compared with light Facebook users. Heavy Twitter users show significantly higher levels of Online Bonding SC than mild and light users.
	Q4b. To what extent do the amounts of Online Bridging Social Capital differ across different types of SNSs employed, considering different levels of intensity of use of SNSs?	 A significant and small to medium interaction effect was found between the amount of time a person spent on a SNS and the type of SNS used. When levels of intensity of use are kept constant, light Facebook users show significantly <i>lower</i> levels of Online Bridging SC than light Twitter users. Heavy Facebook users show significantly <i>higher</i> levels of Online Bridging SC than heavy Twitter users. Within the same type of SNS, all Facebook user groups showed significant differences among them. The more time people spend on Facebook, the greater is their Online Bridging SC. Twitter mild users showed significantly higher levels of Online Bridging SC than heavy and light users.

Table 8.4. Findings Summary for Research Question Q4.

The results in Table 8.4 suggest that the more time a person spends online, the more Online Bonding Social Capital he/she would have independently from the type of SNSs used.

The results undermine the pessimistic view that the Internet reduces or even destroys Social Capital (Putnam, 2000; Norris, 2001) and supports the positive view that the Internet in general, and SNSs in particular, can positively affect Online Social Capital depending on the intensity of use. Yet, the increase in Online Bonding Social Capital is not constant or equal across all the levels of intensity of use (light, mild and heavy), rather it changes accordingly to the type of SNS considered (Facebook or Twitter). Mild and heavy users of Facebook show similar levels of Online Bonding Social Capital but both groups significantly differ from light users. In contrast, heavy Twitter users show significant higher levels of Online Bonding Social Capital. These results support the context acquisition principle of Social Capital, but they also show that Social Capital is affected by the time spent on the online social network considered.

With respect to Online Bridging Social Capital, there is a significant interaction between the amount of time spent on a SNS and the type of SNS used. This indicates that levels of Online Bridging Social Capital reported by Facebook and Twitter users are affected differently by the time spent on those SNSs.

Simple effects analysis revealed that when the intensity of use is held constant, while studying the effects of the type of SNS used, Online Bridging Social Capital is significantly lower for light Facebook users (M = 2.12, SD = 1.05) compared with light Twitter users (M = 2.76, SD = 1.28). Moreover, heavy Facebook Users showed higher levels of Online Bridging Social Capital (M = 4.09, SD = 1.47) compared with heavy Twitter users (M = 3.15, SD = 1.40). Mild users of Facebook (M = 3.49, SD = 1.17) and Twitter (M = 3.60, SD = 1.57) showed similar levels of Online Bridging Social Capital.

These results are partially consistent with the already mentioned "Context Acquisition Principle" (Coleman, 1998; Putnam, 2000). However, as in the case of Online Bonding Social Capital, results are dependent upon the level of intensity of use of the specific SNS considered. Indeed, differences between Facebook and Twitter users on levels of Online Bridging Social Capital are significant for low and heavy levels of usage, but not for mild usage levels. Moreover, the type of SNS used makes a difference

in the amount of Bridging Social Capital owned, according to the amount of time spent on the specific SNS. Indeed, Twitter seems to facilitate the creation of Online Bridging Social Capital at lower levels of usage compared with Facebook. However, Facebook users show higher levels of Online Bridging Social Capital at higher levels of usage, particularly in the heavy category of users.

Furthermore, when analysing differences across intensity of use within the same type of SNS, that is differences in light, mild and heavy users across Facebook and then Twitter separately, we found that the level of Online Bridging Social Capital was significantly different at each level of intensity of use for Facebook users. Indeed, light Facebook users showed significantly lower levels of Online Bridging Social Capital (M = 2.12, SD = 1.05) than mild (M = 3.49, SD = 1.71) and heavy Facebook users (M = 4.09, SD = 1.47).

Mild Facebook users (M = 3.49, SD = 1.71) showed significantly lower levels of Online Bridging Social Capital when compared with heavy Facebook users (M = 4.09, SD = 1.47). This implies that the more time a person spends on Facebook, the more Online Bridging Social Capital he/she would get. This mirrors what we found for Online Bonding Social Capital in Facebook users and supports once again the positive view for which the use of the Internet in general and of Facebook in particular, can positively affect Online Social Capital (Brooks et al., 2011; Burke et al., 2010; Ellison et al., 2007; Ellison et al., 2011; Hofer and Aubert, 2013; Johnston et al., 2013; Lampe et al., 2013; Resnick, 2002), but depending on its intensity of use.

In contrast, for Twitter, only the mild users category showed significantly higher levels of Online Bridging Social Capital (M = 3.60, SD = 1.57) when compared with both light (M = 2.76, SD = 1.28) and heavy (M = 3.15, SD = 1.40) Twitter users. Light and heavy users did not significantly differ in levels of Online Bridging Social Capital. Also in this case, the intensity of use of social media positively affects Online Bridging Social Capital, but only to some extent. The increase in Online Bridging Social Capital is not constant or equal across all the levels of intensity of use. Specifically, light and heavy users of Twitter show similar levels of Online Bridging Social Capital but both groups have significantly lower levels than mild users.

Overall, these results suggest that the interaction between the type of SNS used and its intensity of use must be taken into account when analysing the effects of social media over different types of Social Capital, namely Online Bridging and Online Bonding. Indeed, although the two types of Social Capital show similar behavioural patterns on Facebook with respect to different levels of intensity of use (the more time spent online, the higher is the amount of Online Bonding and Bridging Social Capital reported), the same does not hold true for Twitter. Indeed, for Twitter, an increase in usage levels does not always provide a significant increase in the level of Bridging Social Capital. Moreover, for each level of intensity of use, the differences in the amount of Online Bonding and Bridging Social Capital between Facebook and Twitter can be substantial.

Arguably, the opposing contrasting effects found in previous research regarding the effect of the Internet and of SNSs over Social Capital (see Appendix F) can be due to the fact that those studies do not account for such interaction effect.

8.1.5. RESULTS INTERPRETATION AND DISCUSSION FOR Q5

In Section 7.5 we described the results of two separate non-parametric two-way ANOVAs on the levels of Offline and Online Political Participation reported by users of different SNSs (Facebook and Twitter), categorised according to the intensity of use (time spent online) of the social media employed. Findings are summarised in Table 8.5.

For Offline Political Participation, there is a significant interaction between the intensity of SNS use and the type of SNS. Although small, the interaction effect indicates that levels of Offline Political Participation reported by Facebook and Twitter users are affected differently by the time spent on those SNSs.

A simple effects analysis revealed that Twitter users show significantly higher levels of Offline Political Participation at each level of intensity of use (light, mild and heavy) compared with Facebook users. Therefore, regardless of the level of intensity of use Twitter users are more engaged in Offline Political Participation than Facebook users. This is not surprising, since previous studies investigating the nature of the relationship between Facebook use and Offline Political Participation found either no relationship (Gil de Zuniga et al., 2012; Valenzuela et al., 2009) or even a weak negative relationship (Vitak et al., 2011).

Main Question	Subsidiary Question	Findings
Q5. To what extent do levels of Political Participation differ across different types of SNSs employed, at different levels of their intensity of use?	Q5a. To what extent do levels of Offline Political Participation differ across different types of SNSs, considering different levels of Intensity of use of SNSs?	 A significant interaction effect was found between the amount of time a person spent on a SNS and the type of SNS used. The interaction had small /medium effect size. Across different levels of intensity of use, Twitter users always show significantly higher levels of Offline Political Participation than Facebook users. Within the same type of SNS, Twitter heavy users showed significantly higher levels of Offline Political Participation compared with all other users. No significant differences were found across Facebook users.
	Q5b. To what extent do levels of Online Political Participation differ across different types of SNSs, considering different levels of Intensity of use of SNSs?	 A significant, medium size interaction effect was found between the amount of time spent on a SNS and the type of SNS used. Across different levels of intensity of use, Twitter users show significantly higher levels of Online Political Participation than Facebook users, except for the light category where no differences were found. Within the same type of SNS, Facebook heavy user showed significantly higher levels of Online Political Participation compared with mild and light users. For Twitter, all users categories showed significantly different levels of Online Political Participation. The more time people spend on Twitter the more politically engaged they are online.

There is very limited extant knowledge, restricted mainly to voting, about the relationship between Twitter usage and Offline Political Participation. DiGrazia et al. (2013) and Towner (2013) found that the amount of attention received by candidates on Twitter is a statistically significant indicator of voting. Moreover, a recent study conducted by the Pew Research Center (2015) showed that some of Twitter's intrinsic features and characteristics may facilitate political engagement and especially voting (a traditional offline political activity).

Scholars have argued that the varying opposing effects of SNSs on Offline Political Participation are the result of differences in content, form, and function of those SNSs (Groshek and Dimitrova, 2011; Towner, 2013). Compared with Facebook, which is more focused on entertainment and emotional expression, Twitter seems to be centred around information and news diffusion and it seems to excel in terms of information richness and diversity compared with other SNSs. Indeed, many political news sources, including journalists, media organizations, politicians, civic activists, and ordinary people post news on Twitter. This circumstance, combined with the fact that 46% of Twitter users follow news organizations on Twitter (Pew Research Center, 2015), could arguably be one of the differentiation elements that would help denote the higher levels of Offline Political Participation compared with Facebook. Indeed, exposure to and consumption of political information have been closely related to Offline Political Participation (Gil de Zuniga et al., 2012).

Moreover, Twitter helps people to connect with more strangers than other social media (Zhang et al., 2010), especially compared with Facebook, where online networks predominantly mirror the offline networks of each individual user (Woo-Yoo and Gil de Zuniga, 2014). Connecting with strangers facilitates the creation of heterogeneous networks and "weak ties", therefore facilitating the creation of Bridging Social Capital (Putnam, 2000). Bridging Social Capital and heterogeneous networks are strongly related to political and civic engagement (Lake and Huckfeldt, 1998; McLeod et al., 1999). This suggests that the intrinsic characteristics of the SNS platforms people use to connect with one another affects the nature of the value derived from those relationships (weak ties versus strong ties).

Furthermore, Twitter allows users to remain anonymous as they do not need to disclose themselves to express their opinions (Huberman et al., 2009; Woo-Yoo and Gil

de Zuniga, 2014). Arguably, anonymity could encourage greater participation online because it resolves the social desirability bias and some people may be willing to publicly take a stand that can be opposed by friends and associates (Collins and Butler, 2003). Yet, interactions in social media permeate to the offline world and facilitate face-to-face political coordination that fuels offline political action (Bakker and de Vreese, 2011; Bennett and Segerberg, 2012; Bennett et al., 2014; Eltantawy and Weist, 2011; Gleason, 2013; Howard and Parks, 2012; Lim, 2012, 2013; Theocharis et al., 2015).

With respect to the intensity of use of SNSs, the results of the simple effect analysis suggest that the intensity of use of a specific SNS affects levels of Offline Political Participation within Twitter users.

However, only Twitter heavy users (M = 2.07, SD = 2.35) showed significantly higher levels of Offline Political Participation when compared with both light (M = 1.09, SD = 1.14) and mild (M = 0.93, SD = 1.51) Twitter users. Light and mild users did not differ.

Therefore, we can speculate that the type of SNS used does affect how the time spent on that SNS influences political activities conducted offline. Light, mild and heavy users of Facebook show similar levels of offline political engagement. Hence, the time spent on Facebook does not have any relevant effects on offline political activities. This is in line with empirical evidence provided in previous studies that argued that hours spent online had either a weak (Bakker et al., 2011) or a negative effect on Political Participation (Vitak et al., 2010). This again could mirror the intrinsic characteristics of Facebook, which is seen as a type of SNS centred on recreational and entertainment activities. These activities could arguably take the place of more meaningful civic and political pursuits in a logic of "time displacement", according to which time spent on the Internet reduces the time available to carry out other social behaviours (Nie and Erbring, 2002; Putnam, 1995, 2000). In contrast, the more time people spend on Twitter, the more they seem to participate in offline political activities, denying the pessimistic view of the "time displacement logic" for Twitter.

The reason could be that "*the more time people spend surfing the net, the more likely they are to access political web sites or news*" (Quintelier and Visser, 2008: p. 413), which in turn would increase the chance of engaging in offline political activities (Gil de Zuniga et al., 2012). This is particularly true for Twitter whose content is much more

centred around political news compared with Facebook (Kwak et al., 2010). Moreover, as mentioned earlier, Twitter allows people to form more heterogenous networks as they would connect with more strangers than on Facebook (Zhang et al., 2010). Heterogeneous networks can stimulate people to participate politically and to debate and change their opinions (Druckman and Nelson, 2003; Mutz, 2002) as they would be more exposed to different and even contrasting information sources. The more heterogenous a network is, the more likely people are to participate in politics (Teorell, 2003).

Overall, the findings suggest that the impact of the type of SNS used on Offline Political Participation aligns with the unique intrinsic characteristics and features of each type of social media. Regardless of the intensity of use, Twitter users are more engaged in offline political activities compared with Facebook users. However, within Twitter, users differ in the level of engagement showed according to the time they spend on this SNS. Indeed, findings suggest that the more time people spend on Twitter the more they tend to engage in offline political activities. However, this finding does not hold true when considering Facebook.

With regards to Online Political Participation, this research found a significant interaction between the amount of time spent on a SNS and the type of SNS used. This indicates that levels of Online Political Participation reported on Facebook and Twitter are affected differently by the time spent on those SNSs.

Simple effects analysis revealed a significant effect of Type of SNS used in all intensity of use categories, except the light one. Significant effects were found in the mild category between Facebook (M = 0.78, SD = 1.22) and Twitter users (M = 1.21, SD = 1.58), and in the heavy category where Facebook users (M = 1.40, SD = 1.43) showed significant lower levels of Online Political Participation compared with heavy users of Twitter (M = 3.43, SD = 2.49). These results mirror the rationale provided earlier for Offline Political Participation, based on the intrinsic features and characteristics of Facebook and Twitter. Indeed, Twitter users show higher levels of Online Political Participation compared with Facebook users, but only at the mild and heavy level of usage of the two SNSs. At the light level there are no significant differences. This again could be due to the higher political informational content of Twitter when compared with Facebook, which may boost the online political activities of its users. Indeed, compared with any other SNSs, Twitter is more used by politicians to disseminate information and

carry out online activities like advertising their political platforms and listening to citizens feedback (Golbeck et al., 2009; Grant et al., 2010; Grčić et al, 2017). Yet, it seems that the effect of the type of SNS used kicks in only at the mild level of usage. At lower usage levels, using Facebook or Twitter would make no difference in levels of online political activities performed. Hence, the level of online political engagement in Twitter and Facebook is dependent upon the intensity of use of those SNSs.

Indeed, we found a significant simple effect of levels of intensity of use on both Facebook and Twitter users. Specifically, the level of Online Political Participation was significantly higher for heavy users (M = 1,40, SD = 1.43) compared with both mild (M = 0.78, SD = 1.22) and light (M = 0.88, SD = 1.30) users of Facebook. In contrast, light and mild Facebook users showed non-significant differences in levels of Offline Political Participation. This implies that the more time is spent on Facebook, the higher the engagement in online political activities. However, this is true only if the users reach a very high level of usage.

For Twitter, all the groups were significantly different from each other. Indeed, heavy users (M = 3.43, SD = 2.49) showed significantly higher levels of Online Political Participation when compared with both mild (M = 1.21, SD = 1.58) and light (M = 0.70, SD = 0.50) Twitter users. Moreover, also light (M = 0.70, SD = 0.50) and mild (M = 1.21, SD = 1.58) users significantly differed in levels of Online Political Participation. This implies once again that the more time people spend on Twitter, the more engaged they are in online political activities, mirroring the results of earlier studies (e.g., Hao et al., 2014; Skoric and Kwan, 2011; Skoric and Poor, 2013; Velenzuela et al., 2009).

Overall, the above discussion suggests that the interaction of the type of SNS used with the intensity of use of those SNSs must be considered when studying the effects of social media on Online and Offline Political Participation. While Twitter users generally show higher levels of engagement in both offline and online political activities, their level of online participation tends to be lower (although non-significantly) than that of Facebook users at the light level of usage. Moreover, while different levels of intensity of Twitter use contribute to a significant difference in the amount of online and offline political activities, the same does not hold true for Facebook. Indeed, for the latter, an increase in usage levels does not provide a significant increase on the level of offline political engagement which is similar for light, mild and heavy users. Conceivably, the opposing contrasting effects found in previous studies regarding the effect of the Internet, and SNSs in particular, over Political Participation (Boulianne, 2009; Skoric and Zhu, 2015; Theocharis and Lowe, 2015) can be due to lack of consideration for the above mentioned interaction effect.

8.2. CONTRIBUTION TO THEORY AND PRACTICE

This thesis contributes to theoretical knowledge in different academic fields. First, this research advances prior work in the domain of Engagement and Mobilisation Theories by expanding the Resource Mobilisation Theory framework (McCarthy and Zald, 1977) through the moderating effects of the use of different types of SNSs (Facebook and Twitter) in accordance with the principles of Technology Affordance and Technological Determinism. Indeed, our findings suggest that the relationships between Online Bonding and Bridging Social Capital and Political Participation both Online and Offline works differently in Facebook and Twitter.

Secondly, although several studies had already investigated the links between SNSs and citizens' Political Participation, this thesis differs from those investigations in three ways. Firstly, most extant research focuses heavily on the campaign environment (e.g., Dimitrova et al. 2014; Halpern et al., 2017; Rice et al. 2013) to the point that Boulianne (2019) called for more research outside of the electoral context. This research was conducted in a non-electoral period and thus the contributions of the SNSs use to citizens' Political Participation were assessed independently of the electoral process, which usually sees a rise in Political Participation and therefore might yield biased results (Jensen and Anduiza 2012; Larson 2016). Furthermore, this thesis addressed two conceptual weakness characterising many SNSs and Political Participation studies that is an over-generalised approach to the effects of SNSs use (Boulieanne, 2019; Dimitrova, et al., 2014; Valenzuela, 2017; Woo-Yoo and Gil de Zuniga, 2014) and the failure to consider Political Participation as a multidimensional phenomenon (Boulianne, 2015; Gil de Zuniga et al., 2016). Indeed, many earlier studies either did not account for structural differences of the Political Participation construct or failed to carry out any differential comparisons of the SNSs effects on different forms of Political Participation. Indeed, to the best of our knowledge, only two studies (Halpern et al., 2017; Woo-Yoo and Gil de Zuniga, 2014) compared the effects of different types of SNSs on both Online and Offline Political Participation but did not consider the distinction between the Bonding and Bridging dimensions of Social Capital. This leads to the third differentiating aspect of this thesis, that is the differential analysis of the effects of two dimensions of Online Social Capital (Bonding and Bridging) on two domains of Political Participation (Online/ Offline) rendering a more comprehensive and nuanced picture of the relationships between those variables, while accounting for the different technological affordance of different SNSs and their intensity of use.

Third, this research advances knowledge in the fields of both Online Social Capital and Online and Offline Political Participation by delving into more nuanced connections between them and the use of different SNSs, introducing interaction effects between the different type of SNS used and the time spent on those SNSs which may explain the contrasting empirical evidence in the extant literature (see Chapters 3 and 4). Indeed, our results showed that an interaction effect does exist between the type of SNS employed and the time people spend on that SNS when measuring levels of Bonding/Bridging Social Capital and levels of Offline/Online Political Participation. Our findings highlight the sterility of the debate between different and opposing school of thoughts about the effects of SNSs on Social Capital and Political Participation, namely optimists, normalisers, and pessimists. Contrasting empirical evidence could be due to the lack of consideration of different technological affordances of different SNSs and to a trend towards generalisation of results from one SNS to the entire category of Social Media.

Fourth, this thesis introduces a novel perspective to Political Marketing studies which so far have been heavily criticised for their lack of engagement with theories of democracy, ethics and sociology (Henneberg et al., 2009; O'Shaughnessy, 2001; Ormrod et al., 2013). This shift was advocated by many academics (Henneberg et al., 2004; Ormrod et al., 2013; Savigny and Wring, 2009) and is in line with the Comprehensive Political Marketing (CPM) approach (Lees-Marshment, 2001, 2003) which encourages the creation and integration of Political Marketing Theories with concepts originating from other disciplines. Employing a concept originating from the sociological field like Social Capital as the main element of investigation would provide a new theoretical framework for researchers in Political Marketing that could advance knowledge in engagement and targeting theories meanwhile maximising social, ethical and democratic engagement (see Chapters 3 and 4).

Fifth, this thesis expands on previous SNSs studies in the domain of Media and Communication that have argued that social media affordances facilitate specific political behaviour (Woo-Yoo and Gil de Zuniga, 2014; Halpern, 2017). Integrating the Affordance thesis with the intensity of use of SNSs allowed to test several competing media theories related to media usage against each other (e.g., Time Displacement Theory and Channel Complementarity Theory) in different SNSs and for different types of Political Participation (Online/ Offline).

Moreover, studies on Online Social Capital and Political Participation have been mostly conducted in the US context, while "*empirical work in European countries is still scarce*" (Gustafsson, 2012: p. 1112). Therefore, researchers called for new empirical evidence coming from other contexts, since "*by integrating empirical work made in different national, cultural and demographic context into the existing literature, comparative research will be made easier*" (Gustafsson, 2012: p. 1112). This study answers the call of those researchers and tries to also fill this geographical gap.

With regards to the practical implications of this thesis, contrasting the effects of the affordances of two different SNSs may help practitioners to develop a framework capable of indicating which type of Political Participation (Online/ Offline) can be facilitated by each platform. From a Political Marketing perspective this is relevant in relation to the implementation of effective engagement and targeting strategy of the electorate through the right SNSs for the right type of behaviour. Indeed, our findings suggest that the SNS people use may alter the way in which they participate in the political process affecting the same outcomes of the political process. Understanding whether a specific SNS can enable or inhibit offline or online political behaviour can facilitate marketers' strategic choices with respect to the objectives of their political campaign. Also, understanding how users of different social media have different levels and types of Social Capital would further help Political Marketers for their targeting strategies, given that different types of Social Capital (Bonding/Bridging) have been found to affect Online and Offline Political Participation differently (Boulianne 2015; Skoric and Zhu, 2016).

Finally, political marketers should account for differences in thematic focus and functions of the SNS that they use to reach people since they can affect the relationship between Social Capital and Political Participation. Indeed, the results of the moderation analysis suggest that the relationship between Social Capital and Political Participation is not a homogeneous entity with respect to the type and context of the network in which people are involved online. This should help political marketers to set more efficient targeting strategies.

8.3. LIMITATIONS

Although this thesis followed best practice and academic rigour in implementing the research process, there are limitations that need to be acknowledged. Such limitations range from issues related to theoretical considerations, sampling strategy and crossnational design, to variables measurement and statistical techniques of analysis.

With regards to *theoretical considerations*, our approach corroborates the claim that Social Capital is an antecedent of Political Participation. Although such causal path was established based on theoretical approaches and previous empirical studies (Gil de Zuniga et al., 2012; Zhang and Chia, 2006; Teorell, 2003), we acknowledge that only longitudinal studies can prove directions of influence (Bryman, 2012). However, due to financial constraints, this study employed a cross-sectional design. Moreover, the assumption that collaborative social norms of trust and reciprocity underpins citizens' political behaviour ignores that most political actions are inherently built on conflict over ideas and competition over scarce resources. Yet, following Lin (2001) we conceptualised Social Capital as a resource which people contend and whose accrual would ultimately bring to political mobilisation.

Moreover, we theorised that the impact of SNSs use on Social Capital and Political Participation is structurally caused (based on the principle of technological affordance) rather than content driven. Hence, we did not distinguish between different motivational use of the SNSs (e.g., informational, entertainment, educational, etc.). However, we acknowledge that SNSs might promote Social Capital and Political Participation by providing information (Gil de Zuniga et al., 2012; Mutz, 2006; Valenzuela et al., 2009; Valenzuela et., 2017; Vitak et al., 2011) or when used for civic or political purposes (Dimitrova et al. 2014). Yet it must be noted that our approach was framed in a technological determinism view. Accordingly, the type of information shared and accessed in each social network is inherent to the structure of the SNS itself (e.g., more homogenous information is accessed on Facebook and more diverse in Twitter due

to different algorithm setting of the two SNSs). Consequently, our results of the moderation analysis on the relationship between Social Capital and Political Participation are limited to the structural aspect of the SNSs which, however, can shape multiple factors ranging from information to SNSs' content related aspect. Also, we did not consider any links between Offline and Online Political Participation even though Online Political Participation was found to permeate to the offline world and facilitate face-to-face political coordination that fuels offline political action (Bennett et al., 2014; Eltantawy and Weist, 2011; Gleason, 2013; Howard and Parks, 2012; Lim, 2012, 2013; Theocharis et al., 2015). However, this was beyond the objective of this research whose final aim was to explore the relationship between Online Social Capital and Offline/Online Political Participation. Moreover, although previous research established that citizens' political identity and psychological orientations can exert a strong effect on their willingness to join and participate in civic and political activities we did not control for those variables in conceptual model. This was done to reduce complexity, following the principle of parsimony (Field, 2013).

With regards to *sampling* and *research design*, this study suffers from a self-selection bias due to the administration of an online survey within a cross sectional design and the use of web-panel sample. Indeed, only those people who are willing to enlist in the panel sample could be reached. This posed a limit to the generalisability of the study and therefore to its external validity (Bryman, 2012; Rea and Parker, 2014). Furthermore, the use of online self-completion questionnaires may affect the ecological validity of the research because the very instruments disrupt the "natural habitat" of the respondents (Bryman, 2012). Yet, the degree of ecological validity of online self-completion questionnaires like experiments, focus groups or face-to-face interviews. Furthermore, cross-sectional studies are typically weak in terms of internal validity. However, we reduced internal validity problems by adopting established theoretical frameworks to build our conceptual model. Moreover, as highlighted in Boulianne (2015) and Skoric et al. (2016) these limitations are common in studies in the field of SNSs use, Social Capital and Political Participation which make large use of cross-sectional design, survey instruments and web panel samples.

With regards to *variable measurements*, we must highlight that the intensity of use of SNSs was measured through a self-reported measure of time spent on social media in hours/minutes of usage per day. We acknowledge that this measurement method is not

flawless (Junco, 2013; Olufadi, 2015; Rosen et al., 2013). Indeed, people may not be able to accurately estimate, or may not remember, the amount of time spent on a SNS. Hence, respondents may provide answers which they feel would reflect well on them or others, increasing the "*social desirability bias*", rather than providing the real time spent on digital platforms. Junco (2013) supports this argument, showing that a better method consists of recording the time spent online through a monitoring software installed on the PC of the respondents. Yet this option was not feasible in this study considering the high number of respondents involved in the survey and the researcher's financial constraints. Moreover, Junco (2013) found that the correlation between self-reported time and actual time was significantly high, while the social desirability bias is usually low in selfadministered surveys because of the absence of an interviewer (Bryman, 2012). Yet the results of self-reported measures must always be carefully interpreted.

Moreover, the same variable "Intensity of Use" was trichotomized through the quartile split method to allow for groups comparison to answer research questions Q1, Q3, Q4 and Q5. Notably, trichotomizing a ratio continuous variable presents some drawbacks (Henseler and Fassott, 2010). First, observations that cannot be unambiguously allocated to a single group are dropped off from the analysis. Indeed, the quartiles values are excluded from the analysis as they represent the cut-off points that cannot be unambiguously assigned to any of the groups created. Hence, information is lost. However, such loss of information is a common characteristic of other splitting procedures like the mean or the median split methods. Hence, the choice of the procedure to use should not be focused on the amount of information that could be lost. Rather, the researcher should consider the shape of the distribution of the variable to be split and the possible presence of outliers. Focusing on these aspects, the quartile split method was preferred to the other methods as it is the split procedure suggested when the data is non normally distributed around the mean and there are outliers in the dataset like in this research. Secondly, due to the trichotomization, part of the moderator variable's variance was lost and this could lead to possibly misleading results. However, this is once again a problem of all the split method mentioned above. Hence, the aim here is to acknowledge the limitations of the splitting procedure as a category rather than of the specific method employed.

Moreover, Offline Political Participation did not include the voting activity at the last general election among the measuring items. Yet, the data for this study were gathered in a non-electoral period and thus citizens' Political Participation was assessed independently of the electoral process, which usually sees a rise in Political Participation and therefore is considered to bias survey results (Jensen and Anduiza 2012; Larson 2016). Following Bryman (2012) suggestions, when employing self-administered completion questionnaires, the memory of respondents should not be stretched too much. Considering that data was gathered between January and February 2019 and that the last election before that time was held in June 2017, we dropped the voting item from the Offline Political Participation scale.

Furthermore, although we tried to isolate platform effects to test for moderation of technological affordance of different SNSs we acknowledge that this differentiation is arbitrary and too simplistic. Indeed, according to Boulianne (2019), 52% of Internet users use multiple SNSs, meaning that a "pure" SNS effect is very difficult to measure. Yet this limitation is common to most of the studies on SNSs.

With regards to the statistical techniques of analysis employed, we employed the two-stage approach to answer research question Q2, as recommended by Hair et al. (2017) and Henseler and Chin (2010). Yet this method does not maximise predictive accuracy of the model. Hence, the sensitivity analysis conducted in Section 7.2 on the in-sample and out-of-sample measures of predictive accuracy could be biased as it could provide lower estimates of those metrics. Moreover, prior research indicates that the permutation test is not quite reliable when samples are very different in size like in the case of the analysis conducted to answer research question Q3 (Hair et al., 2018: p. 154). Although a sensitivity analysis was conducted through the employment of the PLS-MGA procedure (Henseler et al., 2009) and the Confidence Set Approach (Sarstedt et al., 2011) which are more robust to unequal sample sizes, it must be acknowledged that also those techniques have some limitations. In particular, the PLS-MGA test is a one-sided test. Using it to test two-sided hypotheses, like in this case, requires subtracting the obtained p value of the test from 1 which would allow to find the effect on the other direction. Yet the bootstrapping distributions are not necessarily symmetrical. Hence the mathematical subtraction of the p value from 1 can provide misleading results (Hair et al., 2018). Therefore, we employed also the Confidence Set Approach which however is most effective when sample sizes are small (Sarstedt et al., 2011). Yet, carrying out a sensitivity analysis through PLS-MGA and Confidence Set Interval combined increased confidence on our findings as those tests yielded identical result for research question Q3.

8.4. FURTHER RESEARCH

Despite the contribution of this thesis to theoretical and practical fields, there is still more room for further research around Online Social Capital and Online and Offline Political Participation.

This study distinguished between Online and Offline Political Participation and assessed if the use of Facebook and Twitter can affect the relationship between Online Bridging and Bonding Social Capital and those two different arrays of political activities. Future studies could distinguish between Offline Bonding and Bridging Social Capital or could investigate the moderating effects of other types of SNSs. In this thesis the Online/Offline distinction of Social Capital was ignored despite empirical evidence suggesting that each of them can facilitate different types of Political Participation (Boulianne, 2015). This is because the main aim of this study was to focus on online interactions. Moreover, only Facebook and Twitter were considered in this research because at the time in which the study was conducted they were the most used SNSs in UK (www. yougov.co.uk, 2018). Not including other SNSs and thus further groups of users in the analysis was a conscious choice necessitated by the complexity of the study. Indeed, the present research dealt with a wide array of issues: it looked at the links between Bonding and Bridging Social Capital and Offline and Online Political Participation and the moderation effects of the Type of SNSs used, their intensity of use and their combined and not combined use. Considering the intricacy of this project, adding other variables and groups of users would have not met the principle of research parsimony (Field, 2013).

Moreover, this study has focused on the individual dimensions of Social Capital and has not looked at how its collective dimension can affect Political Participation. Indeed, Political Participation is also affected by cultural as well as attitudinal factors (Inglehart. 1997; Moy et al., 2012; Putnam, 2000; Ryfe, 2001). This is linked to difficulties in operationalising something as fuzzy as culture, and the resources required to assess its influence on behaviours and attitudes. For instance, Inglehart (1997) was able to assess the effect of cultural factors on participation thanks to the vast data set provided by World Values Surveys, which covered approximately 60,000 participants in 43 countries, mirroring 70% of the world's population. However, not every study can count on such an impressive data set. Considerations of the sampling strategy and methodology adopted in the present study have guided the choice of focusing on the individual level.

Future research into social media may also benefit from comparing multiple populations with different demographic or cultural differences. This type of comparative research may help shed light on the social factors that play a role in influencing different demographics to become civic citizens. For example, there might be different effects of media structures on Social Capital in countries with more authoritarian governments and collectivist cultures. Further research comparing culturally different nations across continents with different media systems and their impact on Social Capital might be insightful. Moreover, this thesis focused on the adult population of the UK, but it is possible that different subsets of the population could exhibit different patterns of Social Capital and/or Political Participation. Future research might focus on specific subgroups of populations; for example, young people could be especially likely to form Social Capital through social media sites because they use them more frequently, while older people, on the other hand, could be more likely to build offline Social Capital and engage offline.

Furthermore, the assessment of the effect of the use of SNSs at the individual level should be more nuanced to capture the diversity of social media use. Thus, the analysis of different online activities facilitated by different types of affordances would allow a better and more differentiated understanding of the SNSs' impact on Social Capital and Political Participation (Bakker and de Vreese, 2011). Indeed, previous studies found that users with greater concern over their privacy limited the amount of information they posted on SNSs, and this in turn affected the amount of Social Capital they possessed (Ko and Kuo, 2009; Vitak, 2012). Those who disclosed more information reported greater Bonding and Bridging Social Capital. Yet, no studies have addressed whether this is true for all SNSs given that they provide different levels of anonymity. While we cannot claim to have adequately measured the impact of SNSs use, we were able to explore and compare the potential of different platforms on Political Participation and Social Capital.

REFERENCES

Adler, P. S. and Kwon, S.W. (2002). 'Social capital: Prospects for a new concept', *Academy of Management Review*, 27: pp. 17–40.

Ahn, J. (2012). 'Teenagers experiences with social network sites: Relationships to bridging and bonding social capital', *The Information Society*, 28(2): pp. 99-109.

Akter, S., Fosso Wamba, S. and Dewan, S. (2017). 'Why PLS-SEM is suitable for complex modelling? An empirical illustration in big data analytics quality', *Production Planning & Control*, 28(11-12): pp. 1011-1021.

Alateeq, D., Alhadi, A., AlAteeq, F., Alsubie, A., AlEesa, D., AlShareef, S. and Omair, A. (2016). 'Psychological well-being among social network users of King Saud University students', European Psychiatry, 33(S1): pp. 571-581.

Albrecht, S. (2006). 'Whose voice is heard in online deliberation?: A study of participation and representation in political debates on the internet', *Information, Communication & Society*, 9(1): pp. 62–82.

Aldridge, S., Halpern, D. and Fitzpatrick, S. (2002). 'Social capital: A discussion paper'. London: Performance and Innovation Unit.

Alesina, A. and Giuliano, P. (2011). 'Family ties and political participation', *Journal of the European Economic Association*, *9*(5): pp. 817-839.

Alessandrini, M. (2006). 'Getting connected: can social capital be virtual?', *Webology*, *3*(4): pp. 1-8.

Almond, G. A. and Verba, S. (1963). 'The civic culture: Political attitudes in five western democracies', *Princeton: Princeton University Press*.

Alves, H., Ferreira, J. and Fernandes, C., I. (2016). 'Customer's operant resources effects on co-creation activities', *Journal of Innovation & Knowledge*, 1(2): pp. 69-80.

American Marketing Association (2004). 'What is Marketing?—The Definition of Marketing', AMA.

Ancu, M. and Cozma, R. (2009). 'MySpace politics: Uses and gratifications of befriending candidates', *Journal of Broadcasting & Electronic Media*, 53: pp. 567-583.

Anderson, J., C. and Gerbing, D., W. (1988). 'Structural equation modeling in practice: A review and recommended two-step approach', *Psychological bulletin*, *103*(3): pp. 411-423.

Anduiza, E., Perea, E. A., Jensen, M. J. and Jorba, L. (2012). 'Digital media and political engagement worldwide: A comparative study', *Cambridge University Press*.

Anim, P.A., Asiedu, F.O., Adams, M., Acheampong, G. and Boakye, E. (2019). 'Mind the gap: to succeed in marketing politics, think of social media innovation'. *Journal of Consumer Marketing*.

Antoniadis, I. and Charmantzi, A. (2016). 'Social network analysis and social capital in marketing: theory and practical implementation', *International journal of technology marketing*, 11: pp. 344-359.

Appel, L., Dadlani, P., Dwyer, M., Hampton, K., Kitzie, V., Matni, Z.A., Moore, P. and Teodoro, R. (2014). 'Testing the validity of social capital measures in the study of information and communication technologies', *Information, Communication & Society*, *17*(4): pp. 398-416.

Araujo, T., Wonneberger, A., Neijens, P. and De Vreese, C. (2017). 'How much time do you spend online? Understanding and improving the accuracy of self-reported measures of internet use', *Communication Methods and Measures*, *11*(3): pp. 173-190.

Ardèvol-Abreu, A., Diehl, T. and Gil de Zuniga, H. (2019). 'Antecedents of internal political efficacy incidental news exposure online and the mediating role of political discussion'. *Politics*, 39(1): pp. 82-100.

Ayios, A., Jeurissen, R., Manning, P. and Spence, L.J. (2014). 'Social capital: a review from an ethics perspective', Business Ethics: A European Review, 23(1): pp.108-124.

Bachmann, I., Kelly, K., Lewis, S. C. and Gil De Zuniga, H. (2010). 'News platform preference: Advancing the effects of age and media consumption on political participation'. *International Journal of Internet Science*

Bagozzi, R.P. (1975). 'Marketing as exchange', Journal of marketing, 39: pp. 32-39.

Bagozzi, R.P. (1978). 'Marketing as exchange: a theory of transactions in the marketplace', *American Behavioral Scientist*, 21: pp. 535-556.

Bagozzi, R.P. and Yi, Y. (2012). 'Specification, evaluation, and interpretation of structural equation models', *Journal of the academy of marketing science*, *40*(1): pp. 8-34.

Baines, P.R. and Egan, J. (2001). 'Marketing and political campaigning: mutually exclusive or exclusively mutual?', *Qualitative Market Research: An International Journal*, 4(1): pp.25 - 34.

Baines, P.R., Brennan, R. and Egan, J. (2003). 'Market Classification and Political Campaigning: Some Strategic Implications', *Journal of political marketing*, 2: pp.47-66.

Bakker, T.P. and De Vreese, C.H. (2011). 'Good news for the future? Young people, Internet use, and political participation', *Communication research*, 38: pp.451-470.

Ballantyne, D. and Varey, R.J. (2006). 'Creating value-in-use through marketing interaction: the exchange logic of relating, communicating and knowing'. *Marketing theory*, 6: pp. 335-348.

Bannon, D.P. (2005). 'Relationship marketing and the political process', *Journal of Political Marketing*, 4: pp.73-90.

Barber, B.R. (1984). *Strong democracy: participatory politics for a new age*, Berkeley: University of California Press.

Barclay, D.W., Higgins, C.A. and Thompson, R. (1995). 'The partial least squares (pls) approach to causal modeling: personal computer adoption and use as an illustration', *Technology Studies*, 2: pp. 285–323.

Barkhuus, L. and Tashiro, J. (2010). 'Student socialization in the age of facebook', In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems* ACM: pp. 133-142.

Baron, R.M., and Kenny, D.A. (1986). 'The moderator–mediator variable distinction in social psychological research: Conceptual, strategic, and statistical considerations', *Journal of Personality and Social Psychology*, 51(6): pp.1173–1182.

Barrett, M. and Brunton-Smith, I.R. (2014). 'Political and civic participation: Towards an integrative perspective', *Journal of Civil Society*, 10 (1): pp. 5-28.

Bauernschuster, S., Falck, O. and Wößmann, L. (2011). 'Surfing Alone? The Internet and Social Capital: Evidence from an Unforeseen Technological Mistake', SOEP papers on Multidisciplinary Panel Data Research (No. 392).

Baumgartner, J.C. and Morris, J.S. (2010). 'MyFaceTube politics: Social networking web sites and political engagement of young adults', *Social Science Computer Review*, 28: pp. 24-44.

Beaumont, E. (2010). 'Political agency and empowerment: Pathways for developing a sense of political efficacy in young adults', in L. R. Sherrod, J. Torney-Purta, and C. A. Flanagan (Eds.), *Handbook of research on civic engagement in youth*, Hoboken, NJ: John Wiley and Sons, Inc: pp. 525-558.

Becker, M.,W., Alzahabi, R. and Hopwood, C.J. (2012). 'Media multitasking is associated with symptoms of depression and social anxiety', *Cyberpsychology, Behavior, and Social Networking*, 16(2): pp. 132–135.

Becker, T.E. (2005). 'Potential problems in the statistical control of variables in organizational research: A qualitative analysis with recommendations', *Organizational research methods*, 8(3): pp. 274-289.

Becker, T.E., Atinc, G., Breaugh, J.A., Carlson, K.D., Edwards, J.R. and Spector, P.E. (2016). 'Statistical control in correlational studies: 10 essential recommendations for organizational researchers', *Journal of Organizational Behavior*, *37*(2): pp.157-167.

Beckman, R. J. and Cook, R. D. (1983).' Outliers'. Technometrics, 25(2): pp. 119-149.

Beldad, A.D. (2021). 'Sustaining social capital online amidst social distancing during the COVID-19 pandemic: web-based communities, their mitigating effects, and associated issues', *International journal of web based communities*, *17*(1): pp.35-52.

Bell, E., Schermer, J., and Vernon, P. (2009). 'The Origins of Political Attitudes and Behaviours: An Analysis Using Twins', *Canadian Journal of Political Science / Revue Canadienne De Science Politique*, 42(4): p. 855-879.

Bennett, S.E. (1984). 'Apathy in America, 1964-1982: A new measure applied to old questions', *Micropolitics*, *3*(4): pp.499-545.

Bennett, W.L. and Iyengar, S. (2008). 'A new era of minimal effects? The changing foundations of political communication'. *Journal of communication*, 58: pp. 707-731.

Bennett, W.L. and Segerberg, A. (2012). 'The logic of connective action: Digital media and the personalization of contentious politics', *Information, Communication & Society*, 15(5): pp. 739–768.

Bennett, W.L., Segerberg, A. and Walker, S. (2014). 'Organization in the crowd: Peer production in large-scale networked protests', *Information, Communication & Society*, 17(2): pp. 232–260.

Bentler, P.M. and Chou, C. (1987). 'Practical issues in structural model modeling', *Sociological Methods and Research*, Vol. 16: pp. 78-117.

Berk, R. (2008). 'Statistical learning from regression perspective', Springer, New York.

Bimber, B. (1994). 'Three Faces of Technological Determinism', in Merritt Roe Smith and Leo Marx (eds), *Does Technology Drive History? The Dilemma of Technological Determinism*, Cambridge, MA: MIT Press: pp. 80–100.

Bimber, B. (1998). 'The Internet and political mobilization', *Social Science Computer Review*, 16: pp. 391-401.

Bimber, B. (1999). 'The Internet and citizen communication with government: does the medium matter?', *Political Communication*, 16(4): pp. 409–428.

Bimber, B. (2001). 'Information and political engagement in America: The search for effects of information technology at the individual level', *Political Research Quarterly*, 54: pp.53-67.

Bimber, B. (2003). 'Information and American democracy: Technology in the evolution of political power', *Cambridge University Press*.

Bimber, B., Flanagin, A.J. and Stohl, C. (2005). 'Reconceptualizing collective action in the contemporary media environment'. *Communication theory*, 15: pp. 365-388.

Bimber, B., Cantijoch Cunill, M., Copeland, L. and Gibson, R. (2015). 'Digital media and political participation: The moderating role of political interest across acts and over time', *Social Science Computer Review*, 33(1): pp. 21-42.

Bishop, P.A. and Herron, R.L. (2015). 'Use and misuse of the likert item responses and other ordinal measures', *International journal of exercise science*, 8(3): pp. 297-302.

Blumenthal, S., (1980). '*The Permanent Campaign: Inside the World of Elite Political Operatives*', Boston, MA: Beacon.

Blumler, J.G. and Kavanagh, D. (1999). 'The Third Age of Political Communication: Influences and Features', *Political Communication* (16): pp. 209-230.

Bode, L. (2012). 'Facebooking it to the polls: A study in online social networking and political behavior', *Journal of Information Technology & Politics*, 9: pp. 352-369.

Bode, L., Vraga, E.K., Borah, P., and Shah, D.V. (2014). 'A new space for political behavior: Political social networking and its democratic consequences', *Journal of Computer-Mediated Communication*, *19*: pp. 414–429.

Bohn, A., Buchta, C., Hornik, K. and Mair, P. (2014). 'Making friends and communicating on Facebook: Implications for the access to social capital', *Social Networks*, *37*: pp. 29-41.

Bollen, K.A., (1984). 'Multiple indicators: Internal consistency or no necessary relationship?', *Quality and Quantity*, *18*(4): pp. 377-385.

Bollen, K.A. (1989). '*Structural Equations with Latent Variables*', New York: Wiley-Interscience Publication.

Bonfadelli, H. (2002). 'The Internet and knowledge gaps: A theoretical and empirical investigation', *European Journal of communication*, 17: pp. 65-84.

Boone, H., N. and Boone, D., A., (2012). 'Analyzing Likert data', *Journal of* extension, 50(2): pp. 1-5.

Borgatti, S.P. and Foster, P.C. (2003). 'The network paradigm in organizational research: A review and typology', *Journal of management*, 29(6): pp. 991-1013.

Boulianne, S. (2009). 'Does Internet Use Affect Engagement? A Meta-Analysis of Research;, *Political Communication*, 26(2): pp. 193-211.

Boulianne, S. (2015). 'Social media use and participation: A meta-analysis of current research', *Information, communication & society*, 18: pp. 524-538.

Boulianne, S. (2019). 'Revolution in the making? Social media effects across the globe', *Information, communication & society*, 22: pp. 39-54.

Boulianne, S. and Theocharis, Y. (2020). 'Young people, digital media, and engagement: A meta-analysis of research', *Social Science Computer Review*, 38: pp. 111-127.

Bourdieu, P. (1986). 'The forms of capital', in J. E. Richardson, *Theory of Research for the Sociology of Education*, Grennword Press: pp. 46-58.

Bourdieu, P. and Wacquant, L. J. (1992). 'An invitation to reflexive sociology', University of Chicago press.

Boyd , D. M. (2010). 'Social Network Sites as Networked Publics: Affordances, Dynamics, and Implications'. In Z. Papacharissi (Ed.), *A networked self: Identity, community, and culture on social network sites*, New York: Routledge: pp. 39–58.

Boyte, H.C. (1995). 'Public opinion as public judgment', *Glasser, Theodore L. y* SALMON.

Brady, H.E., Verba, S. and Schlozman, K.L. (1995). 'Beyond SES: A resource model of political participation', *American Political Science Review*, 89(2): pp. 271-294.

Brady, H.E. (1998). 'Political participation', In: J., P., Robinson, P., R., Shaver and L.,S., Wrightsman (eds.) *Measures of Political Attitudes*. San Diego, CA: Academic Press: pp. 737–801.

Brady, H.E., Schlozman, K.L. and Verba, S. (1999). 'Prospecting for participants: Rational expectations and the recruitment of political activists', *American Political Science Review*, 93: pp. 153-168.

Brandtzæg, P.B. (2010). 'Towards a unified Media-User Typology (MUT): A metaanalysis and review of the research literature on media-user typologies', *Computers in Human Behavior*, 26(5): pp. 940-956.

Brandtzæg, P.B. (2012). 'Social networking sites: Their users and social implications— A longitudinal study'. *Journal of Computer-Mediated Communication*, 17: pp. 467-488.

Breaugh, J.A. and Arnold, J. (2007). 'Controlling nuisance variables by using a matchedgroups design', *Organizational Research Methods*, *10*(3): pp.523-541.

Brooks, B., Welser, H. T., Hogan, B. and Titsworth, S. (2011). 'Socioeconomic status updates: Family SES and emergent social capital in college student Facebook networks', *Information, Communication & Society*, *14*(4): pp. 529-549.

Brooks, B., Hogan, B., Ellison, N., Lampe, C. and Vitak, J. (2014). 'Assessing structural correlates to social capital in Facebook ego networks', *Social Networks*, *38*: pp.1-15.

Brown, J.D. (2011). 'Likert items and scales of measurement', *Statistics*, 15(1): pp.10-14.

Brown, S., Hackley, C., Hunt, S.D., Marsh, C., O'shaughnessy, N., Phillips, B.J., Tonks, D., Miles, C. and Nilsson, T. (2018). 'Marketing (as) Rhetoric: paradigms, provocations, and perspectives', *Journal of Marketing Management*, 34: pp. 1336-1378.

Brundidge, J. and Rice, R. E. (2009). 'Political engagement online: Do the information rich get richer and the like-minded more similar', *Routledge handbook of Internet politics*: pp. 144-156.

Bryman, A. (2012). 'Social research methods', OUP Oxford.

Bucher, T. (2012). 'Want to be on the top? Algorithmic power and the threat of invisibility on Facebook', New media & society, 14: pp. 1164-1180.
Burke, M., Marlow, C. and Lento, T. (2010). 'Social network activity and social wellbeing', *Paper presented at the conference on human factors in computing systems*, Atlanta.

Burke, M., Kraut, R. and Marlow, C. (2011). 'May. Social capital on Facebook: Differentiating uses and users', In *Proceedings of the SIGCHI conference on human factors in computing systems*: pp. 571-580.

Burns, N., Schlozman, K.L. and Verba, S. (2001). '*The Private Roots of Public Action: Gender, Equality, and Political Participation*', Cambridge: Harvard University Press.

Burt, R.S., (1988). 'The stability of American markets', *American journal of Sociology*, 94(2): pp. 356-395.

Burtt, S. (1990). 'The Good Citizen's Psyche: On the Psychology of Civic Virtue', *Polity*, 23: pp. 23-38.

Burt, R.S., (1992). '*Structural holes: The social structure of competition'*, Cambridge, MA: Harvard University Press.

Burt, R.S., (2000). 'The Network Structure of Social Capital', in B. M. Staw, and R. I. Sutton, *Research in Organizational Behavior*, London: Elsevier Science JA: pp. 345-423.

Butler, P. and Collins, N. (1996). 'Strategic analysis in political markets', *European Journal of Marketing*.

Butler, P. and Harris, P. (2009). 'Considerations on the Evolution of Political Marketing Theory', *Marketing Theory*, 9(2): pp. 149-164.

Byrne, B.M. (2012). 'Structural equation modeling with Mplus: Basic concepts, applications, and programming', New York, NY: Routledge Academic.

Calenda, D. and Mosca, L. (2007). 'The political use of the internet: Some insights from two surveys of Italian students', *Information, Community & Society*, 10: pp. 29-47.

Calenda, D. and Meijer, A. (2009). 'Young people, the Internet and political participation: findings of a web survey in Italy, Spain and The Netherlands', *Information, Communication & Society*, *12*(6): pp.879-898.

Callahan, K. (2007). 'Citizen participation: Models and methods', *International Journal of Public Administration*, 30 (11): pp. 1179–1196.

Campbell, D.E. and Kelly, J. S. (2000). 'A simple characterization of majority rule', *Economic Theory*, 15: pp. 689-700.

Campbell, D.E. (2009). 'Civic Engagement and Education: An Empirical Test of the Sorting Model', *American journal of political science*, 53(4): pp. 771-86.

Campbell, D.E. (2013). 'Social Networks and Political Participation', *Annual Review of Political Science*, 16(1): pp. 33-48.

Carifio, J. and Perla, R.J. (2007). 'Ten common misunderstandings, misconceptions, persistent myths and urban legends about Likert scales and Likert response formats and their antidotes', *Journal of Social Sciences*, *3*(3): pp.106-116.

Carifio, J. and Perla, R.J. (2008). 'Resolving the 50-year debate around using and misusing Likert scales', *Medical education*, 42(12): pp. 1150-1152.

Carlisle, J.E. and Patton, R.C. (2013). 'Is social media changing how we understand political engagement? An analysis of Facebook and the 2008 presidential election', *Political research quarterly*, 66: pp. 883-895.

Carpentier, N. and De Cleen, B. (2008). 'Blurring participations and convergences', in N. Carpentier and D. C. Benjamin (Eds.), *Participation and Media Production: Critical Relfections on Content Creation*, Newcastle: Cambridge Scholars Publishing.

Carpentier, N. (2011). '*Media and participation*', Chicago: The University of Chicago Press: p. 195.

Carpentier, N., Dahlgren, P. (2014). 'Histories of mediated participation', *Communication Management Quarterly*, 30,(Special Issue).

Carr, N. (2008). 'Is Google making us stupid?', *Yearbook of the National Society for the Study of Education*, *107*(2): pp. 89-94.

Carrier, L.M., Cheever, N., A., Rosen, L., D., Benitez, S. and Chang, J., (2009). 'Multitasking across generations: Multitasking choices and difficulty ratings in three generations of Americans', *Computers in Human Behavior*, 25: pp. 483–489.

Carson, L. and Elstub, S. (2019). 'Comparing participatory and deliberative democracy', newDemocracy Research and Development Note, newDemocracy Foundation, https://www. newdemocracy. com. au/wp-content/uploads/2019/04/RD-Note-Comparing-Participatoryand-Deliberative-Democracy. pdf, accessed, 11.

Carter, R.L. (2006). 'Solutions for Missing Data in Structural Equation Modeling', *Research & Practice in Assessment*, 1(1): pp. 1-6.

Centola, D. (2011). 'An experimental study of homophily in the adoption of health behavior', *Science*, *334*(6060): pp.1269-1272.

Chae, Y., Lee, S. and Kim, Y. (2019). 'Meta-analysis of the relationship between Internet use and political participation: Examining main and moderating effects', *Asian Journal of Communication*, 29: pp. 35-54.

Chaffee, S.H. (1975). 'Political communication: Issues and strategies for research', *SAGE Publications, Incorporated*.

Chan, M., Wu, X., Hao, Y., Xi, R. and Jin, T. (2012). 'Microblogging, online expression, and political efficacy among young Chinese citizens: The moderating role of information and entertainment needs in the use of Weibo', *Cyberpsychology, Behavior, and Social Networking*, 15: pp. 345-349.

Chan, T.W. and Clayton, M, (2006). 'Should the voting age be lowered to sixteen? Normative and empirical considerations', *Political studies*, *54*(3): pp.533-558.

Chang, H.H. and Chuang, S.S. (2011). 'Social capital and individual motivations on knowledge sharing: Participant involvement as a moderator', *Information & management*, 48(1): pp.9-18.

Childs, S. (2004). 'A feminised style of politics? Women MPs in the House of Commons', *The British Journal of Politics and International Relations*, 6(1): pp. 3-19.

Chin, W.W. (1998). 'The partial least squares approach to structural equation modeling'. In: G. A. Marcoulides (Ed.), *Modern Methods for Business Research*, Mahwah, NJ: Lawrence Erlbaum Associates: pp. 295–358.

Chin, W.W., Marcolin, B.L. and Newsted, P.R. (2003). A partial least squares latent variable modeling approach for measuring interaction effects: Results from a Monte Carlo simulation study and an electronic-mail emotion/adoption study. *Information systems research*, 14(2): pp.189-217.

Chin, W.W., Peterson, R.A. and Brown, S.P. (2008). 'Structural equation modeling in marketing: Some practical reminders', *Journal of marketing theory and practice*, *16*(4): pp.287-298.

Chin, W.W. (2010).' How to write up and report PLS analyses', In *Handbook of partial least squares* (pp. 655-690). Springer, Berlin, Heidelberg.

Christensen, H.S. and Bengtsson, Å. (2011). 'The political competence of Internet participants: Evidence from Finland', *Information, Communication & Society*, 14: pp. 896-916.

Chunly, S. (2019). 'Facebook and political participation in Cambodia: determinants and impact of online political behaviours in an authoritarian state', *South East Asia Research*, 27: pp. 378-397.

Clason, D.L. and Dormody, T.J. (1994). 'Analyzing data measured by individual Likerttype items', *Journal of agricultural education*, *35*: p.31-35.

Cocks, K. and Torgerson, D.J. (2013). Sample size calculations for pilot randomized trials: a confidence interval approach. *Journal of clinical epidemiology*, 66(2): pp.197-201.

Cohen, J. (1988). *Statistical power analysis for the behavioural sciences*, 2nd ed. Hillsdale, NJ: Erlbaum.

Cole, J.I., Suman, M., Schramm, P., Van Bel, D., Lunn, B., Maguire, P., Hanson, K., Singh, R. and Aquino, J., S. (2000). '*The UCLA Internet Report: Surveying the Digital Future*', Los Angeles: UCLA Center for Communication Policy.

Coleman, J.C. (1988). 'Social capital in the creation of human capital', *American Journal* of Sociology, 94: pp. 95–120.

Coleman, J.C. (1990). 'Foundations of Social Theory', Cambridge, MA: Harvard University Press.

Coleman, S. and Blumler, J.G. (2009). 'The Internet and democratic citizenship: Theory, practice and policy', *Cambridge University Press*.

Collins, N. and Butler, P. (2003). 'When Marketing Models Clash with Democracy', *Journal of Public Affairs*, 3(1): pp. 52–62.

Collins, E. and Freeman, J. (2013). 'Do problematic and non-problematic video game players differ in extraversion, trait empathy, social capital and prosocial tendencies?', *Computers in Human Behavior*, 29(5): pp.1933-1940.

Comber, M.K. (2007). 'Political participation and service learning: Civic education as problem and solution', In: D.M. Shea and J.C. Green (eds.) *Fountain of Youth: Strategies and Tactics for Mobilizing America's Young Voters*. Lanham, MD: Rowman & Littlefield: pp. 115–128.

Conge, P.J. (1988). 'The concept of political participation: Toward a definition', *Comparative Politics*, 20 (2): pp. 241–249.

Conway, J.M. and Lance, C.E. (2010). 'What reviewers should expect from authors regarding common method bias in organizational research', *Journal of Business and Psychology*, 25(3): pp.325-334.

Conway, M.M. (2000). '*Political participation in the United States*' (3rd ed.). Washington, DC: CQ Press.

Couldry, N. (2010). 'Why voice matters', London: Sage Publications.

Coulthard, M., Walker, A. and Morgan, A. (2001). 'Assessing People's Perceptions of Their Neighbourhood and Community Involvement (Part 1): A Guide to Questions for Use in the Measurement of Social Capital Based on the General Household Survey Module', *Health Development Agency*.

Curran, J. (2011). 'Media and democracy', Taylor & Francis.

Curran, J., Fenton, N. and Freedman, D. (2012). '*Misunderstanding the internet*', New York: Routledge.

Dahl, R.A. (1956). 'A Preface to Democratic Theory', Chicago, IL: The University of Chicago Press.

Dalton, J.R. and Wattenberg, P. M. (2000). 'Political Change in Advanced Industrial Democracies'.

Dalton, R.J. (2006). 'Citizenship norms and political participation in America: The good news is... the bad news is wrong', *The Center for Democracy phone*, 202: pp. 687-0593.

Dalton, R. J. (2008). 'Citizenship norms and the expansion of political participation', *Political studies*, 56: pp. 76-98.

Dalton, R.J. (2009) 'The Good Citizen: How a Younger Generation is Reshaping American Politics', Washington, DC: CQ Press.

Dalton, R., Van Sickle, A. and Weldon, S. (2010). 'The individual-institutional nexus of protest behaviour', *British journal of political science*: pp. 51-73.

Daly, A.J., Fresno García, M.D. and Bjorklund Jr, P. (2020). 'Social Media in a New Era: Pandemic, Pitfalls, and Possibilities', *American Journal of Education*, *127*(1): pp.143-151.

Dawes, C.T., Loewen, P.J. and Fowler, J.H. (2011). 'Social preferences and political participation', *The Journal of Politics*, 73(3): pp.845-856.

De Cleen, B. and Carpentier, N. (2008). 'Introduction: Blurring participations and convergences'.

De Leeuw, E. D. and D.A. Dillman (2008). 'International handbook of survey methodology', *Taylor & Francis*.

Dean, D. and Croft, R. (2001). 'Friends and relations: long-term approaches to political campaigning', *European Journal of Marketing*, 35(11/12): pp. 1197-1216

Delli Carpini, M.X. and Keeter, S. (1993). 'Measuring political knowledge: Putting first things first', American Journal of Political Science, 1179-1206.

Delli Carpini, M.X., Keeter, S. (1996). 'What Americans Know About Politics and Why It Matters', New Haven, CT: Yale University Press.

Delli Carpini, M.X. (2000). 'Gen. com: Youth, civic engagement, and the new information environment', *Political communication*, 17: pp. 341-349.

Dermody, J. and Scullion, R. (2001). 'Delusions of grandeur? Marketing's contribution to meaningful western political consumption'. *European Journal of Marketing*, 35: pp. 1085-1098.

DeSchon, R.P. (1997). 'Do structural equation models correct for measurement error?', In 1997 Research Methods Forum (Vol. 2). Retrieved from http://division.aomonline.org/rm/1997_forum_structural_equation_models_error.html

Dholakia, N. (2012a). 'Being critical in marketing studies: the imperative of macro perspectives'. *Journal of Macromarketing*, 32: pp. 220-225.

Dholakia, N. (2012b). 'Finanzkapital and consumers: how financialization shaped twentieth century marketing', *Journal of Historical Research in Marketing*, 4: pp. 453–461.

Di Gennaro, C. and Dutton, W. (2006). 'The Internet and the Public: Online and Offline Political Participation in the United Kingdom', *Parliamentary Affairs*, 59(2): pp. 299–313.

Diani, M. (2001). 'Social movement networks: Virtual and real', In: Frank Webster (editor). *Culture and politics in the information age: A new politics?* London: Routledge: pp. 117–128.

Diemer, M.A. (2012). 'Fostering marginalized youths' political participation: Longitudinal roles of parental political socialization and youth socio-political development', *American Journal of Community Psychology*, 50(1-2): pp. 246-256.

Digrazia, J., Mckelvey, K., Bollen, J. and Rojas, F. (2013). 'More tweets, more votes: Social media as a quantitative indicator of political behavior', *PloS one*, *8*(11), p.e79449.

Dijkstra, T.K., and Henseler, J. (2015). Consistent partial least squares path modeling. *MIS Quarterly*, 39(2): pp. 297-316.

Dimitrova, D.V. and Bystrom, D. (2013). 'The effects of social media on political participation and candidate image evaluations in the 2012 Iowa caucuses', *American Behavioral Scientist*, 57: pp. 1568-1583.

Dimitrova, D.V., Shehata, A., Strömbäck, J. and Nord, L. W. (2014). 'The effects of digital media on political knowledge and participation in election campaigns: Evidence from panel data', *Communication research*, 41: pp. 95-118.

Dixon, D.F. and Blois, K.J. (1983). 'Some Limitations of the 4Ps as a Paradigm for Marketing', *Marketing Education Group Annual Conference*, Cranfield Institute of Technology.

Dohmen, T., Falk, A., Huffman, D., Sunde, U. (2009). 'Homo Reciprocans: Survey Evidence on Behavioural Outcomes', *The Economic Journal*, 119 (536): pp. 592–612.

Domholdt, E. (2005). 'Rehabilitation research: principles and applications', Saunders.

Donath, J., and Boyd, D. (2004). 'Public displays of connection', *BT Technology Journal*, 22(4): pp. 71–82.

Downs, A. (1957). 'An economic theory of democracy', *Journal of political economy*, 65(2): pp.135-150.

Drew, D. and Weaver, D. (2006). 'Voter learning in the 2004 presidential election: Did the media matter?' *Journalism & Mass Communication Quarterly*, 83: pp. 25-42.

Druckman, J.N. and Nelson, K.R. (2003). Framing and deliberation: How citizens' conversations limit elite influence. *American Journal of Political Science*, 47(4): pp. 729-745.

Dryzek, J. and List, C. (2003). 'Social choice theory and deliberative democracy: a reconciliation', *British Journal of Political Studies*, 33(1): pp. 1–28.

Duggan, M., Ellison, N.B., Lampe, C., Lenhart, A. and Madden, M. (2015). 'Social media update 2014', *Pew research center*, 19: pp. 1-2.

Dutta-Bergman, M. and Chung, W. (2005). 'The relationship between online political participation and traditional political participation: The theory of channel complementarity', In *Unpublished paper presented at the International Communication Association 2005 Annual Meeting, New York. Available from http://web. ebscohost. com.*

Earl, J. and Kimport, K. (2011). 'Digitally enabled social change: Activism in the internet age', *Mit Press*.

Edvardsson, B. (2005). 'Service quality: beyond cognitive assessment'. *Managing Service Quality: An International Journal.*

Edvardsson, B., Gustafsson, A. and Roos, I. (2005). 'Service Portraits in Service Research: a Critical Review', *International Journal of Service Industry Management*, 16(1): pp. 107-121.

Effing, R., Van Hillegersberg, J. and Huibers, T. (2011). 'Social media and political participation: are Facebook, Twitter and YouTube democratizing our political systems?', In *International conference on electronic participation* Springer, Berlin, Heidelberg: pp. 25-35.

Egan, J. (2005). 'Another False Dawn? The Liberal Democrats 2005', *Journal of Marketing Management*, 21: pp. 959–78.

Egan, J. and Lynch, J. (2012). 'Political MarNeting Through the Lens of Servicedominant Logic', *Journal of Marketing Trends*: pp. 397-386. Eisenstadt, S.N. (1971). 'From generation to generation: Age groups and social structure', *Transaction Publishers*.

Ekman, J., Amna, E. (2012). 'Political participation and civic engagement: towards a new typology', *Human Affairs*, 22: pp. 283–300.

Ekström, A., Jülich, S., Wisselgren, P. and Lundgren, F. (2011). '*History of participatory media: Politics and publics, 1750-2000*', London: Routledge.

Electoral commission (2002). 'Election, 2001: campaign spending', Report, November.

Ellison, N.B., Steinfield, C. and Lampe, C. (2007). 'The benefits of Facebook 'friends:' Social capital and college students' use of online social network sites', *Journal of computer-mediated communication*, 12: pp. 1143-1168.

Ellison, N., B., Steinfield, C. and Lampe, C. (2011). 'Connection strategies: Social capital implications of Facebook-enabled communication practices', *New Media & Society*, 13: pp. 873–892.

Ellison, N.B., Vitak, J., Gray, R. and Lampe, C. (2014). 'Cultivating social resources on social network sites: Facebook relationship maintenance behaviors and their role in social capital processes', *Journal of Computer-Mediated Communication*, 19: pp. 855-870.

Ellison, N.B. and Vitak, J. (2015). 'Social network site affordances and their relationship to social capital processes', *The handbook of the psychology of communication technology*, 32: pp. 205-228.

Eltantawy, N. and Wiest, J.B. (2011). 'The Arab spring Social media in the Egyptian revolution: reconsidering resource mobilization theory', *International journal of communication*, 5: pp.1207-1224.

Engbers, T.A., Thompson, M.F. and Slaper, T.F. (2017). 'Theory and measurement in social capital research', *Social Indicators Research*, *132*(2): pp.537-558.

Enli, G.S. and Skogerbø, E. 2013. 'Personalized campaigns in party-centred politics: Twitter and Facebook as arenas for political communication. Information, communication & society', 16: pp. 757-774.

Ernst, M.D. (2004). 'Permutation methods: A basis for exact inference', *Statistical Science*, 19(4): pp. 676-685.

Evermann, J. and Tate, M. (2010). 'Testing models or fitting models? Identifying model misspecifications in PLS', In *Proceedings of the 31st International Conference on Information Systems (ICIS)*, St. Louis, MS.

Fagerland, M.W. (2012). 'T-tests, non-parametric tests, and large studies—a paradox of statistical practice?', *BMC medical research methodology*, 12(1): pp. 1-7.

Falk, R.F., and Miller, N.B. (1992). 'A primer for soft modeling', Akron, OH: University of Akron Press.

Farrell, D.M. and Wortmann, M. (1987). 'Party strategies in the electoral market: Political marketing in West Germany, Britain and Ireland', *European Journal of Political Research*, 15, 297-318.

Faul, F., Erdfelder, E., Lang, A. G. and Buchner, A. (2007). 'G*Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences', *Behavior Research Methods*, 39: pp. 175–191.

Faul, F., Erdfelder, E., Buchner, A. and Lang, A.G. (2009). 'Statistical power analyses using G* Power 3.1: Tests for correlation and regression analyses', *Behavior research methods*, *41*(4): pp.1149-1160.

Fenton, N. and Barassi, V. (2011). 'Alternative media and social networking sites: The politics of individuation and political participation', The Communication Review, 14: pp. 179-196.

Feys, J. (2016). 'Nonparametric tests for the interaction in two-way factorial designs using R', *The R Journal*, 8(1): pp.367-378.

Field, A. (2013). '*Discovering Statistics using IBM SPSS Statistics*', London: Sage Publication Ltd.

Field, A. (2018). Discovering statistics using IBM SPSS statistics. Sage.

Field, J. (2008) Social Capital, 2nd edn, London: Routledge.

Fine, B. (2001). Social capital versus social theory: Political economy and social science at the turn of the millennium (Vol. 5). Psychology Press.

Fine, B. (2010). Theories of social capital: Researchers behaving badly. Pluto press.

Fine, S.H. (2017). Marketing the public sector: Promoting the causes of public and nonprofit agencies, Routledge.

Finsveen, E., Van Oorschot, W. (2008). 'Access to resources in networks: A theoretical and empirical critique of networks as a proxy for social capital', *Acta Sociologica*, *51*(4): pp. 293-307.

Fiori, K., L., Smith, J., Antonucci, T., C. (2007). 'Social network types among older adults: A multidimensional approach', *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*, 62(6): pp. 322-P330.

Fischer, C.S. (1982). '*To dwell among friends: Personal networks in town and city*', University of chicago Press.

Flap, H. (2002). 'No man is an island: the research programme of a social capital theory'. In O. Favereau and E. Lazega, eds. *Conventions and Structures in Economic Organisations: Markets and Hierarchies. Cheltenham: Edward Elgar.*

Flap, H. and Völker, B. (2004). 'Creation and returns of social capital', (Vol. 9). Routledge.

Ford, I.D., Gadde, L.E., Hakansson, H. and Snehota, I. (2003). '*Managing Business* Networks', John Wiley, Chichester.

Fornell, C.G., and Larcker, D.F. (1981). 'Evaluating structural equation models with unobservable variables and measurement error', *Journal of Marketing Research*, 18(1): pp. 39–50.

Fornell, C.G., Bookstein, F.L. (1982). 'Two structural equation models: LISREL and PLS applied to consumer exit-voice theory', *Journal of Marketing research*: pp. 440-452.

Fowler, J.H., Baker, L.A. and Dawes, C.T. (2008). 'Genetic variation in political participation', *American Political Science Review*, 102(2): pp. 233-248.

Fox, S. (2014). 'Is it time to update the definition of political participation? Political participation in Britain: the decline and revival of civic culture', *Parliamentary Affairs*, 67(2): pp. 495–505.

Franklin, B. (1994). 'Packaging Politics: Political Communications in Britain's Media Democracy', Edward Arnold.

Franklin, M. (2004). 'Voter Turnout and the Dynamics of Electoral Competition in Established Democracies since 1945', Cambridge: Cambridge University Press.

Franzen, A. (2003). 'Social Capital and the Internet: Evidence from Swiss Panel Data', *Kyklos*, 56: pp. 341-360.

Fukuyama, F. (1995). 'Social capital and the global economy', Foreign Aff., 74, p.89.

Fukuyama F. (2000). '*Social Capital and Civil Society*', IMF Working Papers No. 74, March, International Monetary Fund.

Fulkerson, G.M. and Thompson, G.H. (2008). 'The evolution of a contested concept: A meta-analysis of social capital definitions and trends (1988–2006)', *Sociological Inquiry*, 78(4): pp. 536-557.

Fullerton, R.A. (2011). 'Historical methodology: The perspective of a professionally trained Historian turned marketer', *Journal of Historical Research in Marketing*, 3: pp. 436–448.

Fullwood, C., James, B.M. and Chen-Wilson, C.H. (2016). 'Self-Concept Clarity and Online Self-Presentation in Adolescents', *Cyberpsychology, Behavior, and Social Networking*, *19*(12): pp. 716-720.

Gabriel, O.W. (1986). 'Political culture, post-materialism and materialism in the Federal Republic of Germany', Opladen: West German publishing house.

Gagarin, M. and Fantham, E. (2010). '*Oxford encyclopedia of ancient Greece and Rome*' (Vol. I). New York: Oxford University Press.

Gainous, J., Marlowe, A.D. and Wagner, K.M. (2013). 'Traditional cleavages or a new world: Does online social networking bridge the political participation divide?', *International Journal of Politics, Culture, and Society*, *26*(2): pp.145-158.

Galston, W.A. (2001). 'Political knowledge, political engagement, and civic education', *Annual Review of Political Science*, 4(1): pp. 217–234.

Gamble, A. (1974). '*The conservative nation, London*'; Boston: Routledge & Kegan Paul.

Games, P.A. and Howell, J.F. (1976). 'Pairwise multiple comparison procedures with unequal n's and/or variances: a Monte Carlo study', *Journal of Educational Statistics*, 1(2): pp.113-125.

Gan, C. and Wang, W. (2015). 'Uses and gratifications of social media: a comparison of microblog and WeChat', *Journal of Systems and Information Technology*.

Gaskin, J., Godfrey, S. and Vance, A. (2018). 'Successful system use: It's not just who you are, but what you do', *AIS Transactions on Human-Computer Interaction*, 10(2): pp. 57-81.

Gastil, J. and Xenos, M. (2010). 'Of attitudes and engagement: Clarifying the reciprocal relationship between civic attitudes and political participation', *Journal of Communication*, *60*(2): pp. 318-343.

Gefen, D., Straub, D.W. and Boudreau, M.C. (2000). 'Structural Equation Modeling and Regression Guidelines for Research Practice', *Communications of the Association for Information Systems*, 4(7): pp. 2-77.

Gelman, A. and King, G. (1993). 'Why are American presidential election campaign polls so variable when votes are so predictable?', British Journal of Political Science: pp. 409-451.

George, D. and Mallery, P. (2019). '*IBM SPSS statistics 26 step by step: A simple guide and reference*', Routledge.

Giavrimis, P. and Nikolaou, S.M. (2020). 'The Greek University Student's Social Capital during the COVID-19 Pandemic', *European Journal of Education Studies*, 7(8).

Gibson, R.K., Howard, P.E. and Ward, S. (2000). 'Social capital, internet connectedness and political participation: a four-country study', *International Political Science Association*, Québec, Canada.

Gibson, R.K., Lusoli, W. and Ward, S.J. (2005). 'Online Participation in the UK: Testing a contextualised Model of Internet Effects', *British Journal of Politics and International Relations*, 7: pp. 561-583.

Gibson, R.K. and McAllister, I. (2013). 'Online social ties and political engagement', *Journal of Information Technology & Politics*, *10*(1): pp. 21-34.

Gibson, R.K. and Cantijoch, M. (2013). 'Conceptualizing and Measuring Participation in the Age of the Internet: Is Online Political Engagement Really Different to Offline?', *The Journal of Politics*, 75(3): pp. 701-716.

Gil De Zuniga, H., Puig-I-Abril, E. and Rojas, H. (2009). 'Weblogs, traditional sources online and political participation: An assessment of how the Internet is changing the political environment', *New media & society*, 11: pp. 553-574.

Gil de Zuniga, H., Veenstra, A., Vraga, E. and Shah, D. (2010). 'Digital democracy: Reimagining pathways to political participation', *Journal of Information Technology & Politics*, 7(1): pp. 36–51. Gil de Zuniga, H. and Valenzuela, S. (2011). 'The mediating path to a stronger citizenship: Online and offline networks, weak ties and civic engagement'. *Communication Research*, 38(3): p. 397–421.

Gil De Zuniga, H., Jung, N. and Valenzuela, S. (2012). 'Social media use for news and individuals' social capital, civic engagement and political participation'. *Journal of computer-mediated communication*, 17: pp. 319-336.

Gil de Zuniga, H., Bachmann, I., Hsu, S.H. and Brundidge, J. (2013). 'Expressive versus consumptive blog use: Implications for interpersonal discussion and political participation', *International Journal of Communication*, 7: pp. 1538–1559.

Gil de Zuniga, H., Copeland, L. and Bimber, B. (2014a). 'Political consumerism: Civic engagement and the social media connection', *New Media & Society*, 16(3): pp. 488–506.

Gil de Zuniga, H., Molyneux, L. and Zheng, P. (2014b). 'Social media, political expression, and political participation: Panel analysis of lagged and concurrent relationships'. *Journal of Communication*, 64(4): pp. 612-634.

Gil de Zuniga, H., Garcia-Perdomo, V. and McGregor, S.C. (2015). 'What is second screening? Exploring motivations of second screen use and its effect on online political participation', *Journal of communication*, *65*(5): pp. 793-815.

Gil de Zuniga, H., Valenzuela, S. and Weeks, B.E. (2016). 'Motivations for political discussion: Antecedents and consequences on civic engagement', *Human Communication Research*, *42*(4): pp. 533-552.

Gil De Zuniga, H., Barnidge, M. and Scherman, A. (2017). 'Social media social capital, offline social capital, and citizenship: Exploring asymmetrical social capital effects', *Political Communication*, 34: pp. 44-68.

Gittel, R. and Vidal, A. (1998). 'Community organizing: building social capital as a development strategy Sage', *Thousand Oaks*.

Gladwell, M. (2010). '*Small change: Why the revolution will not be tweeted*'. The New Yorker, 4 October.

Gleason, B. (2013). 'Occupy Wall Street: Exploring informal learning about a social movement on Twitter', *American Behavioral Scientist*, *57*(7): pp. 966-982.

Glenville, J.L. and Bienenstock, E.J. (2009). 'A typology for understanding the connections among different forms of social capital' *American Behavioural Scientist*, 52(11): pp. 1507–1530.

Goerres, A. (2007). 'Why are older people more likely to vote? The impact of ageing on electoral turnout in Europe', *British Journal of Politics and International Relations* 9(1): pp. 90–121.

Golbeck, J., Grimes, J.M. and Rogers, A. (2010). 'Twitter use by the US Congress', *Journal of the American society for information science and technology*, *61*(8): pp. 1612-1621.

Graham, T., Broersma, M., Hazelhoff, K. and Van'T Haar, G. (2013). 'Between broadcasting political messages and interacting with voters: The use of Twitter during the 2010 UK general election campaign' *Information, communication & society*, *16*(5): pp.692-716.

Granovetter, M. (1973). 'The strength of weak ties', *American journal of sociology*, 78: pp. 1360-1380.

Granovetter, M. (1974). 'Getting a Job Cambridge', Harvard University Press.

Granovetter, M. (1985). 'Economic action and social structure: The problem of embeddedness', *American journal of sociology*, 91(3): pp.481-510.

Grant, W.J., Moon, B. and Grant, J.B. (2010). 'Digital dialogue? Australian politicians' use of the social network tool Twitter', *Australian Journal of Political Science*, 45(4): pp. 579–604.

Grčić, K., Babac, M.B. and Podobnik, V. (2017). 'Generating politician profiles based on content analysis of social network datasets', *Journal of universal computer science*, 23(3): pp.236-255. Green, J. E. (2010). 'Political participation', In: M. Bevir, ed. *Encyclopedia of political theory*. Vol. 3. Thousand Oaks, CA: Sage: pp. 1068–1073.

Greenhow, C. (2011). 'Online social networks and learning', On the horizon.

Grillo, B. (2013). beppegrillo.it [Online]. Available: www.beppegrillo.it [Accessed 2015].

Grönroos, C. (1990). 'Relationship Marketing: The Strategy Continuum', *Journal of Business Research*, 20: pp. 3-11.

Grönroos, C. (1994). 'From marketing mix to relationship marketing: towards a paradigm shift in marketing', *Asia-Australia Marketing Journal*, 2: pp. 4-29.

Grönroos, C. (1995). 'Relationship marketing: the strategy continuum', *Journal of the academy of marketing science*, 23: pp. 252-254.

Grönroos, C., (2000). 'Service Management and Marketing', Chichester, Wiley.

Grönroos, C. (2006). 'On defining marketing: finding a new roadmap for marketing', *Marketing theory*, 6: pp. 395-417.

Grootaert, C. and Van Bastelaer, T. (2001). 'Understanding and measuring social capital: A synthesis of findings and recommendations from the social capital initiative', (Vol. 24). World Bank, Social Development Family, Environmentally and Socially Sustainable Development Network.

Groshek, J., and Dimitrova, D. (2011). 'A cross-section of voter learning, campaign interest and intention to vote in the 2008 American election: Did Web 2.0 matter', *Communication Studies Journal*, *9*(1): pp. 355-375.

Gueorguieva, V. (2008). 'Voters, MySpace and YouTube: the impact of alternative communication channels on the 2006 election cycle and beyond', *Social Science Computer Review*, 26(3): pp. 288-300.

Gummesson, E., (1987). 'The New Marketing: Developing Long-term Interactive Relationships', *Long Range Planning*, 20(4): pp. 10-20.

Gummesson, E. (1999). 'Total Relationship Marketing: Experimenting With a Synthesis of Research Frontiers', *Australasian Marketing Journal* (AMJ), 7: pp. 72-85.

Gustafsson, N. (2012). 'The subtle nature of Facebook politics: Swedish social network site users and political participation', *New Media & Society*, 14(8): pp. 1286-1303.

Habermas, J. (1989). 'Towards a Communication-Concept of Rational Collective Will-Formation. A Thought-Experiment', *Ratio Juris*, 2: pp. 144-154.

Habermas, J. (1996). 'Between Facts and Norms: Contributions to a Discourse Theory of Law and Democracy', *MIT Press*.

Haenlein, M. and Kaplan, A.M. (2004). 'A beginner's guide to partial least squares analysis', *Understanding statistics*, *3*(4): pp.283-297.

Hair, J.F., Black, W., Babin, B., Anderson, R. E. and Tatham, R. (2006). *Multivariate Data Analysis* ',(6th Edition.). Uppersaddle River: Pearson Prentice Hall.

Hair, J.F., Black, W.C., Babin, B.J. and Anderson, R.E. (2010). *Multivariate Data Analysis A Global Perspective*, New Jersey, Pearson Prentice Hall, Seventh Edition.

Hair, J.F., Ringle, C., M. and Sarstedt, M., (2011). 'PLS-SEM: Indeed a silver bullet', *Journal of Marketing theory and Practice*, *19*(2): pp. 139-152.

Hair, J.F., Sarstedt, M., Ringle, C. M. and Mena, J., A. (2012). 'An assessment of the use of partial least squares structural equation modeling in marketing research', *Journal of the Academy of Marketing Science*, *40*(3): pp. 414-433.

Hair, J.F., Ringle, C.M. and Sarstedt, M. (2013). 'Partial least squares structural equation modeling: Rigorous applications, better results and higher acceptance', *Long range planning*, 46(1-2): pp. 1-12.

Hair, J.F., Risher, J., Sarstedt, M. and Ringle, C. (2019), 'When to use and how to report the results of PLS-SEM', *European Business Review*, (31) No. 1: pp. 2-24.

Hair Jr, J.F., Hult, G.T.M., Ringle, C. and Sarstedt, M. (2017). 'A primer on partial least squares structural equation modeling (PLS-SEM)', Sage publications.

Hair Jr J.F., Sarstedt, M., Ringle, C.M. and Gudergan, S.P. (2018). 'Advanced issues in partial least squares structural equation modeling', Sage publications.

Halpern, D. (2001). 'Moral values, social trust and inequality: can values explain crime?', *British Journal of criminology*, 41: pp. 236-251.

Halpern, D. (2005). 'Social Capital', Cambridge: Polity.

Halpern, D. and Gibbs, J. (2013). 'Social media as a catalyst for online deliberation? Exploring the affordances of Facebook and YouTube for political expression', *Computers in Human Behavior*, *29*(3): pp.1159-1168.

Halpern, D., Valenzuela, S. and Katz, J.E. (2017). 'We face, I tweet: How different social media influence political participation through collective and internal efficacy', *Journal of Computer-Mediated Communication*, 22: pp. 320-336.

Hamilton, A. and Tolbert, C.J. (2012). 'Political engagement and the Internet in the 2008 US presidential elections', *Digital media and political engagement worldwide: A comparative study*, 56.

Hampton, K.N. (2011). 'Comparing bonding and bridging ties for democratic engagement: Everyday use of communication technologies within social networks for civic and civil behaviors', *Information, Communication & Society*, 14: pp. 510-528.

Hampton, K.N., Lee, C.J. and Her, E.J. (2011). 'How new media affords network diversity: Direct and mediated access to social capital through participation in local social settings', *new media & society*, *13*(7): pp.1031-1049.

Hanifan, L.J. (1916). 'The rural school community center. *The Annals of the American Academy of Political and Social Science*, 67(1): pp.130-138.

Hao, X., Wen, N. and George, C. (2014). 'News consumption and political and civic engagement among young people', *Journal of Youth Studies*, 17(9): pp. 1–18.

Harris, P. (2001). 'Machiavelli, Political Marketing and Reinventing Government', *European Journal of Marketing*, 35 (9/10): pp. 1136-1154.

Harrison, T. M. and Barthel, B. (2009). 'Wielding new media in Web 2.0: exploring the history of engagement with the collaborative construction of media products', *New media* & *society*, 11: pp. 155-178.

Harrop, M. (1990). 'Political marketing', Parliamentary Affairs, 43: pp. 277-291.

Hartley, C. and Watson, L., (2014). 'Virtue in Political Thought: On Civic Virtue and Political Liberalism', in Timpe, K. and Boyd, C., A., '*Virtues and their Vices*', Oxford: Oxford University Press.

Hauer, T. (2017). 'Technological determinism and new media', *International Journal of English Literature and Social Sciences*, 2, 239174.

Hay, C. (2002). 'Political analysis', Springer.

Hay, C. (2007). 'Why We Hate Politics'. Cambridge and Malden, MA: Polity Press.

Hayami, Y. (2009). 'Social capital, human capital and the community mechanism: Toward a conceptual framework for economists', *The Journal of Development Studies*, 45(1): pp. 96-123.

Hayat, T., Samuel-Azran, T. and Galily, Y. (2016). 'Al-Jazeera Sport's US Twitter followers: sport-politics nexus?', *Online information review*.

Held, D. (2006). 'Models of Democracy', Cambridge, UK: Polity Press.

Helliwell, J.F. and Putnam, R.D. (2004). 'The social context of well-being', *Philosophical Transactions of The Royal Society B: Biological Sciences*, 35(9): pp. 1435–1446.

Helliwell, J.F. and Putnam, R.D. (2007). 'Education and Social Capital', *Eastern Economic Journal*, 33(1): pp. 1-19.

Hendriks, C.M., Dryzek, J.S. and Hunold, C. (2007). 'Turning up the heat: partisanship in deliberative innovation', *Political Studies*, 55(2): pp. 362–383.

Henn, M., Weinstein, M. and Wring, D. (2002). 'A generation apart? Youth and political participation in Britain', *The British Journal of Politics & International Relations*, 4: pp. 167-192.

Henneberg, S.C., (2002). 'Understanding Political Marketing', in N. J. O'Shaughnessy and S. C. M. Henneberg (eds), *The Idea of Political Marketing*: pp. 93–170. Westport, CT: Praeger.

Henneberg, S.C. (2004). '*The views of an advocatus dei: Political marketing and its critics*', Journal of Public Affairs: An International Journal, 4: pp. 225-243.

Henneberg, S.C. and O'Shaughnessy N.J. (2007). 'Theory and concept development in political marketing: issue and agenda', *Journal of Political Marketing*, 6(2/3): pp. 5-32.

Henneberg, S.C. (2008). 'An epistemological perspective on research in political marketing', *Journal of political marketing*, 7: pp. 151-182.

Henneberg, S.C., Scammell, M. and O'Shaughnessy, N.J. (2009). 'Political marketing management and theories of democracy', *Marketing Theory*, 9(2): pp. 165-188.

Henneberg, S.C. and Ormrod, R.P. (2013). 'The triadic interaction model of political marketing exchange', *Marketing Theory*, 13: pp. 87-103.

Henseler J. and Fassot, G. (2009). 'Testing moderating effects in PLS path models: an illustration of available procedures', In: *Handbook of Partial Least Squares: Concepts, Methods and Applications in Marketing and Related Fields*. Vinzi VE, Chin WW, Henseler J, Wang H (eds), Springer-Verlag: Berlin.

Henseler, J. and Chin, W.W. (2010). 'A comparison of approaches for the analysis of interaction effects between latent variables using partial least squares path modeling', *Structural equation modeling*, 17(1): pp. 82-109.

Henseler, J., Dijkstra, T.K., Sarstedt, M., Ringle, C.M., Diamantopoulos, A., Straub, D.W., Ketchen, D.J., Hair, J.F., Hult, G.T.M. and Calantone, R.J., (2014). 'Common beliefs and reality about partial least squares', *Organizational Research Methods*, *17*(2): pp.182-209.

Henseler, J., Ringle, C.M. and Sarstedt, M., (2015). 'A new criterion for assessing discriminant validity in variance-based structural equation modeling', *Journal of the academy of marketing science*, 43(1): pp.115-135.

Henseler, J., Ringle, C.M. and Sarstedt, M. (2016). 'Testing measurement invariance of composites using partial least squares', *International Marketing Review*, Vol. 33 No. 3: pp. 405-431.

Higgins, J.J. and Tashtoush, S. (1994). 'An aligned rank transform test for interaction', *Nonlinear World 1*, 1994: pp. 201–211.

Hilmer, J.D. (2010). 'The State of Participatory Democratic Theory', *New Political Science*, 32(1): pp. 43-63.

Himelboim, I., Mccreery, S. and Smith, M. (2013). 'Birds of a feather tweet together: Integrating network and content analyses to examine cross-ideology exposure on Twitter', *Journal of computer-mediated communication*, 18: pp. 154-174.

Ho, K.C., Kluver, R. and Yang, C.C., (2003). 'Asia.com: Asia encounters the Internet', London: Routledge.

Hodges, J.L. and Lehmann, E.I. (1962). 'Rank methods for combination of independent experiments in analysis of variance', *Annals of Mathematical Statistics*, 27: pp. 324–335.

Hofer, M. and Aubert, V. (2013). 'Perceived bridgind and bonding social capital on twitter: differentiating between follwers and followees', *Computers in Human Behavior*, 29: pp. 2134-2142.

Holt, K., Shehata, A., Strömbäck, J. and Ljungberg, E. (2013). 'Age and the effects of news media attention and social media use on political interest and participation: Do social media function as leveller?', *European Journal of Communication*, 28: pp. 19-34.

Hołyst, J.A. (2017). 'all. Cyberemotions. Collective Emotions in Cyberspace', Homans, G.C., 1950. *The Human Group*, New York: Har-court, Brace & World.

Honohan, I. (2003). 'Civic Republicanism'. London: Routledge.

Hooghe, M. (2014). 'Defining political participation: How to pinpoint an elusive target?', *Acta Politica*, 49: pp. 338-341.

Hopke, J.E., Gabay, I., Kim, S.C. and Rojas, H. (2016). 'Mobile phones and political participation in Colombia: Mobile Twitter versus mobile Facebook', *Communication and the Public*, 1: pp. 159-173.

Howard, P.N. and Parks, M.R. (2012). 'Social media and political change: Capacity, constraint, and consequence', *Journal of Communication*, 62(2): pp. 359–362.

Huberman, B.A., Romero, D.M. and Wu, F. (2009). 'Social networks that matter: Twitter under the microscope', Available at SSRN 1313405.

Hughes, A. and Dann, S. (2009). 'Political marketing and stakeholder engagement', *Marketing Theory*, 9: pp. 243-256.

Hughes, N. and Burke, J. (2018). 'Sleeping with the frenemy: How restricting 'bedroom use' of smartphones impacts happiness and wellbeing', *Computers in Human Behavior*, 85: pp. 236-244.

Hult, G.T.M., Ketchen, D.J., Griffith, D.A., Finnegan, C.A., Gonzalez-Padron, T., Harmancioglu, N., Huang, Y., Talay, M.B. and Cavusgil, S.T. (2008). 'Data equivalence in cross-cultural international business research: assessment and guidelines', *Journal of International Business Studies*, 39(6): pp.1027-1044.

Hunt, S.D. (2010). 'Doctoral seminars in marketing theory: For incorporating the history of marketing practice and thought', *Journal of Historical Research in Marketing*, 2: pp. 443–456.

Huntington, S.P., Nelson, J.M., (1976). 'No Easy Choice: Political Participation in Developing Countries', Cambridge, Mass.: Harvard University Press.

Iglewicz, B., and Hoaglin, D.C. (1993). '*How to detect and handle outliers*', Milwaukee, WI: ASQC Quality Press.

Inglehart, R. (1977). '*The Silent Revolution. Changing Values and Political Styles among Western Publics*', Princeton: Princeton University Press.

Inglehart, R. (1997). '*Modernisation and Postmodernization: Cultural, Economic, and Political Change in 43 Societies*', Princeton: Princeton University Press.

Ipsos (2018). '*Tech Tracker', Quarterly Release*. Available at: <u>https://www.ipsos.com/sites/default/files/ct/publication/documents/2018-03/ipsos-</u> <u>tech_tracker_q1_2018.pdf</u>. Last access: 10/03/2021.

Israel, G.D., Beaulieu, L.J. and Hartless, G. (2001). 'The influence of family and community social capital on educational achievement', *Rural sociology*, 66: pp. 43-68.

Jenkins, H. (2006). 'Convergence culture', New York: New York University Press.

Jennings, M. and Zeitner, V. (2003). 'Internet use and civic engagement: A longitudinal analysis', *Public Opinion Quarterly*, 67(3): pp. 311-334.

Jensen, M.J. and Anduiza, E. (2012). 'Online political participation in the United States and Spain', *Digital media and political engagement worldwide: A comparative study*: pp.80-101.

Johansen, H.P. (2005). 'Political marketing: More than persuasive techniques, an organizational perspective', *Journal of Political Marketing*, 4, 85-105.

Johnson, T.J. and Kaye, B.K. (2003). 'A boost or bust for democracy? How the web influenced political attitudes and behaviors in the 1996 and 2000 presidential elections', *Harvard international journal of Press/Politics*, 8: pp. 9-34.

Johnston, K., Chen, M., M., Hauman, M. (2013). 'Use, perception and attitude of university students towards Facebook and Twitter', *Electronic Journal Information Systems Evaluation Volume*, *16*(3): pp. 200-210.

Jones, N. (1997). 'Campaign 1997: How the general election was won and lost', London:Indigo.

Jones, T. and Taylor, S.F. 2012). 'Service loyalty: accounting for social capital. Journal of Services Marketing'.

Jstor. Available: https://about.jstor.org/ [Accessed 6 December 2016].

Junco, R. (2013). 'Comparing actual and self-reported measures of Facebook use', *Computers in Human Behaviour*, 29: pp. 626–631.

Jung, N., Kim, Y. and Gil de Zuniga, H. (2011). 'The mediating role of knowledge and efficacy in the effects of communication on political participation', *Mass Communication and Society*, 14(4): pp. 407–430.

Jungherr, A. (2016). 'Twitter use in election campaigns: A systematic literature review', *Journal of information technology & politics*, *13*(1): pp.72-91.

Kaigo, M. (2012). 'Social media usage during disasters and social capital: Twitter and the great east Japan earthquake', *Keio Communication Review*, 34: pp. 19–35.

Kaplan, A.M. and Haenlein, M. (2010). 'Users of the world, unite! The challenges and opportunities of Social Media', *Business horizons*, 53: pp. 59-68.

Kaplan, A.M. and Haenlein, M. (2012). 'Social media: back to the roots and back to the future', *Journal of Systems and Information Technology*.

Karp, J. and Banducci, S. (2008). 'When politics is not just a man's game: Women's representation and political engagement', *Electoral Studies* 27(1): pp. 105–115.

Kaufhold, K., Valenzuela, S., and De Zuniga, H.G. (2010). 'Citizen Journalism and Democracy: How User-Generated News Use Relates to Political Knowledge and Participation', *Journalism and Mass Communication Quarterly*, 87(3): pp. 515–529.

Kaun, A. and Stiernstedt, F. (2014). 'Facebook time: Technological and institutional affordances for media memories', *New Media & Society*, 16: pp. 1154-1168.

Kavanaugh, A., Carroll, J.M., Rosson, M.B., Zin, T.T. and Reese, D.D. (2005). 'Community networks: Where offline communities meet online', *Journal of Computer-Mediated Communication*, *10*(4), p.JCMC10417.

Kavanaugh, A., Kim, B.J., Perez-Quinones, M.A., Schmitz, J. and Isenhour, P. (2008). 'Net gains in political participation: Secondary effects of Internet on community', *Information, Communication & Society*, 11: pp. 933-963. Kay, M. and Wobbrock, J. (2016). 'ARTool: aligned rank transform for nonparametric factorial ANOVAs', R package version 0.11. Retrieved at: <u>https://cran.r-</u>project.org/web/packages/ARTool/. Last access: 15/02/2021.

Keith, R. (1960). 'The Marketing Revolution'. Journal of Marketing, 24: pp. 35-8.

Kelley, D. and Sawilowsky, S.S. (1997). 'Nonparametric alternatives to the f statistic in analysis of variance', *Journal of Statistical Computation and Simulation*, 58(4): pp. 343-359.

Kenny,D.A.(2016).'Moderation',Retrievedat:http://davidakenny.net/cm/moderation.htm. Last access: 23/06/2019

Kenski, K. and Stroud, N.J. (2006). 'Connections between Internet use and political efficacy, knowledge, and participation' *Journal of broadcasting & electronic media*, 50: pp. 173-192.

Kent Jennings, M. and Zeitner, V. (2003). 'Internet use and civic engagement: A longitudinal analysis', *Public Opinion Quarterly*, 67: pp. 311-334.

Kim, J.Y. (2006). 'The impact of Internet use patterns on political engagement: A focus on online deliberation and virtual social capital', *Information Polity*, 11: pp. 35-49.

Kim, K.S. and Kim, Y.C. (2007). 'New and old media uses and political engagement among Korean adolescents', *Asian Journal of Communication*, 17: pp. 342-361.

Kim, Y. and Geidner, N. (2008). 'Politics as friendship: The impact of online social networks on young voters' political behavior', In *annual meeting of the International Communication Association*, Montreal, Quebec, Canada.

Kitschelt, H. (1986). 'Political Opportunity Structures and Political Protest: Anti-Nuclear Movements in Four Democracies', *British Journal of Political Science*, 16(1): pp. 57-85.

Klapper, J.T. (1960). 'The effects of mass communication', New York, NY: The Free Press.

Klein, H.K. and Myers, M.D. (1999). 'A set of principles for conducting and evaluating interpretive field studies in information systems', *MIS quarterly*: pp.67-93.

Klein, J. (2005). 'The perils of the permanent campaign', Time Magazine, 5.

Klein, J. (2006). 'Politics Lost', New York: Doubleday.

Klesner, J.L. (2004). 'Social capital and political participation in Latin America', Paper prepared for delivery at the XXV International Congress of the Latin American Studies Association, Las Vegas.

Knack, S. and Keefer, P. (1997). 'Does social capital have an economic payoff? A crosscountry investigation', *The Quarterly journal of economics*, 112(4): pp.1251-1288.

Knapp, T.R. (1990). 'Treating ordinal scales as interval scales: an attempt to resolve the controversy', *Nursing research*, *39*(2): pp. 121-123.

Knoke, D., Yang, S., (2008). 'Social Network Analysis', Thousand Oaks: Sage Publishing Inc.

Ko, H.C. and Kuo, F.Y. (2009). 'Can blogging enhance subjective well-being through self-disclosure?', *Cyberpsychology & behavior*, 12(1): pp.75-79.

Kock, N. (2015). 'Common method bias in PLS-SEM: A full collinearity assessment approach', *International Journal of e-Collaboration (ijec)*, *11*(4): pp.1-10.

Kotler, P. and Levy, S.J. (1969). 'Broadening the concept of marketing', *Journal of marketing*, 33: pp. 10-15.

Kotler, P. (1975). 'Overview of Political Candidate Marketing', *Advances in Consumer Research*, 2: pp. 761-770.

Kotler, P. and Kotler, N. (1999). 'Political Marketing - Genereting effective candidates, campaigns and causes', in B. Neuman (ed.) '*Handbook of Political Marketing*', Thousand Oaks, CA: Sage.

Kotler, P., Roberto, N. and Lee, N. (2002). 'Social marketing: Improving the quality of *life*' (2nd Ed.), Thousand Oaks, CA: Sage Publications.

Kotler, P. and Keller, K.L. (2005). '*Marketing Management*', 12. Aufl. Englewood Cliffd. NJ: Prentice Hall.

Kotler, P. and Keller, K.L. (2012). 'Prentice Hall video library to accompany Marketing management', *Pearson/Prentice Hall*.

Kraut, R., Kiesler, S., Boneva, B., Cummings, J., Helgeson, V. and Crawford, A. (2002). Internet paradox revisited. *Journal of social issues*, *58*(1): pp.49-74.

Kraut, R., Patterson, M., Lundmark, V., Kiesler, S., Mukopadhyay, T. and Scherlis, W., (1998). 'Internet paradox', *Psychologist*, 53(9): pp. 1017–1031.

Kraut, R., Patterson, M., Lundmark, V., Kiesler, S., Mukophadhyay, T. and Scherlis, W. (1998). 'Internet paradox: A social technology that reduces social involvement and psychological well-being', American psychologist, 53: p. 1017.

Krishna, A. (2002). 'Enhancing political participation in democracies: What is the role of social capital?', *Comparative political studies*, *35*(4): pp.437-460.

Krueger, B.S. (2002). 'Assessing the potential of Internet political participation in the United States: A resource approach', *American Politics Research*, 30: pp. 476–498.

Krueger, B.S. (2006). 'A comparison of conventional and Internet political mobilization', *American Politics Research*, 34: pp. 759-776.

Kwak, H., Lee, C., Park, H. and Moon, S. (2010). 'What is Twitter, a social network or a news media?', In *Proceedings of the 19th international conference on World wide web*: pp. 591-600.

Kwon, S-W. and Adler, P.S. (2014). 'Social Capital: Maturation of a Field of Research' *Academic Management Review*, 39(4): pp. 412-422.

Lake, R.L.D. and Huckfeldt, R. (1998). 'Social capital, social networks, and political participation', *Political Psychology*, 19(3): pp. 567–583.

Lampe, C., Vitak, J. and Ellison, N. (2013). 'Users and nonusers: Interactions between levels of adoption and social capital', In *Proceedings of the 2013 conference on Computer supported cooperative work* :pp. 809-820.

Larcinese, V. (2007). 'Does political knowledge increase turnout? Evidence from the 1997 British general election', *Public Choice*, *131*(3): pp.387-411.

Larcinese, V. (2009). 'Information acquisition, ideology and turnout: theory and evidence from Britain', *Journal of Theoretical politics*, 21(2): pp.237-276.

Larsson, A.O. (2016). 'Online, all the time? A quantitative assessment of the permanent campaign on Facebook', *New media & society*, 18(2): pp.274-292.

Lasswell, H.D. (1948). 'The structure and function of communication in society', *The communication of ideas*, 37: pp. 136-139.

Latan, H. (2018). 'PLS Path Modeling in Hospitality and Tourism Research: The Golden Age and Days of Future Past', In: Ali F, Rasoolimanesh SM and Cobanoglu C (eds) *Applying Partial Least Squares in Tourism and Hospitality Research*. Bingley: Emerald: pp. 53-84.

Laumann, E.O. (1966). 'Prestige and association in an urban community: An analysis of an urban stratification system', Indianapolis: Bobbs-Merrill.

Lazarsfeld, P.F. and Merton, R.K. (1954). 'Friendship as a social process: A substantive and methodological analysis', *Freedom and control in modern society*, 18(1): pp.18-66.

Lee, E., Kim, Y.J. and Ahn, J. (2014). 'How do people use Facebook features to manage social capital?', *Computers in Human Behavior*, *36*: pp.440-445.

Lees-Marshment, J. (2001). '*Political marketing and British political parties*', Manchester: Manchester University Press.

Lees-Marshment, J. (2003). 'Political Marketing: How to Reach That Pot of Gold', *Journal of Political Marketing*, 2(1): pp. 1-32.

Lees-Marshment, J. (2009). 'Political Marketing. Principles and Applications', Routladge.

Lees-Marshment, J. (2019a). 'Marketing scholars and political marketing: the pragmatic and principled reasons for why marketing academics should research the use of marketing in the Political Arena', *Customer Needs and Solutions*, 6: pp. 41-48.

Lees-Marshment, J., Conley, B., Elder, E., Pettitt, R., Raynauld, V. and Turcotte, A. (2019b). '*Political marketing: Principles and applications*', Routledge.

Lees-Marshment, J. and Raynauld, V. (2020). 'The case for ethics in Digital Political Marketing Communication', *Political Communication Ethics: Theory and Practice*.

Leonardi, P.M. and Treem, J.W. (2012). 'Knowledge management technology as a stage for strategic self-presentation: Implications for knowledge sharing in organizations', *Information and Organization*, 22: pp. 37-59.

Levy, Y. and Ellis, T.J. (2006). 'A systems approach to conduct an effective literature review in support of information systems research'.

Lewis, M. (2010). 'A hierarchical regression analysis of the relationship between blog reading, online political activity, and voting during the 2008 presidential campaign', *University of North Texas*.

Li, X., Chen, W. and Popiel, P. (2015). 'What happens on Facebook stays on Facebook? The implications of Facebook interaction for perceived, receiving, and giving social support', *Computers in human behavior*, *51*: pp.106-113.

Li, X., Chen, W., Sang, Y. and Yeon Lee, N. (2018). 'Structure and returns: toward a refined understanding of Internet use and social capital', *Information, Communication & Society*, 22(10): pp.1479-1496.

Lilleker, D.G. and Koc-Michalska, K. (2017). 'What drives political participation? Motivations and mobilization in a digital age', *Political Communication*, 34: pp. 21-43.

Lim, M. (2012). 'Clicks, cabs, and coffee houses: Social media and oppositional movements in Egypt', 2004–2011. *Journal of Communication*, 62(2): pp. 231–248.

Lin, H. and Qiu, L. (2013). 'Two sites, two voices: Linguistic differences between Facebook status updates and tweets', In International Conference on Cross-Cultural Design (pp. 432-440). Springer, Berlin, Heidelberg.

Lim, M. (2013). 'Framing Bouazizi: 'White lies', hybrid network, and collective/connective action in the 2010–11 Tunisian uprising', *Journalism*, 14(7): pp. 921–941.

Lin, N. (1999). 'Building a network theory of social capital', *Connections*, 22(1): pp.28-51.

Lin, N. (2001). 'Social capital: A theory of social structure and action', (No. 19), *Cambridge university press*.

Lin, N. (2008). 'A network theory of social capital', *The handbook of social capital*: pp. 50-69.

Lin, N. and Ao, D. (2008). 'The Invisible Hand of Social Capital: An Exploratory Study', in N. Lin, and B. Erickson, *Social Capital. An International Research Program*, London: Oxford University Press: pp. 107-132.

Lin, T.T. and Chiang, Y.H. (2017). 'Dual screening: Examining social predictors and impact on online and offline political participation among Taiwanese Internet users', *Journal of Broadcasting & Electronic Media*, *61*(2): pp.240-263.

Lindner, R. and Riehm, U. (2011). 'Broadening participation through e-petitions? An empirical study of petitions to the German parliament', *Policy & Internet*, 3: pp. 1-23.

Little, T.D., Lindenberger, U. and Nesselroade, J.R. (1999). 'On selecting indicators for multivariate measurement and modeling with latent variables: When "good" indicators are bad and "bad" indicators are good', *Psychological Methods*, 4: pp. 192–211.

Liu, Y., Venkatanathan, J., Goncalves, K., Karapanos, E. and Kostakos, V., (2014). 'Modeling What Friendship Patterns on Facebook Reveal About Personality and Social Capital', *ACM Transactions on Computer-Human Interaction*, 21(3): pp. 1-20.

Livingstone, S. (2013). 'The participation paradigm in audience research', *The communication review*, 16: pp. 21-30.

Lloyd, J. (2004). 'What The Media Are Doing to Our Politics', London: ConsTable.

Lock, A. and Harris, P. (1996). 'Political Marketing: Vive la Difference!', *European Journal of Marketing*, 30(10/11): pp. 21-31.

Lou, L.L., Yan, Z., Nickerson, A. and McMorris, R. (2012). 'An examination of the reciprocal relationship of loneliness and Facebook use among first-year college students', *Journal of Educational Computing Research*, 46(1): pp. 105-117.

Loury, G. (1992). 'The economics of discrimination: Getting to the core of the problem', *Harvard Journal for African American Public Policy*, *1*(1): pp.91-110.

Luepsen, H. (2017). 'The aligned rank transform and discrete variables: A warning', *Communications in Statistics-Simulation and Computation*, 46: pp. 6923-6936.

Lusch, R.F. and Vargo, S.L. (2006). 'Service-dominant logic: reactions, reflections and refinements', *Marketing theory*, 6: pp. 281-288.

Lusch, R.F., Vargo, S.L. and O'brien, M. (2007). 'Competing through service: Insights from service-dominant logic', *Journal of retailing*, 83: pp. 5-18.

Lusch R.F., Vargo, S.L. and Wessels, G. (2008). 'Toward a Conceptual Foundation for Service Science: Contributions from Service-Dominant Logic', *IBM Systems Journal*, 47(1): pp. 5-14.

Lusch, R.F. and Webster, F.E. Jr. (2011). 'A Stakeholder-Unifying, Co-creation Philosophy of Marketing', *Journal of Macromarketing*, 31 (2): pp. 129-134.

Lusoli, W. and Ward, S. (2004). 'Digital Rank-and-File: Party Activists' Perceptions and Use of the Internet', *The British Journal of Politics and International Relations*, 6: pp. 453-470.

Lynch, R., Baines, P. and Egan, J. (2006). 'Long-term performance of political parties: towards a competitive resource-based perspective', *Journal of Political Marketing*, 5: pp. 71-92.

Maarek, P.J. (2007). 'The Idea of Political Marketing', edited by Nicholas J. O'Shaughnessy and Stephan CM Henneberg *Global Political Campaigning: A Worldwide Analysis of Campaign Professionals and Their Practices*, by Fritz Plasser, with Gunda Plasser: Westport, CT: Praeger, 2002. 272: pp. 98.95 cloth Westport, CT:Praeger,2002.424: pp. 93.95 cloth. Taylor & Francis.

Macedo, S., Alex-Assensoh, Y. and Berry, J. (2005). '*Democracy At Risk: How Political Choices Undermine Citizen Participation And What We Can Do About It*', Washington, DC: Brookings Institution Press.

MacKenzie, S.B., Podsakoff, P.M., Jarvis, C.B. (2005). 'The problem of measurement model misspecification in behavioral and organizational research and some recommended solutions', *Journal of Applied Psychology*, *90*(4): pp. 710-730.

MacKenzie, S.B. and Podsakoff, P.M. (2012). 'Common method bias in marketing: Causes, mechanisms, and procedural remedies', *Journal of retailing*, 88: pp. 542-555.

MacPherson, C.B. (1977). '*The Life and Times of Liberal Democracy*', Oxford: Oxford University Press.

Mair, P. (2005). '*Democracy beyond Parties*'. Centre for the Study of Democracy, Irvine: University of California.

Malhotra, N.K., Nunan, D. and Birks, D.F. (2017). '*Marketing research: an applied approach*', 5th ed. London, UK: Pearson Education Limited.

Margolis, M. and Resnick, D., (2000). '*Politics as usual: The cyberspace revolution*'. Thousand Oaks: Sage Publications.

Marsh, A. (1979). 'The new matrix of political action', Futures, 11: pp. 91-103.

Marsh, D. and Furlong, P. (2002). 'A skin not a sweater: ontology and epistemology in political science', *Theory and methods in political science*, 2: pp. 17-41.

Marshall, T.H. (1952). '*Citizenship and Social Class*', Cambridge: Cambridge University Press.

Martin, A.J. (2012). 'Young People and Politics: Political Engagement in the Anglo-American Democracies', London: Routledge.

Mascheroni, G. (2012). 'Online participation: New forms of civic and political engagement or just new opportunities for networked individualism'.

Mauser, G.A. (1983). '*Political marketing: An approach to campaign strategy*', New York, NY, USA: Praeger.

Maurer, C.C., Bansal, P. and Crossan, M.M. (2011). 'Creating economic value through social values: Introducing a culturally informed resource-based view', *Organization Science*, 22(2): pp.432-448.

McCarthy, E.J. (1960). '*Basic marketing: a managerial approach*', Homewood, Illinois: Richard D. Irwin. Inc. McCarthyBasic Marketing: A Managerial Approach1960.

McCarthy, J.D. and Zald, M.N. (1977). 'Resource mobilization and social movements: A partial theory', *American journal of sociology*, 82: pp. 1212-1241.

McClurg, S.D. (2006). 'The Electoral Relevance of Political Talk: Examining Disagreement and Expertise Effects in Social Networks on Political Participation', *American journal of political science*, 50(3): pp. 737-54.

McCluskey, M.R., Deshpande, S., Shah, D.V. and McLeod, D.M. (2004). 'The efficacy gap and political participation: When political influence fails to meet expectations', *International Journal of Public Opinion Research*, 16: pp. 437–455.

McGrenere, J. and Ho, W. (2000). 'Affordances: Clarifying and evolving a concept', In *Graphics interface* (Vol. 2000): pp. 179-186).

McLeod, J.M., Scheufele, D.A. and Moy, P. (1999). 'Community, communication, and participation: The role of mass media and interpersonal discussion in local political participation', *Political communication*, *16*(3): pp.315-336.

McNair, B. (2017). 'An introduction to political communication', Taylor & Francis. *Mediated Communication*, 12(4).

McPherson, M., Smith-Lovin, L. and Cook, J.M. (2001). 'Birds of a feather: Homophily in social networks', *Annual review of sociology*, 27(1): pp. 415-444.

Melo, D. and Stockemer, D. (2014). 'Age and political participation in Germany, France and the UK: A comparative analysis', *Comparative European Politics* 12(1): pp. 33-53.

Memon, M.A., Cheah, J.H., Ramayah, T., Ting, H., Chuah, F. and Cham, T. H. (2019). 'Moderation analysis: issues and guidelines', *Journal of Applied Structural Equation Modeling*, 3: pp. 1-11.

Michels, A. (2011). 'Innovations in democratic governance: how does citizen participation contribute to a better democracy?', *International Review of Administrative Sciences*, 77: pp. 275-293.

Michels, A. (2012). 'Citizen participation in local policy making: Design and democracy', *International Journal of Public Administration*, 35: pp. 285-292.

Miller, W.E., Miller, A.H., and Schneider, E.J. (1980). 'American National Education Studies Data Source Book, 1952-1978', *Cambridge: Harvard Universitv Press*.

Miyata, K., Kobayashi, T. (2008). 'Causal relationship between Internet use and social capital in Japan', *Asian Journal of Social Psychology*, *11*(1): pp.42-52.

Moeller, J., De Vreese, C., Esser, F. and Kunz, R. (2014). 'Pathway to political participation: The influence of online and offline news media on internal efficacy and turnout of first-time voters', *American behavioral scientist*, *58*(5): pp.689-700.

Molyneux, L., Vasudevan, K. and Gil De Zuniga, H. (2015). 'Gaming social capital: Exploring civic value in multiplayer video games', *Journal of Computer-Mediated Communication*, 20: pp. 381-399.

Moore, S., Shiell, A., Hawe, P. and Haines, V.A. (2005). 'The privileging of communitarian ideas: citation practices and the translation of social capital into public health research', *American journal of public health*, 95(8): pp.1330-1337.

Morozov, E. (2011). '*The net delusion: How not to liberate the world*', Democracy: A Reader, 436.

Morse, S.W. (1993). 'The practice of citizenship: Learn by doing', *The Social Studies* 84(4): pp. 164–168.

Mortimore, R. (2002). 'Why Politics needs marketing'. UK Academy of Marketing Conference on Political Marketing Conference, Aberdeen University, Scotland.
Mou, Y., Atkin, D., Fu, H., Lin, C.A. and Lau, T. (2013). 'The influence of online forum and SNS use on online political discussion in China: Assessing 'spirals of trust''. *Telematics and Informatics*, 30: pp. 359-369.

Moy, P., Manosevitch, E., Stamm, K. and Dunsmore, K. (2005). 'Linking dimensions of internet use and civic engagement', *Journalism and Mass Communication Quarterly*, 82(3): pp. 571–586.

Moy, P., Bimber, B., Rojecki, A., Xenos, M.A. and Iyengar, S. (2012). 'Shifting contours in political communication research', *International Journal of Communication*, 6: pp. 247-254.

Munroe, T. (2002). 'An introduction to politics: Lectures for first-year students', *Canoe Press* (IL).

Musick, M., and Wilson, J. (2008). '*Volunteers: A social profile*', Bloomington, IN: Indiana University Press. Norman, D.

Mutz, D.C. (2002). 'The consequences of cross-cutting social networks for political participation', *American Journal of Political Science*, 46(4) : pp. 838-855.

Mutz, D.C. (2006). '*Hearing the Other Side: Deliberative versus Participatory Democracy*', Cambridge: Cambridge University Press.

Nadkarni, A. and Hofmann, S.G. (2012). 'Why do people use Facebook? Personality and individual differences', 52: pp. 243-249.

Nahapiet, J. and Ghoshal, S. (1998). 'Social Capital, Intellectual Capital, and the Organizational Advantage', *The Academy of Management Review*, Vol. 23, No. 2: pp. 242-266.

Neves, B. (2013). 'Social Capital and Internet Use: the irrelevant, the bad and the good', *Sociology compass*, 7(8): pp. 599-611.

Newman, B.I., (1994), '*The marketing of the president: Political marketing as campaign strategy*', SAGE Publications, Inc., Thousand Oaks.

Newman, B.I. (2001), 'Commentary - Image-manufacturing in the USA: recent US Presidential Elections and beyond', *European Journal of Marketing*, 35(9/10): pp. 966 - 970.

Newstead, T., Dawkins, S., Macklin, R. and Martin, A. (2019). 'We don't need more leaders–We need more good leaders. Advancing a virtues-based approach to leader (ship) development', *The Leadership Quarterly*, 101312.

Nie, N.H., Hillygus, D.S. and Erbring, L. (2002). 'Internet use, interpersonal relations and sociability: A time diary study', in B. Wellman and C. Haythornthwaite (Eds.), *Internet and everyday life*, Oxford: Blackwell: pp. 215- 243.

Niemi, R.G., Craig, S.C. and Mattei, F. (1991). 'Measuring internal political efficacy in the 1988 National Election Study,. *The American Political Science Review*: pp.1407-1413.

Nisbet, M.C. and Scheufele, D.A. (2004). 'Political talk as a catalyst for online citizenship', *Journalism & mass communication quarterly*, 81: pp. 877-896.

Norman, D.A. (1988). 'The psychology of everyday things', Basic books.

Norris, P. (2000). 'A virtuous circle: Political communications in post-industrial societies', Cambridge University Press.

Norris, P. (2001). 'Digital divide: Civic engagement, information poverty, and the Internet worldwide', New York: Cambridge University Press.

Norris, P. (2002). 'Democratic phoenix: Reinventing political activism', *Cambridge University Press*.

Norris, P. (2003). 'Preaching to the converted? Pluralism, participation and party websites', Party politics, 9: pp. 21-45.

Nunnally, J.C. and Bernstein, I.H. (1994). '*Psychometric theory*', New York: McGraw-Hill.

O'Cass, A. (2001). 'Political marketing - An investigation of the political marketing concept and political market orientation in Australian politics', *European Journal of Marketing*, 35 (9/10): pp. 1003 - 1025.

O'Cass, A. (2009). 'A Resource-based View of the Political Party and Value Creation for the Voter-citizen: An Integrated Framework for Political Marketing', *Marketing Theory*, 9(2): pp. 189-208.

O'Leary, R. and Iredale, I. (1976). 'The Marketing Concept: Quo Vadis?', *European Journal of Marketing*, 10(3): pp. 1446–57.

O'Neill, B. (2010). 'The Media's Role in Shaping Canadian Civic and Political Engagement', *Policy and Society*, 29: pp. 37–51.

O'Shaughnessy, N.J. (1990). 'The Phenomenon of Political Marketing', Macmillan, Basingstoke.

O'Shaughnessy, N. (2001). 'The marketing of political marketing', *European Journal of Marketing*.

O'Shaughnessy, N. (2002). 'Toward an ethical framework for political marketing', *Psychology & Marketing*, *19*(12), pp.1079-1094.

Obukhova, E. and Lan, G. (2013). 'Do job seekers benefit from contacts? A direct test with contemporaneous searches', *Management Science*, *59*(10): pp.2204-2216.

O'cass, A. (2009). 'A resource-based view of the political party and value creation for the voter-citizen: an integrated framework for political marketing', *Marketing Theory*, 9: pp. 189-208.

O'leary, R. and Iredale, I. (1976). 'The marketing concept: quo vadis?', *European Journal of Marketing*.

Ohtani, K. (2000). 'Bootstrapping R2 and adjusted R2 in regression analysis', *Economic Modelling*, 17(4): pp. 473-483.

Oliver, W.M., (2000). 'The third generation of community policing: Moving through innovation, diffusion, and institutionalization', *Police quarterly*, *3*(4): pp.367-388.

Olufadi, Y. (2015). 'Social networking time use scale (SONTUS): A new instrument for measuring the time spent on the social networking sites', *Telematics and Informatics*, *33*(2): pp.452-471.

ONS, (2013). 2011 Census: Key Statistics and Quick Statistics for Local Authorities intheUnitedKingdom.Availableat:https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populationestimates/bulletins/keystatisticsandquickstatisticsforlocalauthoritiesintheunitedkingdom/2013-10-11.Last access: 10/03/2021.

ONS, (2019). Population estimates for the UK, England and Wales, Scotland and Northern Ireland: mid-2019. Available at: https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populationestimates/bulletins/annualmidyearpopulationestimates/mid2019estimates. Last access: 10/03/2021.

Orlikowski, W.J. and Baroudi, J.J. (1991). 'Studying information technology in organizations: Research approaches and assumptions', *Information systems research*, 2(1): pp.1-28.

Ormrod, R.P. (2005). 'A conceptual model of political market orientation', Journal of Nonprofit & Public Sector Marketing, 14: pp. 47-64.

Ormrod, R.P. (2007). 'Political market orientation and its commercial cousin: close family or distant relatives?', *Journal of Political Marketing*, 6 (2/3): pp. 69-90.

Ormrod R.P. and Savigny, H. (2012). 'Political market orientation: a framework for understanding relationship structures in political parties', *Party Politics*, 18(4): pp. 487-502.

Ormrod R.P., Henneberg S. and O'Shaughnessy N.J. (2013). '*Political Marketing: Theory and Concepts*', Sage.

Oser, J., Hooghe, M. and Marien, S. (2013). 'Is Online Participation Distinct from Offline Participation? A Latent Class Analysis of Participation Types and Their Stratification', *Political Research Quarterly*, 66(1): pp. 91-101.

O'shaughnessy, N.J. (1990). 'The phenomenon of political marketing', Springer.

O'shaughnessy, N.J. (2002). 'The Marketing of Political Marketing', *The Idea of Political Marketing*, 209.

Ostrom, E. (2000). 'Social capital: a fad or a fundamental concept', *Social capital: A multifaceted perspective*, 172(173): pp.195-98.

Ostrom, E. and Ahn, T.K. (2003). 'A social science perspective on social capital: Social capital and collective action', *Revista Mexicana De Sociologia*, 65(1): pp.155-233.

O'toole, T., Lister, M., Marsh, D., Jones, S. and Mcdonagh, A. (2003). 'Tuning out or left out? Participation and non-participation among young people', *Contemporary politics*, 9: pp. 45-61.

Padilla-Walker, L.M. and Coyne, S.M. (2011). 'Turn that thing off! Parent and adolescent predictors of proactive media monitoring', *Journal of adolescence*, *34*(4): pp.705-715.

Page, T., Putterman, L. and Unel, B. (2005). 'Voluntary association in public goods experiments: Reciprocity, mimicry and efficiency', *The Economic Journal*, 115(506): pp.1032-1053.

Palmer, J. (2002). 'Smoke and mirrors: is that the way it is? Themes in political marketing', *Media, Culture & Society*, 24: pp. 345-363.

Papacharissi, Z. and Mendelson, A. (2011). 'Toward a new (er) sociability: uses, gratifications and social capital on Facebook', *Media perspectives for the 21st century*: p. 212.

Park, C.S. and Karan, K. (2014). 'Unraveling the relationships between smartphone use, exposure to heterogeneity, political efficacy, and political participation: A mediation model approach', *Asian Journal of Communication*, 24(4): pp. 370-389.

Park, C.S. (2019). 'Does too much news on social media discourage news seeking? Mediating role of news efficacy between perceived news overload and news avoidance on social media', *Social Media+ Society*, 5, 2056305119872956.

Park, N., Kee, K.F. and Valenzuela, S. (2009). 'Being immersed in social networking environment: Facebook groups, uses and gratifications, and social outcomes', Cyberpsychology & behavior, 12: pp. 729-733.

Park, T. (2004) 'Jeong, Familism and Interpersonal Trust', Korean Review of Public Administration, 38(6) : pp. 23-45.

Parry, G., Moyser, G., Day, N., (1992). '*Political Participation and Democracy in Britain*', Cambridge: Cambridge University Press.

Pasek, J., More, E. and Romer, D. (2009). 'Realizing the social Internet? Online social networking meets offline civic engagement', *Journal of Information Technology & Politics*, 6(3-4) : pp. 197-215.

Pateman, C. (1970). '*Participation and democratic theory*', Cambridge: Cambridge University Press.

Pattie, C., Seyd, P. and Whiteley, P. (2004). '*Citizenship in Britain: Values, Participation and Democracy*', Cambridge: Cambridge University Press.

Paxton, P. (1999). 'Is social capital declining in the United States? A multiple indicator assessment', *American Journal of Sociology*, *105*(1): pp. 88–127.

Pea, R., Nass, C., Meheula, L., Rance, M., Kumar, A., Bamford, H., Nass, M., Simha, A., Stillerman, B., Yang, S. and Zhou, M. (2012). 'Media use, face-to-face communication, media multitasking, and social well-being among 8-to 12-year-old girls', *Developmental psychology*, 48(2) : pp. 327-336.

Pearce, K.E., Barta, K. and Fesenmaier, M.A. (2015). 'The affordances of social networking sites for relational maintenance in a distrustful society: The case of Azerbaijan', *Social Media*+ *Society*, *1*(2), p.2056305115616150.

Perannagari, K.T. and Chakrabarti, S. (2020). 'Analysis of the literature on political marketing using a bibliometric approach', *Journal of Public Affairs*, 20, e2019.

Perugini, M., Gallucci, M., Presaghi, F. and Ercolani, A.P. (2003). 'The Personal Norm of Reciprocity', *European Journal of Personality*, 17(4): pp. 251-283.

Peters, Y. (2017). 'Political participation, diffused governance, and the transformation of democracy: patterns of change', Routledge.

Petersen, C. and Johnston, K.A. (2015). 'The impact of social media usage on the cognitive social capital of university students', *Informing Science: the International Journal of an Emerging Transdiscipline*, 18: pp. 1-30.

Peterson, M. and Godby, R.W. (2020). 'Citizen Participation in Political Markets: Extending Service-Dominant Logic to Public Policy' *Journal of Public Policy & Marketing*, 39: pp. 412-429.

Petrocchi, N., Asnaani, A., Martinez, A.P., Nadkarni, A. and Hofmann, S.G. (2015). 'Differences between people who use only Facebook and those who use Facebook plus Twitter', *International journal of human-computer interaction*, 31: pp. 157-165.

Petrocchi, N. and Hofmann, S.G. (2015). 'Differences Between People Who Use Only Facebook and Those Who Use Facebook Plus Twitter', *International Journal of Human-Computer Interaction*, 31(2): pp. 157-165.

Pettigrew, T.F. and Tropp, L.R. (2006). 'A meta-analytic test of intergroup contact theory', *Journal of personality and social psychology*, 90(5), p.751.

Pew Research Center (2015). '5 key takeaways about Twitter, Facebook and news use', Available at <u>https://www.pewresearch.org/fact-tank/2015/07/14/5-key-takeaways-about-twitter-facebook-and-news-use/</u>. Last access: 22/11/2020.

Phua, J., Jin, S. V. and Kim, J. J. (2017). 'Uses and gratifications of social networking sites for bridging and bonding social capital: A comparison of Facebook, Twitter, Instagram, and Snapchat', *Computers in human behavior*, 72: pp. 115-122.

Pitas, N. and Ehmer, C. (2020). 'Social Capital in the Response to COVID-19', *American Journal of Health Promotion*, *34*(8): pp.942-944.

Plauche, G.A. (2006). 'De Virtute Libertateque Romanorum. paper presented at the annual conference of the Louisiana Political Science Association at Nicholls State University', Thibodaux, LA on March 3, 2006.

Podsakoff, P.M., MacKenzie, S.B., Lee, J.Y. and Podsakoff, N.P. (2003). 'Common method biases in behavioral research: a critical review of the literature and recommended remedies', *Journal of applied psychology*, 88(5): p.879.

Polat, R.K. (2005). 'The Internet and political participation: Exploring the explanatory links', *European Journal of Communication*, 20: pp. 435–459.

Politica.Nanopress. 2003. politica.nanopress.it [Online]. Available: www.politica.nanopress.it [Accessed 2015].

Portes, A. (1998). 'Social Capital: Its Origins and Applications in Modern Sociology', in *Annual Review of Sociology* (Vol. 24). Annual Reviews.

Press, A.L., Williams, B.A., Press, A.L. and Williams, B.A. (2010). 'The new media environment: An introduction', Wiley-Blackwell Malden.

Putnam, R.D. (1993). 'The prosperous community: Social capital and public life', *The american prospect*, 13.

Putnam, R.D. (1995). 'Tuning in, tuning out: The strange disappearance of social capital in America', *Political Science and Politics*, 28: pp. 664–683.

Putnam, R.D. (2000). 'Bowling alone: The collapse and revival of American community', *Simon and schuster*.

Quan-Haase, A. and Wellman, B., (2004). 'How Does the Internet Affect Social Capital?', in M. Huysman, and V. Wulf, *Social Capital and Information Technology*, Cambridge: The MIT Press: pp. 113-132.

Quan-Haase, A., Cothrel, J. and Wellman, B., (2005). 'Instant messaging for collaboration: A case study of a high-tech firm', *Journal of Computer-Mediated Communication*, 10(4).

Quan-Haase, A. and Young, A.L. (2010). 'Uses and gratifications of social media: A comparison of Facebook and instant messaging', Bulletin of science, technology & society, 30: pp. 350-361.

Quelch, J.A. and Jocz, K.E. (2007). 'Greater Good: How Good Marketing Makes for a Better Democracy', Boston, MA: Harvard Business Press.

Quintelier, E. (2007). 'Differences in political participation between young and old people', *Contemporary Politics*, 13(2): pp. 165-180.

Quintelier, E. and Vissers, S. (2008). 'The effect of Internet use on political participation: An analysis of survey results for 16-year-olds in Belgium', Social science computer review, 26: pp. 411-427.

Rainie, L. and Smith, A. (2012). 'Politics on social Networking Sites', PewResearchCenter.

Rash, W. (1997). 'Politics on the nets: Wiring the political process', WH Freeman & Co.

Rea, L.M. and Parker, R.A. (2014). 'Designing and conducting survey research: A comprehensive guide', John Wiley & Sons.

Reich, S.M., Subrahmanyam, K. and Espinoza, G. (2012). 'Friending, IMing, and hanging out face-to-face: overlap in adolescents' online and offline social networks'. *Developmental psychology*, *48*(2): pp. 356-368.

Reinartz, W., Haenlein, M. and Henseler, J. (2009). 'An empirical comparison of the efficacy of covariance-based and variance-based SEM', *International Journal of research in Marketing*, 26(4): pp. 332-344.

Resnick, P. (2002). 'Beyond bowling together: Sociotechnical capital', *HCI in the New Millenium*: pp. 247-272.

Rice, L.L., Moffett, K.W. and Madupalli, R. (2013). 'Campaign-related social networking and the political participation of college students', *Social Science Computer Review*, 31(2): pp. 257-279.

Riley, C.E., Griffin, C. and Morey, Y. (2010). 'The case for 'everyday politics': Evaluating neo-tribal theory as a way to understand alternative forms of political participation, using electronic dance music culture as an example', *Sociology*, 44(2): pp. 345–363.

Ritchie, R. and Sridharan, S. (2007). 'Marketing in subsistence markets: Innovation through decentralization and externalization. Product and Market Development for Subsistence Marketplaces', *Emerald Group Publishing Limited*.

Ritter, J.A. (2008). 'A national study predicting licensed social workers' levels of political participation: The role of resources, psychological engagement, and recruitment networks', *Social Work*, 53(4): pp.347-357.

Robins, K. and Webster, F. (1999). '*Times of the technoculture: From the information society to the virtual life*', Psychology Press.

Robinson, J.P. and Martin, S. (2010). 'IT Use and Declining Social Capital?', *Journal of Social Science Computer Reviews*, 28(1).

Robinson, J.P., De Haan, J. and Kraut, R. (2006). 'Information Technology and Family Time Displacement'.

Robison, L.J., Schmid, A.A. and Siles, M.E. (2002). 'Is social capital really capital?', *Review of social economy*, 60(1): pp.1-21.

Rodela, T.T., Sultana, A., McKyer, E.L.J., Bhattacharya, S. and Hossain, M.M. (2020). 'Social capital and mental health during the COVID-19 pandemic'.

Rojas, H. and Puig-I-Abril, E. (2009). 'Mobilizers mobilized: Information, expression, mobilization and participation in the digital age', *Journal of Computer-Mediated Communication*, 14: pp. 902-927.

Rönkkö, M., and Evermann, J. (2013). 'A critical examination of common beliefs about partial least squares path modeling', *Organizational Research Methods*, 16: pp. 425–448.

Rosen, L.D., Whaling, K., Carrier, L.M., Cheever, N.A. and Rokkum, J. (2013). 'The media and technology usage and attitudes scale: An empirical investigation', *Computers in human behavior*, 29(6): pp.2501-2511.

Rossiter, J.R. (2002). 'The C-OAR-SE Procedure for Scale Development in Marketing', *International Journal of Research in Marketing*, 19 (4) : pp. 305–35.

Rossiter, J.R. (2008). 'Content Validity of Measures of Abstract Constructs in Management and Organizational Research', *British Journal of Management*, 19 (4) : pp. 380–88.

Rovai, A.P., Baker, J.D. and Ponton, M.K. (2013). 'Social science research design and statistics: A practitioner's guide to research methods and IBM SPSS', Watertree Press, LLC.

Rucht, D., (2007). 'The Spread of Protest Politics' in R., J., Dalton and H.-D., Klingemann (eds), *The Oxford Handbook of Political Behavior*, Oxford: Oxford University Press.

Ryfe, D.M. (2001). History and Political Communication: An Introduction. *Political Communication*, 18(4): pp. 407-420.

Sajuria, J., vanHeerde-Hudson, J., Hudson, D., Dasandi, N. and Theocharis, Y. (2015). 'Tweeting alone? An analysis of bridging and bonding social capital in online networks', *American Politics Research*, *43*(4): pp.708-738.

Saldana, M., McGregor, S.C. and Gil de Zuniga, H. (2015). 'Social media as a public space for politics: Cross-national comparison of news consumption and participatory behaviours in the United States and the United Kingdom', *International Journal of Communication*, 9: pp. 3304–3326.

Salter, K.C. and Fawcett, R.F. (1993).' The art test of interaction: A robust and powerful rank test of interaction in factorial models', *Communications in Statistics: Simulation and Computation*, 22(1): pp. 137–153.

Sarstedt, M. and Mooi, E. (2019). 'A concise guide to market research. The Process, Data, and Methods Using IBM SPSS Statistics', *Springer*.

Sarstedt, M., Henseler, J. and Ringle, C.M. (2011). 'Multigroup analysis in partial least squares (PLS) path modeling: Alternative methods and empirical results', In *Measurement and research methods in international marketing*. Emerald Group Publishing Limited.

Savigny, H. (2008). 'The Problem of Political Marketing', New York: Continuum.

Savigny, H. (2009), 'Political marketing' *Oxford Handbook of British Politics* eds. A. Gamble, M. Flinders, C. Hay and M. Kenny (Oxford: Oxford University Press): pp. 798-817.

Savigny, H. and Wring, D.J. (2009). 'An Ideology of Disconnection: for a critical marketing', *Central European Journal of Communication*, 2(2): pp.251-67.

Sawilowsky, S.S. (1990). 'Nonparametric tests of interaction in experimental design', *Review of Educational Research*, 60(1): pp. 91-126.

Scammell, M. (1999). 'Political marketing: lessons from political science', *Political Studies*, 47: pp. 718-739.

Scammell, M. (2000). 'Media and Democracy: An Introduction Part II', in M. Scammell and H. Semetko (eds) *Media, Journalism and Democracy: A Reader*, Aldershot: Dartmouth-Ashgate.

Scammell, M. (2003). 'Citizen Consumers. Media and the Restyling of Politics: Consumerism, Celebrity and Cynicism' Sage: Thousand Oaks, CA, USA: pp. 117-136.

Scheufele, D.A. and Nisbet, M.C. (2002). 'Being a citizen online: New opportunities and dead ends', *Harvard International Journal of Press/Politics*, 7: pp. 55-75.

Schlozman, K.L., Verba, S. and Brady, H.E. (2010). 'Weapon of the strong? Participatory inequality and the Internet', *Perspectives on Politics*: pp. 487-509.

Schuefele, D.A. and Nisbet, M. (2002). 'Being a citizen online: new opportunities and dead ends', *The Harvard Journal of Press/Politics*, 7(3): pp. 55–75.

Schumaker, R. and Lomax, R., (2004). 'A Beginners Guide to Structural Equation Modeling'. New Jersey: Lawrence Erlbaum Associates.

Schumpeter, J.A. (1976). '*Capitalism, Socialism and Democracy*', New York: Harper & Row.

Seyd, P. and Whiteley, P. (1992). 'Labour's Grass Roots: The Politics of Party Membership'. Oxford: Clarendon Press.

Shah, D.V., Kwak, N. and Holbert, L.R. (2001a). 'Connecting and disconnecting with civic life: Patterns of Internet use and the production of social capital' *Political communication*, 18: pp. 141-162.

Shah, D.V., Mcleod, J.M. and Yoon, S.H. (2001b). 'Communication, context, and community: An exploration of print, broadcast, and Internet influences', *Communication research*, 28: pp. 464-506.

Shama, A. (1976). 'The marketing of political candidates. Journal of the academy of marketing science', 4: pp. 764-777.

Sharma, P.N., Shmueli, G., Sarstedt, M., Danks, N. and Ray, S. (2019a). 'Predictionoriented model selection in partial least squares path modeling', *Decision Sciences*.

Sharma, P.N., Sarstedt, M., Shmueli, G., Kim, K.H. and Thiele, K.O. (2019b). 'PLSbased model selection: The role of alternative explanations in information systems research', *Journal of the Association for Information Systems*, 20(4), Article 4.

Shaw, E., (1994). 'The Labour Party Since 1979: Crisis and Transformation', London, Routledge.

Shmueli, G., Ray, S., Estrada, J.M.V., and Chatla, S.B. (2016). 'The elephant in the room: Predictive performance of PLS models', *Journal of Business Research*, 69(10): pp. 4552-4564

Sigel, R.S. and Hoskin, M.B. (1981). '*The political involvement of adolescents*', Rutgers University Press.

Sirianni, C. and Friedland, L. (1997). 'Civic innovation & American democracy', *Change: The Magazine of Higher Learning*, 29: pp. 14-23.

Skocpol, T. (1995). 'Social policy in the United States: Future possibilities in historical perspective' (Vol. 172). *Princeton University Press*.

Skoric, M.M., Ying, D. and Ng, Y. (2009). 'Bowling online, not alone: Online social capital and political participation in Singapore', *Journal of Computer-Mediated Communication*, 14: pp. 414-433.

Skoric M.M. and Kwan G.C.E. (2011). 'Platforms for mediated sociability and online social capital: the role of Facebook and massively multiplayer online games', *Asian Journal of Communication*, 21(5): pp. 467–484.

Skoric, M.M. and Poor, N. (2013). 'Youth engagement in Singapore: The interplay of social and traditional media', *Journal of Broadcasting & Electronic Media*, 57(2): pp.187-204.

Skoric, M.M. and Zhu, Q. (2016). 'Social media and offline political participation: Uncovering the paths from digital to physical', *International Journal of Public Opinion Research*, 28(3): pp. 415–427.

Skoric, M.M., Zhu, Q., Goh, D. and Pang, N. (2016). 'Social media and citizen engagement: A meta-analytic review', *New Media & Society*, 18: pp. 1817-1839.

Small, M.L. (2009). 'Unanticipated gains: Origins of network inequality in everyday life', Oxford University Press.

Small, T.A. (2011). 'What the hashtag? A content analysis of Canadian politics on Twitter', *Information, communication & society*, *14*(6): pp. 872-895.

Somma, M. (2009). 'Traditional living practices: Return to the villages', In *The transition to sustainable living and practice*. Emerald Group Publishing Limited.

Stanton-Salazar, R.D. and Dornbusch, S.M. (1995). 'Social capital and the reproduction of inequality: Information networks among Mexican-origin high school students', *Sociology of education*: pp.116-135.

Statista (2021a). 'Distribution of Facebook users in the United Kingdom (UK) as of February 2021, by age group and gender', Available at: https://www.statista.com/statistics/1030057/facebook-users-united-kingdom-agegender/. Last access: 10/03/2021.

Statista (2021b). '*Population of the United Kingdom (UK) in 2019, by gender and country*', Available at: <u>https://www.statista.com/statistics/281240/population-of-the-united-kingdom-uk-by-gender/</u>. Last access: 10/03/2021.

Steenkamp, J.B. and Van Trijp, H. (1991). 'The Use Of LISREL In Validating Marketing Constructs', *International Journal of Research in Marketing*, 8 (4): pp. 283-299.

Steinfield, C., Ellison, N.B. and Lampe, C. (2008). 'Social capital, self-esteem, and use of online social network sites: A longitudinal analysis', *Journal of applied developmental psychology*, 29: pp. 434-445.

Street, J., (2003). 'The Celebrity Politician: Political Style and Popular Culture', in J. Corner and D. Pels (eds) *Media and the Restyling of Politics*, London: Sage: pp. 85–98,

Stoker, G., (2006). 'Why Politics Matter: Making Democracy Work', Houndmills and New York: Palgrave Macmillan.

Streukens, S. and Leroi-Werelds, S. (2016). 'Bootstrapping and PLS-SEM: A step-bystep guide to get more out of your bootstrap results', *European Management Journal*, 34(6): pp.618-632.

Sturgis, P., Allum, N. and Smith, P. (2008). 'An experiment on the measurement of political knowledge in surveys', *Public Opinion Quarterly*, 72(1): pp.90-102.

Stutzman, F., Vitak, J., Ellison, N., Gray, R. and Lampe, C. (2012). 'Privacy in interaction: Exploring disclosure and social capital in Facebook', In *Proceedings of the International AAAI Conference on Web and Social Media* (Vol. 6, No. 1).

Su, C.C. and Chan, N.K. (2017). 'Predicting social capital on Facebook: The implications of use intensity, perceived content desirability, and Facebook-enabled communication practices', *Computers in Human Behavior*, 72: pp.259-268.

Sullivan, G.M. and Artino Jr, A.R. (2013). 'Analysing and interpreting data from Likerttype scales', *Journal of graduate medical education*, 5(4): pp.541-542.

Swanson, D.L. and Mancini, P. (1996). 'Politics, media, and modern democracy: An international study of innovations in electoral campaigning and their consequences', *Greenwood Publishing Group*.

Tabachnick, B.G., Fidell, L.S. and Ullman, J.B. (2007). 'Using multivariate statistics', Boston, MA: Pearson.

Tadajewski, M. and Saren, M. (2008). 'The past is a foreign country: amnesia and marketing theory', *Marketing Theory*, 8: pp. 323-338.

Tang, G. and Lee, F.L. (2013). 'Facebook use and political participation: The impact of exposure to shared political information, connections with public political actors, and network structural heterogeneity', *Social science computer review*, 31: pp. 763-773.

Tarrow, S. and Tilly, C. (2007). 'Contentious Politics and Social Movements', in C. Boix and S. C. Stokes (eds), *The Oxford Handbook of Comparative Politics*, Oxford: Oxford University Press.

Taylor, M. (2006). '*Rationality and the ideology of disconnection*', Cambridge: Cambridge University Press.

Tedesco, J.C. (2004). 'Changing the channel: Use of the Internet for communicating about politics', *Handbook of political communication research*, 507-532.

Teorell, J. (2003). 'Linking social capital to political participation: Voluntary associations and networks of recruitment in Sweden', *Scandinavian Political Studies*, 26(1): pp. 49–66.

Teorell, J., Torcal, M. and Montero, J.R. (2007). 'Political Participation. Mapping The Terrain', in J., W., Van Deth, J., R., Montero and A., Westholm (eds), *Citizenship and Involvement in European Democracies. A Comparative Analysis*, London and New York: Routledge.

Theocharis, Y. (2012). 'Cuts, tweets, solidarity and mobilisation: How the internet shaped the student occupations', *Parliamentary affairs*, 65: pp. 162-194.

Theocharis, Y., Lowe, W., Van Deth, J. W. and Garcia-Albacete, G. (2015). 'Using Twitter to mobilize protest action: Online mobilization patterns and action repertoires in the Occupy Wall Street, Indignados, and Aganaktismenoi movements', *Information, Communication & Society*, 18(2): pp. 202–220.

Theocharis, Y. and Lowe, W. (2016). 'Does Facebook increase political participation? Evidence from a field experiment', *Information, Communication & Society*, 19: pp. 1465-1486.

Theocharis, Y. and Van Deth, J.W. (2018). 'The continuous expansion of citizen participation: a new taxonomy', *European Political Science Review*, 10: pp. 139-163.

Timpe, K. and Boyd, C.A. (2014). '*Virtues and their Vices*', Oxford: Oxford University Press.

Tolbert, C.J. and McNeal, R.S. (2003). 'Unravelling the Effects of the Internet on Political Participation?', *Political Research Quarterly*, 56(2): pp. 175-185.

Toriumi, F. and Kamiko, Y. (2017). 'Social Capital on Social Media. In *Reconstruction* of the Public Sphere in the Socially Mediated Age' Springer, Singapore: pp. 73-92.

Towne, W.B. and Herbsleb, J.D. (2012). 'Design considerations for online deliberation systems', *Journal of Information Technology & Politics*, 9: pp. 97-115.

Towner, T.L. (2013). 'All political participation is socially networked? New media and the 2012 election', *Social Science Computer Review*, 31: pp. 527-541.

Tsatsou, P. and Zhao, Y. (2016). 'A 'two-level social capital analysis' of the role of online communication in civic activism: Lessons from the role of Facebook in the sunflower movement', *Social Media*+ *Society*, 2(4), p.2056305116671967.

Turkle, S. (1996). 'Virtuality and its discontents: Searching for community in cyberspace', *American Prospect*: pp. 50-58.

Turner, J.S. and Croucher, S.M. (2014). 'An examination of the relationships among United States college students' media use habits, need for cognition, and grade point average', *Learning, Media and Technology*, *39*(2): pp.199-214.

Uslaner, E.M. (2004). 'Trust, civic engagement, and the Internet', *Political Communication*, 21: pp. 223-242.

Vaccari, C. (2012). 'Online Participation in Italy: Contextual Influences and Political Opportunities', In: Anduiza, E., Jensen, M. J. and Jorba, L. eds. *Digital Media and Political Engagement Worldwide: A Comparative Study*. Cambridge: Cambridge University Press: pp. 138-159.

Valenzuela, S., Park, N. and Kee, K.F. (2009). 'Is there social capital in a social network site?: Facebook use and college students' life satisfaction, trust, and participation', *Journal of computer-mediated communication*, 14: pp. 875-901.

Valenzuela, S., Kim, Y. and Gil De Zuniga, H. (2012). 'Social networks that matter: Exploring the role of political discussion for online political participation', *International journal of public opinion research*, 24: pp. 163-184.

Valenzuela, S. (2013). 'Unpacking the use of social media for protest behavior: The roles of information, opinion expression, and activism', *American behavioral scientist*, *57*(7): pp.920-942.

Valenzuela, S., Arriagada, A. and Scherman, A. (2014). 'Facebook, Twitter, and youth engagement: A quasi-experimental study of social media use and protest behavior using propensity score matching', *International Journal of Communication*, 8, p.25.

Valenzuela, S., Puente, S. and Flores, P.M. (2017). 'Comparing disaster news on Twitter and television: An intermedia agenda setting perspective', *Journal of Broadcasting & Electronic Media*, 61: pp. 615-637.

Valenzuela, S., Correa, T. and Gil De Zuniga, H. (2018). 'Ties, likes, and tweets: Using strong and weak ties to explain differences in protest participation across Facebook and Twitter use', *Political communication*, 35: pp. 117-134.

Van Der Gaag M. and M., Snijders, T.A. (2005). 'The Resource Generator: social capital quantification with concrete items', *Social networks*, 27(1): pp.1-29.

Van der Gaag, M., Snijders, T.A. and Flap, H.D. (2008). 'Position generator measures and their relationship to other social capital measures', *Social capital: An international research program*: pp.27-49.

Van Der Meer, T.W. (2017). 'Political trust and the 'crisis of democracy'. Oxford research encyclopedia of politics.

Van der Stede, W.A., Young, S.M. and Chen C.X. (2005). 'Assessing the quality of evidence in empirical management accounting research: The case of survey studies.' *Accounting, organizations and society* 30(7) : pp. 655-684.

Van Deth, J.W. (1990).' Interest in politics', In M.K. Jennings, J.W. van Deth et al., *Continuities in political action: A longitudinal study of political orientations in three Western democracies*, Berlin/New York: De Gruyter/Aldine: pp. 275–312.

Van Deth, J.W. (2000). 'Interesting but irrelevant. Social capital and the saliency of politics in Western Europe', *European Journal of Political Research*, 37: pp. 115-147.

Van Deth, J.W. (2001). 'Studying political participation: Towards a theory of everything', In *joint sessions of workshops of the European consortium for political research, Grenoble*: pp. 6-11.

Van Deth, J.W. (2014). 'A conceptual map of political participation', *Acta Politica*, 49: pp. 349–67.

Van Laer, J. and Van Aelst, P. (2010). 'Internet and social movement action repertoires: Opportunities and limitations', *Information, Communication & Society*, 13: pp. 1146-1171.

Vargo, S.L. and Lusch, R.F. (2004). 'Evolving to a new dominant logic for marketing', *Journal of marketing*, 68: pp. 1-17.

Vargo, S.L. and Lusch, R.F. (2008). 'From goods to service (s): Divergences and convergences of logics', *Industrial marketing management*, 37: pp. 254-259.

Vargo, S.L. (2008). 'Customer integration and value creation: paradigmatic traps and perspectives', Journal of service research, 11: pp. 211-215.

Verba, S. And Nie, N.H. (1972). '*Participation in America: Social Equality and Political Democracy*'. New York: Harper& Row.

Verba, S., Schlozman, K.L. and Brady, H.E. (1995). 'Voice and equality: Civic voluntarism in American politics', Harvard University Press.

Verba, S., Burns, N. and Schlozman, K.L. (1997). 'Knowing and caring about politics: Gender and political engagement', *The Journal of Politics* 59(4): pp. 1051–1072.

Vesnic-Alujevic, L. (2012). 'Political participation and web 2.0 in Europe: A case study of Facebook', Public relations review, 38: pp. 466-470.

Vinzi, V.E., Trinchera, L. and Amato, S. (2010). 'PLS path modeling: From foundations to recent developments and open issues for model assessment and improvement'. *in Handbook of Partial Least Squares: Concepts, Methods and Application*. Esposito Vinzi, V.; Chin, W.W.; Henseler, J.; Wang, H. (Eds.), Springer: pp. 47-82.

Vissers, S., Hooghe, M., Stolle, D. and Maheo, V.A. (2012). 'The impact of mobilization media on off-line and online participation: Are mobilization effects medium-specific?', *Social Science Computer Review*, 30: pp. 152-169.

Vissers, S. and Stolle, D. (2014). 'The Internet and new modes of political participation: online versus offline participation', *Information, Communication & Society*, *17*(8): pp.937-955.

Vitak, J., Zube, P., Smock, A., Carr, C.T., Ellison, N. and Lampe, C. (2011). 'It's complicated: Facebook users' political participation in the 2008 election', *CyberPsychology, behavior, and social networking*, 14: pp. 107-114.

Vitak, J. (2012). 'Keeping connected in the Facebook age: The relationship between Facebook use, relationship maintenance strategies, and relational outcomes', *Michigan State University. Communication Arts and Sciences-Media and Information Studies.*

Vraga, E.K., Thorson, K., Kligler-Vilenchik, N. and Gee, E. (2015). 'How individual sensitivities to disagreement shape youth political expression on Facebook', *Computers in Human Behavior* [online], 45(4): pp. 281-289.

Wang, S.I. (2007). 'Political Use of the Internet, Political Attitudes and Political Participation', *Asian Journal of Communication*, 17(4): pp. 381–395.

Wang, H. and Wellman, B., (2010). 'Social connectivity in America: Changes in adult friendship network size from 2002 to 2007', *American behavioral scientist*, *53*(8): pp.1148-1169.

Ward, S., Gibson, R. and Lusoli, W. (2003). 'Online participation and mobilisation in Britain: Hype, hope and reality', *Parliamentary affairs*, 56: pp. 652-668.

Warner, B.R. (2018). 'Modeling Partisan Media Effects in the 2014 US Midterm Elections', *Journalism & Mass Communication Quarterly*, 95(3): pp.647-669.

Warren, M.E. (1999). 'Democracy and Trust', Cambridge University Press.

Webb, P. (2000). 'Political Parties: Adapting to the Electoral Market', in P. Dunleavy et al. (eds) *Developments in British Politics*: pp. 151-168.

Webster, J. and Watson, R.T. (2002). 'Analyzing the past to prepare for the future: Writing a literature review', *MIS Quarterly*, 26(2): pp. 13–23.

Weber, L.M., Loumakis, A. and Bergman, J. (2003). 'Who participates and why? An analysis of citizens on the Internet and the mass public', *Social science computer review*, 21: pp. 26-42.

Weeks, B.E., Ardèvol-Abreu, A. and Gil de Zuniga, H. (2017). 'Online influence? Social media use, opinion leadership, and political persuasion', *International Journal of Public Opinion Research*, *29*(2): pp.214-239.

Wellman, B. (1983). 'Network analysis: Some basic principles', *Sociological theory*: pp. 155-200.

Wellman, B., Quan-Haase, A., Witte, J. and Hampton, K. (2001). 'Does the Internet increase, decrease, or supplement social capital? Social networks, participation, and community commitment', *American Behavioral Scientist*, 45(3): pp. 437-456.

Wellman, B. (2004). 'The three ages of internet studies: ten, five and zero years ago', *New media & society*, 6: pp. 123-129.

Whiteley, P. (2011). '*Political Participation in Britain: The Decline and Revival of Civic Culture*'. Basingstoke, UK: Palgrave Macmillan.

Williams, C.B. and Gulati, G.J. (2007). 'Social networks in political campaigns: Facebook and the 2006 midterm elections', annual meeting of the American Political Science Association: pp. 49-62.

Williams, D. (2006). 'On and off the'Net: Scales for social capital in an online era', *Journal of computer-mediated communication*, 11: pp. 593-628.

Williams, D. (2007). 'The Impact of Time Online: Social capital and cyberbalkanization', *CyberPsychology & Behavior*, 10(3): pp. 398–406.

Williams, J. (2012). 'Survey methods in an age of austerity: Driving value in survey design', *International Journal of Market Research*, 54(1): pp.35-47.

Winter-Nielsen, S. (2011). 'Politisk Marketing: Personer, Partier or Praksis', Karnov Group.

Wobbrock, J. O., Findlater, L., Gergle, D. and Higgins, J. (2011). 'The aligned rank transform for nonparametric factorial analyses using only ANOVA procedures', *Computer Human Interaction*: pp. 143–146.

Wolfinger, R.E. and Rosenstone, S.J. (1980). 'Who Votes?', New Haven, CT: Yale University Press.

Wong, K.K.K. (2010). 'Book review: Handbook of partial least squares: Concepts, methods and applications', *Int. Journal of Business Science and Applied Management*, Volume 6, Issue 2, 2011

Woolcock, M. (1998). 'Social capital and economic development: Toward a theoretical synthesis and policy framework', *Theory and society*, *27*(2): pp.151-208.

Woolcock, M. and Narayan, D. (2000). 'Social capital: Implications for development theory, research, and policy', *The world bank research observer*, 15(2): pp.225-249.

Woolcock, M. (2010). 'The rise and routinization of social capital, 1988–2008'. *Annual Review of Political Science*, 13: pp. 469-487.

Woo-Yoo, S. and Gil-De-Zuniga, H. (2014). 'Connecting blog, Twitter and Facebook use with gaps in knowledge and participation', *Communication & Society*, 27: pp. 33-48.

Wortmann, M. (1989). 'Political marketing: a modern party strategy'.

Wray, J.H. (1999). 'Money and Politics', in B. I. Newman (ed.), *Handbook of Political Marketing*, Thousand Oaks: Sage: pp. 741-758.

Xenos, M., Vromen, A. and Loader, B.D. (2014). 'The great equalizer? Patterns of social media use and youth political engagement in three advanced democracies', *Information, Communication & Society*, 17: pp. 151-167.

Yamamoto, M., Kushin, M.J. and Dalisay, F. (2015). 'Social media and mobiles as political mobilization forces for young adults: Examining the moderating role of online political expression in political participation', *New Media & Society*, *17*(6): pp.880-898.

Yang, H. and Dehart, J.L. (2016). 'Social media use and online political participation among college students during the US election 2012', *Social Media+ Society*, 2, 2056305115623802.

Yoder, C. and Stutzman, F. (2011). 'Identifying social capital in the Facebook interface', In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*: pp. 585-588.

Yoon, K., Pinkleton, B., E. and Ko, W. (2005). 'Effects of Negative Political Advertising on Voting Intention: An Exploration of the Roles of Involvement and Source Credibility in the Development of Voter Cynicism', *Journal of Marketing Communications*, 11(2): pp. 95–112.

Yung, B., Leung, L. and Y.M., (2014). 'Facebook as change? Political engagement in semi-democratic Hong Kong in its transition to universal suffrage', *Journal of Asian Public Policy*, 7(3): pp. 291-305.

Zald, M.N. and McCarthy, J.D. (2002). 'The resource mobilization research program: progress, challenges, and transformation', in J. Berger and M. Zelditch (eds), *New Directions in Contemporary Sociological Theory*, Lanham, MD: Rowman & Littlefield: pp. 147–71.

Zaller, J.R. (1992). '*The nature and origins of mass opinion*', Cambridge: Cambridge University Press.

Zhan, L., Sun, Y., Wang, N. and Zhang, X. (2016). 'Understanding the influence of social media on people's life satisfaction through two competing explanatory mechanisms', *Aslib Journal of Information Management*, *68*(3): pp. 347-361.

Zhang, W. and Chia, S.C. (2006). 'The effects of mass media use and social capital on civic and political participation', *Communication Studies*, 57: pp. 277-297.

Zhang, W., Johnson, T.J., Seltzer, T. and Bichard, S.L. (2010). 'The revolution will be networked: The influence of social networking sites on political attitudes and behavior' *Social Science Computer Review*, 28(1), pp.75-92.

Zhang, W. and Wang, R. (2010). 'Interest-oriented versus relationship-oriented social network sites in China', *First Monday*.

Zhang, W. and Gearhart, S. (2015). 'The Effects of Internet use and Internet efficacy on offline and online engagement', *Online Journal of Communication and Media Technologies*, 5: p.147.

Zhao, X., Tang, Q., Liu, S. and Liu, F. (2016). 'Social capital, motivations, and mobile coupon sharing', *Industrial Management & Data Systems*.

Zhong, Z.J. (2011). 'The effects of collective MMORPG (Massively Multiplayer Online Role-Playing Games) play on gamers' online and offline social capital', *Computers in Human Behavior*, 27: pp. 2352–2363.

Zhong, Z.J. (2014). 'Civic engagement among educated Chinese youth: The role of SNS (Social Networking Services), bonding and bridging social capital', *Computers & Education*, 75: pp. 263-273.

Zukin, C., Keeter, S., Andolina, M., Jenkins, K. and Delli Carpini, M.X. (2006). 'A New Engagement? Political Participation, Civic Life, and the Changing American Citizen', Oxford: Oxford University Press.